## SPECIFICATIONS AND OTHER BIDDING AND CONTRACT DOCUMENTS

ALAMEDA COUNTY PROJECT # 2043 HIGHLAND HOSPITAL – KORET BUILDING EXHAUST FAN AND WATERPROOFING 1331 EAST 31<sup>ST</sup> STREET Oakland, California

## MANDATORY PRE-BID SITE VISIT AND MEETING

Tuesday July 7, 2020 at 2:00pm Location: Remote (Zoom Meeting) Contact Person: Hadi Hadjarzadeh Hadi.Hadjar@acgov.org (510) 220-3445

> ALAMEDA COUNTY GENERAL SERVICES AGENCY TECHNICAL SERVICES DEPARTMENT 1401 LAKESIDE DRIVE, #800 OAKLAND, CALIFORNIA

PHONE: 510-208-9589 FAX: 510-208-3995

Architecture and Engineering Mazzetti 3600 American River Drive, Suite 203 Sacramento, CA 95864-5950 PHONE: 916-978-4122

> Simpson Gumpertz & Heger 100 Pine Street, Suite 1600 San Francisco, CA 94111 PHONE: 415-495-3700



**DOCUMENT 00 01 09** 

## **SUMMARY BIDDING CALENDAR**

**NOTICE** – THIS SUMMARY IS FOR INFORMATIONAL PURPOSES ONLY AND DOES NOT LIST ALL DATES, OR TIMES IN THE BIDDING DOCUMENTS. The dates and times listed may not be relied upon or enforced. This summary does not form a part of the contract documents and does not establish contractual obligations. All bidders and contractors must refer to the actual documents for all applicable dates, times and time periods.

<u>Event</u>	<u>Date</u>	<u>Reference</u>		
Contract Documents Available	June 18, 2020	00 11 16 Notice to Bidders		
Pre-Bid Conference & Site Visit	July 7, 2020	00 11 16 Notice to Bidders		
Last Day for Receipt of Requests	10 business days before date for	00 21 13 Instructions to Bidders		
for Substitutions before Receipt of	Receipt of Bids			
Bids				
Last Day for Receipt of Questions	10 business days before date for	00 21 13 Instructions to Bidders		
	Receipt of Bids			
Last Day to Submit Bid Protest	5 <sup>th</sup> Business Day from Date of	00 21 13 Instructions to Bidders		
	Notice of Intent to Award			
Receipt of Bids and Bid Opening	July 17, 2020 By 2:00 p.m.	00 11 16 Notice to Bidders		
Last day to submit ECOP forms	(2) business days following the	00 22 19 Supplemental		
101A, 101B and 102	bid opening	Instructions to Bidders-		
		Enhanced Construction		
		Outreach Program		
Estimated Date of Notice of Award	September 4, 2020	00 51 00 Notice of Award		
Signing of Contract	7 Calendar days after Notice of	00 11 16 Notice to Bidders		
	Award	00.01.10.7		
Submit Post-Award Documents	7 Calendar days after Notice of	00 21 13 Instructions to Bidders		
Y D G 1 T D	Award	00.56.00 F		
Last Day to Submit Escrow Bid	7 Calendar days after Notice of	00 56 00 Escrow Bid		
Documentation	Award	Documentation		
Contract Duration	210 calendar days	00 52 13 Agreement Form –		
		Stipulated Sum (Single-Prime		
		Contract)		
Contract Duration Begins	Date in Notice Proceed	00 55 00 Notice to Proceed		
Contract Duration Ends	Date in Notice Proceed	00 55 00 Notice to Proceed		
Last Day to Submit Preliminary	10 <sup>th</sup> Business Day following	00 55 00 Notice to Proceed		
Schedule, etc. per Notice to	Notice to Proceed			
Proceed				

END OF DOCUMENT

#### **DOCUMENT 00 01 10**

#### TABLE OF CONTENTS

This Table of Contents describes the documents provided to Bidders: Specifications and Other Bidding and Contract Documents for the Highland Hospital – Koret Building Exhaust Fan and Waterproofing Project. Please refer to the all Division 00, Division 01, Specifications, and Drawings in order to understand the complete requirements for the Project.

## DIVISION 00 — PROCUREMENT AND CONTRACTING REQUIREMENTS

00 00 00 PROCUREMENT AND CONTRACTING REQUIREMENTS

#### INTRODUCTORY INFORMATION

00 01 01	Project Title Page
<b>00 01 01</b>	I I U JULU I IIIU I agu

00 01 09 Summary Bidding Calendar

00 01 10 Table of Contents

00 01 15 List of Drawings

## PROCUREMENT REQUIREMENTS

00 10 00	SOLICITATION
\\\\\\\\\\\\\\\\\\\\\\\\\\\\\\\\	SOLICITATION

00 11 00 Advertisements and Invitations

00 11 16 Notice to Bidders

00 20 00 Instructions for Procurement

00 21 00 Instructions

00 21 13 Instructions to Bidders

00 22 19 Supplementary Instructions to Bidders – Enhanced Construction Outreach Program

00 30 00 AVAILABLE INFORMATION

00 31 00 Available Project Information

00 31 19 Existing Conditions Information

00 40 00 PROCUREMENT FORMS AND SUPPLEMENTS

00 41 00 Bid Forms

00 41 13 Bid Form – Stipulated Sum (Single-Prime Contact)

00 43 00 Procurement Form Supplements

00 43 13 Bid Security Form

00 43 36 Designated Subcontractors List

00 45 00 Representations and Certifications

00 45 01 Site-Visit Certification

00 45 13 Non-Collusion Affidavit

00 45 26 Worker's Compensation Certification

00 45 46 Governmental Certifications

00 45 46.01 Prevailing Wage and Related Labor Requirements Certification

ALAMEDA COUNTY GSA-CP

## Alameda County, General Services Agency

**PROJECT # 2043** 

Highland Hospital – Koret Building Exhaust Fan and Waterproofing

00 45 46.04 Hazardous Materials Certification 00 45 46.06 Imported Materials Certification

## CONTRACTING REQUIREMENTS

00 50 00 CONTRACTING FORMS AND SUPPLEMENTS

00 51 00 Notice of Award

00 51 13 Notice of Intent to Award

00 52 00 Agreement Forms

00 52 13 Agreement Form – Stipulated Sum (Single-Prime Contract)

00 52 13.1 Debarment and Suspension Certification Form

00 55 00 Notice to Proceed

00 56 00 Escrow of Bid Documentation

00 57 00 Escrow Agreement for Security Deposits in Lieu of Retention

00 60 00 PROJECT FORMS

00 61 00 Bond Forms

00 61 13 Performance and Payment Bond Forms

00 61 13.13 Performance Bond Form

**00 61 13.16 Payment Bond Form** 

00 65 00 Closeout Forms

00 65 19 Certificate of Completion Forms

00 65 19.26 Final Settlement Certificate Form

00 65 36 Warranty Form

00 70 00 CONDITIONS OF THE CONTRACT

00 72 00 General Conditions

00 72 13 General Conditions – Stipulated Sum (Single-Prime Contract)

00 73 00 Supplementary Conditions

00 73 13 Special Conditions

00 73 49 Project Stabilization/Community Benefits Agreement (PSCBA)

00 73 56 Hazardous Materials Procedures and Requirements

#### **DIVISION 01 — GENERAL REQUIREMENTS**

01 00 00 GENERAL REQUIREMENTS

01 10 00 Summary

01 11 00 Summary of Work

01 20 00 PRICE AND PAYMENT PROCEDURES

01 22 00 Unit Prices and Alternates

01 26 00 Contract Modification Procedures

01 29 00 Payment Procedures

01 30 00 Administrative Requirements

01 31 00 Project Management and Coordination

01 31 19 Project Meetings

01 32 16 Schedules and Reports

ALAMEDA COUNTY GSA-CP

Page 2 of 3

TABLE OF CONTENTS DOCUMENT 00 01 10

## **Alameda County, General Services Agency**

**PROJECT # 2043** 

Highland Hospital – Koret Building Exhaust Fan and Waterproofing

## 01 33 00 Submittal Procedures

01 40 00 QUALITY REQUIREMENTS 01 41 00 Regulatory Requirements

01 42 00 **References** 

01 42 13 Abbreviations and Acronyms

01 42 16 Definitions and Reference Standards

01 43 00 Quality Assurance – materials and equipment

01 45 00 Quality Control

#### 01 50 00 TEMPORARY FACILITIES AND CONTROLS

01 60 00 PRODUCT REQUIREMENTS

01 62 00 Product Options and Substitutions

01 65 00 Delivery, Storage and Handling

01 70 00 EXECUTION AND CLOSEOUT REQUIREMENTS

01 71 00 Examination and Preparation

01 71 23 Field Engineering

01 73 00 Execution

01 73 29 Cutting and Patching

01 74 19 Construction Waste Management

01 76 00 Protecting Installed Construction

01 77 00 Construction Closeout

01 78 00 Closeout Submittals

01 78 23 Operation and Maintenance Instructions

**01 78 36** Warranties

01 78 39 Project Record Documents

01 90 00 LIFE CYCLE ACTIVITIES

01 91 00 Commissioning

01 91 13 General Commissioning Requirements

#### **SPECIFICATIONS**

07 92 00 Joint Sealants

09 96 53 Elastomeric Coating

#### **DRAWINGS**

KORET BUILDING EMERGENCY DEPARTMENT EXHAUST FAN BUILDING K (Koret Building) EXTERIOR MAINTENANCE

## DOCUMENT 00 01 15

## **LIST OF DRAWINGS**

#### KORET BUILDING EMERGENCY DEPARTMENT EXHAUST FAN DRAWINGS

Sheet number	<u>Description</u>
G0.0	Project Cover Sheet
M0.0	Mechanical Legends, Abbreviations, Notes, and Schedules
M0.1	Mechanical Specifications
M1.0	Lobby Air Flow Mechanical Plan
M5.1	Mechanical Details
S1.0	Seismic Bracing of Suspended Piping, Conduit, and Ductwork

## BUILDING K (Koret Building) EXTERIOR MAINTENANCE DRAWINGS

Sheet number	<u>Description</u>
A010	Project Information, Sheet Index, Key Plan, Symbols
A104	Existing Plan – 4 <sup>th</sup> Floor
A105	Existing Plan – 5 <sup>th</sup> Floor
A106	Existing Plan – 6 <sup>th</sup> Floor
A107	Existing Plan – 7 <sup>th</sup> Floor
A108	Existing Roof Plan
A200	Existing Elevations
A201	Existing Elevations
A202	Existing Elevations
A900	Existing Wall Sections
A901	Scope of Work – Partial Elevations
A902	Scope of Work – Typical Details
A910	Existing Details
A911	Existing Details

END OF DOCUMENT

#### **DOCUMENT 00 11 16**

#### **NOTICE TO BIDDERS**

- Notice is hereby given that The County of Alameda General Services Agency ("GSA")
   Purchasing Department ("County" or "Owner") will receive sealed bids for the following project, Bid No. 2043 ("Project" or "Contract"):
   Highland Hospital Koret Building Exhaust Fan and Waterproofing (Project Name)
- 2. Sealed Bids will be received until 2:00 p.m., July 17, 2020, at 1401 Lakeside Drive, Purchasing Department Room #609, Oakland, California, at or after which time the bids will be opened and publicly read aloud. Any claim by a bidder of error in its bid must be made in compliance with \$5100 et seq. of the Public Contract Code. Any bid that is submitted after this time will be considered non-responsive and returned to the bidder.
- 3. The Project consists of:
  - Koret Building Emergency Department Exhaust Fan Project Installation of an exterior, pad-mounted exhaust fan and all associated work including, but not limited to site, architectural, mechanical, electrical, and seismic anchorage, to provide code-required airflows (supply and exhaust) in the existing Emergency Department Waiting Room.

    Koret Building Exterior Maintenance Project Preparation and elastomeric coating at

cement plaster assemblies and sealant replacement to address water intrusion at the Koret Building exterior.

The budgetary estimate for the scope of work is approximately \$2.7M. The time to complete this project is 210 calendar days from the Notice to Proceed.

- 4. All bids shall be on the Bid Form Document 00 41 13 provided by the County. Each bid must conform to and be responsive to all pertinent Contract Documents, including, but not limited to, the Instructions to Bidders Document 00 21 13 and the Supplemental Instructions to Bidders -Enhanced Construction Outreach Program Document 00 22 19.
- 5. Bidders are strongly encouraged to review the Supplementary Instructions to Bidders Enhanced Construction Outreach Program Document (ECOP) 00 22 19 and to begin their outreach efforts prior to the initial mandatory project job walk. The list of bidders solicited for this project include but are not limited to all those construction contractors listed in the GSA Small, Local & Emerging Program Vendor Query database located at <a href="http://www.acgov.org/sleb\_query\_app/gsa/sleb/query/slebmenu.jsp">http://www.acgov.org/sleb\_query\_app/gsa/sleb/query/slebmenu.jsp</a>.
- 6. To bid on this Project, the Bidder is required to possess the following State of California Contractor Licenses:

### B – General Building Contractor

The Bidder's license must remain active and in good standing throughout the term of the Contract.

- 7. A bid bond by an admitted surety insurer on the form provided by the County, cash, or a cashier's check or a certified check, drawn to the order of the County of Alameda, in the amount of ten percent (10%) of the total bid price, shall accompany the Bid Form, as a guarantee that the Bidder will, within seven (7) calendar days after the date of the Notice of Award, enter into a contract with the County for the performance of the services as stipulated in the bid.
- 8. The successful Bidder shall be required to furnish a 100% Performance Bond and a 100% Payment Bond if it is awarded the contract for the Work.
- 9. The successful Bidder may substitute securities for any monies withheld by the County to ensure performance under the Contract, in accordance with the provisions of section 22300 of the Public Contract Code.
- 10. The Contractor and all Subcontractors under the Contractor shall pay all workers on all work performed pursuant to this Contract not less than the general prevailing rate of per diem wages and the general prevailing rate for holiday and overtime work as determined by the Director of the Department of Industrial Relations, State of California, for the type of work performed and the locality in which the work is to be performed within the boundaries of the County, pursuant to sections 1770 et seq. of the California Labor Code. Prevailing wage rates are also available from the County or on the Internet at: <a href="http://www.dir.ca.gov">http://www.dir.ca.gov</a>.
- 11. This project is subject to compliance monitoring and enforcement by the Department of Industrial Relations. The following requirements apply to this bid and contract:
  - A. No contractor or subcontractor may be listed on a bid proposal for a public works project (submitted on or after March 1, 2015) unless registered with the Department of Industrial Relations pursuant to Labor Code section 1725.5 [with limited exceptions from this requirement for bid purposes only under Labor Code section 1771.1(a)].
  - B. No contractor or subcontractor may be awarded a contract for public work on a public works project (awarded on or after April 1, 2015) unless registered with the Department of Industrial Relations pursuant to Labor Code section 1725.5.
- 12. The Work performed pursuant to this Contract will be subject to the requirements of the "PROJECT STABILIZATION/COMMUNITY BENEFITS AGREEMENT for the COUNTY OF ALAMEDA" as described in Project Stabilization/ Community Benefit

## **Alameda County General Services Agency**

Highland Hospital – Koret Building Exhaust Fan and Waterproofing

Document 00 73 49. In consideration of the award of a Contract to perform the Work, the Contractor agrees to be party to and bound by the "PROJECT STABILIZATION/COMMUNITY BENEFITS AGREEMENT for the COUNTY OF ALAMEDA". Contractor agrees to execute the "PROJECT STABILIZATION/COMMUNITY BENEFITS AGREEMENT for the COUNTY OF ALAMEDA" Letter of Assent and shall require all of its subcontractors, of whatever tier, to become similarly bound for all work within the scope of this Contract by signing an identical Letter of Assent.

13. A remote mandatory pre-bid conference via Zoom will be held on July 7, 2020, at 2.00 PM. Participants should plan on logging into the Zoom meeting 5-10 minutes before the meeting begins, as they will be required to register with their e-mail, name, company and contact information, The pre-bid conference is expected to take approximately 90 minutes. Zoom attendance will be emailed to participants. Please note that this meeting will be recorded. To Join Zoom Meeting:

https://vanir.zoom.us/j/81534258722?pwd=NENrY2tBbjN6aG1Xa0MzRGluc2tGUT09 Meeting

ID: 815 3425 8722 Password: 817904

One tap mobile

- +16699006833,,81534258722#,,,,0#,,817904# US (San Jose)
- +12532158782,,81534258722#,,,,0#,,817904# US (Tacoma)

#### Dial by your location

- +1 669 900 6833 US (San Jose)
- +1 253 215 8782 US (Tacoma)
- +1 346 248 7799 US (Houston)
- +1 312 626 6799 US (Chicago)
- +1 929 205 6099 US (New York)
- +1 301 715 8592 US (Germantown)

Meeting ID: 815 3425 8722

Password: 817904

Find your local number: https://vanir.zoom.us/u/kcD1MLuMx8

- 14. Contract Documents are available on June 18, 2020, for review at the site office of Vanir Construction Management, Highland Hospital, Wing B1, California. (650) 392-9698. In addition, Contract Documents are available for bidders' review at the locations shown on Attachment 1 to this Document.
- 15. The County has found and determined that the following item(s) shall be used on this Project based on the purpose(s) indicated. (Public Contract Code section 3400(b)): A particular material, product, thing, or service is designated by specific brand or trade name for the following purpose(s):
- (1) Not Applicable
- 16. It is County policy to minimize the expenditure of County funds on goods and services produced by any entity which buys, sell, leases or distributes commodities and/or professional services to (1) the government of Burma; or (2) any entity organized under the laws of Burma; or (3) any entity which does business with any private or public entity located in Burma, or conducts operations in Burma. Contractors are urged to comply with the policy in making purchases and subcontracts. (ref. Alameda County, Cal., Adm. Code tit.4, §4.32.050(B),(F))

#### 17. INTENTIONALLY LEFT BLANK

- 18. The County reserves the right to reject any and all bids and/or waive any irregularity in any bid received. If the County awards the Contract, the security of unsuccessful bidder(s) shall be returned within sixty (60) days from the time the award is made. Unless otherwise required by law, no bidder may withdraw its bid for ninety (90) days after the date of the bid opening.
- 19. The County shall award the Contract, if it awards it at all, to the lowest responsive responsible bidder based on:
- A. The base bid amount only.

Determination of the responsible bidder with the lowest responsive bid will also be subject to the terms of the Supplemental Instructions to bidders Enhanced Construction Outreach Program Document 00 22 19.

END OF DOCUMENT

				ATTACHMENT 1
PLAN ROOM ADVERTISING LIST				
1	Bay Area Builders Exchange** 3055 Alvarado Street San Leandro, CA 94577 Phone: (510) 483-8880 ;Fax: (925) 685-3424 Email: planroom@bayareabx.com (This is a merger of Builders Exchange of Alameda County and Contra Costa Builders Exchange 5/18/15.)		2	San Francisco Builders Exchange 850 South Van Ness Avenue San Francisco, CA 94110 Phone: (415) 282-8220 Fax: (415) 821-0363 Email: djohnsonsf@sbcglobal.net
3	Dodge Data and Analytics (Dodge Plan Room, formerly McGraw-Hill Construction Dodge) (Online) 3315 Central Avenue Hot Springs Arkansas (AR) 71913 (Contact: Gerry McCarthy) 626-531-6818; Fax: 626-226-1623 Email gerry.mccarthy@construction.com		4	Small Business Exchange 703 Market Street, Suite 1000 San Francisco, CA 94103 Phone: (415) 778-6250 Fax: (415) 778-6255 Email: sbe@sbeinc.com
<u> </u>	Central California Builders Exchange 1244 N. Mariposa St. Fresno, Ca 93703 Phone (559) 237-1831; Fax (559) 264-2532 Email: megan@cencalbx.com		6	County of Alameda Current Contracting Opportunities Website located at <a href="http://www.acgov.org/gsa_app/gsa/purchas-ing/bid_content/contractopportunities.jsp">http://www.acgov.org/gsa_app/gsa/purchas-ing/bid_content/contractopportunities.jsp</a>
7	The Blue Book Building & Construction Netwo Contact: Amanda Limitone, Project Communic Phone: (855) 805-2560, ext.3145; Email: alimi	ation	Spe	cialist
8	Reed Construction Data** – Online/Electronic Plan Room 30 Technology Parkway South, Suite 100 Norcross, GA 30092-2912 Phone: (770) 209-3396 Jeannie Kwan; Fax (Addenda only): (800) 303-8629; Fax (Notice to Bidders/IFB): (800) 642-2437; Email (addenda only): <a href="mailto:docprocessing@reedbusiness.com">docprocessing@reedbusiness.com</a> -Send requests to advertise to above address/fax/phone- Local Email: <a href="mailto:jeannie.kwan@reedbusiness.com">jeannie.kwan@reedbusiness.com</a> (EPR: <a href="http://www.reedconstructiondata.com">http://www.reedconstructiondata.com</a> )			
9	East Bay Blue Print & Supply Co. 1745 Fourteenth Ave Oakland, CA 94606 Phone: (510) 261-2990 - Sandy Petty Email: ebbp@eastbayblueprint.com			
<u> </u>	Construction Bidboard, Inc.(Online)** 11622 El Camino Real, Suite 100 San Diego, CA 92130 800-479-5314 phone; 619-688-0585 fax (Contact Dorothy Ellithorpe dellithorpe@ebiAlternate: planroom@ebidboard.com* ebidbo			

<sup>\*</sup> Plans/Specs must be sent to individual Plan Rooms to ensure posting at that location.
\*\*Construction trade journals specified for alternate bidding procedures for projects between \$25,000 and \$125,000 minimum advertising requirements. County policy is to post all construction projects over \$25,000 in all listed Plan Rooms, Press/Newspaper Publications and Local Chambers of Commerce/Trade Organizations

#### **DOCUMENT 00 21 13**

## **INSTRUCTIONS TO BIDDERS**

Bidders shall follow the instructions in this document, and shall submit all documents, forms, and information required for consideration of a Bid.

County will evaluate information submitted by the apparent low Bidder and, if incomplete or unsatisfactory to County, Bidder's bid may be rejected at the sole discretion of County.

- 1. Bids are requested for a general construction contract, or work described in general, for the following project ("Project" or "Contract"):
  - Highland Hospital Koret Building Exhaust Fan and Waterproofing
- 2. County will receive sealed Bids from Bidders as stipulated in the Notice to Bidders Document 00 11 16.
- 3. Bidders must submit Bids on Bid Form 00 41 13 and all other required County forms. Bids not submitted on the County's required forms shall be deemed non-responsive and shall not be considered. Additional sheets required to fully respond to requested information are permissible.
- 4. Bidders must supply all information required by each Bid Document. Bids must be completed in full. County reserves the right in its sole discretion to reject any Bid as non-responsive as a result of any error or omission in the Bid. Bidders must complete and submit all of the following documents with Bid Form Document 00 41 13:
  - a. Bid Bond on Bid Security Form Document 00 43 13 or other security
  - b. Designated Subcontractors List Document 00 43 36
  - c. Site-Visit Certification Document 00 45 01, if a site visit was required
  - d. Non-Collusion Affidavit Document 00 45 13
  - e. Enhanced Construction Outreach Program Package as required by Supplementary Instructions to Bidders Enhanced Construction Outreach Program Document 00 22 19.
  - f. Completed Debarment Form, Document 00 52 13.1.
- 5. Bidders must submit with their Bids cash, a cashier's check or a certified check payable to County, or a Bid Bond of not less than ten percent (10%) of amount of base Bid, plus all additive alternates. Required form of corporate surety, Bid Security Form, is provided by County and must be used and fully completed by Bidders choosing to provide a Bid Bond as security. The Surety on Bidder's Bid Bond must be an insurer admitted in the State of California and authorized to issue surety bonds in the State of California. Bids

- submitted without necessary bid security will be deemed non-responsive and will not be considered.
- 6. If Bidder to whom Contract is awarded shall for <u>SEVEN (7)</u> calendar days after the date of the Notice of Award, fail or neglect to enter into Contract and submit required bonds, insurance certificates, and all other required documents, County may deposit Bid Bond, cash, cashier's check, or certified check for collection, and proceeds thereof may be retained by County as liquidated damages for failure of Bidder to enter into Contract, in the sole discretion of County. It is agreed that calculation of damages County may suffer as a result of Bidder's failure to enter into the Contract would be extremely difficult and impractical to determine and that the amount of the Bidder's required bid security shall be the agreed and conclusively presumed amount of damages.
- 7. Bidders must submit with the Bid the Designated Subcontractors List for those subcontractors who will perform any portion of Work, including labor, rendering of service, or specially fabricating and installing a portion of the Work or improvement according to detailed drawings contained in the plans and specifications, in excess of one half of one percent (0.5%) of total Base Bid. Failure to submit this list when required by law shall result in Bid being deemed non-responsive and the Bid will not be considered.
- 8. If a mandatory pre-bid conference and site visit ("Site Visit") is requested as referenced in the Instructions to Bidders, then Bidders must submit the Site-Visit Certification with their Bid. County will transmit to all prospective Bidders of record such Addenda as County in its discretion considers necessary in response to questions arising at the Site Visit. Oral statements shall not be relied upon and will not be binding or legally effective. Addenda issued by the County as a result of the Site Visit, if any shall constitute the sole and exclusive record and statement of the results of the Site Visit.
- 9. Bidders shall submit the Non-Collusion Affidavit with their Bids. Bids submitted without the Non-Collusion Affidavit shall be deemed non-responsive and will not be considered.
- 10. Bids shall be clearly written without erasure or deletions. County reserves the right to reject any Bid containing erasures or deletions.
- 11. Bidders shall not modify Bid Form 00 41 13 or qualify their Bids. Bidders shall not submit to the County a scanned, re-typed, word-processed, or otherwise recreated version of Bid Form 00 41 13 or other County-provided document.
- 12. The successful Bidder and all its subcontractors shall pay all workers on all work performed pursuant to this Contract not less than the general prevailing rate of per diem wages and the general prevailing rate for holiday and overtime work as determined by the Director of the Department of Industrial Relations, State of California, for the type of

- work performed and the locality in which the work is to be performed within the boundaries of the County, pursuant to sections 1770 et seq. of the California Labor Code.
- 13. Submission of Bid signifies Contractor agreement to be party to and bound by the "PROJECT STABILIZATION/COMMUNITY BENEFITS AGREEMENT for the COUNTY OF ALAMEDA". Contractor agrees to execute the "PROJECT STABILIZATION/COMMUNITY BENEFITS AGREEMENT for the COUNTY OF ALAMEDA" Letter of Assent and shall require all of its subcontractors, of whatever tier, to become similarly bound for all work within the scope of this Contract by signing an identical Letter of Assent.
- 14. Submission of Bid signifies careful examination of Contract Documents and complete understanding of the nature, extent, and location of Work to be performed. Bidders must complete the tasks listed below as a condition to bidding, and submission of Bid shall constitute the Bidder's express representation to County that Bidder has fully completed the following:
  - a. Bidder has visited the Site and has examined thoroughly and understood the nature and extent of the Contract Documents, Work, Site, locality, actual conditions, as-built conditions, and all local conditions and federal, state and local laws, and regulations that in any manner may affect cost, progress, performance, or furnishing of Work or that relate to any aspect of the means, methods, techniques, sequences, or procedures of construction to be employed by Bidder and safety precautions and programs incident thereto;
  - b. Bidder has conducted or obtained and has understood all examinations, investigations, explorations, tests, reports, and studies that pertain to the subsurface conditions, as-built conditions, underground facilities, and all other physical conditions at or contiguous to the Site or otherwise that may affect the cost, progress, performance, or furnishing of Work, as Bidder considers necessary for the performance or furnishing of Work at the Contract Sum, within the Contract Time, and in accordance with the other terms and conditions of Contract Documents, including specifically the provisions of the General Conditions; and no additional examinations, investigations, explorations, tests, reports, studies, or similar information or data are or will be required by Bidder for such purposes;
  - c. Bidder has correlated its knowledge and the results of all such observations, examinations, investigations, explorations, tests, reports, and studies with the terms and conditions of the Contract Documents;
  - d. Bidder has given County prompt written notice of all conflicts, errors, ambiguities, or discrepancies that it has discovered in or among the Contract Documents and the actual conditions, and the written resolution thereof by County is acceptable to Bidder;

- e. Bidder has made a complete disclosure in writing to County of all facts bearing upon any possible interest, direct or indirect, that Bidder believes any representative of County or other officer or employee of County presently has or will have in this Contract or in the performance thereof or in any portion of the profits thereof;
- f. Bidder must, prior to bidding, perform the work, investigations, research, and analysis required by this document and that Bidder represents in its Bid Form 00 41 13 and the Agreement that it performed prior to bidding. Bidders are charged with all information and knowledge that a reasonable bidder would ascertain from having performed this required work, investigation, research, and analysis. Bid prices must include entire cost of all work "incidental" to completion of the Work.
- g. Conditions Shown on the Contract Documents: Information as to underground conditions, as-built conditions, or other conditions or obstructions, indicated in the Contract Documents, e.g., on Drawings or in Specifications, has been obtained with reasonable care, and has been recorded in good faith. However, County only warrants, and Bidder may only rely, on the accuracy of limited types of information.
  - (1) As to above-ground conditions or as-built conditions shown or indicated in the Contract Documents, there is no warranty, express or implied, or any representation express or implied, that such information is correctly shown or indicated. This information is verifiable by independent investigation and Bidder is required to make such verification as a condition to bidding. In submitting its Bid, Bidder shall rely on the results of its own independent investigation. In submitting its Bid, Bidder shall not rely on County-supplied information regarding above-ground conditions or asbuilt conditions.
  - As to any subsurface condition shown or indicated in the Contract Documents, Bidder may rely only upon the general accuracy of actual reported depths, actual reported character of materials, actual reported soil types, actual reported water conditions, or actual obstructions shown or indicated. County is not responsible for the completeness of such information for bidding or construction; nor is County responsible in any way for any conclusions or opinions of Bidder drawn from such information; nor is County responsible for subsurface conditions that are not specifically shown (for example, County is not responsible for soil conditions in areas contiguous to areas where a subsurface condition is shown).

- h. Conditions Shown in Reports and Drawings Supplied for Informational Purposes: Reference is made to the document entitled Geotechnical Data, and the document entitled Existing Conditions Information, for identification of:
  - (1) Subsurface Conditions: Those reports of explorations and tests of subsurface conditions at or contiguous to the Site that have been utilized by Architect in preparing the Contract Documents; and
  - (2) Physical Conditions: Those drawings of physical conditions in or relating to existing surface or subsurface structures at or contiguous to the Site that has been utilized by Architect in preparing the Contract Documents.
  - (3) These reports and drawings are <u>not</u> Contract Documents and, except for any "technical" data regarding subsurface conditions specifically identified in Geotechnical Data and Existing Conditions Information, and underground facilities data, Bidder may not in any manner rely on the information in these reports and drawings. Subject to the foregoing, Bidder must make its own independent investigation of all conditions affecting the Work and must not rely on information provided by County.
- 15. Bidders may examine any available "as-built" drawings of previous work by giving County reasonable advance notice. County will not be responsible for accuracy of "asbuilt" drawings. The document entitled Existing Conditions Information applies to all supplied "as-built" drawings.
- 16. Copies of the general prevailing rates of per diem wages for each craft, classification, or type of worker needed to execute the Contract, as determined by Director of the State of California Department of Industrial Relations, are on file at the County's principal office. Prevailing wage rates are also available from the County or on the internet at (http://www. dir.ca.gov)
- 17. All questions about the meaning or intent of the Contract Documents are to be directed in writing, including by e-mail, to County. Interpretations or clarifications considered necessary by County in response to such questions will be issued in writing by Addenda faxed, mailed, or delivered to all parties recorded by County as having received the Contract Documents. Questions received less than <u>TEN(10)</u> business days prior to the date for opening Bids may not be answered. Only questions answered by formal written Addenda will be binding. Oral and other interpretations or clarifications will be without legal effect.
- 18. Addenda may also be issued to modify other parts of the Contract Documents as deemed advisable by County.

- 19. Each Bidder must acknowledge each Addendum in its Bid Form 00 41 13 by number or its Bid shall be considered non-responsive. Addenda shall be part of the Contract Documents. A complete listing of Addenda may be secured from County.
- 20. Bids shall be based on products and systems specified in Contract Documents or listed by name in Addenda. County is not responsible and/or liable in any way for a Bidder's damages and/or claims related, in any way, to that Bidder's basing its bid on any requested substitution that County has not approved. Bidders and materials suppliers who submit requests for substitutions prior to the award of the Contract must do so in writing and in compliance with Public Contract Code section 3400. All requests must comply with the following:
  - a. County must receive any request for substitution a minimum of <u>TEN (10)</u> business days prior to bid opening.
  - b. Requests for substitutions shall contain sufficient information to assess acceptability of product or system and impact on Project, including, without limitation, the requirements specified in the Special Conditions and the Specifications. Insufficient information shall be grounds for rejection of substitution.
  - c. Approved substitutions shall be listed in Addenda. County reserves the right not to act upon submittals of substitutions until after bid opening.
  - d. Substitutions may be requested after Contract has been awarded only if indicated in and in accordance with requirements specified in the Special Conditions and the Specifications.
- 21. All Bids must be sealed, and marked with name and address of the Bidder and the Project Number, Bid number, Bid package, and time of bid opening. Bids will be received as indicated in the Notice to Bidders.
  - a. Mark envelopes with the name of the Project.
  - b. Bids must be submitted at the place and by date and time shown in the Instructions to Bidders.
  - c. Bids must contain all documents as required herein.
- 22. Bids will be opened Remotely via Zoom immediately after the time indicated for receipt of bids.
- 23. This Contract may include alternates. Alternates are defined as alternate products, materials, equipment, systems, methods, or major elements of the construction that may, at the County's option and under terms established in the Contract and pursuant to section 20103.8 of the Public Contract Code, be selected for the Work. County shall award the

- Contract, if it awards it at all, to the lowest responsive responsible bidder based on the criteria as indicated in these contract documents.
- 24. Time for Completion: County may issue a Notice to Proceed within <u>NINETY (90)</u> calendar days from the date of the Notice of Award. Once Contractor has received the Notice to Proceed, Contractor shall complete the Work within the period of time indicated in the Contract Documents.
  - a. In the event that County desires to postpone issuing the Notice to Proceed beyond the 90-day period above, it is expressly understood that with reasonable notice to the Contractor, County may postpone issuing the Notice to Proceed.
  - b. It is further expressly understood by Bidder that Contractor shall not be entitled to any claim of additional compensation as a result of the postponement of the issuance of the Notice to Proceed beyond the 90-day period. If the Contractor believes that a postponement of issuance of the Notice to Proceed will cause a hardship to the Contractor, the Contractor may terminate the Contract. Contractor's termination due to a postponement beyond this 90-day period shall be by written notice to County within <u>TEN (10)</u> calendar days after receipt by Contractor of County's notice of postponement.
  - c. It is further understood by Bidder that in the event that Contractor terminates the Contract as a result of postponement by County, County shall only be obligated to pay Contractor for the Work that Contractor had performed at the time of notification of postponement and which County had in writing authorized Contractor to perform prior to issuing a Notice to Proceed.
  - d. Should the Contractor terminate the Contract as a result of a notice of postponement, County shall have the authority to award the Contract to the next lowest responsive responsible bidder.
- 25. The Bidder to whom Contract is awarded shall execute and submit the following documents by 5:00 p.m. of the **SEVENTH** (7<sup>TH</sup>) calendar day following the date of the Notice of Award. Failure to properly and timely submit these documents entitles County to reject the bid as non-responsive.
  - a. Agreement: To be executed by successful Bidder. Submit four (4) copies, each bearing an original signature.
  - b. Escrow of Bid Documentation: This must include all required documentation. See the document Escrow of Bid Documentation for more information.
  - c. Performance Bond (100%): On the form provided in the Contract Documents and fully executed as indicated on the form.

- d. Payment Bond (100%) (Contractor's Labor and Material Bond): On the form provided in the Contract Documents and fully executed as indicated on the form.
- e. Insurance Certificates and Endorsements as required.
- f. Workers' Compensation Certification.
- g. Prevailing Wage and Related Labor Requirements Certification.
- h. Hazardous Materials Certification.
- i. Contractor's Safety Plan specifically adapted for the Project.
- j. Executed "PROJECT STABILIZATION/COMMUNITY BENEFITS AGREEMENT for the COUNTY OF ALAMEDA" Letter of Assent (For Projects over \$1,000,000 only)
- 26. Any Bid protest by any Bidder regarding any other Bid must be submitted in writing to the County's GSA–Office of Acquisition Policy, ATTN: Contract Compliance Officer, located at 1401 Lakeside Drive, 10th Floor, Oakland, CA 94612, Fax: (510) 208-9720, before 5:00 p.m. of the FIFTH (5th) business day following the date of issuance of the Notice of Intent to Award, not the date received by the Bidder. A Bid protest received after 5:00 p.m. is considered received as of the next business day.
  - a. The Bid protest must contain a complete statement of the reasons and facts for the protest.
  - b. The protest must refer to the specific portions of all documents that form the basis for the protest.
  - c. The protest must include the name, address, email address, fax number and telephone number of the person representing the protesting party.
  - d. The County Agency/Department will notify all bidders of the protest as soon as possible.
  - e. Upon receipt of written protest, GSA–Office of Acquisition Policy, or designee, will review and evaluate the protest and issue a written decision. The GSA–Office of Acquisition Policy, may, at its discretion, investigate the protest, obtain additional information, provide an opportunity to settle the protest by mutual agreement, and/or schedule a meeting(s) with the protesting Bidder and others (as appropriate) to discuss the protest. The decision on the bid protest will be issued at least ten (10) business days prior to the Board hearing or GSA award date.

- f. The decision will be communicated by e-mail, fax, or US Postal Service mail, and will inform the bidder whether or not the recommendation to the Board of Supervisors or GSA in the Notice of Intent to Award is going to change. A copy of the decision will be furnished to all Bidders affected by the decision. As used in this paragraph, a Bidder is affected by the decision on a Bid protest if a decision on the protest could have resulted in the Bidder not being the apparent successful Bidder on the Bid.
- The decision of the GSA-Office of Acquisition Policy on the bid protest may be g. appealed to the Auditor-Controller's Office of Contract Compliance & Reporting (OCCR) located at 1221 Oak St., Room 249, Oakland, CA 94612, Fax: (510) 272-6502 unless the OCCR determines that it has a conflict of interest in which case an alternate will be identified to hear the appeal and all steps to be taken by OCCR will be performed by the alternate. The Bidder whose Bid is the subject of the protest, all Bidders affected by the GSA-Office of Acquisition Policy's decision on the protest, and the protestor have the right to appeal if not satisfied with the GSA-Office of Acquisition Policy's decision. All appeals to the Auditor-Controller's OCCR shall be in writing and submitted within five (5) business days following the issuance of the decision by the GSA-Office of Acquisition Policy, not the date received by the Bidder. An appeal received after 5:00 p.m. is considered received as of the next business day. An appeal received after the FIFTH (5th) business day following the date of issuance of the decision by the GSA-Office of Acquisition Policy shall not be considered under any circumstances by the GSA or the Auditor-Controller OCCR.
- h. The appeal shall specify the decision being appealed and all the facts and circumstances relied upon in support of the appeal.
- i. In reviewing protest appeals, the OCCR will not re-judge the proposal(s). The appeal to the OCCR shall be limited to review of the procurement process to determine if the contracting department materially erred in following the Bid or, where appropriate, County contracting policies or other laws and regulations.
- j. The appeal to the OCCR also shall be limited to the grounds raised in the original protest and the decision by the GSA-Office of Acquisition Policy. As such, a Bidder is prohibited from stating new grounds for a Bid protest in its appeal. The Auditor-Controller (OCCR) shall only review the materials and conclusions reached by the GSA-Office of Acquisition Policy or department designee, and will determine whether to uphold or overturn the protest decision.
- k. The Auditor's Office may overturn the results of a bid process for ethical violations by Procurement staff, County Selection Committee members, subject matter experts, or any other County staff managing or participating in the competitive bid process, regardless of timing or the contents of a bid protest.

- 1. The decision of the Auditor-Controller's OCCR is the final step of the appeal process. A copy of the decision of the Auditor-Controller's OCCR will be furnished to the protestor, the Bidder whose Bid is the subject of the Bid protest, and all Bidders affected by the decision.
- m. The County will complete the Bid protest/appeal procedures set forth in this paragraph before a recommendation to award the Contract is considered by the Board of Supervisor or GSA.
- n. The procedures and time limits set forth in this paragraph are mandatory and are each Bidder's sole and exclusive remedy in the event of Bid Protest. A Bidder's failure to timely complete both the Bid protest and appeal procedures shall be deemed a failure to exhaust administrative remedies. Failure to exhaust administrative remedies, or failure to comply otherwise with these procedures, shall constitute a waiver of any right to further pursue the Bid protest, including filing a Government Code Claim or legal proceedings.
- 27. A responsive bid is a solicited bid that has been determined to be in conformance with the conditions, completion or delivery requirements, and specifications detailed in the solicitation for bid. Responsive bids are those submitted on time; contain complete information, and required submittals and/or supporting documentation.
- 28. A responsible bidder is defined by the California Public Contract Code section 1103 as "a bidder who has demonstrated the attribute of trustworthiness, as well as quality, fitness, capacity, and experience to satisfactorily perform this public works contract."
- 29. County reserves the right to reject any or all bids, including without limitation the right to reject any or all nonconforming, non-responsive, unbalanced, or conditional bids, to rebid, and to reject the bid of any bidder if County believes that it would not be in the best interest of County to make an award to that bidder, whether because the bid is not responsive or the bidder is unqualified or of doubtful financial ability or fails to meet any other pertinent standard or criteria established by County. County also reserves the right to waive inconsequential deviations not involving price, time, or changes in the Work. For purposes of this paragraph, an "unbalanced bid" is one having nominal prices for some work items and/or enhanced prices for other work items.
- 30. Discrepancies between written words and figures, or words and numerals, will be resolved in favor of the figures or numerals.
- 31. Prior to the award of Contract, County reserves the right to consider the responsibility of the Bidder. County may conduct investigations as County deems necessary to assist in the evaluation of any bid and to establish the responsibility, including, without limitation, qualifications and financial ability of Bidders, proposed subcontractors, suppliers, and

## **Alameda County General Services Agency**

PROJECT #2043

Highland Hospital – Koret Building Exhaust Fan and Waterproofing

other persons and organizations to perform and furnish the Work in accordance with the Contract Documents to County's satisfaction within the prescribed time.

END OF DOCUMENT

#### **DOCUMENT 00 22 19**

# SUPPLEMENTAL INSTRUCTIONS TO BIDDERS ENHANCED CONSTRUCTION OUTREACH PROGRAM (ECOP) (Not Required for Construction Projects 125K and under)

#### **GENERAL**

#### 1. PURPOSE

- 1.1 It is the express purpose of the Enhanced Construction Outreach Program (ECOP) to encourage the participation in County of Alameda construction projects with:
  - 1.1.1 Minority Owned Business Enterprise (MBE),
  - 1.1.2 Woman Owned Business Enterprise (WBE),
  - 1.1.3 Local Business Enterprise (LBE) and
  - 1.1.4 Small Local Business Enterprise (SLBE)

And to ensure that all contracting firms receive an equal opportunity to bid and receive work for this project. The ECOP encourages the inclusion of small businesses in this contract in accordance with Public Contract Code § 2002.

- 1.2 By submitting a bid, Bidders acknowledge and agree to all Document 00 22 19 provisions contained herein.
- 1.3 In the event of conflict between the terms of this Section 00 22 19 and the PROJECT STABILIZATION / COMMUNITY BENEFITS AGREEMENT for the COUNTY OF ALAMEDA, (Document 00 73 49B) the terms of the PROJECT STABILIZATION / COMMUNITY BENEFITS AGREEMENT for the COUTY OF ALAMEDA shall take priority.

#### 2. APPLICATION

- 2.1 The provisions outlined in this Section 00 22 19 apply to this contract for the construction of the above-referenced project. This project is funded solely with local dollars, and these provisions shall apply to all work performed under any contract awarded as a result of this competitive process.
- 2.2 To be considered for a contract award, any bidder who fails to meet ECOP goals identified herein shall be required to demonstrate to the satisfaction of the County that good faith efforts (GFEs) were made in accordance with the criteria listed in Section 7.9, GFE 1-9. Failure of the bidder to demonstrate a good faith effort may result in the bid being deemed non-responsive.

#### 3. **DEFINITIONS**

3.1 LOCAL BUSINESS ENTERPRISE (LBE)

ALAMEDA COUNTY GSA-CP

SUPPLEMENTARY INSTRUCTIONS TO BIDDERS with PSCBA ENHANCED CONSTRUCTION OUTREACH PROGRAM DOCUMENT 00 22 19

Page 1 of 16 Rev 10/10/19

3.1.1 For the purposes of this program, a Local Business Enterprise means a business that is a firm or dealer with fixed offices located in, and having a street address within the County and holds a valid business license issued by the County or a city within the County for at least 6 months prior to the date upon which a request for sealed bids or proposals is issued.

## 3.2 MINORITY OR WOMEN BUSINESS ENTERPRISE (MWBE)

- 3.2.1 For the purposes of this program, an MWBE is a Small Business Enterprise (SBE), as that term is defined by the State of California, that meets both of the following criteria:
  - 3.2.1.1 At least 51 percent of the business is owned by one or more minority persons or women, or in the case of any business whose stock is publicly held, at least 51 percent of the stock is owned by one or more minority persons or women; and
  - 3.2.1.2 Whose management and daily business operations are controlled by one or more minority persons or women.
- 3.2.2 An MWBE must be certified as such. Valid certification documentation must be provided with the bid response. The County will honor certifications from the following agencies:
  - 3.2.2.1 DBE/ACDBE certification from any California Unified Certification Program (CUCP) member agency, as long as it meets the County's MBE/WBE certification criteria.
  - 3.2.2.2 MBE certification from Western Regional Minority Supplier Development Council (WRMSDC)
  - 3.2.2.3 WBE certification from Women's Business Enterprise National Council (WBENC)
  - 3.2.2.4 SMBE/SWBE certification from the California Dept. of Transportation (CalTrans)
  - 3.2.2.5 MBE/WBE certification from the City of Los Angeles
  - 3.2.2.6 MBE/WBE certification from the California Public Utilities Commission (CPUC) through the Supplier Clearinghouse

## 3.3 <u>MINORITY PERSON</u>

- 3.3.1 Minority person, for purposes of this section, means Black Americans, Hispanic Americans, Native Americans, Asian-Pacific Americans, and Subcontinent Asian Americans.
- 3.4 SMALL BUSINESS ENTERPRISE (SBE)

ALAMEDA COUNTY GSA-CP

SUPPLEMENTARY INSTRUCTIONS TO BIDDERS with PSCBA ENHANCED CONSTRUCTION OUTREACH PROGRAM DOCUMENT 00 22 19

Page 2 of 16 Rev 10/10/19

- 3.4.1 For the purposes of this program, an SBE meets the current State of California definition of a small business, which is one that:
  - 3.4.1.1 Must be independently owned and operated;
  - 3.4.1.2 Cannot be dominant in its field of operation;
  - 3.4.1.3 Must have its principal office located in California;
  - 3.4.1.4 Must have its owners (or officers in the case of a corporation) domiciled in California; and
  - 3.4.1.5 Together with its affiliates, be either:
    - 3.4.1.5.1 A business with 100 or fewer employees, and an average annual gross receipts of \$15 million or less over the previous three tax years, or
    - 3.4.1.5.2 A manufacturer with 100 or fewer employees.
  - 3.4.1.6 An SBE must be certified or recognized as such by organizations whose certification is accepted by the California Department of General Services or by local agencies identified by the County of Alameda to have effective certification programs. Validation of the current certification by one of the following local agencies must be provided with the bid response:
    - 3.4.1.6.1 Alameda County Transportation Commission (Alameda CTC)
    - 3.4.1.6.2 California Department of General Services (DGS)
    - 3.4.1.6.3 Port of Oakland
    - 3.4.1.6.4 City of Oakland
    - 3.4.1.6.5 <u>and, when the State SBE definition is met,</u> Alameda County (SLEB certification)
- 3.5 <u>SMALL LOCAL BUSINESS ENTERPRISE (S/LBE)</u>
  - 3.5.1 For the purposes of this program, a Small Local Business Enterprise is defined by the County of Alameda and means a business that meets the SBE definition above, and is a firm or dealer with fixed offices located in, and having a street address within the County, and holds a valid business license issued by the County or a city within the County.

Page 3 of 16

#### 4. ENHANCED CONSTRUCTION OUTREACH PROGRAM (ECOP) GOALS

## 4.1 MBE PARTICIPATION SUBCONTRACTING – 15% GOAL

- 4.1.1 The MBE element of the ECOP program shall include subcontractors, manufacturers, suppliers, and truckers in calculating achievement of the MBE goal. Any contractor who fails to meet the MBE goals described herein must demonstrate to the satisfaction of the County of Alameda that a good faith effort was made to meet these goals in order to be considered for a contract award.
  - 4.1.1.1 The County shall further require that in order to be awarded a contract, a prime contractor must show that a good faith effort was made to provide at least 15% of the total contract amount to MBE subcontractors, manufacturers, suppliers, and truckers.
  - 4.1.1.2 The MBE goals must be achieved by the use of MBE subcontractors, manufacturers, suppliers, and/or truckers. If the Contractor plans to perform all the work with the Contractor's own forces, the goal will still apply and must be achieved by the use of suppliers, manufacturers, and/or truckers.
  - 4.1.1.3 A certified MBE prime contractor **may not** apply the percentage of the prime contractor's work toward meeting the goals as set forth above. An MBE subcontractor meeting the definition of both an MBE and a WBE **may not** be used to achieve both MBE and WBE required goals. The percentage of MBE firms utilized for the project described herein can only be applied to either MBE or WBE required goals. For purposes of meeting the MBE goals for this project, each participating MBE must be identified as an MBE.
  - 4.1.1.4 Prime contractors are strongly encouraged to sub-contract with S/LBE certified MBEs to meet the goals.

## 4.2 WBE PARTICIPATION SUBCONTRACTING – 5% GOAL

- 4.2.1 The WBE element of the ECOP program shall include subcontractors, manufacturers, suppliers, and truckers in calculating achievement of the WBE goal. Any contractor who fails to meet the WBE goals described herein must demonstrate to the satisfaction of the County of Alameda that a good faith effort was made to meet these goals in order to be considered for a contract award.
  - 4.2.1.1 The County shall further require that in order to be awarded a contract; a prime contractor must show that a good faith effort was made to provide at least 5% of the total contract amount to WBE subcontractors, manufacturers, suppliers, and/or truckers.

ALAMEDA COUNTY GSA-CP

Page 4 of 16 Rev 10/10/19

- 4.2.1.2 The WBE goals must be achieved by the use of subcontractors, manufacturers, suppliers, and/or truckers. If the Contractor plans to perform all the work with the Contractor's own forces, the goal will still apply and must be achieved by the use of manufacturers, suppliers, and/or truckers.
- 4.2.1.3 A certified WBE prime contractor **may not** apply the percentage of the prime contractor's work toward meeting the goals as set forth above. A WBE subcontractor meeting the definition of both an MBE and a WBE **may not** be used to achieve both the MBE and WBE required goals. The percentage of WBE firms utilized for the project described herein can only be applied to either MBE or WBE required goals. For purposes of meeting the WBE goals for this project, each participating WBE must be identified as a WBE.
- 4.2.1.4 Prime contractors are strongly encouraged to sub-contract with S/LBE certified WBEs to meet the goals.

#### 4.3 LBE PARTICIPATION GOALS –60% GOAL

- 4.3.1 The LBE element of the ECOP program shall include subcontractors, manufacturers, suppliers and/or truckers in calculating achievement of the LBE goal. Any contractor who fails to meet the LBE goals described herein must demonstrate to the satisfaction of the County of Alameda that a good faith effort was made to meet these goals in order to be considered for a contract award.
  - 4.3.1.1 The County shall further require that in order to be awarded a contract, a prime contractor must show that a good faith effort was made to provide at least 60% of the total contract amount to an LBE.
  - 4.3.1.2 The LBE prime contractor may count a portion, or all of its work towards meeting the goal and/or the LBE goal may be achieved by the use of subcontractors, manufacturers, suppliers, and/or truckers.

#### 4.4 S/LBE PARTICIPATION - 20% GOAL

- 4.4.1 The S/LBE element of the ECOP program shall include subcontractors, manufacturers, suppliers and/or truckers in calculating achievement of the S/LBE goal. Any contractor who fails to meet the S/LBE goals described herein must demonstrate to the satisfaction of the County of Alameda that a good faith effort was made to meet these goals in order to be considered for a contract award.
  - 4.4.1.1 The County shall further require that in order to be awarded a contract; a prime contractor must show that a good faith effort was made to provide at least 20% of the total contract amount to an S/LBE.

ALAMEDA COUNTY GSA-CP

Page 5 of 16

Rev 10/10/19

4.4.1.2 An S/LBE prime contractor may count a portion or all of its work towards meeting the goal and/or the S/LBE goal may be achieved by the use of subcontractors, manufacturers, suppliers, and/or truckers. For purposes of meeting this goal, the 20% S/LBE participation may also be counted toward achieving the 60% LBE participation goal.

## 5. SMALL BUSINESS ENTERPRISE 5% BID PREFERENCE

5.1 Prime contractors who are certified small local businesses (S/LBE) shall be eligible to receive a 5% bid preference (maximum financial value shall be \$150,000). Prime contractors that subcontract with certified small local businesses (S/LBE) (in accordance with the Public Contract Code 2002) for a minimum 40% of the contract amount will also be eligible to receive this 5% bid preference. This bid preference shall be applied by multiplying the total Base Bid amount by .95 to determine the bid amount for comparison purposes.

## 6. <u>HIRING OF LOCAL APPRENTICES, YOUTH, UNEMPLOYED AND UNDEREMPLOYED</u> RESIDENTS (FOR PROJECTS OVER \$125K)

6.1 The County of Alameda strongly encourages the hiring of local apprentices, youth, unemployed, and under-employed County residents to complete the work required for this project. Those firms that can demonstrate the ability and willingness to provide jobs required to complete this project to local apprentices, youth, unemployed and underemployed County residents should include such evidence in their bid response.

## 7. GOOD FAITH EFFORTS, ECOP PACKAGE SUBMITTALS, AND EVALUATION PROCEDURES

- 7.1 It is required that bidders exercise a good faith effort to secure the participation, as set forth in the specifications, of M/W/S/LBE subcontractors, manufacturers, suppliers and/or truckers on the project. Achievement of the ECOP goals shall constitute prima facie evidence of a Good Faith Effort (GFE). The failure of any bidder to make a good faith effort to achieve the specified participation of M/W/S/LBE subcontractors, manufacturers, suppliers and/or truckers may be grounds for determining that the bid is non-responsive.
- 7.2 In order to be considered for an award, responsible bidders must submit documentation to support the ECOP goals met and the GFEs made. The documentation submitted by each bidder shall be referred to as the ECOP Package.
- 7.3 ECOP Package shall include, but not limited to, ECOP Form 101A, 101B, 102A, 102B and 102C (provided separately as Excel fillable forms) and supporting documentation verifying ECOP goals met and GFEs made. The ECOP Package is to ge submitted no later than 2:00 p.m. on the second business day following bid opening.

ALAMEDA COUNTY GSA-CP

SUPPLEMENTARY INSTRUCTIONS TO BIDDERS with PSCBA ENHANCED CONSTRUCTION OUTREACH PROGRAM DOCUMENT 00 22 19

Page 6 of 16 Rev 10/10/19

- 7.3.1 The individual dollar amounts to be subcontracted to the M/W/S/LBE listed in the bidder's proposal will be listed on the S/LBE Participation Information ECOP Forms 101A and 101B and the M/WBE Subcontractor Participation Information ECOP Forms 102A, 102B.
- 7.3.2 ECOP Forms 101A, 101B, 102A 102B, 102C (Excel fillable forms), signature page, and supporting documentation shall be delivered to the County.
- 7.4 Upon request from the County, M/W/S/LBE subcontractors, manufacturers, suppliers and/or truckers who bid to a responsible bidder are required to provide the amounts of their bids to the County for the purposes of verification after bids are opened. This information shall be certified by a principal of the subcontracting firm. To the extent permitted by law, the information provided by the subcontractors, manufacturers, suppliers and/or truckers will be treated as proprietary, and will be solely for the use of County staff or its agents.
- 7.5 Each ECOP Package will be reviewed and evaluated by the County in a timely manner. Bidders must meet the ECOP goals *OR* make GFEs (see section 7.9) in order for their bid to be deemed responsive.
- 7.6 The ECOP Package must be complete, submitted on a flash drive, and contain legible supporting documents:
  - 7.6.1 ECOP Forms 101A, 101B, 102A,102B, and 102C to be completed electronically and submitted on a flash drive along with the hard copy signature page and supporting documentation.
  - 7.6.2 Supporting certification documentation for the prime contractor and each subcontractor, manufacturer, supplier and/or trucker M/W/S/LBEs submitted in the order they are listed on the ECOP forms and **must be submitted as hardcopy.** 
    - 7.6.2.1 To be considered towards meeting the ECOP goals bidders must submit:
      - 7.6.2.1.1 Acceptable certifying documentation for the prime contractor and its subcontractors, manufacturers, suppliers and/or truckers, as applicable (for example, local business license with proof of issue and expiration date, certification letters with expiration dates).
      - 7.6.2.1.2 Upon request, evidence that manufacturers, suppliers, and/or truckers are providing goods or services to subcontractors (for example, letter of intent, agreement, etc.).
  - 7.6.3 Documents evidencing those good faith efforts that were made, submitted in the order listed in the table below with the corresponding item number (1-9) noted on each document.
  - 7.6.4 Upon request, evidence of M/W/S/LBE participation (copies of bids, agreements, etc.) for all listed subcontractors, manufacturers, suppliers, and/or truckers that are *not* directly contracting with them (for example, material suppliers to subcontractors).

ALAMEDA COUNTY GSA-CP

SUPPLEMENTARY INSTRUCTIONS TO BIDDERS with PSCBA ENHANCED CONSTRUCTION OUTREACH PROGRAM DOCUMENT 00 22 19

Page 7 of 16 Rev 10/10/19

7.7 The County reserves the right, as it may deem appropriate and necessary, to contact responsible bidders during the evaluation process for clarification and/or submission of additional ECOP Goals or GFE documentation.

## 7.8 <u>ECOP GOALS / GOOD FAITH EFFORTS REQUIRED</u>

Listed in the table (below) are examples of acceptable documentation to support a determination that ECOP goals have been met

	ECOP GOALS	EXAMPLES OF ACCEPTABLE DOCUMENTATION
1	60% Local Business Enterprise (LBE) LBE participation may consist of the Prime Contractor and Subcontractors and may count towards the LBE, SBE, MBE and/or WBE ECOP goals.	<ul> <li>Business license issued by the County of Alameda or a City within the County of Alameda and proof of date issued (which is at least 6 months prior to the date bids were solicited). OR</li> <li>Certification letter from an acceptable certifying agency showing a local address and issuance/expiration dates.</li> </ul>
2	20% Certified Small Business Enterprise (SBE) Certified SBEs must be Local (S/LBE) to be considered. S/LBE participation may consist of the Prime Contractor and Subcontractors and may count towards the LBE, SBE, MBE and/or WBE ECOP goals.  An SBE meets the LBE definition above and the current State definition of a small business that is <100 employees and <\$15 Million annual gross revenues (over the last three years).	<ul> <li>Same as LBE         <i>PLUS</i> <ul> <li>Current certification document or letter with SBE designation</li> </ul> </li> </ul>
3	15% Minority-Owned Business Enterprise (MBE)  Subcontractors  MBEs are defined per PCC 2000(e)(1), (e)(2), and (f) and are not required to be LBEs. An MWBE may count towards only MBE or WBE participation (not both); however, a local MBE may count towards both LBE and S/LBE ECOP goals.  An MBE is a minority-owned business certified by one of the agencies listed below. An MBE can also be an SBE or LBE for purposes of meeting the SBE or LBE subcontracting goals, but an MBE cannot also be considered a WBE.	Current certification document, letter, etc., with MBE designation

Page 8 of 16 Rev 10/10/19

## 4 5% Woman-Owned Business Enterprise (WBE) <u>Subcontractors</u>

WBEs are defined per PCC 2000(e)(1), (e)(2), and (f) and are not required to be LBEs. An MWBE may count towards only MBE or WBE participation (not both); however, a local WBE may count both towards the LBE and S/LBE ECOP goals.

A WBE is a woman-owned business certified by one of the agencies listed below. A WBE can also be an SBE or LBE for purposes of meeting the SBE or LBE subcontracting goals, but a WBE cannot also be considered an MBE.  Current certification document, letter, etc., with WBE designation

7.9 The <u>examples</u> of GFE Indicators listed in the table below and suggested samples and are not meant to be mandatory or exclusionary. Other documentation may be acceptable as long as it evidences a GFE.

	Required Good Faith Effort Indicators	Examples of Acceptable Documentation
1.	The bidder attended mandatory pre-solicitation or pre-bid meetings that were scheduled by the local agency to inform all bidders of the ECOP requirements for the project for which the contract will be awarded.	Copy of pre-bid meeting sign-in sheet     (which is e-mailed to attendees and     available on County Current Contracting     Opportunities website listed below). The     name of the firm must be listed. <a href="http://www.acgov.org/gsa_app/gsa/purchasing/bid">http://www.acgov.org/gsa_app/gsa/purchasing/bid</a> content/contractopportunities.jsp
2.	The bidder identified and selected specific items of the project for which the contract will be awarded to be performed by M/W/S/LBEs to provide an opportunity for participation by those enterprises.	Copy of advertisements, certified letters, successfully completed faxes and/or other notices to M/W/S/LBEs with selected specific items identified.
3.	The bidder advertised, not less than ten (10) calendar days before the date the bids are opened, in one or more local daily or weekly newspapers, trade association publications, minority or trade-oriented publications, or trade journals for M/W/S/LBEs that are interested in participating in the project.	<ul> <li>Copy of advertisements placed showing publication name and date, and dated receipts.</li> <li>Dated receipt with ad copy.</li> </ul>

Page 9 of 16

Rev 10/10/19

- 4. The bidder provided written notice of his or her interest in bidding on the contract to the number of M/W/S/LBEs required to be notified by the project specifications not less than ten (10) calendar days prior to the opening of bids. The bidder may utilize the list of certified local business enterprises in the online County Small Local Emerging Business (SLEB) Vendor Query System located at <a href="http://www.acgov.org/sleb\_query\_app/gsa/sleb\_/query/slebmenu.jsp">http://www.acgov.org/sleb\_query\_app/gsa/sleb\_/query/slebmenu.jsp</a>.

  The minimum number of M/W/S/LBE firms required to be notified is three (3) for each item of the project selected to be performed by an M/W/S/LBE, where an M/W/S/LBE
- Copy of dated notice, complete distribution list(s) and evidence of distribution (proof of faxes, e-mails sent, etc.)
- Undelivered faxes do not count toward the effort to meet the minimum requirement
- Trades and specialties, in addition to M/W/S/LBE designation, must be clearly identified to meet the minimum requirement by using certification letter or source documentation

5. The bidder followed up initial solicitations of interest by contacting the enterprises to determine with certainty whether the enterprises were interested in performing specific items of the project.

subcontractor has not been secured for that

- Successfully completed telephone log containing specific dates, name of caller, person contacted and comments (i.e., why not bidding, information sent to/date)
- 6. The bidder provided interested M/W/S/LBEs with information about the plans, specifications, and requirements for the selected subcontracting or material supply work.
- Copy of published advertisements, letters, successfully completed faxes, etc. with M/W/S/LBE name/contact information including the required information or directions on how to obtain it and the date the information was provided
- Agenda, meeting notes, etc. including specific topics discussed, M/W/S/LBE firm names and contact persons in attendance that received information, and the location and date information was provided
- 7. The bidder requested assistance from local and small business and minority and women community organizations; local and small, minority and women contractor groups, local, state, or federal M/W/S/LBE assistance offices, or other organizations that provide assistance in recruitment and placement of M/W/S/LBEs.
- Copy of dated written request and response (letter, successfully completed fax, e-mail, etc.)
- Or 2<sup>nd</sup> written request to follow-up, if needed. Phone log is not acceptable.

item.

Page 10 of 16 Rev 10/10/19

8.	The bidder negotiated in good faith with the M/W/S/LBEs and did not unjustifiably reject as unsatisfactory bids prepared by any M/W/S/LBEs as determined by GSA	•	Copies or list of all bids and a spreadsheet listing all bids with firm name, contact person, bid items(s), bid price, M/W/S/LBE classification, and comments re-selection or rejection M/W/S/LBE bids accepted and included in bid response
9.	Where applicable, the bidder advised and made efforts to assist interested M/W/S/LBEs in obtaining bonds, lines of credit, or insurance required by either the GSA or the contractor.	•	Copy of advertisements or other notices with specifics referencing willingness to assist M/W/S/LBEs  Agenda, meeting notes including presenter's name and title, specific topics discussed, handouts, etc., name of M/W/S/LBE firms in attendance, contact persons who received advice, location, and date advice was provided

7.10 The performance by a bidder of the GFE Indicators specified in the table above shall create a rebuttable presumption, affecting the burden of producing evidence, that a bidder has made a good faith effort to comply with the goals and requirements relating to participation by M/W/S/LBEs.

#### 8. JOINT VENTURES

8.1 Whenever a joint venture occurs involving either a prime or non-prime (for example, subcontractors, manufacturers, suppliers, and truckers) M/W/S/LBE firm at any level of contracting, trucking, manufacturing, or supplying, the prime contractor shall provide the County with a full account of the nature of ownership interests, the basis for creation of the joint venture, and the particular financial participation and administrative responsibilities of the interested parties. In evaluating the prime contractor's effort, the M/W/S/LBE percentage that is to be attributed to a joint venture shall be determined by multiplying the percentage of the total contract amount that is to be performed by the joint venture times the percentage of actual financial participation in the joint venture represented by the M/W/S/LBE business.

#### 9. NONDISCRIMINATION

9.1 The Contractor shall comply with the Americans with Disabilities Act and Title VII of the Civil Rights Act of 1964 and shall not, in regard to any position for which an employee or applicant for employment is qualified, discriminate against any employee or applicant for employment because of race, creed, color, disability, sex, sexual orientation, political affiliation, or by any other non-merit factors be otherwise subjected to discrimination. The Contractor shall apply the ECOP that ensures applicants are employed, and that employees are treated during employment without regard to their race, age, religion, Vietnam Era Veteran's status, political affiliation, or any other non-merit factors. Such action shall include, but not be limited to, the following: employment, upgrading, demotion or transfer; recruitment or recruitment advertising; layoff or termination; rates of pay or other terms of

ALAMEDA COUNTY GSA-CP

Page 11 of 16 Rev 10/10/19

- compensation; and selection for training, including apprenticeship. The Contractor agrees to post in conspicuous places, available to employees and applicants for employment, notices setting forth the provisions of this nondiscrimination clause.
- 9.2 Contractor shall, in all solicitations or advertisements for employees placed on behalf of the County, state that all qualified applicants will receive consideration for employment without regard to race, creed, color, disability, sex, sexual orientation or national origin, age, religion, Vietnam Era Veteran's status, political affiliation, or any other non-merit factors.

#### **SECTION II**

## CONTRACT COMPLIANCE REQUIREMENTS

#### 1. APPLICATION

1.1 The following provisions shall apply to all contracts subject ECOP.

#### 2. ALAMEDA COUNTY CONTRACT COMPLIANCE SYSTEM

- 2.1 Alameda County utilizes the Elation Systems contract compliance application as part of its commitment to assist contractors to comply with certain legal and contractual requirements. The Elation Systems, a secure web-based computer system, was implemented to monitor compliance and to track and report M/W/S/LBE participation in County contracts.
- 2.2 The prime contractor and all participating local and M/W/S/LBE subcontractors awarded contracts as a result of the bid process for this project are required to use the Elation System to submit ECOP information including, but not limited to, weekly certified payrolls, monthly progress payment reports and other information related to M/W/S/LBE participation. Use of the Elation System, support and training are available at no charge to prime and subcontractors participating in County contracts.
- 2.3 Upon contract award:
  - 2.3.1 The County will provide contractors and subcontractors participating in any contract awarded as a result of this bid process, a code that will allow them to register and use the Elation System free of charge.
  - 2.3.2 Contractors should schedule a representative from their office/company, along with each of their subcontractors, to attend Elation Systems training.
    - 2.3.2.1 Free multi-agency Elation Systems one-hour training sessions require reservations and are held monthly in the Pleasanton, California area.
- 2.4 It is the Contractor's responsibility to ensure that they and their subcontractors are registered and trained as required to utilize Elation Systems.

ALAMEDA COUNTY GSA-CP

Page 12 of 16 Rev 10/10/19

- 2.4.1 For systems support visit Elation Systems online at <a href="http://www.elationsys.com/">http://www.elationsys.com/</a> or contact them at (925) 924-0340.
- 2.4.2 If you have questions regarding the utilization of the Elation Systems, please contact the Project Manager.

#### 3. MEETINGS

3.1 After the award of the contract and prior to beginning work, the County may hold a pre-construction conference at which a representative of the Contractor and of each subcontractor must attend. As it becomes necessary during the course of the contract, the County may call meetings of the Contractor and pertinent subcontractors.

#### 4. INFORMATION AND RECORDS

- 4.1 For the purposes of determining compliance with this program, the Contractor shall provide the County with access to all records and documents that relate to M/W/S/LBE participation. To the extent permitted by applicable law, proprietary information will be safeguarded.
- 4.2 The Contractor must submit the following information to the County on Alameda County approved forms. All subcontractor submittals must be through the prime contractor.
  - 4.2.1 S/LBE Participation Information and M/WBE Subcontractor Participation Information, (ECOP Forms 101A,101B, 102A, and 102B provided separately as Excel fillable forms) submitted no later than 2:00 p.m. on the second business day following bid opening.

#### 5. SUBSTITUTION OF M/W/S/LBE FIRMS

5.1 Substitution of other **firms** (subcontractors at any level, manufacturers, suppliers and/or truckers) for those listed in the proposal on the sheet entitled M/WBE Subcontractor Participation Information or S/LBE Participation Information shall not be made without prior approval of the County, and shall be in accordance with State or Federal law where applicable.

#### **SECTION III**

#### NON-COMPLIANCE WITH ECOP

#### 1. APPLICATION

1.1 The following provisions shall apply to all contracts subject to ECOP.

#### 2. DETERMINATION OF NON-COMPLIANCE

2.1 During the performance of the contract, if the General Services Agency has reason to believe or finds that the Contractor has not met the ECOP requirements in the contract, the Director of the General Services Agency (or the Director's designee) shall hold a meeting with the Contractor for the purpose of determining

ALAMEDA COUNTY GSA-CP

SUPPLEMENTARY INSTRUCTIONS TO BIDDERS with PSCBA ENHANCED CONSTRUCTION OUTREACH PROGRAM DOCUMENT 00 22 19

Page 13 of 16 Rev 10/10/19

whether the Contractor is out of compliance. If after the meeting, the Contractor is found to be out of compliance, the Contractor will be notified of a public hearing. The public hearing may be held before the Board of Supervisors with a minimum five calendar-day notice given to the Contractor. If the Board of Supervisors finds that there has been a violation, the County will notify the Contractor in writing of the sanctions to be imposed by the Board.

#### 3. SANCTIONS

- 3.1 A finding at the public hearing that there has been a violation of the ECOP requirements of the contract shall be cause for the Board of Supervisors to impose any or all of the following sanctions:
  - 3.1.1 Withhold an additional ten percent (10%) of all further contract progress payments until the Contractor provides evidence satisfactory to the Board of Supervisors that the condition of noncompliance has been corrected.
  - 3.1.2 Suspend the contract until such time as the Contractor provides evidence satisfactory to the Board of Supervisors that the condition of noncompliance has been corrected.
  - 3.1.3 Terminate the contract and collect appropriate damages from the Contractor.
  - 3.1.4 Declare that the Contractor is not a responsible bidder, and is ineligible to make bids on future County contracts for a stated period of time or until the Contractor can demonstrate to the satisfaction of the Board of Supervisors that the violation has been corrected.

#### **SECTION IV**

### 1. OUTREACH

- 1.1 To promote the ECOP goals and assist contractors and subcontractors in their efforts to develop the relationships they may require to meet the ECOP goals for this project, and the County will
  - 1.1.1 E-mail the Notice to Bidders to vendors in the County Vendor Database and other sources. Advertise the project once a week for at least two consecutive weeks in a newspaper of general circulation in the county where the project is located, trade organizations and chambers of commerce, and plan rooms. Notice of this project will also be posted on the County Current Contracting Opportunities and Calendar of Events websites (see website URL addresses below).
  - 1.1.2 Incorporate a networking and informational component in the mandatory bid walk/site visit.
  - 1.1.3 Provide information about the project, the ECOP, and other current and upcoming projects at the bid conference/networking meeting.
  - 1.1.4 E-mail the list of attendees from the mandatory bid walk to each attendee when issuing the first Addendum for the Project and post the attendance and first Addendum on the Current Contracting Opportunities website.

ALAMEDA COUNTY GSA-CP

Page 14 of 16 Rev 10/10/19

#### 2. CONTRACTOR RESOURCES

The following sources may be contacted for assistance in soliciting M/W/S/LBE participation:

#### **Asian American Contractors Association**

Juliana Choy Sommer, President (415) 642-1818 www.aaca-sf.com

# Western Regional Minority Supplier Development Council (WRMSDC) – MBE certifications only

80 Swan Way, Suite 245 Oakland, CA 94621 (510) 686-2555

www.wrmsdc.org

# Women's Business Enterprise National Council (WBENC)

WBE certifications only - www.wbenc.org

# California Public Utilities Commission (CPUC) The Supplier Clearinghouse

MBE and WBE vendors and certifications - www.thesupplierclearinghouse.com

### **Alameda County Transportation Commission**

LBE and SLBE vendors and certifications - <a href="https://www.alamedactc.org/get-involved/contract-equity/">https://www.alamedactc.org/get-involved/contract-equity/</a>

Visit the following County of Alameda GSA websites for

CERTIFIED SMALL LOCAL VENDORS	http://www.acgov.org/sleb_query_app/gsa/sleb/query/slebres ultlist.jsp?smEmInd=C
CURRENT CONTRACT OPPORTUNITIES	http://www.acgov.org/gsa_app/gsa/purchasing/bid_content/contractopportunities.jsp
UPCOMING CONTRACT OPPORTUNITIES	http://www.acgov.org/gsa_app/gsa/purchasing/bid_content/fu_turecontractopportunities.jsp
CALENDAR OF EVENTS	http://www.acgov.org/calendar_app/DisplayListServlet?site =Internet&ag=GSA&ty=PUR
COUNTY OF ALAMEDA HOME PAGE	http://www.acgov.org/index.htm

#### **BIDDER INFORMATION AND ACCEPTANCE**

(Submit a hardcopy of this completed page along with all hardcopy ECOP supporting documentation.)

ALAMEDA COUNTY GSA-CP

SUPPLEMENTARY INSTRUCTIONS TO BIDDERS with PSCBA ENHANCED CONSTRUCTION OUTREACH PROGRAM DOCUMENT 00 22 19

Page 15 of 16 Rev 10/10/19

The undersigned has read and agrees to the Supplementary Instructions to Bidders – Enhanced Construction Outreach Program, Document 00 22 19 of the Bid packet and declares that the ECOP Forms 101A, 101B, 102A, 102B and 102C (Excel Fillable Forms provided separately) have been completed accurately by the Prime Firm submitting the bid.

Official Name of Bidder:			
Street Address Line 1:			
Street Address Line 2:			
City:	State:	Zip Code:	
Webpage:			
Type of Entity / Organizational Structure (check	cone):		
Corporation	Join	t Venture	
Limited Liability Partnership	Part	nership	
Limited Liability Corporation	Non	-Profit / Church	
Other:			
Jurisdiction of Organization Structure:			
Date of Organization Structure:			
Federal Tax Identification Number:			
Primary Contact Information:			
Name / Title:			
Telephone Number:			
E-mail Address:			
SIGNATURE:			
Name and Title of Signer:			
Dated this day of		20	

#### **DOCUMENT 00 31 19**

### **EXISTING CONDITIONS INFORMATION**

# 1. Summary

This document describes existing conditions at or near the Project, and use of information available regarding existing conditions. This document is **not** part of the Contract Documents. See General Conditions for definition(s) of terms used herein.

# 2. Reports and Information on Existing Conditions

- 2.1. Documents providing a general description of the Site and conditions of the Work may have been collected by County its consultants, contractors, and tenants. These documents may include previous contracts, contract specifications, tenant improvement contracts, as-built drawings, utility drawings, and information regarding underground facilities.
- 2.2. Information regarding existing conditions may be inspected at the County offices or the Construction Manager's offices, if any, and copies may be obtained at cost of reproduction and handling upon Bidder's agreement to pay for such copies. These reports, documents, and other information are **not** part of the Contract Documents.
- 2.3. Information regarding existing conditions may also be included in the Project Manual, but shall **not** be considered part of the Contract Documents.
- 2.4. The reports and other data or information regarding existing conditions and underground facilities at or contiguous to the Project are the following:
  - 2.4.1. Hazardous Material Reports described are <u>not</u> part of the Contract Documents. However, for Bidder's convenience, a copy of the surveys have been provided with the Project Manual. Reports are provided immediately following this document.
    - 2.4.1.1. Project: R1207159; Highland Hospital, 1411 East 31st Street, Oakland, CA EML ID: 2403939
    - 2.4.1.2. Project: R1197G03; Highland Hospital/1411 East 31st St./Oakland, CA., K. Bldg. EML ID: 2403934
    - 2.4.1.3. Project: R1197G03; Highland Hospital, 1411 East 31st Street, Oakland, CA EML ID: 2403937
    - 2.4.1.4. Project: R1207159; Highland Hospital/1411 East 31st St./Oakland, CA. EML ID: 2403936

ALAMEDA COUNTY GSA-CP

**EXISTING CONDITIONS INFORMATION** 

Highland Hospital – Koret Building Exhaust Fan and Waterproofing

#### 3. Use of Information

- 3.1. Information regarding existing conditions was obtained only for use of County and its consultants, contractors, and tenants for planning and design and is **not** part of the Contract Documents.
- 3.2. County does not warrant, and makes no representation regarding, the accuracy or thoroughness of any information regarding existing conditions. Bidder represents and agrees that in submitting a bid it is not relying on any information regarding existing conditions supplied by County.
- 3.3. Under no circumstances shall County be deemed to warrant or represent existing above-ground conditions, as-built conditions, or other actual conditions, verifiable by independent investigation. These conditions are verifiable by Contractor by the performance of its own independent investigation, which Contractor must perform as a condition to bidding, and Contractor should not and shall not rely on this information or any other information supplied by County regarding existing conditions.
- 3.4. Any information shown or indicated in the reports and other data supplied herein with respect to existing underground facilities at or contiguous to the Project may be based upon information and data furnished to County by the County's employees and/or consultants or builders of such underground facilities or others. County does not assume responsibility for the completeness of this information, and Bidder is solely responsible for any interpretation or conclusion drawn from this information.
- 3.5. County shall be responsible only for the general accuracy of information regarding underground facilities, and only for those underground facilities that are owned by County, and only where Bidder has conducted the independent investigation required of it pursuant to the Instructions to Bidders, and discrepancies are not apparent.

#### 4. Investigations/Site Examinations

- 4.1. Before submitting a Bid, each Bidder is responsible for conducting or obtaining any additional or supplementary examinations, investigations, explorations, tests, studies, and data concerning conditions (surface, subsurface, and underground facilities) at or contiguous to the Site or otherwise, that may affect cost, progress, performance, or furnishing of Work or that relate to any aspect of the means, methods, techniques, sequences, or procedures of construction to be employed by Bidder and safety precautions and programs incident thereto or that Bidder deems necessary to determine its Bid for performing and furnishing the Work in accordance with the time, price, and other terms and conditions of the Contract Documents.
- 4.2. On request, County will provide each Bidder access to the Site to conduct such examinations, investigations, explorations, tests, and studies, as each Bidder deems necessary for submission of a Bid. Bidders must fill all holes and clean up and restore the Site to its former condition upon completion of its explorations, investigations, tests,

ALAMEDA COUNTY GSA-CP

**EXISTING CONDITIONS INFORMATION** 

and studies. Such investigations and Site examinations may be performed during any and all Site visits indicated in the Notice to Bidders and only under the provisions of the Contract Documents, including, but not limited to, proof of insurance and obligation to indemnify against claims arising from such work, and County's prior approval.

#### **DOCUMENT 00 41 13**

# BID FORM – STIPULATED SUM (SINGLE-PRIME CONTRACT)

10:	The County of Alameda						
From:							
	(Proper Name of Bidder)						
to Bidd Supple have be to perfe	dersigned declares that the Contract Documents including Document 00 11 16, the Instructions to Bidders Documentary Instructions to Bidders – Construction Outreacter read and agrees and proposes to furnish all necessary orm and furnish all work in accordance with the terms a tents, including, without limitation, the Drawings and S.	cument 00 21 13 and the ch Program Document 00 22 19 cy labor, materials, and equipment and conditions of the Contract					
	PROJECT: Highland Hospital – Koret Building Exl	naust Fan and Waterproofing					
	ect" or "Contract") and will accept in full payment for the nount, all taxes included:	nat Work the following total lump					
		\$					
		\$					
		\$					
	dollars	\$					
TOTA	L BASE BID						

Descriptions of alternates are primarily scope definitions and do not necessarily detail the full range of materials and processes needed to complete the construction.

- 1. <u>Unit Prices.</u> The Bidder's Base Bid includes the following unit prices, which the Bidder must provide and County may, at its discretion, utilize in valuing additive and/or deductive change orders;
  - A. Exterior Maintenance Project: glass lite caulking/LF.
  - B. Exterior Maintenance Project: sealant at window perimeters/LF.

ALAMEDA COUNTY GSA-CP

BID FORM – STIPULATED SUM (SINGLE-PRIME CONTRACT) DOCUMENT 00 41 13 Rev 3/17/15

- C. Exterior Maintenance Project: window and metal spandrel panel joints.
- D. Exterior Maintenance Project: Elastomeric paint/SF.
- 2. The undersigned has reviewed the Work outlined in the Contract Documents and fully understands the scope of Work required in this Bid, understands the construction and project management function(s) is described in the Contract Documents, and that each Bidder who is awarded a contract shall be in fact a prime contractor, not a subcontractor, to County, and agrees that its Bid, if accepted by County, will be the basis for the Bidder to enter into a contract with County in accordance with the intent of the Contract Documents.
- 3. The undersigned has notified County in writing of any discrepancies or omissions or of any doubt, questions, or ambiguities about the meaning of any of the Contract Documents, and has contacted the Construction Manager before bid date to verify the issuance of any clarifying Addenda.
- 4. The undersigned agrees to commence work under this Contract on the date established in the Contract Documents and to complete all work within the time specified in the Contract Documents.
- 5. The liquidated damages clause of the General Conditions and Agreement is hereby acknowledged.
- 6. It is understood that County reserves the right to reject this bid and that the bid shall remain open to acceptance and is irrevocable for a period of ninety (90) days.
- 7. The following documents are attached hereto:
  - a. Bid Bond on Bid Security Form Document 00 43 13 or other security
  - b. Designated Subcontractors List Document 00 43 36
  - c. Site-Visit Certification Document 00 45 01, if a site visit was required
  - d. Non-Collusion Affidavit Document 00 45 13
  - e. Enhanced Construction Outreach Program Package (Due no later than 2:00 pm, second business day following bid opening as required by as required by Supplemental Instructions to Bidders Enhanced Construction Outreach Program Document 00 22 19
  - f. Completed Debarment Form, Document 00 52 13.1

Highland Hospital – Koret Building Exhaust Fan and Waterproofing

Receipt and acceptance of the following addenda is hereby acknowledged:

No, Dated	No, Dated
No, Dated	No, Dated
No, Dated	No, Dated
No, Dated	No, Dated

- 8. Bidder acknowledges that the license required for performance of the Work is an active California State Contractor's license with a "B" General Building Contractor classification and each Sub-Contractor posses a valid, current California contractor's license for the Project services for which it intends to perform.
- 9. The undersigned hereby certifies that Bidder is able to furnish labor that can work in harmony with all other elements of labor employed or to be employed on the Work.
- 10. The Bidder represents that it is competent, knowledgeable and has special skills with respect to the nature, extent, and inherent conditions of the Work to be performed. Bidder further acknowledges that there are certain peculiar and inherent conditions existent in the construction of the Work that may create, during the Work, unusual or peculiar unsafe conditions hazardous to persons and property.
- 11. Bidder expressly acknowledges that it is aware of such peculiar risks and that it has the skill and experience to foresee and to adopt protective measures to adequately and safely perform the Work with respect to such hazards.
- 12. Bidder expressly acknowledges that it is aware that if a false claim is knowingly submitted (as the terms "claim" and "knowingly" are defined in the California False Claims Act, Cal. Gov. Code, §12650 et seq.), County will be entitled to civil remedies set forth in the California False Claim Act. It may also be considered fraud and the Contractor may be subject to criminal prosecution.
- 13. The undersigned Bidder certifies that it is, at the time of bidding, and shall be throughout the period of the contract, licensed by the State of California to do the type of work required under the terms of the Contract Documents. Bidder further certifies that it is

ALAMEDA COUNTY GSA-CP

BID FORM – STIPULATED SUM (SINGLE-PRIME CONTRACT) DOCUMENT 00 41 13 Rev 3/17/15

Highland Hospital – Koret Building Exhaust Fan and Waterproofing

regularly engaged in the general class and type of work called for in the Contract Documents.

14. The undersigned Bidder certifies that it is not, at the time of bidding, on the California Department of General Services (DGS) list of persons determined to be engaged in investment activities in Iran or otherwise in violation of the Iran Contracting Act of 2010 (Public Contract Code Section 2200-2208).

Furthermore, Bidder hereby certifies to County that all representations, certifications, and statements made by Bidder, as set forth in this bid form, are true and correct and are made under penalty of perjury pursuant to the laws of California.

Dated this	day of			20
Name of Bidder				
Type of Organization				
Signed by				
Title of Signer				
Address of Bidder				
Taxpayer's Identification No	of Bidder			
Telephone Number				
Fax Number				
E-mail		Web page		
Contractor's License No(s):	No.:	Class:	Expiration Date:	
	No.:	Class:	Expiration Date:	
	No.:	Class:	Expiration Date:	
Department of Industrial Rel	lations Regist	ration Number	::	
If Bidder is a corporation, af	fix corporate	seal.		
Name of Corporation:				
			DID EODM	CTIDIII /

ALAMEDA COUNTY GSA-CP

BID FORM – STIPULATED SUM (SINGLE-PRIME CONTRACT) DOCUMENT 00 41 13 Rev 3/17/15

PROJECT #2043

**Alameda County, General Services Agency** Highland Hospital – Koret Building Exhaust Fan and Waterproofing

President:	 		
Secretary:			
Treasurer:			
Manager:			

**DOCUMENT 00 43 13** 

# **BID SECURITY FORM**

# (Bid Bond)

(Note: If Bidder is providing a bid bond as its bid security, Bidder must use this form, NOT a surety company form.)

KNOW ALL PERSONS BY THESE PRESENTS:	
That the undersigned, as	as Principal ("Principal"),
and	as Surety ("Surety"),
a corporation organized and existing under and by virtue and authorized to do business as a surety in the State of C unto the County of Alameda, State of California as Oblig	California, are held and firmly bound
	(\$)
lawful money of the United States of America, for the pa	syment of which sum well and truly to

lawful money of the United States of America, for the payment of which sum well and truly to be made, we, and each of us, bind ourselves, our heirs, executors, administrators, successors, and assigns, jointly and severally, firmly by these presents.

THE CONDITION OF THIS OBLIGATION IS SUCH that whereas the Principal has submitted a bid to County for all Work specifically described in the accompanying bid;

Now, therefore, if the Principal is awarded the Contract and, within the time and manner required under the Contract Documents, after the prescribed forms are presented to Principal for signature, enters into a written contract, in the prescribed form in accordance with the bid, and files two bonds, one guaranteeing faithful performance and the other guaranteeing payment for labor and materials as required by law, and meets all other conditions to the contract between the Principal and the Obligee becoming effective, or if the Principal shall fully reimburse and save harmless the Obligee from any damage sustained by the Obligee through failure of the Principal to enter into the written contract and to file the required performance and labor and material bonds, and to meet all other conditions to the Contract between the Principal and the Obligee becoming effective, then this obligation shall be null and void; otherwise, it shall be and remain in full force and effect. The full payment of the sum stated above shall be due immediately if Principal fails to execute the Contract within seven (7) days of the date of the County's Notice of Award to Principal.

Surety, for value received, hereby stipulates and agrees that no change, extension of time, alteration or addition to the terms of the Contract or the call for bids, or to the work to be performed thereunder, or the specifications accompanying the same, shall in any way affect its obligation under this bond, and it does hereby waive notice of any such change, extension of

**PROJECT # 2043** 

Highland Hospital – Koret Building Exhaust Fan and Waterproofing

time, alteration or addition to the terms of the Contract or the call for bids, or to the work, or to the specifications.

In the event suit is brought upon this bond by the Obligee and judgment is recovered, the Surety shall pay all costs incurred by the Obligee in such suit, including a reasonable attorneys' fee to be fixed by the Court.

If the County awards the bid, the security of unsuccessful bidder(s) shall be returned within sixty (60) days from the time the award is made. Unless otherwise required by law, no bidder may withdraw its bid for ninety (90) days after the date of the bid opening.

	ment has been duty executed by the Principal and Surety
(Affix Corporate Seal)	Principal
	By
(Affix Corporate Seal)	Surety
	By
	Name of California Agent of Surety
	Address of California Agent of Surety
	Telephone Number of California Agent of Surety

Bidder must attach Power of Attorney and Certificate of Authority for Surety and a Notarial Acknowledgment for all Surety's signatures. The California Department of Insurance must authorize the Surety to be an admitted Surety Insurer.

#### **DOCUMENT 00 43 36**

#### **DESIGNATED SUBCONTRACTORS LIST**

PROJECT: Highland Hospital – Koret Building Exhaust Fan and Waterproofing

Bidder must list hereinafter the name and location of each subcontractor who will be employed, and the kind of Work that each will perform if the Contract is awarded to the Bidder. Bidder acknowledges and agrees that under Public Contract Code section 4100, et seq., it must clearly set forth below the name and location of each subcontractor who will perform work or labor or render service to the Bidder in or about the construction of the Work in an amount in excess of one-half of one percent (1/2 of 1%) of Bidder's total Bid, and that as to any Work that Bidder fails to list, Bidder agrees to perform that portion itself or be subjected to penalty under applicable law.

In case more than one subcontractor is named for the same kind of Work, state the portion that each will perform. Vendors or suppliers of materials only do not need to be listed.

If further space is required for the list of proposed subcontractors, additional sheets showing the required information, as indicated below, shall be attached hereto and made a part of this document.

				Subcontractor Numbers.	's License
Name and City of Subcontractor (1)	Description of Work: Reference to Contract Items (1)	Prices Under Subcontract (1)	State of California Contractor's License (1)	Dept. of Industrial Relations Registration No. (DIR) (1)	Business License and City/Count y Issued (1)
	(D) 11				

(Bidder to attach additional sheet(s) if necessary)

(1) Submit this information with sealed bid.

**DOCUMENT 00 45 01** 

# **SITE-VISIT CERTIFICATION**

# For Projects Where A Site Visit Was Mandatory

	t Name)	Highland Hos	pitai – Korei	Building Exn	aust Fan and	waterproofing	
Check	whichever opt	ion applies:					
	conditions rel		ection and lab	oor. I fully un	derstand the f	y acquainted wi facilities, difficu	
	construction a	Work and becand all the labor. The labor and restrictions a	me fully acq Bidder's repr	uainted with the esentative full	ne conditions y understood	the facilities,	te of
Manage damage	er, and all of the, or omissions	fies the County neir respective of s, related to con presentative's v	officers, age aditions that o	nts, employees could have bee	, and consulta	•	n
•	under penalty d correct.	y of perjury und	der the laws	of the State of	California tha	at the foregoing	; is
Date:							
Proper	Name of Bidd	er:					
Signatu	ıre:						
Print N	ame:						
Title:							

**DOCUMENT 00 45 13** 

# NON-COLLUSION AFFIDAVIT Public Contract Code Section 7106

### TO BE EXECUTED BY BIDDER AND SUBMITTED WITH BID

STATE OF CALIFORNIA	)	-	
COUNTY OF ALAMEDA	) s )	SS.	
			being first
duly sworn deposes and says t			
of	, tl	he Bidder making the foregoi	ing Bid that the Bid is not
made in the interest of, or on b			
association, organization, or co			
the Bidder has not directly or			
sham bid, and has not directly		•	•
bidder or anyone else to put in		•	•
Bidder has not in any manner,	•		
conference with anyone to fix overhead, profit, or cost eleme		•	•
advantage against the County		1 '	•
contained in the bid are true;	•		
submitted his or her Bid price			•
information or data relative th	•		
partnership, company associat			
thereof to effectuate a collusiv			
I certify and declare ur all the foregoing information i			of the State of California that and correct.
Date:			
Proper Name of Bidder:			
Signature:			
Print Name:			
Title:			

(ATTACH NOTARIAL ACKNOWLEDGMENT FOR THE ABOVE SIGNATURE)

**Alameda County, General Services Agency** Highland Hospital – Koret Building Exhaust Fan and Waterproofing

# **DOCUMENT 00 45 26**

# **WORKERS' COMPENSATION CERTIFICATION**

PROJECT/C and	CONTRACT NO.: 2043 between County of Alameda (the "County" or the "Owner")  (the "Contractor" or the "Bidder")			
	ct" or the "Project").			
Labor Code	section 3700 in relevant part provides:			
	y employer except the State shall secure the payment of compensation in one or of the following ways:			
a.	By being insured against liability to pay compensation by one or more insurers duly authorized to write compensation insurance in this state.			
b.	By securing from the Director of Industrial Relations a certificate of consent to self-insure, which may be given upon furnishing proof satisfactory to the Director of Industrial Relations of ability to self-insure and to pay any compensation that may become due to his employees.			
be insured as accordance v	of the provisions of section 3700 of the Labor Code which require every employer to gainst liability for workers' compensation or to undertake self-insurance in with the provisions of that code, and I will comply with such provisions before g the performance of the Work of this Contract.			
Date:				
Proper Name	e of Contractor:			
Signature:				
Print Name:				
Title:				
Labor Code,	ce with Article 5 - commencing at section 1860, chapter 1, part 7, division 2 of the the above certificate must be signed and filed with the awarding body prior to any Work under this Contract.)			

Highland Hospital – Koret Building Exhaust Fan and Waterproofing

DOCUMENT 00 45 46.01

# PREVAILING WAGE AND RELATED LABOR REQUIREMENTS CERTIFICATION

	2043 between County of Alameda (the "County" or the "Owner")
and	(the "Contractor" or the "Bidder")
(the "Contract" or the "Project	t'').
regarding prevailing wages, be	form to the State of California Public Works Contract requirements enefits, on-site audits with 48-hours notice, payroll records, and ment requirements, for all Work on the above Project.
including the Elation Systems Hiring Program compliance of County's requirements, and sh and remain competent in the u County Contract Compliance	unty will use the Alameda County Contract Compliance System, Inc. program, to monitor contract labor compliance and Local conitoring. Contractor shall use these Compliance Systems to meet hall participate in training as directed by County in order to become use of the Compliance Systems. Costs associated with the Alameda System, including the Elation Systems, Inc. programs shall be not increase the cost of the Contract.
Date:	
Proper Name of Contractor:	
Signature:	
Print Name:	
Title:	

END OF DOCUMENT

ALAMEDA COUNTY GSA-CP

Highland Hospital – Koret Building Exhaust Fan and Waterproofing

# DOCUMENT 00 45 46.04

# **HAZARDOUS MATERIALS CERTIFICATION**

		2043 ("Contract" or "Project") between the County of ("Contractor")	f Alameda			
1.	polychlorinated biphen Environmental Protecti material, or any other r rules, or regulations ("I incorporated in any wa	fies that no Asbestos, or Asbestos-Containing Materia yl (PCB), or any material listed by the federal or state on Agency or federal or state health agencies as a hazanaterial defined as being hazardous under federal or st New Hazardous Material"), shall be furnished, installe y into the Project or in any tools, devices, clothing, or on of Contractor's work on the Project for County.	ardous ate laws, d, or			
2.		fies that it has instructed its employees with respect to azards, risks, and liabilities.	the above-			
3.	Asbestos and/or asbestos-containing material shall be defined as all items containing detectable amounts of, but not limited to, chrysotile, crocidolite, amosite, anthophyllite, tremolite, and actinolite.					
4.	Any disputes involving the question of whether or not material is New Hazardous Material shall be settled by electron microscopy or other appropriate and recognized testing procedure, at the County's determination. The costs of any such tests shall be paid by Contractor if the material is found to be New Hazardous Material.					
5.	All Work or materials found to be New Hazardous Material or Work or material installed with equipment containing "New Hazardous Material" will be immediately rejected and this Work will be removed at Contractor's expense at no additional cost to the County.					
6.	Contractor has read and understood the document Hazardous Materials Procedures & Requirements, and shall comply with all the provisions outlined therein.					
Date:	-					
Proper	Name of Contractor:					
Signatu	ıre: _					
Print N	Jame:					
Title:	-	END OF DOCUMENT				

DOCUMENT 00 45 46.06

# **IMPORTED MATERIALS CERTIFICATION**

PROJECT/CONT	RACT NO.: <u>2043</u> between Count	•	,
"Project").			
deliver and/or sup Fill shall satisfy al to the statutes and	executed by the Contractor and by the ply any soils, aggregate, or related requirements of any environment guidelines of the California Environment ("CEQA").	d materials ("Fill") to the ntal review of the Project	Project Site. All performed pursuant
harmless the Counvolunteers pursuan	ent permitted by California law, on ty, its agents, representatives, off that to the indemnification provision im(s) connected with providing, of	icers, consultants, emplons in the Contract Docum	yees, trustees, and nents for, without
Certification of:	☐ Delivery Firm/Transporter	□ Supplier	☐ Manufacturer
	□ Wholesaler	□ Broker	□ Retailer
	□ Distributor	□ Other	
Type of Entity	<ul><li>□ Corporation</li><li>□ Limited Partnership</li><li>□ Sole Proprietorship</li></ul>	<ul><li>☐ General Partnership</li><li>☐ Limited Liability Co</li><li>☐ Other</li></ul>	ompany
Name of firm ("Fi	rm"):		
Mailing address: _			
Addresses of branch	ch office used for this Project:		
If subsidiary, name	e and address of parent company:		

By my signature below, I hereby certify that I am aware of section 25260 of the Health and Safety Code and the sections referenced therein regarding the definition of hazardous material. I further certify on behalf of the Firm that all soils, aggregates, or related materials provided, delivered, and/or supplied or that will be provided, delivered, and/or supplied by this Firm to the Project Site are free of any and all hazardous material as defined in section 25260 of the Health

Title:

**PROJECT # 2043** 

Highland Hospital – Koret Building Exhaust Fan and Waterproofing

and Safety Code. I further certify that I am authorized to make this certification on behalf of the Firm.

Date:

Proper Name of Contractor:

Signature:

Print Name:

**DOCUMENT 00 51 00** 

# **NOTICE OF AWARD**

Dated:			20			
То:						
То:	(Contractor)					
10.	(Address)					
From:	The County of	Alameda				
	PROJECT:				(Project	Name)
("Proje	ect" or "Contrac	et").				
<u>ONE</u> :	by action of the	County's Boa	ard of Supervisors	on	the	
Superv		pursuant	to a delegation of	ununomy by the ex	salley 5 Boa	
				Dolla		
	. ,		,	except Drawings) a red separately or other		

f Additional copies are available at cost of reproduction.

The Bidder to whom Contract is awarded shall execute and submit the following documents by 5:00 p.m. of the **SEVENTH** (7<sup>TH</sup>) Calendar day following the date of the Notice of Award. Failure to properly and timely submit these documents entitles County to reject the bid as nonresponsive.

- Agreement: To be executed by successful Bidder. Submit four (4) copies, each a. bearing an original signature.
- Escrow of Bid Documentation: This must include all required documentation. b. See the document Escrow of Bid Documentation for more information.
- Performance Bond (100%): On the form provided in the Contract Documents and c. fully executed as indicated on the form.
- Payment Bond (100%) (Contractor's Labor and Material Payment Bond): On the d. form provided in the Contract Documents and fully executed as indicated on the form.

- e. Insurance Certificates and Endorsements as required.
- f. Workers' Compensation Certification.
- g. Prevailing Wage and Related Labor Requirements Certification.
- h. Hazardous Materials Certification.
- i. Imported Materials Certification.
- j. Completed, signed Document 00 52 13.1 Debarment And Suspension Certification Form

Failure to comply with these conditions within the time specified will entitle County to consider your bid abandoned, to annul this Notice of Award, and to declare your Bid Security forfeited, as well as any other rights the County may have against the Contractor.

After you comply with those conditions, County will return to you one fully signed counterpart of the Agreement.

The Cour	nty of Alam	eda	
BY:			
NAME:			
TITLE:			

**Alameda County, General Services Agency** Highland Hospital – Koret Building Exhaust Fan and Waterproofing

DOCUMENT 00 51 13

# NOTICE OF INTENT TO AWARD

1,01101	01 11(121(1 1 0 11)(11112	
[DATE]		
[NAME] [COMPANY] [ADDRESS] [CITY, STATE ZIP]		
SUBJECT: INTENT TO AWARD A	A CONTRACT FOR PROJECT	7 NO
Thank you for your participation and interevaluation and acceptance of the bids sul Supervisors that the contract for project [CONTRACTOR].	bmitted, GSA will be recommend	ing to its Board of
GSA's recommendation to accept and averaged Tuesday, [DATE]. Upon Board approver review. A signed contract will be issued alone County is unable to pay for goods and services	al a draft contract will be submitteng with an Agreement Form and Notice	d to the awardee for
Below is a summary of all bids/proposal	s received for this project.	
Bidder	Location	Bid Price*
(*) – A 5% bid preference (stipulated in the evaluated at \$ which male		
Please refer to Document 00 21 13 - Inst	ructions to Bidders for additional	information.
For information on other contracting opp http://www.acgov.org/gsa/purchasing/bio Upcoming Contracting Opportunities and	<u>d_content/FutureContractOpportu</u>	
http://www.acgov.org/gsa/purchasing/bio	<pre>d_content/ContractOpportunities.j</pre>	sp for Current
Contracting Opportunities.	The County of Alameda	
	BY:	
	NAME:	

END OF DOCUMENT

TITLE:

**DOCUMENT 00 52 13** 

# AGREEMENT FORM – STIPULATED SUM (SINGLE-PRIME CONTRACT)

THIS	S AGREEMENT IS MADE AND I	ENTERED INTO THIS $\_\_$	DAY OF
	, 20, by and between the Cou	unty of Alameda ("County	") and
	<u> </u>	("Contractor"	') ("Agreement").
	NESSETH: That the parties heretoents do covenant and agree with each	•	d and agreed, and by these
1.	The Work: Contractor agrees to and material necessary to perform work of		
	PROJECT:		(Project Name)
	("Project" or "Contract" or "Wo	ork")	

for which the Drawings and Specifications are identified by the signature of the parties to this Agreement. It is understood and agreed that the Work shall be performed and completed as required in the Contract Documents including, without limitation, the Drawings and Specifications, under the direction and supervision of, and subject to, the approval of County or its authorized representative.

- 2. The Contract Documents: The complete Contract consists of all Contract Documents as defined in the General Conditions and incorporated herein by this reference. Any and all obligations of the County and Contractor are fully set forth and described in the Contract Documents. All Contract Documents are intended to cooperate so that any Work called for in one and not mentioned in the other or vice versa is to be executed the same as if mentioned in all Contract Documents.
- 3. Interpretation of Contract Documents: Should any question arise concerning the intent or meaning of Contract Documents, including the Drawings or Specifications, the question shall be submitted to the County for interpretation. If a conflict exists in the Contract Documents, modifications, beginning with the most recent, shall control over this Agreement, which shall control over the Special Conditions, which shall control over the General Conditions, which shall control over the remaining Division 00 documents, which shall control over Division 01 Documents, which shall control over Division 02 through Division 49 documents, which shall control over figured dimensions, which shall control over large-scale drawings, which shall control over small-scale drawings. In no case shall a document calling for lower quality and/or quantity material or workmanship control. The decision of County in the matter shall be final.

- **4. Time For Completion**: It is hereby understood and agreed that the work under this contract shall be completed within one hundred and forty (210) consecutive calendar days ("Contract Time") from the date specified in the County's Notice to Proceed.
- 5. Completion-Extension Of Time: Should the Contractor fail to complete this Contract, and the Work provided herein, within the time fixed for completion, due allowance being made for the contingencies provided for herein, the Contractor shall become liable to the County for all loss and damage that the County may suffer on account thereof. The Contractor shall coordinate its work with the Work of all other contractors. County shall not be liable for delays resulting from Contractor's failure to coordinate its Work with other contractors in a manner that will allow timely completion of Contractor's Work. Contractor shall be liable for delays to other contractors caused by Contractor's failure to coordinate its Work with the work of other contractors.
- 6. Liquidated Damages: Time is of the essence for all work under this Agreement. It is hereby understood and agreed that it is and will be difficult and/or impossible to ascertain and determine the actual damage that County will sustain in the event of and by reason of Contractor's delay; therefore, Contractor agrees that it shall pay to the County the sum of five thousand (\$5,000) per day as liquidated damages for each and every day's delay beyond the time herein prescribed in finishing the Work. It is hereby understood and agreed that this amount is not a penalty.

In the event any portion of the liquidated damages is not paid to County, County may deduct that amount from any money due or that may become due the Contractor under this Agreement. County's right to assess liquidated damages is as indicated herein and in the General Conditions.

The time during which the Contract is delayed for cause as hereinafter specified may extend the time of completion for a reasonable time as County may grant. This provision does not exclude the recovery of damages for delay by either party under other provisions in the Contract Documents.

7. Indemnity: To the fullest extent permitted by law (including, without limitation, California Civil Code Section 2782), Contractor shall defend (with legal counsel reasonably acceptable to the County), indemnify and hold harmless County and its officers, agents, departments, officials, representatives and employees (collectively "Indemnitees") from and against any and all claims, loss, cost, damage, injury (including, without limitation, injury to or death of an employee of Contractor or its Subcontractors), expense and liability of every kind, nature and description (including, without limitation, incidental and consequential damages, court costs, attorneys' fees, litigation expenses and fees of expert consultants or expert witnesses incurred in connection therewith and costs of investigation) which arises out of or is in any way connected to the performance of this agreement (collectively "Liabilities") except where such Liabilities are caused solely by the negligence or willful misconduct of any indemnitee. The County may participate in the defense of any such claim without relieving Contractor of any obligation hereunder. This

indemnification, defense, and hold harmless obligation includes any failure or alleged failure by Contractor to comply with any provision of law or the Contract Documents, including, without limitation, any stop notice actions, or liens by the California Department of Labor Standards Enforcement. This indemnity obligation shall be for the full amount of all damage to County, including defense costs, and shall not be limited by any insurance limits.

- 7.1 Contractor shall defend (with legal counsel reasonably acceptable to the County), indemnify and hold harmless the Indemnitees from all loss, cost, damage, expense, liability or claims, in law or in equity, including attorneys' fees, court costs, litigation expenses and fees of expert consultants or expert witnesses, that may at any time arise for any infringement of the patent rights, copyright, trade secret, trade name, trademark, service mark or any other proprietary right of any person or persons in consequence of the use by County, or any of the other Indemnitees, of articles or Services to be supplied in the performance of this Agreement.
- 7.2 Contractor shall place in its subcontracting agreements and cause its Subcontractors to agree to indemnities and insurance obligations in favor of County and other Indemnitees in the exact form and substance of those contained in this Agreement. Contractor shall require all subcontractors to comply with all indemnification and insurance requirements of this agreement, including, without limitation, Exhibit C. Contractor shall verify subcontractor's compliance.
- 8. Loss Or Damage: County and its authorized representatives shall not in any way or manner be answerable or suffer loss, damage, expense, or liability for any loss or damage that may happen to the Work, or any part thereof, or in or about the same during its construction and before acceptance, and the Contractor shall assume all liabilities of every kind or nature arising from the Work, either by accident, negligence, theft, vandalism, or any cause whatever; and shall hold County and its authorized representatives harmless from all liability of every kind and nature arising from accident, negligence, or any cause whatever.
- **9. Insurance and Bonds**: Contractor shall provide all required certificates of insurance, and payment and performance bonds as evidence thereof.
- **10. Prosecution of Work**: If the Contractor should neglect to prosecute the Work properly or fail to perform any provisions of this contract, County, may, pursuant to the General Conditions and without prejudice to any other remedy it may have, make good such deficiencies and may deduct the cost thereof from the payment then or thereafter due the Contractor.
- 11. Authority of Architect: Contractor hereby acknowledges that the Architect has authority to approve and/or stop Work if the Contractor's Work does not comply with the requirements of the Contract Documents and all applicable laws. The Contractor shall be liable for any delay caused by its non-compliant Work.
- **12. Assignment of Contract**: Neither the Contract, nor any part thereof, nor any moneys due or to become due thereunder, may be assigned by the Contractor without the written approval of the County, nor without the written consent of the Surety on the Contractor's

Performance Bond (the "Surety"), unless the Surety has waived in writing its right to notice of assignment.

- 13. Classification Of Contractor's License: Contractor hereby acknowledges that it currently holds valid Type B Contractor's license(s) issued by the State of California, Contractor's State Licensing Board, in accordance with division 3, chapter 9, of the Business and Professions Code and in the classification called for in the Contract Documents.
- 14. It is County policy to minimize the expenditure of County funds on goods and services produced by any entity which buys, sell, leases or distributes commodities and/or professional services to (1) the government of Burma; or (2) any entity organized under the laws of Burma; or (3) any entity which does business with any private or public entity located in Burma, or conducts operations in Burma. Contractors are urged to comply with the policy in making purchases and subcontracts. (ref. Alameda County, Cal., Adm. Code Title.4, §4.32.050(B),(F))
- 15. Payment of Prevailing Wages: The Contractor and all Subcontractors under the Contractor shall pay all workers on all Work performed pursuant to this Contract not less than the general prevailing rate of per diem wages and the general prevailing rate for holiday and overtime work as determined by the Director of the Department of Industrial Relations, State of California, for the type of work performed and the locality in which the work is to be performed within the boundaries of County, pursuant to sections 1770 et seq. of the California Labor Code.
- 16. Contract Price: In consideration of the foregoing covenants, promises, and agreements on the part of the Contractor, and the strict and literal fulfillment of each and every covenant, promise, and agreement, and as compensation agreed upon for the Work and construction, erection, and completion as aforesaid, County covenants, promises, and agrees that it will well and truly pay and cause to be paid to the Contractor in full, and as the full Contract Price and compensation for construction, erection, and completion of the Work hereinabove agreed to be performed by the Contractor, the following price:

	Dollar	S
(\$	<u>)</u> ,	

in lawful money of the United States, which sum is to be paid according to the schedule provided by the Contractor and accepted by County and subject to additions and deductions as provided in the Contract. This amount supersedes any previously stated and/or agreed to amount(s).

17. Severability: If any term, covenant, condition, or provision in any of the Contract Documents is held by a court of competent jurisdiction to be invalid, void or unenforceable, the remainder of the provisions in the Contract Documents shall remain in full force and effect and shall in no way be affected, impaired, or invalidated thereby.

# 18. The terms of the Agreement supersede the terms of the Purchase Order.

#### **DOCUMENT 00 52 13.1**

# COUNTY OF ALAMEDA DEBARMENT AND SUSPENSION CERTIFICATION FORM

The bidder, under penalty of perjury, certifies that, except as noted below, bidder, its Principal, and any named and unnamed subcontractor:

- Is not currently under suspension, debarment, voluntary exclusion, or determination of ineligibility by any federal agency;
- Has not been suspended, debarred, voluntarily excluded or determined ineligible by any federal agency within the past three years;
- Does not have a proposed debarment pending; and
- Has not been indicted, convicted, or had a civil judgment rendered against it by a court of
  competent jurisdiction in any matter involving fraud or official misconduct within the
  past three years.

II there	are any	exceptions i	to this cert	incation, i	nsert the e	exceptions in	i the following	g space

Exceptions will not necessary result in denial of award, but will be considered in determining bidder responsibility. For any exception noted above, indicate below to whom it applies, initiating agency, and dates of action.

Notes: Providing false information may result in criminal prosecution or administrative sanctions. The above certification is part of the Proposal. Signing this Proposal on the signature portion thereof shall also constitute signature of this Certification.

BIDDER:	
DD D ICIDA I	TOTAL C
PRINCIPAL:	TITLE:
SIGNATURE:	DATE:

ALAMEDA COUNTY GSA-TSD

DEBARMENT AND SUSPENSION CERTIFICATION FORM DOCUMENT 00 52 13.1 **Alameda County, General Services Agency** Highland Hospital – Koret Building Exhaust Fan and Waterproofing

DOCUMENT 00 55 00

# NOTICE TO PROCEED

Dated:	, 20	
TO: <u>(Co</u>	ntractor)	
ADDRESS		
PROJECT:		(Project Name)
	CONTRACT NO.:ctor ("Contract").	between County of Alameda
on under the C date of com You must s	a are notified that the Contract Time under the, 20 By that date, you a contract Documents. In accordance with the Appletion is, 20  ubmit the following documents by 5:00 p.m. on the date of this Notice to Proceed:	are to start performing your obligations greement executed by Contractor, the
a.	Contractor's preliminary schedule of const	truction.
b.	Contractor's preliminary schedule of value	es for all of the Work.
c.	Contractor's preliminary schedule of subm Product Data, and Samples submittals	
d.	Contractor's Safety Plan specifically adapt	ted for the Project.
e.	A complete subcontractors list, including t facsimile number, California State Contract and monetary value of all Subcontracts.	-

**PROJECT # 2043** 

Highland Hospital – Koret Building Exhaust Fan and Waterproofing

Thank you. We look forward to a very successful Project.

COUNTY OF ALAMEDA

#### **DOCUMENT 00 56 00**

# **ESCROW OF BID DOCUMENTATION**

# 1. Requirement to Escrow Bid Documentation

- a. Contractor shall submit, within **SEVEN** (7) calendar days after the date of the Notice of Award, one copy of all documentary information received or generated by Contractor in preparation of bid prices for this Contract, as specified herein. This material is referred to herein as "Escrow Bid Documentation." The Escrow Bid Documentation of the Contractor will be held in escrow for the duration of the Contract.
- b. Contractor agrees, as a condition of award of the Contract, that the Escrow Bid Documentation constitutes all written information used in the preparation of its bid, and that no other written bid preparation information shall be considered in resolving disputes or claims. Contractor also agrees that nothing in the Escrow Bid Documentation shall change or modify the terms or conditions of the Contract Documents.
- c. The Escrow Bid Documentation will not be opened by County except as indicated herein. The Escrow Bid Documentation will be used only for the resolution of change orders and claims disputes.
- d. Contractor's submission of the Escrow Bid Documentation, as with the bonds and insurance documents required, is considered an essential part of the Contract award. Should the Contractor fail to make the submission within the allowed time specified above, County may deem the Contractor to have failed to enter into the Contract, and the Contractor shall forfeit the amount of its bid security, accompanying the Contractor's bid, and County may award the Contract to the next lowest responsive responsible bidder.
- e. NO PAYMENTS WILL BE MADE, NOR WILL COUNTY ACCEPT PROPOSED CHANGE ORDERS UNTIL THE ABOVE REQUIRED INFORMATION IS SUBMITTED AND APPROVED.
- f. The Escrow Bid Documentation shall be submitted in person by an authorized representative of the Contractor to County.

# 2. Ownership of Escrow Bid Documentation

a. The Escrow Bid Documentation is, and shall always remain, the property of Contractor, subject to review by County, as provided herein.

b. Escrow Bid Documentation constitute trade secrets, not known outside Contractor's business, known only to a limited extent and only by a limited number of employees of Contractor, safeguarded while in Contractor's possession, extremely valuable to Contractor, and could be extremely valuable to Contractor's competitors by virtue of it reflecting Contractor's contemplated techniques of construction. Subject to the provisions herein, County agrees to safeguard the Escrow Bid Documentation, and all information contained therein, against disclosure to the fullest extent permitted by law.

#### 3. Format and Contents of Escrow Bid Documentation

- a. Contractor may submit Escrow Bid Documentation in its usual cost-estimating format; a standard format is not required. The Escrow Bid Documentation shall be submitted in the English language.
- b. Escrow Bid Documentation must clearly itemize the estimated costs of performing the work of each bid item contained in the bid schedule, separating bid items into sub-items as required to present a detailed cost estimate and allow a detailed cost review. The Escrow Bid Documentation shall include all subcontractor bids or quotes, supplier bids or quotes, quantity takeoffs, crews, equipment, calculations of rates of production and progress, copies of quotes from subcontractors and suppliers, and memoranda, narratives, add/deduct sheets, and all other information used by the Contractor to arrive at the prices contained in the bid proposal. Estimated costs should be broken down into Contractor's usual estimate categories such as direct labor, repair labor, equipment ownership and operation, expendable materials, permanent materials, and subcontract costs as appropriate. Plant and equipment and indirect costs should be detailed in the Contractor's usual format. The Contractor's allocation of indirect costs, contingencies, markup, and other items to each bid item shall be identified.
- c. All costs shall be identified. For bid items amounting to less than \$10,000, estimated unit costs are acceptable without a detailed cost estimate, provided that labor, equipment, materials, and subcontracts, as applicable, are included and provided that indirect costs, contingencies, and markup, as applicable, are allocated.
- d. Bid Documentation provided by County should not be included in the Escrow Bid Documentation unless needed to comply with the following requirements.

#### 4. Submittal of Escrow Bid Documentation

a. The Escrow Bid Documentation shall be submitted by the Contractor in a sealed container within **SEVEN** (7) calendar days after the date of the Notice of Award. The container shall be clearly marked on the outside with the Contractor's name, date of submittal, project name and the words "Escrow Bid Documentation –

Intended to be opened in the presence of Authorized Representatives of Both County and Contractor".

- b. By submitting Escrow Bid Documentation, Contractor represents that the material in the Escrow Bid Documentation constitutes all the documentary information used in preparation of the bid and that the Contractor has personally examined the contents of the Escrow Bid Documentation container and has found that the documents in the container are complete.
- c. If Contractor's proposal is based upon subcontracting any part of the work, each subcontractor whose total subcontract price exceeds 5 percent of the total contract price proposed by Contractor, shall provide separate Escrow Documents to be included with those of Contractor. Those documents shall be opened and examined in the same manner and at the same time as the examination described above for Contractor.
- d. If Contractor wishes to subcontract any portion of the Work after award, County retains the right to require Contractor to submit Escrow Documents for the Subcontractor before the subcontract is approved.

### 5. Storage, Examination and Final Disposition of Escrow Bid Documentation

- a. The Escrow Bid Documentation will be placed in escrow, for the life of the Contract, in a mutually agreeable institution. The cost of storage will be paid by Contractor for the duration of the project until final Contract payment. The storage facilities shall be the appropriate size for all the Escrow Bid Documentation and located conveniently to both County's and Contractor's offices.
- b. The Escrow Bid Documentation shall be examined by both County and Contractor, at any time deemed necessary by either County or Contractor, to assist in the negotiation of price adjustments and change orders or the settlement of disputes and claims. In the case of legal proceedings, Escrow Bid Documentation shall be used subject to the terms of an appropriate protective order if requested by Contractor and ordered by a court of competent jurisdiction. Examination of the Escrow Bid Documentation is subject to the following conditions:
  - (1) As trade secrets, the Escrow Bid Documentation is proprietary and confidential to the extent allowed by law.
  - (2) County and Contractor shall each designate, in writing to the other party **SEVEN** (7) calendar days prior to any examination, the names of representatives who are authorized to examine the Escrow Bid Documentation. No other person shall have access to the Escrow Bid Documentation.

- (3) Access to the documents may take place only in the presence of duly designated representatives of the County and Contractor. If Contractor fails to designate a representative or appear for joint examination on **SEVEN** (7) calendar days notice, then the County representative may examine the Escrow Bid Documents alone upon an additional **THREE** (3) calendar days notice if a representative of the Contractor does not appear at the time set.
- (4) If a subcontractor has submitted sealed information to be included in the Escrow Bid Documents, access to those documents may take place only in the presence of a duly designated representative of the County, Contractor and that subcontractor. If that subcontractor fails to designate a representative or appear for joint examination on **SEVEN** (7) calendar days notice, then the County representative and/or the Contractor may examine the Escrow Bid Documentation without that subcontractor present upon an additional **THREE** (3) calendar days notice if a representative of that subcontractor does not appear at the time set.
- c. The Escrow Bid Documentation will be returned to Contractor at such time as the Contract has been completed and final settlement has been achieved.

DOCUMENT 00 57 00

### **ESCROW AGREEMENT FOR** SECURITY DEPOSITS IN LIEU OF RETENTION

Public Contact Code Section 22300

	Escrow Agreement ("Escrow Agreement") is made and entered into this day of, 20, by and between County of Alameda, whose address is, California ("County"); and
busir	, whose place of , ("Contractor"); and
	, a state or federally ered bank in the state of California, whose place of business is located at, ("Escrow Agent").
For t	he consideration hereinafter set forth, County, Contractor, and Escrow Agent agree as ws:
1.	Pursuant to section 22300 of Public Contract Code of the State of California, which is hereby incorporated by reference, Contractor has the option to deposit securities with Escrow Agent as a substitute for retention earnings required to be withheld by County pursuant to the Construction Contract No entered into between County and Contractor for the [Name of Project], in the amount of dated,, 20, (the "Contract"). Alternatively, on written request of Contractor, County shall make payments of the retention earnings directly to Escrow Agent. When Contractor deposits the securities as a substitute for Contract earnings, Escrow Agent shall notify County within ten (10) calendar days of the deposit. The market value of the securities at the time of substitution and at all times from substitution until the termination of the Escrow Agreement shall be at least equal to the cash amount then required to be withheld as retention under terms of Contract between County and Contractor.  Securities shall be held in name of County of Alameda, and shall designate Contractor as beneficial owner.
2.	County shall make progress payments to Contractor for those funds which otherwise would be withheld from progress payments pursuant to Contract provisions, provided that Escrow Agent holds securities in form and amount specified above.
3.	When County makes payment of retention earned directly to Escrow Agent, Escrow Agent shall hold them for the benefit of Contractor until the time that the escrow created under this Escrow Agreement is terminated. Contractor may direct the investment of the

payments into securities. All terms and conditions of this Escrow Agreement and the rights and responsibilities of the Parties shall be equally applicable and binding when County pays Escrow Agent directly.

- 4. Contractor shall be responsible for paying all fees for the expenses incurred by Escrow Agent in administering the Escrow Account, and all expenses of County. These expenses and payment terms shall be determined by County, Contractor, and Escrow Agent.
- 5. Interest earned on securities or money market accounts held in escrow and all interest earned on that interest shall be for sole account of Contractor and shall be subject to withdrawal by Contractor at any time and from time to time without notice to County.
- 6. Contractor shall have the right to withdraw all or any part of the principal in the Escrow Account only by written notice to Escrow Agent accompanied by written authorization from County to Escrow Agent that County consents to withdrawal of amount sought to be withdrawn by Contractor.
- 7. County shall have the right to draw upon the securities and/or withdraw amounts from the Escrow Account in event of default by Contractor. Upon seven (7) days written notice to Escrow Agent from County of the default, if applicable, Escrow Agent shall immediately convert the securities to cash and shall distribute the cash as instructed by County.
- 8. Upon receipt of written notification from County certifying that the Contract is final and complete, and that Contractor has complied with all requirements and procedures applicable to the Contract, Escrow Agent shall release to Contractor all securities and interest on deposit less escrow fees and charges of the Escrow Account. The escrow shall be closed immediately upon disbursement of all monies and securities on deposit and payments of fees and charges.
- 9. Escrow Agent shall rely on written notifications from County and Contractor pursuant to Paragraphs 5 through 8, inclusive, of this Escrow Agreement and County and Contractor shall hold Escrow Agent harmless from Escrow Agent's release and disbursement of securities and interest as set forth above.
- 10. Names of persons who are authorized to give written notice or to receive written notice on behalf of County and on behalf of Contractor in connection with the foregoing, and exemplars of their respective signatures are as follows:

On behalf of County:	On behalf of Contractor:
Title	Title
Name	Name
Signature ALAMEDA COUNTY GSA-CP	Signature ESCROW AGREEMENT FOR SECURITY DEPOSITS IN LIEU OF RETENTION
Page 2 of 3	DOCUMENT 00 57 00

Address	Address
On behalf of Escrow Agent:	
Title	
Name	
Signature	
Address	
At the time of Escrow Accour Agent a fully executed of this Agreen	nt is opened, County and Contractor shall deliver to Escrownent.
IN WITNESS WHEREOF, the officers on the date first set forth above	e parties have executed this Agreement by their proper we.
On behalf of County:	On behalf of Contractor:
Title	Title
Name	Name
Signature	Signature
Address	Address
On behalf of Escrow Agent:	
Title	
Name	
Signature	
Address	

END OF DOCUMENT

ALAMEDA COUNTY GSA-CP

ESCROW AGREEMENT FOR SECURITY DEPOSITS IN LIEU OF RETENTION **DOCUMENT 00 57 00** 

DOCUMENT 00 61 13.13

## PERFORMANCE BOND FORM (100% of Contract Price)

(Note: Bidders must use this form, NOT a surety company form.)

KNOW ALL PERSONS BY THESE PRESENTS:

That WHEREAS, the Board of Supervisors of the County of Alameda ("County")
and, ("Principal)" have entered into a contract
for the furnishing of all materials and labor, services and transportation, necessary, convenient, and proper to perform the following project:
(Project Name)
("Project" or "Contract")
which Contract dated, 20, and all of the Contract Documents attached to or forming a part of the Contract, are hereby referred to and made a part hereof, and
And WHEREAS, said Principal is required under the terms of the Contract to furnish a bond for the faithful performance of the Contract;
NOW, THEREFORE, the Principal and
("Surety") are held and firmly bound unto the Board of County in the penal sum of DOLLARS (\$ ), lawful money of the United
States, for the payment of which sum well and truly to be made we bind ourselves, our
heirs, executors, administrators, successors, and assigns jointly and severally, firmly by
these presents, to perform all the work required to complete the Project and to pay to
County all damages County incurs as a result of the Principal's failure to perform all the
Work required to complete the Project.

The condition of the obligation is such that, if the above bounden Principal, his or its heirs, executors, administrators, successors, or assigns, shall in all things stand to and abide by, and well and truly keep and perform the covenants, conditions, and agreements in the Contract and any alteration thereof made as therein provided, on his or their part to be kept and performed at the time and in the intent and meaning, and shall indemnify and save harmless County, its trustees, officers and agents, as therein stipulated, then this obligation shall become null and void, otherwise it shall be and remain in full force and virtue.

And the Surety, for value received, hereby stipulates and agrees that no change, extension of time, alteration, or addition to the terms of the contract or to the work to be performed thereunder or the specifications accompanying the same shall in any way affect its

obligation on this bond, and it does hereby waive notice of any such change, extension of time, alteration, or addition to the terms of the Contract or to the work or to the specifications.

IN WITNESS WHEREOF, two (2) identical which shall for all purposes be deemed an or the Principal and Surety above named, on the	riginal thereof, have been duly executed by
(Affix Corporate Seal)	Principal
	By
	Surety
	By
	Name of California Agent of Surety
	Address of California Agent of Surety
Surety	Telephone Number of California Agent of

Bidder must attach a Notarial Acknowledgment for all Surety's signatures and a Power of Attorney and Certificate of Authority for Surety. The California Department of Insurance must authorize the Surety to be an admitted surety insurer.

END OF DOCUMENT

#### **Alameda County, General Services Agency**

Highland Hospital – Koret Building Exhaust Fan and Waterproofing

DOCUMENT 00 61 13.16

# PAYMENT BOND FORM Contractor's Labor & Material Payment Bond (100% of Contract Price)

(Note: Bidders must use this form, NOT a surety company form.)

KNOW ALL PERSONS BY THESE PRESENTS:

That WHEREAS, the Board of Supervisors	
	(Project Name)
("Project")	
which Contract dated, 20 to or forming a part of the Contract, are hereby reference.	_, and all of the Contract Documents attached erred to and made a part hereof, and
WHEREAS, pursuant to law and the Contra upon the performance of the work, to file a good at Contract is awarded in an amount equal to 100 per to secure the claims to which reference is made in 9564 of the Civil Code of California, and division 2	cent (100%) of the Maximum Contract Value, sections 9000, 9100, 9356 through 9560, and
NOW, THEREFORE, WE, the Principal and are held and firmly bound unto all laborers, material statutes in the sum of lawful money of the United States, being a sum not terms of Contract, for the payment of which sum wour heirs, executors, administrators, successors, or presents.	Dollars (\$),  t less than the total amount payable by the  yell and truly to be made, we bind ourselves,

The condition of this obligation is that if the Principal or any of his or its subcontractors, of the heirs, executors, administrators, successors, or assigns of any, all, or either of them shall fail to pay for any labor, materials, provisions, provender, or other supplies, used in, upon, for or about the performance of the work contracted to be done, or for any work or labor thereon of any kind, or for amounts due under the Unemployment Insurance Act with respect to such work or labor, that the Surety will pay the same in an amount not exceeding the amount herein above set forth, and also in case suit is brought upon this bond, will pay a reasonable attorney's fee to be awarded and fixed by the Court, and to be taxed as costs and to be included in the judgment therein rendered.

ALAMEDA COUNTY GSA-CP Page 1 of 2 Rev. 3/20/13

#### **Alameda County, General Services Agency**

Highland Hospital – Koret Building Exhaust Fan and Waterproofing

It is hereby expressly stipulated and agreed that this bond shall inure to the benefit of any and all persons, companies, and corporations entitled to file claims under sections 9000, 9100, 9356 through 9560, and 9564 of the Civil Code, so as to give a right of action to them or their assigns in any suit brought upon this bond.

Should the condition of this bond be fully performed, then this obligation shall become null and void; otherwise it shall be and remain in full force and affect.

And the Surety, for value received, hereby stipulates and agrees that no change, extension of time, alteration, or addition to the terms of Contract or the specifications accompanying the same shall in any manner affect its obligations on this bond, and it does hereby waive notice of any such change, extension, alteration, or addition.

Affix Corporate Seal)	
	Principal
	Ву
	Surety
Affix Corporate Seal)	Ву
	Name of California Agent of Surety
	Address of California Agent of Surety

Bidder must attach a Notarial Acknowledgment for all Surety's signatures and a Power of Attorney and Certificate of Authority for Surety. The California Department of Insurance must authorize the Surety to be an admitted surety insurer.

END OF DOCUMENT

ALAMEDA COUNTY GSA-CP Page 2 of 2 Rev. 3/20/13

### DOCUMENT 00 65 19.26

### FINAL SETTLEMENT CERTIFICATE FORM

("Agr	eement and Release") IS MA	REEMENT AND RELEASE OF CLAIMS DE AND ENTERED INTO THIS DAY nd between the County of Alameda ("County") and	
	, 20 oy us	("Contractor"), whose place of	
busine	ess is		
	RECITALS:		
1.		red into PROJECT/CONTRACT NO.: the County of Alameda, California.	
2.	The Work under the Contract has been completed.		
NOW	, THEREFORE, it is mutuall	y agreed between County and Contractor as follows:	
		<u>AGREEMENT</u>	
3.	Contractor will only be asse	essed liquidated damages as detailed below:	
	Original Contract Sum	\$	
	Modified Contract Sum	\$	
	Payment to Date	\$	
	Liquidated Damages	\$	
	Payment Due Contractor	\$	
4.	undisputed sum of \$ Cents) under the Co	reof, County shall forthwith pay to Contractor the(	
5.	outstanding claims in dispur work under the Contract, ex- continuing obligations desc in executing this Agreemen effective as a full, final and of action, obligations, costs	and hereby agrees that there are no unresolved or the against County arising from the performance of accept for the claims described in Paragraph 6 and with ribed in Paragraph 8. It is the intention of the parties that and Release that this Agreement and Release shall be general release of all claims, demands, actions, causes acceptable expenses, damages, losses and liabilities of all its respective agents, employees, inspectors,	

- assignees and transferees except for the Disputed Claim is set forth in Paragraph 6 and continuing obligations described in Paragraph 8 hereof.
- 6. The following claims are disputed (hereinafter, the "Disputed Claims") and are specifically excluded from the operation of this Agreement and Release:

Claim No. Description of Claim Amount of Claim Date Claim Submitted

#### [Insert information, including attachment if necessary]

- 7. Consistent with California Public Contract Code section 7100, Contractor hereby agrees that, in consideration of the payment set forth in Paragraph 4 hereof, Contractor hereby releases and forever discharges County, all its agents, employees, inspectors, assignees, and transferees from any and all liability, claims, demands, actions, or causes of action of whatever kind or nature arising out of or in any way concerned with the Work under the Contract.
- **8.** Guarantees and warranties for the Work, and any other continuing obligation of Contractor, shall remain in full force and effect as specified in the Contract Documents.
- 9. To the furthest extent permitted by California law, Contractor shall defend, indemnify, and hold harmless the County, its agents, representatives, officers, consultants, employees, trustees, and volunteers (the "indemnified parties") from any and all losses, liabilities, claims, suits, and actions of any kind, nature, and description, including, but not limited to, attorneys' fees and costs, directly or indirectly arising out of, connected with, or resulting from the performance of the Contract unless caused wholly by the sole negligence or willful misconduct of the indemnified parties.
- **10.** Contractor hereby waives the provisions of California Civil Code section 1542 which provides as follows:

A GENERAL RELEASE DOES NOT EXTEND TO CLAIMS WHICH THE CREDITOR DOES NOT KNOW OR SUSPECT TO EXIST IN HIS OR HER FAVOR AT THE TIME OF EXECUTING THE RELEASE, WHICH IF KNOWN BY HIM OR HER MUST HAVE MATERIALLY AFFECTED HIS OR HER SETTLEMENT WITH THE DEBTOR.

11. The provisions of this Agreement and Release are contractual in nature and not mere recitals and shall be considered independent and severable. If any such provision or any part thereof shall be at any time held invalid in whole or in part under any federal, state, county, municipal, or other law, ruling, or regulations, then such provision, or part thereof, shall remain in force and effect to the extent

permitted by law, and the remaining provisions of this Agreement and Release shall also remain in full force and effect, and shall be enforceable.

**12.** All rights of County shall survive completion of the Work or termination of Contract, and execution of this Release.

\* \* \* CAUTION: THIS IS A RELEASE - READ BEFORE EXECUTING \* \* \*

COUNTY OF ALAMEDA
TITLE:
NAME:
SIGNATURE:
CONTRACTOR
TITLE:
NAME:
SIGNATURE:

END OF DOCUMENT

### DOCUMENT 00 65 36

### **WARRANTY FORM**

		actor") hereby agrees that the _
Alameda ("County") for the fol	("Work" of Contractor) which Contractoring project:	ctor has installed for County of
DD O IE CE		(Project Name)
("Project" or "Contract")		
	nce with the requirements of the Contra e requirements of the Contract Docume	
workmanship or material togeth with such replacement within a defined in Public Contract Code	ir or replace any or all of such Work the ner with any other adjacent Work that ner period of year(s) e section 7107, subdivision (c), ordinary date of completion is	nay be displaced in connection from the date of completion as y wear and tear and unusual
reasonable period of time, as de notified in writing by County, t	's failure to comply with the above-mentermined by County, but not later than the undersigned authorizes the County that expense of the undersigned. The under	seven (7) days after being to proceed to have said defects
Date:		
Proper Name of Contractor:		
Signature:		
Print Name:		
Title:		
Representatives to be contacted	for service subject to terms of Contrac	et:
NAME:		
ADDRESS:		•
PHONE NO.:		

END OF DOCUMENT

### SECTION 00 72 13

### **GENERAL CONDITIONS**

#### **TABLE OF CONTENTS**

		<u>Page</u>
1.	CONTRACT TERMS AND DEFINITIONS	8
1.1	Definitions	8
1.2	Laws Concerning The Contract	11
1.3	No Oral Agreements	11
1.4	No Assignment	12
1.5	Notice And Service Thereof	12
1.6	No Waiver	12
1.7	Substitutions For Specified Items	12
1.8	Materials and Work	12
2.	COUNTY	14
2.1	Occupancy	14
3.	ARCHITECT	14
3.1	Role and Authority	14
3.2	Interpretations	14
3.3	Laws 14	
3.4	Communications	14
4.	CONSTRUCTION MANAGER	14
4.1	Role and Authority	14
4.2	Authority to Reject	15
4.3	If No Construction Manager	15
5.	INSPECTIONS AND TESTS	15
5.1	Tests and Inspections	15

5.2	Costs for After Hours and/or Off Site Inspections	15
6.	CONTRACTOR	16
6.1	Status of Contractor	16
6.2	Contractor's Supervision	16
6.3	Duty to Provide Fit Workers	17
6.4	Purchase of Materials and Equipment.	17
6.5	Documents On Work Site	17
6.6	Preservation of Records	18
6.7	Integration of Work	18
6.8	Obtaining of Permits and Licenses	19
6.9	Work to Comply With Applicable Laws and Regulations	19
6.10	Safety/Protection of Persons and Property	20
6.11	Working Evenings and Weekends	22
6.12	Badge Policy For Contractors	22
6.13	County Drug Policy - Drug Free Work Place	23
6.14	Cleaning Up	23
7.	SUBCONTRACTORS	23
7.1	Contractor Shall Provide Subcontractor Information	24
7.2	No Contractual Relationship Between County and Subcontractors	24
7.3	Contractor Binds Every Subcontractor by Terms of Contract	24
7.4	No Waiver of Obligations	24
7.5	Contractor to Familiarize Itself with Laws	24
7.6	Subcontractor Substitutions	24
7.7	Subcontractor Coordination	25
7.8	Subcontractor Relations	25
7.9	Assignment or Termination	25

8.	OTHER CONTRACTS/CONTRACTORS	25
8.1	County Right to Perform	25
8.2	Protection of Work	25
8.3	Coordination with Other Work	25
8.4	Measurement of Work Performed	25
8.5	Knowledge of Other Work	25
8.6	No Exclusive Occupancy of Site	26
9.	DRAWINGS AND SPECIFICATIONS	26
9.1	List of all Drawings	26
9.2	Technical and Trade Words	26
9.3	Trade Name or Trade Term	26
9.4	The Naming of any Material and/or Equipment Shall Mean Furnishing	26
9.5	Contract Documents are Complementary	26
9.6	Drawings and Specifications are Intended to Comply With All Laws	26
9.7	Plans, Drawings, Designs, Specifications are County Property	27
9.8	Order of Precedence	27
9.9	Resolution of Discrepancy or AmbiguityF	27
9.10	0 County Clarification	27
10.	CONTRACTOR'S SUBMITTALS AND SCHEDULES	27
10.1	1 Schedule of Work, Schedule of Submittals, and Schedule of Values	27
10.2	2 Monthly Progress Schedule(s)	30
10.3	3 Material Safety Data Sheets (MSDS)	30
11.	SITE ACCESS, CONDITIONS, AND REQUIREMENTS	30
11.1	1 Site Investigation	30
11.2	2 Soils Investigation Report	30
11.3	3 Access to Work	31

11.4 Layout and Field Engineering	31
11.5 Utilities	31
11.6 Sanitary Facilities	31
11.7 Surveys	31
11.8 Regional Notification Center	31
11.9 Existing Utility Lines	32
11.10 Notification	32
11.11 Hazardous Materials	33
11.12 No Signs	33
12. TRENCHES	33
12.1 Trenches Greater Than Five Feet	33
12.2 Excavation Safety	33
12.3 No Tort Liability of County	33
12.4 No Excavation Without Permits	33
12.5 Discovery of Hazardous Waste and/or Unusual Conditions	33
13. INSURANCE AND BONDS	32
13.1 Insurance	34
13.2 Contract Security - Bonds	36
14. WARRANTY/GUARANTEE/INDEMNITY	37
14.1 Warranty/Guarantee	37
14.2 Indemnity	37
15. <b>TIME</b>	39
15.1 Notice to Proceed	39
15.2 Computation of Time / Adverse Weather	39
15.3 Hours of Work	40
15.4 Progress and Completion	40

15.5	Expeditious Completion	40
16.	EXTENSIONS OF TIME – LIQUIDATED DAMAGES	40
16.1	Liquidated Damages	40
16.2	Excusable Delay	41
16.3	No Additional Compensation for Delays Within Contractor's Control	41
16.4	Float or Slack in the Schedule	42
17.	CHANGES IN THE WORK	42
17.1	No Changes Without Authorization.	42
17.2	Architect Authority to Order Minor Changes	42
17.3	Change Orders	43
17.4	Construction Change Directives	43
17.5	Force Account Directives	43
17.7	Proposed Change Order	44
17.6	Price Request	44
17.8	Format for Proposed Change	45
17.9	Change Order Certification	46
17.10	O Determination of Change Order Cost	46
17.1	1 Allowable Costs	46
17.1	2 Deductive Change Orders	47
17.13	3 Discounts, Rebates, and Refunds	48
17.1	4 Accounting Records	48
17.1	5 Notice Required	48
17.1	6 Applicability to Subcontractors	48
17.1	7 Alteration to Change Order Language	48
17.13	8 Failure of Contractor to Execute Change Order	48
18.	REQUEST FOR INFORMATION	48

18.1	Coordination of Contract Work	49
18.2	Reference Contract Documents	49
18.3	Contractor Responsible For Costs	49
19.	PAYMENTS	49
19.1	Contract Price	49
19.2	Applications for Progress Payments	49
19.3	Progress Payments	52
19.4	Decisions to Withhold Payment	53
19.5	Subcontractor Payments	55
20.	COMPLETION OF THE WORK	56
20.1	Completion	56
20.2	Close-Out Procedures.	56
21.	FINAL PAYMENT AND RETENTION	58
21.1	Final Payment	58
21.2	Prerequisites for Final Payment	59
21.3	Retention	59
21.4	Substitution of Securities	60
22.	UNCOVERING OF WORK	60
23.	NONCONFORMING WORK, CORRECTION OF WORK AND COUNTY'S RIGHT TO PERFORM WORK	
23.1	Nonconforming Work	60
23.2	Correction of Work	60
23.3	County's Right to Perform Work	61
24.	TERMINATION AND SUSPENSION	62
24.1	County's Right to Terminate Contractor for Cause	62
24.2	Termination of Contractor for Convenience	64

#### **PROJECT # 2043**

24.3	Emergency Termination of Public Contracts Act of 1949	64
25.	CLAIMS AND DISPUTES	65
25.1	Performance During Claim Process.	65
25.2	Definition of Claim	65
25.3	Claim Presentations	65
25.4	Claim Resolution	66
26.	LABOR, WAGE & HOUR, APPRENTICE, AND RELATED PROVISIONS	67
26.1	Wage Rates, Travel, and Subsistence	68
26.2	Hours of Work	69
26.3	Payroll Records	70
26.4	Apprentices	72
26.5	Non-Discrimination	73
26.6	Labor First Aid	73
27.	MISCELLANEOUS	73
27.1	Assignment of Antitrust Actions	73
27.2	Excise Taxes	74
27.3	Taxes	74
27.4	Shipments	74

#### **GENERAL CONDITIONS**

#### 1. CONTRACT TERMS AND DEFINITIONS

#### 1.1 Definitions

Wherever used in the Contract Documents, the following terms shall have the meanings indicated, which shall be applicable to both the singular and plural thereof:

- **1.1.1** Adverse Weather: Shall be only weather that satisfies all of the following conditions: (1) unusually severe precipitation, sleet, snow, hail, heat, or cold conditions in excess of the norm for the location and time of year it occurred, (2) unanticipated, and (3) at the Project.
- **1.1.2** Approval, Approved, and/or Accepted: Refer to written authorization, unless stated otherwise.
- **1.1.3** Architect: The individual, partnership, corporation, joint venture, or any combination thereof, named as Architect, who will have the rights and authority assigned to the Architect in the Contract Documents. The term Architect means the County's Architect on this Project or the Architect's authorized representative.
- **1.1.4** Architect's Supplemental Instruction: A document prepared by the Architect to provide supplemental instructions or interpretations or to order minor changes in the work not involving an adjustment in the Contract Amount or Contract Time.
- **1.1.5** Bidder: A contractor who intends to provide a bid to the County to perform the Work of this Contract.
- **1.1.6** Change Order: A written order to the Contractor authorizing an addition to, deletion from, or revision in the Work, and/or authorizing an adjustment in the Contract Price or Contract Time.
- **1.1.7** Construction Change Directive: A written order prepared and issued by the County, the Construction Manager, and/or the Architect and signed by the County and the Architect, directing a change in the Work.
- **1.1.8** Construction Manager: The individual, partnership, corporation, joint venture, or any combination thereof, or its authorized representative, named as such by the County. If no Construction Manager is used on the Project that is the subject of this contract, then all references to Construction Manager herein shall be read to refer to County.
- **1.1.9** Construction Schedule: The progress schedule of construction of the Project as provided by Contractor and approved by County.
- **1.1.10** Contract, Contract Documents: The Contract consists exclusively of the documents evidencing the agreement of the County and Contractor, identified as the Contract Documents. The Contract Documents consist of the following documents:
  - **1.1.10.1** Notice to Bidders
  - **1.1.10.2** Instructions to Bidders
  - **1.1.10.3** Supplementary Instructions to Bidders Construction Outreach Program

#### **Alameda County, General Services Agency**

Highland Hospital – Koret Building Exhaust Fan and Waterproofing

1.1.10.4	Bid Form
1.1.10.5	Bid Security Form
1.1.10.6	Designated Subcontractors List
1.1.10.7	Site-Visit Certification, if a site visit was required.
1.1.10.8	Non-Collusion Affidavit
1.1.10.9	Workers' Compensation Certification
1.1.10.10	Prevailing Wage and Related Labor Requirements Certification
1.1.10.11	Construction Outreach Program Certifications
1.1.10.12	Hazardous Materials Certification
1.1.10.13	Imported Materials Certification
1.1.10.14	Notice of Award
1.1.10.15	Agreement
1.1.10.16	Notice to Proceed
1.1.10.17	Escrow of Bid Documentation
1.1.10.18	Escrow Agreement for Security Deposits in Lieu of Retention
1.1.10.19	Performance Bond
1.1.10.20	Payment Bond (Contractor's Labor & Material Payment Bond)
1.1.10.21	General Conditions
1.1.10.22	Special Conditions
1.1.10.23	Hazardous Materials Procedures and Requirements
1.1.10.24	Divisions 01 through 49
1.1.10.25	All Plans, Technical Specifications, and Drawings
1.1.10.26	Any and all addenda to any of the above documents
1.1.10.27	Any and all change orders or written modifications to the above documents if approved in writing by the County.

- **1.1.11** Contract Price: The total monies payable to the Contractor under the terms and conditions of the Contract Documents.
- **1.1.12** Contract Time: The time period stated in the Agreement for the completion of the Work.
- **1.1.13** Contractor: The person or persons identified in the Agreement as contracting to perform the Work to be done under this Contract, or the legal representative of such a person or persons.
- **1.1.14** County: County of Alameda, acting through its Board of Supervisors or any of its authorized agents. The County may, at any time:

- 1.1.14.1 Direct the Contractor to communicate with or provide notice to the Construction Manager or the Architect on matters for which the Contract Documents indicate the Contractor will communicate with or provide notice to the County; and/or
- **1.1.14.2** Direct the Construction Manager or the Architect to communicate with or direct the Contractor on matters for which the Contract Documents indicate the County will communicate with or direct the Contractor.
- **1.1.15** Daily Job Report(s): Daily Project reports prepared by the Contractor's employee(s) who are present on Site, which shall include the information required herein.
  - **1.1.16** Day(s): Unless otherwise designated, day(s) means calendar day(s).
- **1.1.17** Drawings: (or "Plans") The graphic and pictorial portions of the Contract Documents showing the design, location, scope and dimensions of the work, generally including plans, elevations, sections, details, schedules, sequence of operation, and diagrams.
- **1.1.18** Force Account Directive: A process that may be used when the County and the Contractor cannot agree on a price for a specific portion of work or before the Contractor prepares prices for a specific portion of work and whereby the Contractor performs the work as indicated herein on a time and materials basis.
- **1.1.19** Premises: The real property owned by the County on which the Site is located.
- **1.1.20** Product(s): New material, machinery, components, equipment, fixtures, and systems forming the Work, including existing materials or components required and approved by the County for reuse.
- **1.1.21** Product Data: Illustrations, standard schedules, performance charts, instructions, brochures, diagrams, and other information furnished by the Contractor to illustrate a material, product, or system for some portion of the Work.
- **1.1.22** Project: The planned undertaking as provided for in the Contract Documents.
- 1.1.23 Program Manager: The individual, partnership, corporation, joint venture, or any combination thereof, or its authorized representative, named as such by the County. If no Program Manager is designated for Project that is the subject of this Contract, then all references to Project Manager herein shall be read to refer to County.
- **1.1.24** Provide: Shall include "provide complete in place," that is, "furnish and install," and "provide complete and functioning as intended in place" unless specifically stated otherwise.
- 1.1.25 Request for Information: A written request prepared by the Contractor requesting that the Architect provide additional information necessary to clarify or amplify an item in the Contract Documents that the Contractor believes is not clearly shown or called for in the Drawings or Specifications or other portions of the Contract Documents, or to address problems that have arisen under field conditions.

- **1.1.26** Request for Substitution: A request by Contractor to substitute an equal or superior material, product, thing, or service for a specific material, product, thing, or service that has been designated in the Contract Documents by a specific brand or trade name.
- **1.1.27** Safety Orders: Written and/or verbal orders for construction issued by the California Division of Industrial Safety ("CalOSHA") or by the United States Occupational Safety and Health Administration ("OSHA").
- **1.1.28** Safety Plan: Contractor's safety plan specifically adapted for the Project. Contractor's Safety Plan shall comply with all provisions regarding Project safety, including all applicable provisions in these General Conditions.
- **1.1.29** Samples: Physical examples that illustrate materials, products, equipment, finishes, colors, or workmanship and that, when approved in accordance with the Contract Documents, establish standards by which portions of the Work will be judged.
- **1.1.30** Shop Drawings: All drawings, prints, diagrams, illustrations, brochures, schedules, and other data that are prepared by the Contractor, a subcontractor, manufacturer, supplier, or distributor, that illustrate how specific portions of the Work shall be fabricated or installed.
  - **1.1.31** Site: The Project site as shown on the Drawings.
- **1.1.32** Specifications: That portion of the Contract Documents, Division 01 through Division 49, and all technical sections, and addenda to all of these, if any, consisting of written descriptions and requirements of a technical nature of materials, equipment, construction methods and systems, standards, and workmanship.
- **1.1.33** Subcontractor: A contractor and/or supplier who is under contract with the Contractor or with any other subcontractor, regardless of tier, to perform a portion of the Work of the Project.
- **1.1.34** Submittal Schedule: The schedule of submittals as provided by Contractor and approved by County.
- **1.1.35** Surety: The person, firm, or corporation that executes as surety the Contractor's Performance Bond and Payment Bond, and must be a California admitted surety insurer as defined in the Code of Civil Procedure Section 995.120.
- **1.1.36** Work: All labor, materials, equipment, components, appliances, supervision, coordination, and services required by, or reasonably inferred from, the Contract Documents, that are necessary for the construction and completion of the Project.

#### 1.2 Laws Concerning The Contract

Contract is subject to all provisions of the Constitution and laws of California governing, controlling, or affecting County, or the property, funds, operations, or powers of County, and such provisions are by this reference made a part hereof. Any provision required by law to be included in this Contract shall be deemed to be inserted.

#### 1.3 No Oral Agreements

No oral agreement or conversation with any officer, agent, or employee of County, either before or after execution of Contract, shall affect or modify any of the terms or obligations contained in any of the documents comprising the Contract.

#### 1.4 No Assignment

Contractor shall not assign this Contract or any part thereof including, without limitation, any services or money to become due hereunder without the prior written consent of the County. Assignment without County's prior written consent shall be null and void. Any assignment of money due or to become due under this Contract shall be subject to a prior lien for services rendered or material supplied for performance of work called for under this Contract in favor of all persons, firms, or corporations rendering services or supplying material to the extent that claims are filed pursuant to the Civil Code, Code of Civil Procedure, Government Code, Labor Code, and/or Public Contract Code, and shall also be subject to deductions for liquidated damages or withholding of payments as determined by County in accordance with this Contract. Contractor shall not assign or transfer in any manner to a Subcontractor or supplier the right to prosecute or maintain an action against the County.

#### 1.5 Notice And Service Thereof

**1.5.1** Any notice from one party to the other or otherwise under Contract shall be in writing and shall be dated and signed by the party giving notice or by a duly authorized representative of that party. Any notice shall not be effective for any purpose whatsoever unless served in one of the following manners:

- **1.5.1.1** If notice is given by personal delivery thereof, it shall be considered delivered on the day of delivery.
- **1.5.1.2** If notice is given by overnight delivery service, it shall be considered delivered on (1) day after date deposited, as indicated by the delivery service.
- **1.5.1.3** If notice is given by depositing same in United States mail, enclosed in a sealed envelope, it shall be considered delivered three (3) days after date deposited, as indicated by the postmarked date.
- **1.5.1.4** If notice is given by registered or certified mail with postage prepaid, return receipt requested, it shall be considered delivered on the day the notice is signed for.

#### 1.6 No Waiver

The failure of County in any one or more instances to insist upon strict performance of any of the terms of this Contract or to exercise any option herein conferred shall not be construed as a waiver or relinquishment to any extent of the right to assert or rely upon any such terms or option on any future occasion. No action or failure to act by the County, Architect, or Construction Manager shall constitute a waiver of any right or duty afforded the County under the Contract, nor shall any action or failure to act constitute an approval of or acquiescence in any breach thereunder, except as may be specifically agreed in writing.

#### 1.7 Substitutions For Specified Items

See Special Conditions.

#### 1.8 Materials and Work

- **1.8.1** Except as otherwise specifically stated in this Contract, Contractor shall provide and pay for all materials, labor, tools, equipment, transportation, supervision, temporary constructions of every nature, and all other services, management, and facilities of every nature whatsoever necessary to execute and complete this Contract within the Contract Time.
- **1.8.2** Unless otherwise specified, all materials shall be new, and the best of their respective kinds and grades as noted or specified and workmanship shall be of good quality.
- **1.8.3** Materials shall be furnished in ample quantities and at such times as to ensure uninterrupted progress of Work and shall be stored properly and protected as required.
- **1.8.4** For all materials and equipment specified or indicated in the Drawings, the Contractor shall provide all labor, materials, equipment, and services necessary for complete assemblies and complete working systems, functioning as intended. Incidental items not indicated on Drawings, nor mentioned in the Specifications, that can legitimately and reasonably be inferred to belong to the Work described, or be necessary in good practice to provide a complete assembly or system, shall be furnished as though itemized here in every detail. In all instances, material and equipment shall be installed in strict accordance with each manufacturer's most recent published recommendations and specifications.
- **1.8.5** Contractor shall, after award of Contract by County and after relevant submittals have been approved, place orders for materials and/or equipment as specified so that delivery of same may be made without delays to the Work. Contractor shall, upon demand from County, present documentary evidence showing that orders have been placed.
- **1.8.6** County reserves the right but has no obligation, for any neglect in complying with the above instructions, to place orders for such materials and/or equipment as it may deem advisable in order that the Work may be completed at the date specified in the Agreement, and all expenses incidental to the procuring of said materials and/or equipment shall be paid for by Contractor or withheld from payment(s) to Contractor.
- 1.8.7 Contractor warrants good title to all material, supplies, and equipment installed or incorporated in Work and agrees upon completion of all Work to deliver the Site to County, together with all improvements and appurtenances constructed or placed thereon by it, and free from any claims, liens, or charges. Contractor further agrees that neither it nor any person, firm, or corporation furnishing any materials or labor for any work covered by the Contract shall have any right to lien any portion of the Premises or any improvement or appurtenance thereon, except that Contractor may install metering devices or other equipment of utility companies or of political subdivision, title to which is commonly retained by utility company or political subdivision. In the event of installation of any such metering device or equipment, Contractor shall advise County as to owner thereof.
- **1.8.8** Nothing contained in this Article, however, shall defeat or impair the rights of persons furnishing materials or labor under any bond given by Contractor for their protection or any rights under any law permitting such protection or any rights under any law permitting such persons to look to funds due Contractor in hands of County (e.g., Stop

#### **Alameda County, General Services Agency**

Highland Hospital – Koret Building Exhaust Fan and Waterproofing

Notices), and this provision shall be inserted in all subcontracts and material contracts and notice of its provisions shall be given to all persons furnishing material for work when no formal contract is entered into for such material.

**1.8.9** Title to new materials and/or equipment for the Work of this Contract and attendant liability for its protection and safety shall remain with Contractor until incorporated in the Work of this Contract and accepted by County. No part of any materials and/or equipment shall be removed from its place of storage except for immediate installation in the Work of this Contract. Contractor shall keep an accurate inventory of all materials and/or equipment in a manner satisfactory to County or its authorized representative and shall, at the County's request, forward it to the County.

#### 2. COUNTY

#### 2.1 Occupancy

County reserves the right to occupy portions of the Project at any time before completion. Neither the County's Final Acceptance, the making of Final Payment, any provision in Contract Documents, nor the use or occupancy of the Work, in whole or in part, by County shall constitute acceptance of Work not in accordance with the Contract Documents nor relieve the Contractor or the Contractor's Performance Bond Surety from liability with respect to any warranties or responsibility for faulty or defective Work or materials, equipment and workmanship incorporated therein.

#### 3. ARCHITECT

#### 3.1 Role and Authority

The Architect shall represent County during the Project and will observe the progress and quality of the Work on behalf of County. Architect shall have the authority to act on behalf of County to the extent expressly provided in the Contract Documents and to the extent determined by County. Architect shall have authority to reject materials, workmanship, and/or the Work whenever rejection may be necessary, in Architect's reasonable opinion, to ensure the proper execution of the Contract.

#### 3.2 Interpretations

Architect shall, with County and on behalf of County, determine the amount, quality, acceptability, and fitness of all parts of the Work, and interpret the Specifications, Drawings, and shall, with County, interpret all other Contract Documents.

#### **3.3 Laws**

Architect shall have all authority and responsibility established by law, including Title 24 of the California Code of Regulations.

#### **3.4 Communications**

Contractor shall provide County and the Construction Manager with a copy of all written communication between Contractor and Architect at the same time as that communication is made to Architect, including, without limitation, all RFIs, correspondence, submittals, claims, and proposed change orders.

#### 4. CONSTRUCTION MANAGER

#### 4.1 Role and Authority

If a construction manager is used on this Project ("Construction Manager" or "CM"), the Construction Manager will provide administration of the Contract on the County's behalf. After execution of the Contract and Notice to Proceed, all correspondence, and/or instructions from Contractor and/or County shall be forwarded through the Construction Manager. The Construction Manager will not be responsible for and will not have control or charge of construction means, methods, techniques, sequences, or procedures or for safety precautions in connection with the Work, which shall all remain the Contractor's responsibility.

#### 4.2 Authority to Reject

The Construction Manager, however, will have authority to reject materials and/or workmanship not conforming to the Contract Documents, as determined by the County and/or the Architect. The Construction Manager shall also have the authority to require special inspection or testing of any portion of the Work, whether it has been fabricated, installed or fully completed. Any decision made by the Construction Manager, in good faith, shall not give rise to any duty or responsibility of the Construction Manager to the Contractor, any Subcontractor, their agents, employees, or other persons performing any of the Work. The Construction Manager shall have free access to any or all parts of the Work at any time.

#### 4.3 If No Construction Manager

If the County does not use a Construction Manager on this Project, all references to Construction Manager or CM shall be read as County.

#### 5. INSPECTIONS AND TESTS

#### **5.1 Tests and Inspections**

- **5.1.1** The County will select an independent testing laboratory to conduct tests. Selection of the materials required to be tested shall be by the laboratory or the County's representative and not by the Contractor. The Contractor shall notify the County's representative a sufficient time in advance of its readiness for required observation or inspection.
- **5.1.2** The Contractor shall notify the County's representative a sufficient time in advance of the manufacture of material to be supplied under the Contract Documents that must by terms of the Contract Documents be tested, in order that the County may arrange for the testing of same at the source of supply. This notice shall be, at a minimum, seventy-two (72) hours prior to the manufacture of the material that needs to be tested.
- **5.1.3** Any material shipped by the Contractor from the source of supply prior to having satisfactorily passed such testing and inspection or prior to the receipt of notice from said representative that such testing and inspection will not be required, shall not be incorporated into and/or onto the Project.
- **5.1.4** The County will select and pay testing laboratory costs for all tests and inspections. Costs of tests of any materials found to be not in compliance with the Contract Documents shall be paid for by the County and reimbursed by the Contractor or deducted from the Contract Price.

#### 5.2 Costs for After Hours and/or Off Site Inspections

If the Contractor performs Work outside the County's regular working hours or requests the County to perform inspections off Site, costs of any inspections required outside regular working hours or off Site shall be borne by the Contractor and may be invoiced to the Contractor by the County or the County may deduct those expenses from the next Progress Payment.

#### 6. CONTRACTOR

Contractor shall construct the Work for the Contract price including any adjustment(s) to the Contract Price pursuant to provisions herein regarding changes to the Contract Price. Except as otherwise noted, Contractor shall provide and pay for all labor, materials, equipment, permits, fees, licenses, facilities, transportation, taxes, and services necessary for the proper execution and completion of the Work, except as indicated herein.

#### **6.1 Status of Contractor**

- **6.1.1** Contractor is and shall at all times be deemed to be an independent contractor and shall be wholly responsible for the manner in which it and its Subcontractors perform the services required of it by the Contract Documents. Nothing herein contained shall be construed as creating the relationship of employer and employee, or principal and agent, between the County, or any of the County's employees or agents, and Contractor or any of Contractor's Subcontractors, agents or employees. Contractor assumes exclusively the responsibility for the acts of its employees as they relate to the services to be provided during the course and scope of their employment. Contractor, its agents, its employees and its Subcontractors shall not be entitled to any rights or privileges of County employees. County shall be permitted to monitor the Contractor's activities to determine compliance with the terms of this Contract.
- **6.1.2** As required by law, Contractor and all Subcontractors shall be properly licensed and regulated by the Contractor's State License Board, 3132 Bradshaw Road, Post Office Box 2600, Sacramento, California 98826, http://www.cslb.ca.gov.

#### **6.2 Contractor's Supervision**

- **6.2.1** At all times during progress of the Work, while any work is being performed, Contractor shall keep on the Premises, and at all other locations where any Work related to the Contract is being performed, a competent project manager and construction superintendent who are employees of the Contractor, to whom the County does not object and at least one of whom shall be fluent in English, written and verbal.
- **6.2.2** The project manager and construction superintendent shall both speak fluently the predominant language of the Contractor's employees fluently. All workers shall be sufficiently competent in English to respond to inquiries and instructions and give directions concerning matters of safety and concerning the identification and location of site foremen, the Contractor's construction superintendent, and the Contractor's project manager.
- **6.2.3** Before commencing the Work herein, Contractor shall give written notice to County of the name and relevant credentials of its project manager and construction superintendent. Neither the Contractor's project manager nor construction superintendent shall be changed except with prior written notice to County and County's approval, unless the Contractor's project manager and/or construction superintendent proves to be unsatisfactory to Contractor, County, any of the County's employees, agents,

the Construction Manager, or the Architect, in which case, Contractor shall notify County in writing. The Contractor's project manager and construction superintendent shall each represent Contractor, and all directions given to Contractor's project manager and/or construction superintendent shall be as binding as if given to Contractor.

**6.2.4** Contractor shall give efficient supervision to Work, using its best skill and attention. Contractor shall carefully study and compare all Contract Documents, Drawings, Specifications, and other instructions and shall at once report to County, Construction Manager, and Architect any error, inconsistency, or omission that Contractor or its employees and Subcontractors may discover, in writing. The Contractor shall have responsibility for discovery of errors, inconsistencies, or omissions.

#### **6.3 Duty to Provide Fit Workers**

- **6.3.1** Contractor and Subcontractor(s) shall at all times enforce strict discipline and good order among their employees and shall not employ or work any unfit person or anyone not skilled in work assigned to that person. It shall be the responsibility of Contractor to ensure compliance with this requirement. County may require Contractor to permanently remove unfit persons from Project Site.
- **6.3.2** Any person in the employ of Contractor or Subcontractor(s) whom County may deem incompetent or unfit shall be excluded from working on the Project and shall not again be employed on the Project except with the prior written consent of County.
- **6.3.3** The Contractor shall furnish labor that can work in harmony with all other elements of labor employed or to be employed in the Work.
- **6.3.4** If Contractor intends to make any change in the name or legal nature of the Contractor's entity, Contractor must first notify the County. The County shall determine if Contractor's intended change is permissible while performing this Contract.

#### 6.4 Purchase of Materials and Equipment

The Contractor is required to order, obtain, and store materials and equipment sufficiently in advance of its Work at no additional cost or advance payment from County to assure that there will be no delays.

#### 6.5 Documents On Work Site

- **6.5.1** Contractor shall at all times keep on the Work Site, or such other location as County may authorize in writing one legible copy of all Contract Documents, including Addenda and Change Orders, and titles 19 and 24 of the California Code of Regulations, the specified edition(s) of the Uniform Building Code, all approved Drawings, Plans, Schedules, and Specifications, and all codes and documents referred to in the Specifications, and made part thereof. These documents shall be kept in good order and available to County, Construction Manager, Architect, Architect's representatives, and all authorities having jurisdiction. Contractor shall be acquainted with and comply with the provisions of these titles as they relate to this Project. Contractor shall also be acquainted with and comply with all California Code of Regulations provisions relating to conditions on this Project. Contractor shall coordinate with Architect and Construction Manager.
- **6.5.2** Daily Job Reports. Contractor shall maintain, at a minimum, at least one (1) set of Daily Job Reports on the Project. These must be prepared by the Contractor's

#### Alameda County, General Services Agency

Highland Hospital – Koret Building Exhaust Fan and Waterproofing

employee(s) who are present on Site and must include, at a minimum, the following information:

6.5.2.1	A brief description of all Work performed on that day. This shall include a listing of what was done, which contractors were on site that day, and where on the site the work was performed.
6.5.2.2	A summary of all other pertinent events and/or occurrences on that day.
6.5.2.3	The weather conditions on that day.
6.5.2.4	A list of all Subcontractor(s) working on that day,
6.5.2.5	A list of each Contractor employee working on that day and the total hours worked for each employee.
6.5.2.6	A complete list of all equipment on Site that day, whether in use or not.
6.5.2.7	All complete list of all materials, supplies, and equipment delivered on that day.
6.5.2.8	A complete list of all inspections and tests performed on that day.
6.5.2.9	Each day Contractor shall provide a copy of the previous day's Daily Job Report to the County or the County's Construction Manager.

#### **6.6 Preservation of Records**

The County shall have the right to examine and audit all Daily Job Reports or other Project records of Contractor's project manager(s), project superintendent(s), and/or project foreperson(s), all certified payroll records and/or related documents including, without limitation, payroll, payment, timekeeping and tracking documents; all books, estimates, records, contracts, documents, bid documents, bid cost data, subcontract job cost reports, and other data of the Contractor, any Subcontractor, and/or supplier, including computations and projections related to bidding, negotiating, pricing, or performing the Work or Contract modification, in order to evaluate the accuracy, completeness, and currency of the cost, manpower, coordination, supervision, or pricing data at no additional cost to the County. These documents may be duplicative and/or be in addition to any Bid Documents held in escrow by the County. The Contractor shall make available at its office at all reasonable times the materials described in this paragraph for the examination, audit or reproduction until three (3) years after final payment under this Contract. Notwithstanding the provisions above, Contractor shall provide any records requested by any governmental agency if available, after the time set forth above.

#### **6.7 Integration of Work**

- **6.7.1** Contractor shall do all cutting, fitting, patching, and preparation of Work as required to make its several parts come together properly, to fit it to receive or be received by work of other contractors, and to coordinate tolerances to various pieces of work, showing upon, or reasonably implied by, the Drawings and Specifications for the completed structure, and shall conform them as County and/or Architect may direct.
- **6.7.2** All cost caused by defective or ill-timed Work shall be borne by Contractor, inclusive of repair work.

**6.7.3** Contractor shall not endanger any work performed by it or anyone else by cutting, excavating, or otherwise altering work and shall not cut or alter work of any other contractor except with the consent of County.

#### **6.8 Obtaining of Permits and Licenses**

Contractor shall secure and pay for all permits, licenses, and certificates necessary for prosecution of Work before the date of the commencement of the Work or before the permits, licenses, and certificates are legally required to continue the Work without interruption. The Contractor shall obtain and pay, only when legally required, for all licenses, permits, inspections, and inspection certificates required to be obtained from or issued by any authority having jurisdiction over any part of the Work included in the Contract. All final permits, licenses, and certificates shall be delivered to County before demand is made for final payment.

#### 6.9 Work to Comply With Applicable Laws and Regulations

**6.9.1** Contractor shall give all notices and comply with the following specific laws, ordinances, rules, and regulations and all other applicable laws, ordinances, rules, and regulations bearing on conduct of Work as indicated and specified, including but not limited to the appropriate statutes and administrative code sections. If Contractor observes that Drawings and Specifications are at variance therewith, or should Contractor become aware of the development of conditions not covered by Contract Documents that will result in finished Work being at variance therewith, Contractor shall promptly notify County in writing, including by e-mail, and any changes deemed necessary by County shall be made as provided in Contract for changes in Work.

6.9.1.1	National Electrical Safety Code, U. S. Department of Commerce
6.9.1.2	National Board of Fire Underwriters' Regulations
6.9.1.3	Uniform Building Code, latest addition, and the California Code of Regulations, title 24, including amendments
6.9.1.4	Manual of Accident Prevention in Construction, latest edition, published by A.G.C. of America
6.9.1.5	Industrial Accident Commission's Safety Orders, State of California
6.9.1.6	Regulations of the State Fire Marshall (title 19, California Code of Regulations) and Pertinent Local Fire Safety Codes
6.9.1.7	Americans with Disabilities Act
6.9.1.8	Government Code of the State of California
6.9.1.9	Labor Code of the State of California, Division 2, part 7, Public Works and Public Agencies
6.9.1.10	Public Contract Code of the State of California
6.9.1.11	California Art Preservation Act
6.9.1.12	U. S. Copyright Act
6.9.1.13	U. S. Visual Artists Rights Act

- **6.9.2** Contractor shall comply with all applicable mitigation measures, if any, adopted by any public agency with respect to this Project pursuant to the California Environmental Quality Act (Public Resources Code Section 21000 et. Seq.)
- **6.9.3** If Contractor performs any Work that it knew, or through exercise of reasonable care should have known, to be contrary to any applicable laws, ordinance, rules, or regulations, Contractor shall bear all costs arising therefrom.
- **6.9.4** Where Specifications or Drawings state that materials, processes, or procedures must be approved by the State Fire Marshall, or other body or agency, Contractor shall be responsible for satisfying requirements of such bodies or agencies.

#### 6.10 Safety/Protection of Persons and Property

- **6.10.1** The Contractor will be solely and completely responsible for conditions of the Work Site, including safety of all persons and property during performance of the Work. This requirement will apply continuously and not be limited to normal working hours.
- **6.10.2** The wearing of hard hats will be mandatory at all times for all personnel on Site. Contractor shall supply sufficient hard hats to properly equip all employees and visitors.
- **6.10.3** Any construction review of the Contractor's performance is not intended to include a review of the adequacy of the Contractor's safety measures in, on, or near the Work Site.
- **6.10.4** Implementation and maintenance of safety programs shall be the sole responsibility of the Contractor.
- **6.10.5** The Contractor shall furnish to the County a copy of the Contractor's safety plan within the time frame indicated in the Contract Documents and specifically adapted for the Project.
- **6.10.6** Contractor shall be responsible for all damages to persons or property that occur as a result of its fault or negligence in connection with the prosecution of this Contract and shall take all necessary measures and be responsible for the proper care and completion and final acceptance by County. All Work shall be solely at Contractor's risk with the exception of damage to the Work caused by "acts of God" as defined in Public Contract Code Section 7105.
- 6.10.7 Contractor shall take, and require Subcontractors to take, all necessary precautions for safety of workers on the Project and shall comply with all applicable federal, state, local, and other safety laws, standards, orders, rules, regulations, and building codes to prevent accidents or injury to persons on, about, or adjacent to premises where Work is being performed and to provide a safe and healthful place of employment. Contractor shall furnish, erect, and properly maintain at all times, all necessary safety devices, safeguards, construction canopies, signs, nets, barriers, lights, and watchmen for protection of workers and the public and shall post danger signs warning against hazards created by such features in the course of construction.
- **6.10.8** Hazards Control Contractor shall store volatile wastes in covered metal containers and remove them from the Site daily. Contractor shall prevent

the accumulation of wastes that create hazardous conditions. Contractor shall provide adequate ventilation during use of volatile or noxious substances.

- **6.10.9** Contractor shall designate a responsible member of its organization on the Project, whose duty shall be to post information regarding protection and obligations of workers and other notices required under occupational safety and health laws, to comply with reporting and other occupational safety requirements, and to protect the life, safety, and health of workers. Name and position of person so designated shall be reported to County by Contractor.
- **6.10.10** Contractor shall correct any violations of safety laws, rules, orders, standards, or regulations. Upon the issuance of a citation or notice of violation by the Division of Occupational Safety and Health, Contractor shall correct such violation promptly.
- **6.10.11** Contractor shall comply with any County stormwater requirements that are approved by the County and applicable to the Project, at no additional cost to the County.
- **6.10.12** In an emergency affecting safety of life or of work or of adjoining property, Contractor, without special instruction or authorization, shall act, at its discretion, to prevent such threatened loss or injury. Any compensation claimed by Contractor on account of emergency work shall be determined by agreement.
- **6.10.13** All connections to public utilities and/or existing on-site services shall be made and maintained in such a manner as to not interfere with the continuing use of same by the County during the entire progress of the Work.
- **6.10.14** Contractor shall provide such heat, covering, and enclosures as are necessary to protect all Work, materials, equipment, appliances, and tools against damage by weather conditions, such as extreme heat, cold, rain, snow, dry winds, flooding, or dampness.
- **6.10.15** The Contractor shall protect and preserve the Work from all damage or accident, providing any temporary roofs, window and door coverings, boxing, or other construction. The Contractor shall be responsible for existing structures, walks, roads, trees, landscaping, and/or improvements in working areas; and shall provide adequate protection therefor. If temporary removal is necessary of any of the above items, or damage occurs due to the Work, the Contractor shall replace same at his expense with same kind, quality, and size of Work or item damaged. This shall include any adjoining property of the County and others.
- **6.10.16** Contractor shall take adequate precautions to protect existing roads, sidewalks, curbs, pavements, utilities, adjoining property, and structures (including, without limitation, protection from settlement or loss of lateral support), and to avoid damage thereto, and repair any damage thereto caused by construction operations.
- **6.10.17** Contractor shall confine apparatus, the storage of materials, and the operations of workers to limits indicated by law, ordinances, permits, or directions of Architect, and shall not interfere with the Work or unreasonably encumber Premises or overload any structure with materials. Contractor shall enforce all instructions of County and Architect regarding signs, advertising, fires, and smoking, and require that all workers comply with all regulations while on Project Site.

- **6.10.18** Contractor, Contractor's employees, Subcontractors, Subcontractors' employees, or any person associated with the Work shall conduct themselves in a manner appropriate for a public site. No verbal or physical contact with the public, neighbors, or tenants, or profanity, or inappropriate attire or behavior will be permitted. County may require Contractor to permanently remove non-complying persons from Project Site.
- **6.10.19** Contractor shall take care to prevent disturbing or covering any survey markers, monuments, or other devices marking property boundaries or corners. If such markers are disturbed, Contractor shall have a civil engineer, registered as a professional engineer in California, replace them at no cost to County.
- **6.10.20** In the event that the Contractor enters into any agreement with owners of any adjacent property to enter upon the adjacent property for the purpose of performing the Work, Contractor shall fully indemnify, defend, and hold harmless each person, entity, firm, or agency that owns or has any interest in adjacent property. The form and content of the agreement of indemnification shall be approved by the County prior to the commencement of any Work on or about the adjacent property. The Contractor shall also indemnify the County as provided in the indemnification provision herein. These provisions shall be in addition to any other requirements of the owners of the adjacent property.

#### 6.11 Working Evenings and Weekends

Contractor may be required to work evenings and/or weekends at no additional cost to the County. Contractor shall give the County seventy-two (72) hours notice prior to performing any evening and/or weekend work. Contractor shall perform all evening and/or weekend work only upon County's approval and in compliance with all applicable rules, regulations, laws, and local ordinances including, without limitation, all noise and light limitations. Contractor shall reimburse the County for any expenses necessitated by the Contractor's evening and/or weekend work.

#### **6.12** Badge Policy For Contractors

All Contractors doing work for Alameda County will provide their workers with identification badges. These badges will be worn by all members of the Contractor's staff who are working in a County facility.

- **6.12.1** Badges must be filled out in full and contain the following information:
  - **6.12.1.1** Name of Contractor and Contractor's Company logo, if any
  - **6.12.1.2** Name and front facial photograph of Employee
  - **6.12.1.3** Contractor's address and phone number
  - **6.12.1.4** Name and phone number of Project Manager (County)
- **6.12.2** Badges are to be worn when the Contractor or his/her employees are on site and must be visible at all times. Contractors must inform their employees that they are required to allow County employees to review the information on the badges upon request

**6.12.3** Failure to display identification badges as required by this policy may result in the assessment of fines against the Contractor.

#### 6.13 County Drug Policy - Drug-Free Work Place

6.13.1 Contractor, Contractor's employees, and Contractor's Subcontractors and their employees shall comply with the County's policy of maintaining a drug-free workplace. Neither Contractor/Subcontractor nor Contractor's/Subcontractor's employees shall unlawfully manufacture, distribute, dispense, possess or use controlled substances, as defined in 21 U.S. Code Section 812, including marijuana, heroin, cocaine, and amphetamines, at any County facility or work site. If Contractor or any employee of contractor is convicted or pleads nolo contendere to a criminal drug statute violation occurring at a County facility or work site, the Contractor within five (5) calendar days thereafter shall notify the head of the County department/agency for which the contract services are performed. Violation of this provision shall constitute a material breach of this contract.

#### 6.14 Cleaning Up

- 6.14.1 The Contractor shall provide all services, labor, materials, and equipment necessary for protecting the Work, all Project occupants, furnishings, equipment, and building structure from damage until its completion and final acceptance by County. Dust barriers shall be provided to isolate dust and dirt from construction operations. At completion of the Work and portions thereof, Contractor shall clean to the original state any areas beyond the Work area that become dust laden as a result of the Work. The Contractor must erect the necessary warning signs and barricades to ensure the safety of all Project occupants. The Contractor at all times must maintain good housekeeping practices to reduce the risk of fire damage and must make a fire extinguisher, fire blanket, and/or fire watch, as applicable, available at each location where cutting, braising, soldering, and/or welding is being performed and locations where there is an increased risk of fire.
- **6.14.2** Contractor at all times shall keep Premises free from debris such as waste, rubbish, and excess materials and equipment caused by the Work. Contractor shall not leave debris under, in, or about the Premises, but shall promptly remove same from the Premises on a daily basis. If Contractor fails to clean up, County may do so, and the cost thereof shall be charged to Contractor. If Contract is for work on an existing facility, Contractor shall also perform specific clean-up on or about the Premises upon request by the County as it deems necessary for the continuing use of the facility. Contractor shall comply with all related provisions of the Specifications.
- **6.14.3** If the Construction Manager, Architect, or County observes the accumulation of trash and debris, the County will give the Contractor a 24-hour written notice to mitigate the condition.
- **6.14.4** Should the Contractor fail to perform the required clean-up, or should the clean-up be deemed unsatisfactory by the County, the County will then perform the clean-up. All cost associated with the clean-up work (including all travel, payroll burden, and costs for supervision) will be deducted from the Contract Price, or County may withhold those amounts from payment(s) to Contractor.

#### 7. SUBCONTRACTORS

#### 7.1 Contractor Shall Provide Subcontractor Information

Contractor shall provide the County with information for all Subcontracts as indicated in the Contractor's Submittals and Schedules Section herein.

#### 7.2 No Contractual Relationship Between County and Subcontractors

No contractual relationship exists between the County and any Subcontractor, Supplier, or sub-subcontractor supplier, or sub-subcontractor by reason of this Contract.

#### 7.3 Contractor Binds Every Subcontractor by Terms of Contract

Contractor agrees to bind every Subcontractor by terms of Contract as far as those terms are applicable to Subcontractor's work. If Contractor shall subcontract any part of this Contract, Contractor shall be as fully responsible to County for acts and omissions of any Subcontractor and of persons either directly or indirectly employed by any Subcontractor, as it is for acts and omissions of persons directly employed by Contractor. The divisions or sections of the Specifications are not intended to control the Contractor in dividing the Work among Subcontractors or limit the work performed by any trade.

### 7.4 No Waiver of Obligations

County's consent to, or approval of, or failure to object to, any Subcontractor under this Contract shall not in any way relieve Contractor of any obligations under this Contract, and no such consent shall be deemed to waive any provisions of this Contract.

#### 7.5 Contractor to Familiarize Itself with Laws

Contractor is directed to familiarize itself with sections 4100 through 4114 of the Public Contract Code of the State of California, as regards subletting and subcontracting, and to comply with all applicable requirements therein. In addition, Contractor is directed to familiarize itself with Sections 1720 through 1861 of the Labor Code of the State of California, as regards the payment of prevailing wages and related issues, and to comply with all applicable requirements therein all including, without limitation, Section 1775 and the Contractor's and Subcontractors' obligations and liability for violations of prevailing wage law and other applicable laws.

#### 7.6 Subcontractor Substitutions

No Contractor whose Bid is accepted shall, without consent of the awarding authority and in full compliance with section 4100, et seq, of the Public Contract Code, including, without limitation, sections 4107, 4107.5, and 4109 of the Public Contract Code, either:

- **7.6.1** Substitute any person as a Subcontractor in place of the Subcontractor designated in the original Bid; or
- **7.6.2** Permit any Subcontract to be assigned or transferred, or allow any portion of the Work to be performed by anyone other than the original Subcontractor listed in the Bid; or
- **7.6.3** Sublet or subcontract any portion of the Work in excess of one-half of one percent (1/2 of 1%) of the Contractor's total bid as to which his original bid did not designate a Subcontractor.

#### **Alameda County, General Services Agency**

Highland Hospital – Koret Building Exhaust Fan and Waterproofing

#### 7.7 Subcontractor Coordination

The Contractor shall be responsible for the coordination of the trades, Subcontractors, sub-subcontractors, and material or equipment suppliers working on the Project.

#### 7.8 Subcontractor Relations

Contractor is solely responsible for settling any differences between the Contractor and its Subcontractor(s) or between Subcontractors.

#### 7.9 Assignment or Termination

Contractor must include in all of its subcontracts the assignment provisions as indicated in the Termination section of these General Conditions.

#### 8. OTHER CONTRACTS/CONTRACTORS

#### 8.1 County Right to Perform

County reserves the right to let other contracts, and/or to perform work with its own forces, in connection with the Project. Contractor shall afford other County and other contractors' reasonable opportunity for introduction and storage of their materials and execution of their work and shall properly coordinate and connect Contractor's Work with the work of County and other contractors.

#### 8.2 Protection of Work

In addition to Contractor's obligation to protect its own Work, Contractor shall protect the work of County and any other contractor that Contractor encounters while working on the Project.

#### 8.3 Coordination with Other Work

If any part of Contractor's Work depends for proper execution or results upon work of County or any other contractor, the Contractor shall inspect and promptly report to the County in writing, including by e-mail, before proceeding with its Work any defects in County's or any other contractor's work that render Contractor's Work unsuitable for proper execution and results. Contractor shall be held accountable for damages to County for County's or any other contractor's work that Contractor failed to inspect or should have inspected. Contractor's failure to inspect and report shall constitute Contractor's acceptance of all County's or other contractor's work as fit and proper for reception of Contractor's Work, except as to defects that may develop in County's or other contractor's work after execution of Contractor's Work.

#### 8.4 Measurement of Work Performed

To ensure proper execution of its subsequent work, Contractor shall measure and inspect work already in place and shall at once report to the County in writing, including by e-mail, any discrepancy between that executed work and the Contract Documents.

#### 8.5 Knowledge of Other Work

Contractor shall ascertain to its own satisfaction the scope of the Project and nature of any County-performed work or other contracts that have been or may be awarded

Highland Hospital – Koret Building Exhaust Fan and Waterproofing

by County in prosecution of the Project to the end that Contractor may perform this Contract in light of the other contracts, if any.

## 8.6 No Exclusive Occupancy of Site

Nothing herein contained shall be interpreted as granting to Contractor exclusive occupancy of the Site, the Premises, or of the Project. Contractor shall not cause any unnecessary hindrance or delay to the use and/or operation(s) of the Premises and/or to County or any other contractor working on the Project. If simultaneous execution of any contract or operation is likely to cause interference with performance of Contractor's Contract, Contractor shall coordinate with those contractor(s), person(s), and/or entity(s) and shall notify the County of the resolution.

#### 9. DRAWINGS AND SPECIFICATIONS

#### 9.1 List of all Drawings

A complete list of all Drawings that form a part of the Contract is to be found as an index on the Drawings themselves, and/or may be provided to the Contractor and/or in the Table of Contents.

#### 9.2 Technical and Trade Words

Materials or Work described in words that so applied have a well-known technical or trade meaning shall be deemed to refer to recognized standards unless noted otherwise.

#### 9.3 Trade Name or Trade Term

It is not the intention of this Contract to go into detailed descriptions of any materials and/or methods commonly known to the trade under "trade name" or "trade term." The mere mention or notation of "trade name" or "trade term" shall be considered a sufficient notice to Contractor that it will be required to complete the work so named, complete, finished, and operable, with all its appurtenances, according to the best practices of the trade.

## 9.4 The Naming of any Material and/or Equipment Shall Mean Furnishing

The naming of any material and/or equipment shall mean furnishing and installing of same, including all incidental and accessory items thereto and/or labor, therefore, as per best practices of the trade(s) involved, unless specifically noted otherwise.

#### 9.5 Contract Documents are Complementary

Contract Documents are complementary, and what is called for by one shall be binding as if called for by all. As such, Drawings and Specifications are intended to be fully cooperative and to agree. However, if Contractor observes that Drawings and Specifications are in conflict, Contractor shall promptly notify County and Architect in writing, including by e-mail, and any necessary changes shall be made as provided in the Contract Documents.

## 9.6 Drawings and Specifications are Intended to Comply With All Laws

Drawings and Specifications are intended to comply with all laws ordinances, rules, and regulations of constituted authorities having jurisdiction, and where referred to in the Contract Documents, the laws, ordinances, rules, and regulations shall be considered as

a part of the Contract within the limits specified. Contractor shall bear all expense of correcting work done contrary to said laws, ordinances, rules, and regulations.

# 9.7 Plans, Drawings, Designs, Specifications are County Property

All copies of Plans, Drawings, Designs, Specifications and copies of other incidental architectural and engineering work, or copies of other Contract Documents furnished by County, are the property of County. They are not to be used by Contractor in other work and, with the exception of signed sets of Contract Documents, are to be returned to County on request at completion of Work, or may be used by County as it may require without any additional costs to County. Neither the Contractor nor any Subcontractor, or material or equipment supplier shall own or claim copyright in the Drawings, Specifications, and other documents prepared by the Architect. County hereby grants the Contractor, Subcontractors, sub-subcontractors, and material or equipment suppliers a limited license to use applicable portions of the Drawings prepared for the Project in the execution of their Work under the Contract Documents.

#### 9.8 Order of Precedence

In the case of discrepancy or ambiguity in the Contract Documents, the order of precedence in the Agreement shall prevail.

## 9.9 Resolution of Discrepancy or Ambiguity

However, in the case of discrepancy or ambiguity solely between and among the Drawings and Specifications, the discrepancy or ambiguity shall be resolved in favor of the interpretation that will provide County with the functionally complete and operable Project described in the Drawings and Specifications.

## 9.10 County Clarification

In case of ambiguity, conflict, or lack of information, County will furnish clarifications with reasonable promptness. Should any clarification, in the opinion of Contractor, cause an increase in the Contract Price, Contractor may request a change in the Contract Price and/or Contract. Within seven (7) days after receipt of the interpretation or request, Contractor to submit to the Construction Manager a detailed description of the contract requirements that were exceeded and the resulting change in cost.

## 10. CONTRACTOR'S SUBMITTALS AND SCHEDULES

Refer to Section 01 33 00 "Submittal Requirements." Contractor's submittals shall comply with the provisions and requirements of the Specifications.

# 10.1 Schedule of Work, Schedule of Submittals, and Schedule of Values

Within TEN (10) calendar days after the date of the Notice to Proceed (unless otherwise specified in the Specifications), the Contractor shall prepare and submit to the County for review, in a form supported by sufficient data to substantiate its accuracy as the County may require:

### 10.1.1 Preliminary Schedule

A preliminary schedule of construction indicating the starting and completion dates of the various stages of the Work, including any information and following any form as may be specified in the Specifications. Once approved by County, this shall become the

Construction Schedule. This schedule shall include and identify all tasks that are on the Project's critical path with a specific determination of the start and completion of each critical path task as well as all contract milestones and each milestone's completion date(s) as may be required by the County.

# 10.1.2 Preliminary Schedule of Values

A preliminary schedule of values for all of the Work, which must include quantities and prices of items aggregating the Contract Price and must subdivide the Work into component parts in sufficient detail to serve as the basis for progress payments during construction. This preliminary schedule of values shall include, at a minimum, the following information, and the following structure:

- **10.1.2.1** Divided into at least the following categories:
  - **10.1.2.1.1** Overhead and profit;
  - **10.1.2.1.2** Supervision;
  - **10.1.2.1.3** General conditions;
  - **10.1.2.1.4** Layout;
  - **10.1.2.1.5** Mobilization;
  - **10.1.2.1.6** Submittals;
  - **10.1.2.1.7** Bonds and insurance;
  - **10.1.2.1.8** Close-out documentation;
  - **10.1.2.1.9** Demolition;
  - **10.1.2.1.10** Installation;
  - **10.1.2.1.11** Rough-in;
  - **10.1.2.1.12** Finishes;
  - **10.1.2.1.13** Testing;
  - **10.1.2.1.14** Punch list and acceptance.
- **10.1.2.2** Divided by each of the following areas:
  - **10.1.2.2.1** Site work;
  - **10.1.2.2.2** By each building;
  - **10.1.2.2.3** By each floor.
- **10.1.3** The preliminary schedule of values shall not provide for values any greater than the following percentages of the Contract value:
  - **10.1.3.1** Mobilization and layout combined to equal not more than 1%;
  - Submittals, samples and shop drawings combined to equal not more than 3%, bonds and insurance combined to equal not more than 2%.
  - 10.1.3.3 Closeout documentation shall have a value in the preliminary schedule of not less than 5%.

- **10.1.4** Notwithstanding any provision of the Contract Documents to the contrary, payment of the Contractor's overhead, supervision, general conditions costs, and profit, as reflected in the Cost Breakdown, shall be paid by the County in equal installments, based on percentage complete, with the disbursement of Progress Payments and the Final Payment.
- 10.1.5 Contractor shall certify that the preliminary schedule of values, as submitted to the County, is accurate and reflects the costs as developed in preparing Contractor's bid. The preliminary schedule of values shall be subject to the County's review and approval of the form and content thereof. In the event that the County objects to any portion of the preliminary schedule of values, the County shall notify the Contractor, in writing, including by e-mail, of the County's objection(s) to the preliminary schedule of values. Within five (5) calendar days of the date of the County's written objection(s), Contractor shall submit a revised preliminary schedule of values to the County for review and approval. The foregoing procedure for the preparation, review, and approval of the preliminary schedule of values shall continue until the County has approved the entirety of the preliminary schedule of values.
- 10.1.6 Once the preliminary schedule of values is approved by the County, this shall become the Schedule of Values. The Schedule of Values shall not be thereafter modified or amended by the Contractor without the prior consent and approval of the County, which may be granted or withheld in the sole discretion of the County.

## 10.1.7 Preliminary Schedule of Submittals

A preliminary schedule of submittals, including Shop Drawings, Product Data, and Samples submittals. Once approved by County, this shall become the Submittal Schedule. All submittals shall be forwarded to the County by the date indicated on the approved Submittal Schedule unless an earlier date is necessary to maintain the Construction Schedule, in which case those submittals shall be forwarded to the County so as not to delay the Construction Schedule.

#### 10.1.8 Safety Plan

Contractor's Safety Plan specifically adapted for the Project. Contractor's Safety Plan shall comply with the following requirements:

- 10.1.8.1 All applicable requirements of California Division of Industrial Safety ("CalOSHA") and/or of the United States Occupational Safety and Health Administration ("OSHA").
- **10.1.8.2** All provisions regarding Project safety, including all applicable provisions of these General Conditions.
- **10.1.8.3** Contractor's Safety Plan shall be in English and in the language(s) of the Contractor's and its Subcontractors' employees.

### 10.1.9 Complete Subcontractor List

The name, address, telephone number, facsimile number, California State Contractors License number, classification, and monetary value of all Subcontracts for parties furnishing labor, material, or equipment for completion of the Project.

## 10.1.10 General Requirements

Highland Hospital – Koret Building Exhaust Fan and Waterproofing

- 10.1.10.1 Contractor must provide all schedules both in hard copy and electronically, in a format (e.g., Microsoft Project or Primavera) approved in advance by the County.
- 10.1.10.2 The County will review the schedules submitted, and the Contractor shall make changes and corrections in the schedules as requested by the County and resubmit the schedules until approved by the County.
- 10.1.10.3 The County shall have the right at any time to revise the schedule of values if, in the County's sole opinion, the schedule of values does not accurately reflect the value of the Work performed.
- **10.1.10.4** All submittals and schedules must be approved by the County before Contractor can rely on them as a basis for payment.

## 10.2 Monthly Progress Schedule(s)

- 10.2.1 Upon request by the County, Contractor shall provide Monthly Progress Schedule(s) to the County. A Monthly Progress Schedule shall update the approved Construction Schedule or the last Monthly Progress Schedule, showing all work completed and to be completed. The monthly Progress Schedule shall be sent within the timeframe requested by the County and shall be in a format acceptable to the County and contain a written narrative of the progress of work that month and any changes, delays, or events that may affect the work. The process for County approval of the Monthly Progress Schedule shall be the same as the process for approval of the Construction Schedule.
- **10.2.2** Contractor shall also submit Monthly Progress Schedule(s) with all payment applications.

#### 10.3 Material Safety Data Sheets (MSDS)

Contractor is required to ensure Material Safety Data Sheets are available in a readily accessible place at the Work Site for any material requiring a Material Safety Data Sheet per the Federal "Hazard Communication" standard, or employees right to know law. The Contractor is also required to ensure proper labeling on substance brought onto the job site and that any person working with the material or within the general area of the material is informed of the hazards of the substance and follows proper handling and protection procedures. Two additional copies of the Material Safety Data Sheets shall also be submitted directly to the County.

#### 11. SITE ACCESS, CONDITIONS, AND REQUIREMENTS

#### 11.1 Site Investigation

Before bidding on this Work, Contractor shall make a careful investigation of the Site and thoroughly familiarize itself with the requirements of the Contract. By the act of submitting a bid for the Work included in this Contract, Contractor shall be deemed to have made a complete study and investigation, and to be familiar with and accepted the existing conditions of the Site.

#### 11.2 Soils Investigation Report

11.2.1 When a soils investigation report obtained from test holes at Site is available, that report shall be available to the Contractor but shall not be a part of this Contract. Any information obtained from that report or any information given on

Highland Hospital – Koret Building Exhaust Fan and Waterproofing

Drawings as to subsurface soil condition or to elevations of existing grades or elevations of underlying rock is approximate only, is not guaranteed, does not form a part of this Contract, and Contractor may not rely thereon. By submitting its bid, Contractor acknowledges that it has made a visual examination of Site and has made whatever tests Contractor deems appropriate to determine underground condition of the soil.

11.2.2 Contractor agrees that no claim against County will be made by Contractor for damages and hereby waives any rights to damages if, during progress of Work, Contractor encounters subsurface or latent conditions at Site materially differing from those shown on Drawings or indicated in Specifications, or for unknown conditions of an unusual nature that differ materially from those ordinarily encountered in the work of the character provided for in Plans and Specifications, except as indicated in the provisions of these General Conditions regarding trenches, trenching, and/or existing utility lines.

#### 11.3 Access to Work

County and its representatives shall at all times have access to Work wherever it is in preparation or progress, including storage and fabrication. Contractor shall provide safe and proper facilities for such access so that County's representatives may perform their functions.

#### 11.4 Layout and Field Engineering

- 11.4.1 All field engineering required for layout of this Work and establishing grades for earthwork operations shall be furnished by Contractor at its expense. This Work shall be done by a qualified, California-registered civil engineer approved in writing by County and Architect. Any required "Record" drawings of Site development shall be prepared by the approved civil engineer.
- 11.4.2 The Contractor shall be responsible for having ascertained pertinent local conditions such as location, accessibility, and general character of the Site and for having satisfied itself as to the conditions under which the Work is to be performed. County shall not be liable for any claim for allowances because of Contractor's error or negligence in acquainting itself with the conditions at the Site.
- 11.4.3 Contractor shall protect and preserve established benchmarks and monuments and shall make no changes in locations without the prior written approval of County. Contractor shall replace any benchmarks or monuments that are lost or destroyed subsequent to proper notification of County and with County's approval.

#### 11.5 Utilities

Utilities shall be provided as indicated in the Specifications.

## 11.6 Sanitary Facilities

Sanitary facilities shall be provided as indicated in the Specifications.

# 11.7 Surveys

Contractor shall provide surveys done by a California-licensed civil engineer surveyor to determine locations of construction, grading, and site work as required to perform the Work.

## 11.8 Regional Notification Center

The Contractor, except in an emergency, shall contact the appropriate regional notification center at least two (2) days prior to commencing any excavation if the excavation will be conducted in an area or in a private easement that is known, or reasonably should be known, to contain subsurface installations other than the underground facilities owned or operated by the County, and obtain an inquiry identification number from that notification center. No excavation shall be commenced and/or carried out by the Contractor unless an inquiry identification number has been assigned to the Contractor or any Subcontractor, and the Contractor has given the County the identification number. Any damages arising from Contractor's failure to make appropriate notification shall be at the sole risk and expense of the Contractor. Any delays caused by failure to make appropriate notification shall be at the sole risk of the Contractor and shall not be considered for an extension of the Contract time.

## 11.9 Existing Utility Lines

- 11.9.1 Pursuant to Government Code Section 4215, County assumes the responsibility for removal, relocation, and protection of main or trunk utility lines and facilities located on the Construction Site at the time of commencement of construction under this Contract with respect to any such utility facilities that are not identified in the Plans and Specifications. Contractor shall not be assessed for liquidated damages for delay in completion of the Project caused by failure of County or the owner of a utility to provide for removal or relocation of such utility facilities.
- 11.9.2 Locations of existing utilities provided by County shall not be considered exact, but approximate within reasonable margin and shall not relieve Contractor of responsibilities to exercise reasonable care nor costs of repair due to Contractor's failure to do so. County shall compensate Contractor for the costs of locating, repairing damage, not due to the failure of Contractor to exercise reasonable care, and removing or relocating such utility facilities not indicated in the Plans and Specifications with reasonable accuracy, and for equipment necessarily idle during such work.
- 11.9.3 No provision herein shall be construed to preclude assessment against Contractor for any other delays in completion of the Work. Nothing in this Article shall be deemed to require County to indicate the presence of existing service laterals, appurtenances, or other utility lines, with the exception of main or trunk utility lines. Whenever the presence of these utilities on the Site of the construction Project can be inferred from the presence of other visible facilities, such as buildings, meter junction boxes, on or adjacent to the Site of the construction.
- 11.9.4 If Contractor, while performing Work under this Contract, discovers utility facilities not identified by County in Contract Plans and Specifications, Contractor shall immediately notify the County and the utility in writing. The cost of repair for damage to above-mentioned visible facilities without prior written notification to the County shall be borne by the Contractor.

#### 11.10 Notification

Contractor understands, acknowledges and agrees that the purpose for prompt notification to the County pursuant to these provisions is to allow the County to investigate the condition(s) so that the County shall have the opportunity to decide how the County desires to proceed as a result of the condition(s). Accordingly, failure of Contractor to promptly notify the County in writing, including by e-mail, pursuant to these provisions,

Highland Hospital – Koret Building Exhaust Fan and Waterproofing

shall constitute Contractor's waiver of any claim for damages or delay incurred as a result of the condition(s).

#### 11.11 Hazardous Materials

Contractor shall comply with all provisions and requirements of the Contract Documents related to hazardous materials including, without limitation, Hazardous Materials Procedures, and Requirements.

# 11.12 No Signs

Neither the Contractor nor any other person or entity shall display any signs not required by law or the Contract Documents at the Site, fences trailers, offices, or elsewhere on the Site without specific prior written approval of the County.

#### 12. TRENCHES

## 12.1 Trenches Greater Than Five Feet

Pursuant to Labor Code Section 6705, if the Contract Price exceeds \$25,000 and involves the excavation of any trench or trenches five (5) feet or more in depth, the Contractor shall, in advance of excavation, promptly submit to the County and/or a registered civil or structural engineer employed by the County or Architect, a detailed plan showing the design of shoring for protection from the hazard of caving ground during the excavation of such trench or trenches.

## 12.2 Excavation Safety

If such plan varies from the Shoring System Standards established by the Construction Safety Orders, the plan shall be prepared by a registered civil or structural engineer, but in no case shall such plan be less effective than that required by the Construction Safety Orders. No excavation of such trench or trenches shall be commenced until said plan has been accepted by the County or by the person to whom authority to accept has been delegated by the County.

# 12.3 No Tort Liability of County

Pursuant to Labor Code section 6705, nothing in this Article shall impose tort liability upon the County or any of its employees.

#### 12.4 No Excavation Without Permits

The Contractor shall not commence any excavation Work until it has secured all necessary permits including the required CalOSHA excavation/shoring permit. Any permits shall be prominently displayed on the Site prior to the commencement of any excavation.

## 12.5 Discovery of Hazardous Waste and/or Unusual Conditions

- **12.5.1** Pursuant to Public Contract Code Section 7104, if the Work involves digging trenches or other excavations that extend deeper than four feet below the Surface, the Contractor shall promptly, and before the following conditions are disturbed, notify the County, in writing, including by e-mail, of any:
  - **12.5.1.1** Material that the Contractor believes may be material that is hazardous waste, as defined in Section 25117 of the Health and Safety

- Code that is required to be removed to a Class I, Class II, or Class III disposal site in accordance with provisions of existing law.
- **12.5.1.2** Subsurface or latent physical conditions at the Site differing from those indicated.
- 12.5.1.3 Unknown physical conditions at the Site of any unusual nature, different materially from those ordinarily encountered and generally recognized as inherent in work of the character provided for in the Contract.
- 12.5.2 The County shall promptly investigate the conditions, and if it finds that the conditions do materially so differ, or do involve hazardous waste, and cause a decrease or increase in the Contractor's cost of, or the time required for, performance of any part of the Work, shall issue a Change Order under the procedures described herein.
- 12.5.3 In the event that a dispute arises between County and the Contractor whether the conditions materially differ, or involve hazardous waste, or cause a decrease or increase in the Contractor's cost of, or time required for, performance of any part of the Work, the Contractor shall not be excused from any scheduled completion date provided for in the Contract, but shall proceed with all work to be performed under the Contract. The Contractor shall retain any and all rights provided either by Contract or by law that pertains to the resolution of disputes and protests.

## 13. INSURANCE AND BONDS

#### 13.1 Insurance

All insurance required of Contractor and/or its Subcontractor(s) shall be in amounts set forth in the Special Conditions and include the provisions as set forth herein.

#### 13.1.1 Commercial General Liability and Automobile Liability Insurance

- 13.1.1.1 Contractor shall procure and maintain, during the life of this Contract, Commercial General Liability Insurance and Automobile Liability Insurance that shall protect Contractor, County, Construction Manager(s), and Architect(s) from all claims for bodily injury, property damage, personal injury, death, advertising injury, and medical payments arising from operations under this Contract. Contractor shall ensure that Products Liability and Completed Operations coverage and Fire Damage Liability is included within the above policies and within the required limits, or Contractor shall procure and maintain these coverages separately.
- 13.1.1.2 Subcontractor: Contractor shall require its Subcontractors if any, to procure and maintain similar Commercial General Liability Insurance and Automobile Liability Insurance with minimum limits equal to the amount required of the Contractor. Contractor shall verify Subcontractor's compliance.

#### 13.1.2 Excess Liability Insurance

13.1.2.1 Contractor shall procure and maintain, during the life of this Contract, Excess Liability Insurance that shall protect Contractor, County, Construction Manager(s), and Architect(s) in amounts and including

- the provisions as set forth in the Special Conditions, and that complies with all requirements for Commercial General Liability and Automobile Liability and Employers' Liability Insurance.
- 13.1.2.2 Subcontractor: Contractor shall require its Subcontractor(s) if any, to procure and maintain similar Excess Liability Insurance with minimum limits equal to the amount required of the Contractor. Contractor shall verify Subcontractor's compliance.

## 13.1.3 Workers' Compensation and Employers' Liability Insurance

- 13.1.3.1 In accordance with provisions of Section 3700 of the California Labor Code, the Contractor and every Subcontractor shall be required to secure the payment of compensation to its employees.
- Contractor shall procure and maintain, during the life of this Contract, 13.1.3.2 Workers' Compensation Insurance and Employers' Liability Insurance for all of its employees engaged in work under this Contract, on/or at the Site of the Project. This coverage shall cover, at a minimum, medical and surgical treatment, disability benefits, rehabilitation therapy, and survivors' death benefits. Contractor shall require its Subcontractor(s) if any, to procure and maintain Workers' Compensation Insurance and Employers' Liability Insurance for all employees of Subcontractor(s). Any class of employee or employees not covered by a Subcontractor's insurance shall be covered by Contractor's insurance. If any class of employee or employees engaged in Work under this Contract, on or at the Site of the Project, is not protected under the Workers' Compensation Statute, Contractor shall provide, or shall cause a Subcontractor to provide, adequate insurance coverage for the protection of any employee(s) not otherwise protected before any of those employee(s) commence work.

#### 13.1.4 Builder's Risk Insurance: Builder's Risk "All-Risk" Insurance.

Contractor shall procure and maintain, during the life of this Contract, Builder's Risk (Course of Construction), or similar first party property coverage acceptable to the County, issued on a replacement cost value basis. The cost shall be consistent with the total replacement cost of all insurable Work of the Project included within the Contract Documents. Coverage is to insure against all risks of accidental physical loss and shall include without limitation the perils of vandalism and/or malicious mischief (both without any limitation regarding vacancy or occupancy), sprinkler leakage, civil authority, sonic disturbance, earthquake, flood, collapse, wind, fire, war, terrorism, lightning, smoke, and rioting. Coverage shall include debris removal, demolition, increased costs due to enforcement of all applicable ordinances and/or laws in the repair and replacement of damaged and undamaged portions of the property, and reasonable costs for the Architect's and engineering services and expenses required as a result of any insured loss upon the Work and Project, including completed Work and Work in Progress, to the full insurable value thereof.

# 13.1.5 Proof of Carriage of Insurance and Other Requirements: Endorsements and Certificates.

- 13.1.5.1 Contractor shall not commence Work nor shall it allow any Subcontractor to commence Work under this Contract, until Contractor and its Subcontractor(s) have procured all required insurance and Contractor has delivered in duplicate to the County complete endorsements (or entire insurance policies) and certificates indicating the required coverages have been obtained, and the County has approved these documents.
- **13.1.5.2** Endorsements, certificates, and insurance policies shall include the following:

## **13.1.5.2.1** A clause stating:

"This policy shall not be amended, canceled or modified and the coverage amounts shall not be reduced until notice has been mailed to County, Architect, and Construction Manager stating date of amendment, modification, cancellation or reduction. Date of amendment, modification, cancellation or reduction may not be less than thirty (30) days after date of mailing notice."

- **13.1.5.2.2** Language stating, in particular, those insured, extent of insurance, location and operation to which insurance applies, expiration date, to whom cancelation and reduction notice will be sent, and length of notice period.
- All endorsements, certificates and insurance policies shall state that County, its Supervisors, employees, and agents, Construction Manager(s), and Architect(s) are named additional insureds under all policies except Workers' Compensation Insurance and Employers' Liability Insurance. Contractor's and Subcontractors' insurance policy(s) shall be primary and non-contribution to any insurance or self-insurance maintained by County, its Supervisors, employees and/or agents, Construction Manager(s), and/or Architect(s). All endorsements shall waive any right to subrogation against any of the named additional insureds.
- **13.1.5.4** All policies shall be written on an occurrence form.

# 13.2 Contract Security - Bonds

- **13.2.1** Contractor shall furnish two surety bonds issued by a California admitted surety insurer as follows:
  - Performance Bond: A bond in an amount at least equal to one hundred percent (100%) of Contract Price as security for faithful performance of this Contract.
  - Payment Bond: A bond in an amount at least equal to one hundred percent (100%) of the Contract Price as security for payment of persons performing labor and/or furnishing materials in connection with this Contract.
    - 13.2.2 Cost of bonds shall be included in the Bid and Contract Price.

13.2.3 All bonds related to this Project shall be in the forms set forth in these Contract Documents and shall comply with all requirements of the Contract Documents, including, without limitation, the bond forms.

#### 14. WARRANTY/GUARANTEE/INDEMNITY

# 14.1 Warranty/Guarantee

- **14.1.1** The Contractor shall obtain and preserve for the benefit of the County, manufacturer's warranties on materials, fixtures, and equipment incorporated into the Work.
- **14.1.2** In addition to guarantees required elsewhere, Contractor shall, and hereby does guarantee and warrant all Work furnished on the job against all defects for a period of TWO (2) years after the later of the following dates:
  - **14.1.2.1** The date of completion as defined in Public Contract Code Section 7107, subdivision (c),
  - **14.1.2.2** The commissioning date for the Project, if any.
- 14.1.3 At the County's sole option, Contractor shall repair or replace any and all of that Work, together with any other Work that may be displaced in so doing, that may prove defective in workmanship and/or materials within the warranty period specified in Section 00 65 36 Warranty Form, from date of completion as defined above without expense whatsoever to County. In the event of failure of Contractor and/or Surety to commence and pursue with diligence said replacements or repairs within ten (10) days after being notified in writing, including by e-mail, Contractor and Surety hereby acknowledge and agree that County is authorized to proceed to have defects repaired and made good at expense of Contractor and/or Surety who hereby agree to pay costs and charges therefore immediately on demand. Said notice period shall be forty-eight (48) hours for components essential to operation of the facility, including without limitation fire alarms, water, heat, security systems, and electrical systems.
- 14.1.4 If, in the opinion of County, defective work creates a dangerous condition or requires immediate correction or attention to prevent further loss to County or to prevent interruption of operations of County, County will attempt to give the notice required above. If Contractor or Surety cannot be contacted or does not comply with County's request for correction within a reasonable time as determined by County, County may, notwithstanding the above provision, proceed to make any and all corrections and/or provide attentions the County believes are necessary. The costs of correction or attention shall be charged against Contractor and Surety of the guarantees provided in this Article or elsewhere in this Contract.
- 14.1.5 The above provisions do not in any way limit the guarantees on any items for which a longer guarantee is specified or on any items for which a manufacturer gives a guarantee for a longer period. Contractor shall furnish to County all appropriate guarantee or warranty certificates as indicated in the Specifications or upon request by County.
- **14.1.6** Nothing herein shall limit any other rights or remedies available to County.

#### 14.2 Indemnity

- 14.2.1 To the fullest extent permitted by California law, the Contractor shall indemnify, defend with legal counsel reasonably acceptable to the County, keep and hold harmless the County and its consultants, the Architect and its consultants, the Construction Manager and its consultants, separate contractors, and their respective board members, officers, representatives, contractors, agents, and employees, in both individual and official capacities ("Indemnitees"), against all suits, claims, liabilities, damages, losses, and expenses caused by, arising out of, resulting from, or incidental to, the performance of the Work under this Contract by the Contractor or its Subcontractors to the full extent allowed by the laws of the State of California, and not to any extent that would render these provisions void or unenforceable, including, without limitation, any such suit, claim, damage, loss, or expense attributable to, without limitation, bodily injury, sickness, disease, death, alleged patent violation or copyright infringement, or to injury to or destruction of tangible property (including damage to the Work itself) including the loss of use resulting therefrom, except to the extent caused solely by the negligence, or willful misconduct of the Indemnitees. The County may participate in the defense of any such claim without relieving Contractor of any obligation hereunder. This agreement and obligation of the Contractor shall not be construed to negate, abridge, or otherwise reduce any right or obligation of indemnity that would otherwise exist as to any party or person described herein. This indemnification, defense, and hold harmless obligation includes any failure or alleged failure by Contractor to comply with any provision of law or the Contract Documents, including, without limitation, any stop notice actions, or liens by the California Department of Labor Standards Enforcement. This indemnity obligation shall be for the full amount of all damage to County, including defense costs, and shall not be limited by any insurance limits.
- 14.2.2 The Contractor shall give prompt notice to the County in the event of any injury (including death), loss or damage included herein. Without limitation of the provisions herein, if the Contractor's agreement to indemnify, defend, and hold harmless the Indemnitees as provided herein against liability for damage arising out of bodily injury to persons or damage to property caused by or resulting from the negligence of any of the Indemnitees shall to any extent be or be determined to be void or unenforceable, it is the intention of the parties that these circumstances shall not otherwise affect the validity or enforceability of the Contractor's agreement to indemnify, defend, and hold harmless the rest of the Indemnitees, as provided herein, and in the case of any such suits, claims, damages, losses, or expenses caused in part by the default, negligence, or act or omission of the Contractor, any Subcontractor, anyone directly or indirectly employed by any of them or anyone for whose acts any of them may be liable, and in part by any of the Indemnitees, the Contractor shall be and remain fully liable on its agreements and obligations herein to the full extent permitted by law.
- 14.2.3 In any and all claims against any of the Indemnitees by any employee of the Contractor, any Subcontractor, anyone directly or indirectly employed by any of them or anyone for whose acts any of them may be liable, the Contractor's indemnification obligation herein shall not be limited in any way by any limitation on the amount or type of damages, compensation, or benefits payable by or for the Contractor or any Subcontractor under workers' compensation acts, disability benefit acts, or other employee benefit acts.
- **14.2.4** Contractor shall place in its Subcontractor agreements and cause its Subcontractors to agree to indemnities and insurance obligations in favor of County and

other Indemnities in the exact form and substance of those contained in these General Conditions (00 72 13). Contractor shall require all Subcontractors to comply with all indemnification and insurance requirements of this agreement. Contractor shall verify Subcontractor's compliance.

#### **15.** TIME

#### 15.1 Notice to Proceed

- **15.1.1** County may issue a Notice to Proceed within three (3) months from the date of the Notice of Award. Once Contractor has received the Notice to Proceed, Contractor shall complete the Work within the period of time indicated in the Contract Documents.
- 15.1.2 In the event that the County desires to postpone issuing the Notice to Proceed beyond this 3-month period, it is expressly understood that with reasonable notice to the Contractor, the County may postpone issuing the Notice to Proceed. It is further expressly understood by Contractor that Contractor shall not be entitled to any claim of additional compensation as a result of the postponement of the issuance of the Notice to Proceed.
- 15.1.3 If the Contractor believes that a postponement of issuance of the Notice to Proceed will cause a hardship to Contractor, Contractor may terminate the Contract. Contractor's termination due to a postponement shall be by written notice to County within ten (10) days after receipt by Contractor of County's notice of postponement. It is further understood by Contractor that in the event that Contractor terminates the Contract as a result of postponement by the County, the County shall only be obligated to pay Contractor for the Work that Contractor had performed at the time of notification of postponement. Should Contractor terminate the Contract as a result of a notice of postponement, County shall have the authority to award the Contract to the next lowest responsive, responsible bidder.

#### 15.2 Computation of Time / Adverse Weather

- **15.2.1** The Contractor will only be allowed a time extension for Adverse Weather conditions if requested by Contractor and only if all of the following conditions are met:
  - 15.2.1.1 The weather conditions constitute Adverse Weather, as defined herein and further specified in the Special Conditions;
  - 15.2.1.2 Contractor can verify that the Adverse Weather caused delays in excess of seventy-five percent (75%) for at least five hours, of the normal labor and equipment force toward completion of the day's current controlling item(s) on the latest accepted schedule;
  - **15.2.1.3** The Contractor's crew is dismissed as a result of the Adverse Weather; and
  - **15.2.1.4** The number of days of delay for the month exceeds those indicated in the Special Conditions.
- 15.2.2 A day-for-day extension will only be allowed for those days in excess of those indicated in the Special Conditions. Weather delay time extensions to the contract period will be non-compensable.

Highland Hospital – Koret Building Exhaust Fan and Waterproofing

- 15.2.3 The Contractor shall work seven (7) days per week, if necessary, irrespective of inclement weather, to maintain access and the Construction Schedule, and to protect the Work under construction from the effects of Adverse Weather, all at no further cost to the County.
- 15.2.4 The Contract Time has been determined with consideration given to the average climate weather conditions prevailing in the County in which the Project is located.

### 15.3 Hours of Work

#### 15.3.1 Sufficient Forces

Contractor and Subcontractors shall continuously furnish sufficient forces to ensure the prosecution of the Work in accordance with the Construction Schedule.

# 15.3.2 Performance During Working Hours

Work shall be performed during regular working hours as permitted by the appropriate governmental agency except that in the event of an emergency, or when required to complete the Work in accordance with job progress, Work may be performed outside of regular working hours with the advance written consent of the County and approval of any required governmental agencies.

# 15.4 Progress and Completion

#### 15.4.1 Time of the Essence

Time limits stated in the Contract Documents are of the essence to the Contract. By executing the Agreement, the Contractor confirms that the Contract Time is a reasonable period for performing the Work.

#### 15.4.2 No Commencement Without Insurance

The Contractor shall not commence operations on the Project or elsewhere prior to the effective date of insurance and bonds. The date of commencement of the Work shall not be changed by the effective date of such insurance. If Contractor commences Work without insurance and bonds, all Work is performed at Contractor's peril and shall not be compensable until and unless Contractor secures bonds and insurance pursuant to the terms of the Contract Documents and subject to County claim for damages.

## 15.5 Expeditious Completion

The Contractor shall proceed expeditiously with adequate forces and shall achieve Completion within the Contract Time.

## 16. EXTENSIONS OF TIME – LIQUIDATED DAMAGES

# 16.1 Liquidated Damages

Contractor and County hereby agree that the exact amount of damages for failure to complete the Work within the time specified is extremely difficult or impossible to determine. If the Work is not completed within the time specified in the Contract Documents, it is understood that the County will suffer damage. It being impractical and unfeasible to determine the amount of actual damage, it is agreed the Contractor shall pay to County as fixed and liquidated damages, and not as a penalty, the amount set forth in the

Agreement for each calendar day of delay in completion. Contractor and its Surety shall be liable for the amount thereof pursuant to Government Code Section 53069.85.

# 16.2 Excusable Delay

- 16.2.1 Contractor shall not be charged for liquidated damages because of any delays in completion of Work which are not the fault of Contractor or its Subcontractors, including acts of God as defined in Public Contract Code Section 7105, acts of enemy, epidemics, and quarantine restrictions. Contractor shall, within five (5) calendar days of beginning of any delay, notify County in writing of causes of delay including documentation and facts explaining the delay. County shall review the facts and extent of any delay and shall grant extension(s) of time for completing Work when, in its judgment, the findings of fact justify an extension. Extension(s) of time shall apply only to that portion of Work affected by delay, and shall not apply to other portions of Work not so affected. An extension of time may only be granted if Contractor has timely submitted the Construction Schedule as required herein.
- 16.2.2 Contractor shall notify the County pursuant to the claims provisions in these General Conditions of any anticipated delay and its cause. Following submission of a claim, the County may determine whether the delay is to be considered avoidable or unavoidable, how long it continues, and to what extent the prosecution and completion of the Work might be delayed thereby.
- 16.2.3 In the event the Contractor requests an extension of Contract Time for unavoidable delay, such request shall be submitted in accordance with the provisions in the Contract Documents governing changes in Work. When requesting time, requests must be submitted with full justification and documentation. If the Contractor fails to submit justification, it waives its right to a time extension at a later date. Such justification must be based on the official Construction Schedule as updated at the time of occurrence of the delay or execution of Work related to any changes to the Scope of Work. Any claim for delay must include the following information as support, without limitation:
  - **16.2.3.1** The duration of the activity relating to the changes in the Work and the resources (manpower, equipment, material, etc.) required to perform the activities within the stated duration.
  - 16.2.3.2 Specific logical ties to the Contract Schedule for the proposed changes and/or delay showing the activity/activities in the Construction Schedule that are affected by the change and/or delay.
  - **16.2.3.3** A recovery schedule must be submitted.

## 16.3 No Additional Compensation for Delays Within Contractor's Control

- 16.3.1 Contractor is aware that governmental agencies, including, without limitation, the Department of General Services, gas companies, electrical utility companies, water companies, and other agencies may have to approve Contractor-prepared drawings or approve a proposed installation. Accordingly, Contractor shall include in its bid, time for possible review of its drawings and for reasonable delays and damages that may be caused by such agencies. Thus, Contractor is not entitled to make a claim for damages or delays arising from the review of Contractor's drawings.
- **16.3.2** Contractor shall only be entitled to compensation for delay when all of the following conditions are met:

Highland Hospital – Koret Building Exhaust Fan and Waterproofing

- **16.3.2.1** The County is responsible for the delay:
- **16.3.2.2** The delay is unreasonable under the circumstances involved;
- **16.3.2.3** The delay was not within the contemplation of County and Contractor; and
- **16.3.2.4** Contractor complies with the claims procedure of the Contract Documents.

#### 16.4 Float or Slack in the Schedule

Float or slack is the amount of time between the early start date and the late start date, or the early finish date and the late finish date, of any of the activities in the schedule. Float or slack is not for the exclusive use of or benefit of either the County or the Contractor, but its use shall be determined solely by the County.

#### 17. CHANGES IN THE WORK

## 17.1 No Changes Without Authorization

- 17.1.1 There shall be no change whatsoever in the Drawings, Specifications, or in the Work without an executed Change Order or a written Construction Change Directive authorized by the County as herein provided. County shall not be liable for the cost of any extra work or any substitutions, changes, additions, omissions, or deviations from the Drawings and Specifications unless the County's governing board has authorized the same and the cost thereof has been approved in writing by Change Order or Construction Change Directive. No extension of time for performance of the Work shall be allowed hereunder unless claim for such extension is made at the time changes in the Work are ordered, and such time duly adjusted in writing in the Change Order or Construction Change Directive. The provisions of the Contract Documents shall apply to all such changes, additions, and omissions with the same effect as if originally embodied in the Drawings and Specifications.
- 17.1.2 Contractor shall perform immediately all work that has been authorized by a fully executed Change Order or Construction Change Directive. Contractor shall be fully responsible for any and all delays and/or expenses caused by Contractor's failure to expeditiously perform this Work.
- 17.1.3 Should any Change Order result in an increase in the Contract Price, the cost of that Change Order shall be agreed to, in writing, in advance by Contractor and County and be subject to the monetary limitations set forth in Public Contract Code Section 20137. In the event that Contractor proceeds with any change in Work without a Change Order executed by the County or Construction Change Directive, Contractor waives any claim for additional compensation or time for that additional work.
- 17.1.4 Contractor understands, acknowledges, and agrees that the reason for County authorization is so that County may have an opportunity to analyze the Work and decide whether the County shall proceed with the Change Order or alter the Project so that a change in Work becomes unnecessary.

#### 17.2 Architect Authority to Order Minor Changes

The Architect will have authority to order minor changes in the Work not involving any adjustment in the Contract Price, or an extension of the Contract Time, or a

change that is inconsistent with the intent of the Contract Documents. These changes shall be effected by written Change Order, Construction Change Directive, or by Architect's response(s) to RFI(s).

## 17.3 Change Orders

- 17.3.1 A Change Order is a written instrument prepared and issued by the County and/or the Architect and signed by the County (as authorized by the County's Board of Supervisors), the Contractor, and the Architect, stating their agreement regarding all of the following:
  - **17.3.1.1** A description of a change in the Work;
  - 17.3.1.2 The amount of the adjustment in the Contract Price, if any; and
  - **17.3.1.3** The extent of the adjustment in the Contract Time, if any.

## 17.4 Construction Change Directives

- 17.4.1 A Construction Change Directive is a written order prepared and issued by the County, the Construction Manager, and/or the Architect and signed by the County and the Architect, directing a change in the Work. The County may as provided by law, by Construction Change Directive and without invalidating the Contract, order changes in the Work consisting of additions, deletions, or other revisions. Any dispute as to the sum of the Construction Change Directive or timing of payment shall be resolved pursuant to the Payment and Claims and Disputes provisions herein.
- 17.4.2 The County may issue a Construction Change Directive in the absence of agreement on the terms of a Change Order.

#### 17.5 Force Account Directives

- 17.5.1 When work, for which a definite price has not been agreed upon in advance, is to be paid for on a force account basis, all direct costs necessarily incurred and paid by the Contractor for labor, material, and equipment used in the performance of that Work, shall be subject to the approval of the County and compensation will be determined as set forth herein.
- 17.5.2 The County will issue a Force Account Directive to proceed with the Work on a force account basis, and a not-to-exceed budget will be established by the County.
- 17.5.3 All requirements regarding direct cost for labor, labor burden, material, equipment, and markups on direct costs for overhead and profit described in this section shall apply to Force Account Directives. However, the County will only pay for actual costs verified in the field by the County or its authorized representative(s) on a daily basis.
- 17.5.4 The Contractor shall be responsible for all cost related to the administration of Force Account Directive. The markup for overhead and profit for Contractor modifications shall be full compensation to the Contractor to administer Force Account Directive.
- 17.5.5 The Contractor shall notify the County or its authorized representative(s) at least twenty-four (24) hours prior to proceeding with any of the force account work. Furthermore, the Contractor shall notify the County when it has consumed

eighty percent (80%) of the budget and shall not exceed the budget unless specifically authorized in writing by the County. The Contractor will not be compensated for force account work in the event that the Contractor fails to timely notify the County regarding the commencement of force account work, or exceeding the force account budget.

- daily basis, submit a daily force account report on a form supplied by the County no later than 5:00 p.m. each day. The report shall contain a detailed itemization of the daily labor, material, and equipment used on the force account work only. The names of the individuals performing the force account work shall be included in the daily force account reports. The type and model of equipment shall be identified and listed. The County will review the information contained in the reports, and sign the reports no later than the next work day, and return a copy of the report to the Contractor for their records. The County will not sign, nor will the Contractor receive compensation for work the County cannot verify. The Contractor will provide a weekly force account summary indicating the status of each Force Account Directive in terms of percent complete of the not-to-exceed budget and the estimated percent complete of the work.
- 17.5.7 In the event the Contractor and the County reach a written agreement on a set cost for the work while the work is proceeding based on a Force Account Directive, the Contractor's signed daily force account reports shall be discontinued, and all previously signed reports shall be invalid.

## 17.6 Price Request

#### 17.6.1 Definition of Price Request

A Price Request ("PR") is a written request prepared by the Architect requesting the Contractor to submit to the County and the Architect an estimate of the effect of a proposed change in the Work on the Contract Price and the Contract Time.

## 17.6.2 Scope of Price Request

A Price Request shall contain adequate information, including any necessary Drawings and Specifications, to enable Contractor to provide the cost breakdowns required herein. The Contractor shall not be entitled to any additional compensation for preparing a response to a Price Request, whether ultimately accepted or not.

## 17.7 Proposed Change Order

#### 17.7.1 Definition of Proposed Change Order

A Proposed Change Order ("PCO") is a written request prepared by the Contractor requesting that the County and the Architect issue a Change Order based upon a proposed change to the Work.

#### 17.7.2 Changes in Contract Price

A PCO shall include breakdowns pursuant to the revisions herein to validate any change in Contract Price.

#### 17.7.3 Changes in Time

A PCO shall also include any changes in time required to complete the Project. Any additional time requested shall not be the number of days to make the proposed change, but must be based upon the impact to the Construction Schedule as defined in the

Highland Hospital – Koret Building Exhaust Fan and Waterproofing

Contract Documents. If Contractor fails to request a time extension in a PCO, then the Contractor is thereafter precluded from requesting time and/or claiming a delay.

#### 17.7.4 Unknown and/or Unforeseen Conditions

If Contractor submits a PCO requesting an increase in Contract Price and/or Contract Time that is based at least partially on Contractor's assertion that Contractor has encountered unknown and/or unforeseen condition(s) on the Project, then Contractor shall base the PCO on provable information that, beyond a reasonable doubt and to the County's satisfaction, demonstrates that the unknown and/or unforeseen condition(s) were actually unknown and/or unforeseen and that the condition(s) were reasonably unknown and/or unforeseen. If not, the County shall deny the PCO and the Contractor shall complete the Project without any increase in Contract Price and/or Contract Time based on that PCO.

# 17.8 Format for Proposed Change

17.8.1 The following format shall be used as applicable by the County and the Contractor (e.g. Change Orders, PCO's) to communicate proposed additions and deductions to the Contract, supported by attached documentation. In no case shall the Contractor's total mark-up exceed 26.5%.

	SUBCONTRACTOR PERFORMED WORK	ADD	<u>DEDUCT</u>
a)	Material (attach itemized quantity and unit cost plus sales tax)		
b)	Add Labor (attach itemized hours and rates, fully encumbered)		
c)	Add Equipment (attach suppliers' invoice)		
d)	Subtotal		
e)	Add Subcontractor's overhead and profit, not to exceed ten percent (10%) of item (d)		
f)	Subtotal		
g)	Add Contractor's overhead and profit, not to exceed ten percent (10%) of Item (f)		
h)	Subtotal		
i)	Add Bond and Insurance, not to exceed one percent (1%) of Item (h)		
j)	TOTAL		
k)	<u>Time</u>	Days	
	CONTRACTOR PERFORMED WORK	ADD	DEDUCT
	Material (attach itemized quantity and unit cost plus sales tax)		
a)			

Highland Hospital – Koret Building Exhaust Fan and Waterproofing

b)	Add Labor (attach itemized hours and rates, fully encumbered)	
c)	Add Equipment (attach suppliers' invoice)	
d)	Subtotal	
e)	Add Contractor's overhead and profit, not to exceed ten percent (10%) of item (d)	
f)	Subtotal	
g)	Add Bond and Insurance, not to exceed one percent (1%) of Item (f)	
h)	TOTAL	
i)	<u>Time</u>	Days

## 17.9 Change Order Certification

**17.9.1** All Change Orders and PCOs must include the following certification by the Contractor:

- 17.9.1.1 The undersigned Contractor approves the foregoing as to the changes, if any, and the Contract Price specified for each item and as to the extension of time allowed, if any, for completion of the entire Work as stated herein, and agrees to furnish all labor, materials, and service, and perform all work necessary to complete any additional work specified for the consideration stated herein. Submission of sums which have no basis in fact or which Contractor knows are false are at the sole risk of Contractor and may be a violation of the False Claims Act set forth under Government Code section 12650 et seq. It is understood that the changes herein to the Contract shall only be effective when approved by the Board of Supervisors.
- 17.9.1.2 It is expressly understood that the value of the extra Work or changes expressly includes any and all of the Contractor's costs and expenses, both direct and indirect, resulting from additional time required on the Project or resulting from delay to the Project. Any costs, expenses, damages, or time extensions not included are deemed waived.

#### 17.10 Determination of Change Order Cost

The amount of the increase or decrease in the Contract Price from a Change Order, if any, shall be determined by one or more of the following ways as applicable to a specific situation and at the County's discretion:

- **17.10.1** County acceptance of a PCO;
- **17.10.2** By unit prices contained in Contractor's original bid;
- **17.10.3** By agreement between County and Contractor.

#### 17.11 Allowable Costs

Allowable costs for any change order shall be limited to the following:

- **17.11.1** Costs of labor, including social security, Medicare, and unemployment insurance, fringe benefits required workers' compensation insurance.
- **17.11.2** Costs of first line supervision labor, including labor burden as described in paragraph 1. "First-Line Supervision" shall mean a working foreman or lead craft worker other than the project superintendent;
- 17.11.3 Actual costs of the project superintendent associated with any period of compensable delay caused by issuance of the change order. In the absence of a compensable delay, all of the project superintendent's time is considered to have been paid for as part of the overhead;
  - **17.11.4** Actual costs of materials, including sales tax and delivery;
- 17.11.5 Rental costs of machinery and equipment, exclusive of small tools, whether rented from the Contractor or others. For Contractor and Subcontractor-owned equipment, payment will be made at rental rates listed for equipment in California Department of Transportation official equipment rental rate schedule. For rental equipment, payment will be made based on actual rental invoices. Rental rates paid shall be deemed to cover cost of fuel, oil, lubrication, supplies, small tools, necessary attachments, repairs and maintenance of any kind, depreciation, storage, insurance and all incidentals;
- **17.11.6** Overhead and Profit as specified below. "Overhead" shall include the following:
  - 17.11.6.1 Preparation of all paperwork related to changes in the Work, including field review, estimating and cost breakdown; coordination and supervision, both office and field, including the project superintendent; vehicles including has and maintenance; small tools, incidentals and consumables; engineering, detailing, and revisions to shop drawings and as-built drawings; general office expense; extended and unabsorbed home office overhead; warranty, all taxes; and all other expenses not specifically described in items 17.11.1 through 17.11.5.
  - **17.11.6.2** The actual costs of insurance premiums required by this contract and associated with the change order work will be reimbursed by the County
- 17.11.7 Upon receipt of a PCO/Change Directive, the Contractor shall promptly proceed with the change in the Work involved and advise the Construction Manager within seven (7) calendar days of the Contractor's agreement or disagreement with the method, if any, provided in the PCO/Change Directive for determining the proposed adjustment in the Contract Sum or Contract Time.
- **17.11.8** Failure to respond to and return a PCO/Change Directive to the County within (7) days indicates the Contractor's agreement therewith, including adjustment in Contract Sum and Contract Time or the method for determining them. Such agreement shall be effective immediately and shall be recorded as a Change Order.

## 17.12 Deductive Change Orders

All deductive Change Order(s) must be prepared pursuant to the provisions herein. If Contractor offers a proposed amount for a Deductive Change Order(s), Contractor shall include a minimum of five percent (5%) total profit and overhead to be deducted with the amount of the work of the Change Order(s). If Subcontractor work is involved, Subcontractors shall also include a minimum of five percent (5%) profit and overhead to be deducted with the amount of its deducted work. Any deviation from this provision shall not be allowed.

#### 17.13 Discounts, Rebates, and Refunds

For purposes of determining the cost, if any, of any change, addition, or omission to the Work hereunder, all trade discounts, rebates, refunds, and all returns from the sale of surplus materials and equipment shall accrue and be credited to the Contractor, and the Contractor shall make provisions so that such discounts, rebates, refunds, and returns may be secured, and the amount thereof shall be allowed as a reduction in the Contractor's cost in determining the actual cost of construction for purposes of any change, addition, or omission in the Work as provided herein.

#### 17.14 Accounting Records

With respect to portions of the Work performed by Change Orders and Construction Change Directives, the Contractor shall keep and maintain cost-accounting records satisfactory to the County, which shall be available to the County on the same terms as any other books and records the Contractor is required to maintain under the Contract Documents.

# 17.15 Notice Required

If the Contractor desires to make a claim for an increase in the Contract Price or any extension of the Contract Time for completion, it shall notify the County pursuant to the provisions herein. No claim shall be considered unless made in accordance with this subparagraph. Contractor shall proceed to execute the Work, even though, the adjustment may not have been agreed upon. Any change in the Contract Price or extension of the Contract Time resulting from such claim shall be authorized by a Change Order.

## 17.16 Applicability to Subcontractors

Any requirements under this Article shall be equally applicable to Change Orders, or Construction Change Directives issued to Subcontractors by the Contractor to the extent as required by the Contract Documents.

# 17.17 Alteration to Change Order Language

Contractor shall not alter Change Orders or reserve time in Change Orders. Contractor shall execute finalized Change Orders and proceed under the provisions herein with proper notice.

# 17.18 Failure of Contractor to Execute Change Order

Contractor shall be in default of the Contract if Contractor fails to execute a Change Order when the Contractor agrees with the addition and/or deletion of the Work in that Change Order.

# 18. REQUEST FOR INFORMATION

18.1 The Contractor shall coordinate the Work so that dimensions are verified, and clarifications that may affect the work are identified to allow for resolution without delaying the Work. The Contractor is responsible to submit a Request for Information as soon as the issue requiring clarification is identified. The Contractor shall be responsible for any delay in the construction progress due to any untimely submission of a Request for Information for A/E's review. Non-receipt of a Request for Information, or proceeding with Work pertaining to the Request for Information shall be construed as relieving the County of any Claim for added cost or extension of time.

#### 18.2 Reference Contract Documents

Any Request for Information shall reference all applicable Contract Document(s), including Specification section(s), detail(s), page number(s), drawing number(s), and sheet number(s), etc. The Contractor shall make suggestions and interpretations of the issue raised by each Request for Information. A Request for Information cannot modify the Contract Price, Contract Time, or the Contract Documents.

# **18.3** Contractor Responsible For Costs

Contractor shall be responsible for any costs incurred for professional services which County may deduct from any amounts owing to the Contractor if a Request for Information requests an interpretation or decision of a matter where the information sought is equally available to the party making the request. County, at its sole discretion, shall deduct from and/or invoice Contractor for all the professional services arising herein.

#### 19. PAYMENTS

#### 19.1 Contract Price

The Contract Price is stated in the Agreement and, including authorized adjustments, is the total amount payable by the County to the Contractor for performance of the Work under the Contract Documents.

## 19.2 Applications for Progress Payments

## 19.2.1 Procedures for Applications for Progress Payments

- 19.2.1.1 Not before the fifth (5th) day of each calendar month during the progress of the Work, Contractor shall submit to the County and the Architect an itemized Application for Payment for operations completed in accordance with the Schedule of Values. Such application shall be notarized, if required and supported by the following or each portion thereof unless waived by the County in writing:
  - **19.2.1.1.1** The amount paid to the date of the Application to the Contractor, to all its Subcontractors, and all others furnishing labor, material, or equipment for its Contract;
  - **19.2.1.1.2** The amount being requested under the Application for Payment by the Contractor on its own behalf and separately stating the amount requested on behalf of each of the Subcontractors and all others furnishing labor, material, and equipment under the Contract;

- **19.2.1.1.3** The balance that will be due to each of such entities after said payment is made;
- **19.2.1.1.4** A certification that the Record Drawings and annotated Specifications are current;
- **19.2.1.1.5** Itemized breakdown of work done for the purpose of requesting partial payment;
- **19.2.1.1.6** An updated and acceptable construction schedule in conformance with Section 10.1 above:
- **19.2.1.1.7** The additions to and subtractions from the Contract Price and Contract Time:
- **19.2.1.1.8** A total of the retentions held;
- **19.2.1.1.9** Material invoices, evidence of equipment purchases, rentals, and other support and details of cost as the County may require from time to time;
- **19.2.1.1.10** The percentage of completion of the Contractor's Work by line item;
- **19.2.1.1.11** Schedule of Values updated from the preceding Application for Payment;
- **19.2.1.1.12** A duly completed and executed conditional waiver and release upon progress payment compliant with Civil Code Section 3262 from the Contractor and each subcontractor of any tier and supplier to be paid from the current progress payment;
- **19.2.1.1.13** A duly completed and executed unconditional waiver and release upon progress payment compliant with Civil Code Section 3262 from the Contractor and each subcontractor of any tier and supplier that was paid from the previous progress payment; and
- **19.2.1.1.14** A certification by the Contractor of the following:

The Contractor warrants title to all Work performed as of the date of this payment application. The Contractor further warrants that all Work performed as of the date of this payment application is free and clear of liens, claims, security interests, or encumbrances in favor of the Contractor, Subcontractors, material and equipment suppliers, workers, or other persons or entities making a claim by reason of having provided labor, materials, and equipment relating to the Work, except those of which the County has been informed.

- **19.2.2** The Contractor shall be subject to the False Claims Act set forth under Government Code Section 12650 et seq., for information provided with any Application for Progress Payment.
  - 19.2.3 Prerequisites for Progress Payments

- 19.2.3.1 First Payment Request: The following items, if applicable, must be completed before the County will accept and/or process the Contractor's first payment request:
  - **19.2.3.1.1** Installation of the Project sign;
  - **19.2.3.1.2** Installation of field office;
  - **19.2.3.1.3** Installation of temporary facilities and fencing;
  - **19.2.3.1.4** Schedule of Values;
  - **19.2.3.1.5** Contractor's Construction Schedule in conformance with Section 10.1.1.1 above;
  - **19.2.3.1.6** Schedule of unit prices, if applicable;
  - 19.2.3.1.7 Submittal Schedule;
  - **19.2.3.1.8** Receipt by Architect of all submittals due as of the date of the payment application;
  - **19.2.3.1.9** Copies of necessary permits;
  - **19.2.3.1.10** Copies of authorizations and licenses from governing authorities;
  - **19.2.3.1.11** Initial progress report;
  - 19.2.3.1.12 Surveyor qualifications;
  - **19.2.3.1.13** Written acceptance of County's survey of rough grading, if applicable;
  - **19.2.3.1.14** List of all Subcontractors, with names, license numbers, telephone numbers, and Scope of Work;
  - **19.2.3.1.15** All bonds and insurance endorsements:
  - **19.2.3.1.16** Resumes of Contractor's project manager, and if applicable, job site secretary, record documents recorder, and job site superintendent; and
  - **19.2.3.1.17** Safety plan.
- 19.2.3.2 Second Payment Request. The County will not process the second payment request until and unless all submittals and Shop Drawings have been accepted for review by the Architect, and Contractor's Schedule has been accepted as in compliance with Section 10.1.1.1 above.
- 19.2.3.3 No Waiver of Criteria. Any payments made to Contractor where criteria set forth herein have not been met shall not constitute a waiver of said criteria by County. Instead, such payment shall be construed as a good faith effort by County to resolve differences so Contractor may pay its Subcontractors and suppliers. Contractor agrees that failure to submit such items may constitute a breach of contract by Contractor and may subject Contractor to termination.

#### 19.3 Progress Payments

## 19.3.1 County's Approval of Application for Payment

- **19.3.1.1** Upon receipt of an Application for Payment, the County shall act in accordance with both of the following:
  - **19.3.1.1.1** Each Application for Payment shall be reviewed by the County as soon as practicable after receipt for the purpose of determining that the Application for Payment is a proper Application for Payment.
  - 19.3.1.1.2 Any Application for Payment determined not to be a proper Application for Payment suitable for payment shall be returned to the Contractor as soon as practicable, but not later than seven (7) calendar days, after receipt. An Application for Payment returned pursuant to this paragraph shall be accompanied by a document setting forth in writing, including by e-mail, the reasons why the Application for Payment is not proper. The number of days available to the County to make a payment without incurring interest pursuant to this section shall be reduced by the number of days by which the County exceeds this seven-day return requirement.
  - **19.3.1.1.3** An Application for Payment shall be considered properly executed if funds are available for payment of the Application for Payment, and payment is not delayed due to an audit inquiry by the financial officer of the County.
- 19.3.1.2 The County's review of the Contractor's Application for Payment will be based on the County's and the Architect's observations at the Site and the data comprising the Application for Payment that the Work has progressed to the point indicated and that, to the best of the County's and the Architect's knowledge, information, and belief, the quality of the Work is in accordance with the Contract Documents. The foregoing representations are subject to:
  - **19.3.1.2.1** Observation of the Work for general conformance with the Contract Documents,
  - **19.3.1.2.2** Results of subsequent tests and inspections,
  - **19.3.1.2.3** Minor deviations from the Contract Documents correctable prior to completion, and
  - **19.3.1.2.4** Specific qualifications expressed by the Architect.
- 19.3.1.3 County's approval of the certified Application for Payment shall be based on Contractor complying with all requirements for a fully complete and valid certified Application for Payment.

#### 19.3.2 Payments to Contractor

19.3.2.1 Within thirty (30) days after approval of the Application for Payment, Contractor shall be paid a sum equal to ninety percent (90%) of the value of the Work performed (as verified by Architect and certified

by Contractor) up to the last day of the previous month, less the aggregate of previous payments and amount to be withheld. The value of the Work completed shall be Contractor's best estimate. No inaccuracy or error in said estimate shall operate to release the Contractor, or any Surety upon any bond, from damages arising from such Work, or from the County's right to enforce each and every provision of this Contract, and the County shall have the right subsequently to correct any error made in any estimate for payment.

- 19.3.2.2 The Contractor shall not be entitled to have any payment requests processed, or be entitled to have any payment made for Work performed, so long as any lawful or proper direction given by the County concerning the Work or any portion thereof remains incomplete.
- 19.3.2.3 If the County fails to make any progress payment within thirty (30) days after receipt of an undisputed and properly submitted Application for Payment by the Contractor, the County shall pay interest to the Contractor equivalent to the legal rate set forth in subdivision (a) of Section 685.010 of the Code of Civil Procedure.

#### **19.3.3 No Waiver**

No payment by County hereunder shall be interpreted so as to imply that County has inspected, approved, or accepted any part of the Work. Notwithstanding any payment, the County may enforce each and every provision of this Contract. The County may correct or require correction of any error subsequent to any payment.

#### 19.3.4 Removal of Liens

- 19.3.4.1 If a lien or a claim based on a stop notice of any nature should at any time be filed against the Work or any County property, by any entity that has supplied material or services at the request of the Contractor, Contractor and Contractor's Surety shall promptly, on demand by County and at Contractor's and Surety's own expense, take any and all action necessary to cause any such lien or a claim based on a stop notice to be released or discharged immediately therefrom.
- 19.3.4.2 If the Contractor fails to furnish to the County within ten (10) calendar days after demand by the County, satisfactory evidence that a lien or a claim based on a stop notice has been so released, discharged, or secured, the County may discharge such indebtedness and deduct the amount required therefor, together with any and all losses, costs, damages, and attorney's fees and expense incurred or suffered by County from any sum payable to Contractor under the Contract.

# 19.4 Decisions to Withhold Payment

## 19.4.1 Reasons to Withhold Payment

The County may withhold payment in whole, or in part, to the extent reasonably necessary to protect the County if, in the County's opinion, the representations to the County required herein cannot be made. The County may withhold payment, in

Highland Hospital – Koret Building Exhaust Fan and Waterproofing

whole, or in part, to such extent as may be necessary to protect the County from loss because of, but not limited to:

- **19.4.1.1** Defective Work not remedied within the time frames noted in Section 14 hereof of written notice to Contractor;
- **19.4.1.2** Stop Notices, or other liens served upon the County as a result of the Contract;
- **19.4.1.3** Liquidated damages assessed against the Contractor;
- 19.4.1.4 The cost of completion of the Contract, if there exists reasonable doubt that the Work can be completed for the unpaid balance of the Contract Price or by the completion date;
- **19.4.1.5** Damage to the County or other contractor(s);
- **19.4.1.6** Unsatisfactory prosecution of the Work by the Contractor;
- **19.4.1.7** Failure to store and properly secure materials;
- 19.4.1.8 Failure of the Contractor to submit, on a timely basis, proper, sufficient, and acceptable documentation required by the Contract Documents, including, without limitation, a Construction Schedule, Schedule of Submittals, Schedule of Values, Monthly Progress Schedules, Shop Drawings, Product Data and Samples, Proposed product lists, executed Change Orders, and/or properly completed Elation updates;
- **19.4.1.9** Failure of the Contractor to maintain Record Drawings;
- **19.4.1.10** Erroneous estimates by the Contractor of the value of the Work performed, or other false statements in an Application for Payment;
- **19.4.1.11** Unauthorized deviations from the Contract Documents;
- **19.4.1.12** Failure of the Contractor to prosecute the Work in a timely manner in compliance with the Construction Schedule established progress schedules, and/or completion dates;
- **19.4.1.13** Failure to properly pay prevailing wages as defined in Labor Code section 1720 et seq., and/or failure to comply with any other Labor Code requirements,
- **19.4.1.14** Failure to properly maintain or clean up the Site;
- **19.4.1.15** Payments to indemnify, defend, or hold harmless the County;
- 19.4.1.16 Any payments due to the County, including but not limited to payments for failed tests, utility changes or permits;
- **19.4.1.17** Failure to pay Subcontractor(s) or supplier(s) as required by law and by the Contract Documents;
- **19.4.1.18** Contractor is otherwise in breach, default, or in substantial violation of any provision of this Contract.
  - 19.4.2 Reallocation of Withheld Amounts

- 19.4.2.1 County may, at its discretion, apply any withheld amount to pay outstanding claims or obligations as defined herein. In so doing, County shall make such payments on behalf of Contractor. If any payment is so made by County, then that amount shall be considered a payment made under Contract by County to Contractor and County shall not be liable to Contractor for any payment made in good faith. These payments may be made without prior judicial determination of claim or obligation. County will render Contractor an accounting of funds disbursed on behalf of Contractor.
- 19.4.2.2 If Contractor defaults or neglects to carry out the Work in accordance with the Contract Documents or fails to perform any provision thereof, County may, after FORTY-EIGHT (48) hours written notice to the Contractor and, without prejudice to any other remedy, make good such deficiencies. The County shall adjust the total Contract Price by reducing the amount thereof by the cost of making good such deficiencies. If County deems it inexpedient to correct Work that is damaged, defective, or not done in accordance with Contract provisions, an equitable reduction in the Contract Price (of at least one hundred twenty-five percent (125%) of the estimated reasonable value of the nonconforming Work) shall be made therefor.

## 19.4.3 Payment After Cure

When Contractor removes the grounds for declining approval, payment shall be made for amounts withheld because of them. No interest shall be paid on any retainage or amounts withheld due to the failure of the Contractor to perform in accordance with the terms and conditions of the Contract Documents.

#### 19.5 Subcontractor Payments

#### 19.5.1 Payments to Subcontractors

No later than seven (7) days after receipt, or pursuant to Business and Professions Code Section 7108.5 and Public Contract Code Section 7107, the Contractor shall pay to each Subcontractor, out of the amount paid to the Contractor on account of such Subcontractor's portion of the Work, the amount to which said Subcontractor is entitled. The Contractor shall, by appropriate agreement with each Subcontractor, require each Subcontractor to make payments to its Sub-subcontractors in a similar manner.

## 19.5.2 No Obligation of County for Subcontractor Payment

The County shall have no obligation to pay or to see to the payment of, money to a Subcontractor except as may otherwise be required by law.

## 19.5.3 Joint Checks

County shall have the right in its sole discretion if necessary for the protection of the County, to issue joint checks made payable to the Contractor and Subcontractors and material or equipment suppliers. The joint check payees shall be responsible for the allocation and disbursement of funds included as part of any such joint payment. In no event shall any joint check payment be construed to create any contract between the County and a Subcontractor of any tier, any obligation from the County to such Subcontractor, or rights in such Subcontractor against the County.

#### 20. COMPLETION OF THE WORK

#### 20.1 Completion

- **20.1.1** County will accept completion of the Contract and have the Notice of Completion recorded when the entire Work shall have been completed to the satisfaction of County.
- **20.1.2** The Work may only be accepted as complete by action of the County Board of Supervisors.
- 20.1.3 County, at its sole option, may accept completion of Contract and have the Notice of Completion recorded when the entire Work shall have been completed to the satisfaction of County, except for minor corrective items, as distinguished from incomplete items. If Contractor fails to complete all minor corrective items within thirty (30) days after the date of the County's acceptance of completion, County shall withhold from the final payment one hundred fifty percent (150%) of an estimate of the amount sufficient to complete the corrective items, as determined by County, until the item(s) are completed.
- **20.1.4** At the end of the thirty-five (35) day period, if there are any items remaining to be corrected, County may elect to proceed as provided herein related to adjustments to Contract Price, and/or County's right to perform the Work of the Contractor.

#### **20.2** Close-Out Procedures

#### **20.2.1 Punch List**

The Contractor shall notify the Architect when Contractor considers the Work complete. Upon notification, Architect will prepare a list of minor items to be completed or corrected ("Punch List"). The Contractor and/or its Subcontractors shall proceed promptly to complete and correct items on the Punch List. Failure to include an item on Punch List does not alter the responsibility of the Contractor to complete all Work in accordance with the Contract Documents.

## **20.2.2** Close-Out Requirements

#### **20.2.2.1** Utility Connections

Buildings shall be connected to water, gas, sewer, and electric services, complete and ready for use. Service connections shall be made, and existing services reconnected.

#### 20.2.2.2 Record Drawings

- **20.2.2.2.1** Contractor shall provide exact "as-built" Record Drawings of the Work upon completion of the Project as indicated in the Specifications.
- 20.2.2.2 Contractor is liable and responsible for any and all inaccuracies in as-built Record Drawings, even if inaccuracies become evident at a future date.
- **20.2.2.2.3** Upon completion of the Work and as a condition precedent to approval of final payment, Contractor shall

obtain the Architect's approval of the corrected prints and employ a competent draftsman to transfer the "asbuilt" information to the most current version of Autocad that is, at that time, currently utilized for plan check submission by either the County, the Construction Manager and/or the Architect, and submit electronic files. When completed, Contractor shall deliver corrected electronic files acceptable to County with Autocad file to the County.

**20.2.2.2.4** Maintenance Manuals: Contractor shall prepare all operation and maintenance manuals and date as indicated in the Specifications.

## **20.3** Final Inspection

- 20.3.1 Contractor shall comply with Punch List procedures as provided herein, and maintain the presence of a Project Superintendent and Project Manager until the Punch List is complete to ensure proper and timely completion of the Punch List. Under no circumstances shall Contractor demobilize its forces prior to completion of the Punch List. Upon receipt of Contractor's written notice that all of the Punch List items have been fully completed, and the Work is ready for final inspection and acceptance, Architect and Construction Manager will inspect the Work and shall submit to Contractor and County a final inspection report noting the Work, if any, required in order to complete in accordance with the Contract Documents. Absent unusual circumstances, this report shall consist of the Punch List items not yet satisfactorily completed.
- **20.3.2** Upon Contractor's completion of all items on the Punch List and any other uncompleted portions of the Work, the Contractor shall notify the County and Architect, who shall again inspect such Work. If the Architect finds the Work complete and acceptable under the Contract Documents, the Architect will notify Contractor, who shall then jointly submit to the Architect and the County its final Application for Payment.

# **20.3.3** Final Inspection Requirements

Before calling for final inspection, Contractor shall determine that the following have been performed:

20.3.3.1	The Work has been completed.
20.3.3.2	All life-safety items are completed and in working order.
20.3.3.3	Mechanical and Electrical Work are complete and tested, fixtures are in place, connected, and ready for tryout.
20.3.3.4	Electrical circuits scheduled in panels and disconnect switches labeled.
20.3.3.5	Painting and special finishes complete.
20.3.3.6	Doors complete with hardware, cleaned of protective film, relieved of sticking or binding, and in working order.
20.3.3.7	Tops and bottoms of doors sealed.
20.3.3.8	Floors waxed and polished as specified.

Highland Hospital – Koret Building Exhaust Fan and Waterproofing

20.3.3.9	Broken glass replaced and glass cleaned.
20.3.3.10	Grounds cleared of Contractor's equipment, raked clean of debris,

and trash removed from Site.

**20.3.3.11** Work cleaned, free of stains, scratches, and other foreign matter, of damaged and broken material, replaced.

**20.3.3.12** Finished and decorative work shall have marks, dirt, and superfluous labels removed.

**20.3.3.13** Final cleanup, as provided herein.

## **20.4** Costs of Multiple Inspections

More than two (2) requests of the County to make a final inspection shall be considered an additional service of County, Architect, and/or Construction Manager, and all subsequent costs will be invoiced to Contractor and if funds are available, withheld from remaining payments.

## 20.5 Partial Occupancy or Use Prior to Completion

# 20.5.1 County's Rights

The County may occupy or use any completed or partially completed portion of the Work at any stage. The County and the Contractor shall agree in writing to the responsibilities assigned to each of them for payments, security, maintenance, heat, utilities, damage to the Work, insurance, the period for correction of the Work, and the commencement of warranties required by the Contract Documents. Any dispute as to responsibilities shall be resolved pursuant to the Claims and Disputes provisions herein, with the added provision that during the dispute process, the County shall have the right to occupy or use any portion of the Work that it needs or desires to use.

#### 20.5.2 Inspection Prior to Occupancy or Use

Immediately prior to partial occupancy or use, the County, the Contractor, and the Architect shall jointly inspect the area to be occupied or portion of the Work to be used in order to determine and record the condition of the Work.

#### **20.5.3 No Waiver**

Unless otherwise agreed upon, partial or entire occupancy or use of a portion or portions of the Work shall not constitute beneficial occupancy or acceptance of the Work not complying with the requirements of the Contract Documents.

#### 21. FINAL PAYMENT AND RETENTION

# 21.1 Final Payment

Upon receipt and approval of a valid and final Application for Payment, the Architect will issue a final Certificate of Payment. The County shall thereupon jointly inspect the Work and either accept the Work as complete or notify the Architect and the Contractor in writing of reasons why the Work is not complete. Upon acceptance of the Work of the Contractor as fully complete (that, absent unusual circumstances, will occur when the Punch List items have been satisfactorily completed), the County shall record a

Notice of Completion with the County Recorder, and the Contractor shall, upon receipt of final payment from the County, pay the amount, due Subcontractors.

# 21.2 Prerequisites for Final Payment

The following conditions must be fulfilled prior to Final Payment:

- **21.2.1** A full and final waiver or release of all Stop Notices in connection with the Work shall be submitted by Contractor, including a release of Stop Notice in recordable form, together with (to the extent permitted by law) a copy of the full and final release of all Stop Notice rights.
  - 21.2.1.1 A duly completed and executed conditional waiver and release upon final payment compliant with Civil Code section 3262 from the Contractor and each subcontractor of any tier and supplier to be paid from the current progress payment;
  - 21.2.1.2 A duly completed and executed unconditional waiver and release upon progress payment compliant with Civil Code section 3262 from the Contractor and each subcontractor of any tier and supplier that was paid from the previous progress payment; and
  - 21.2.1.3 The Contractor shall have made all corrections to the Work that are required to remedy any defects therein, to obtain compliance with the Contract Documents or any requirements of applicable codes and ordinances, or to fulfill any of the orders or directions of County required under the Contract Documents.
- **21.2.2** Each Subcontractor shall have delivered to the Contractor all written guarantees, warranties, applications, and bonds required by the Contract Documents for its portion of the Work.
- **21.2.3** Contractor must have completed all requirements set forth under "Close-Out Procedures," Including, without limitation, an approved set of complete "asbuilt" Record Drawings.
- **21.2.4** Architect shall have issued its written approval that final payment can be made.
- **21.2.5** The Contractor shall have delivered to the County all manuals and materials required by the Contract Documents.
- **21.2.6** The Contractor shall have completed final clean up as provided herein.

#### 21.3 Retention

- **21.3.1** The retention, less any amounts disputed by the County or that the County has the right to withhold pursuant to provisions herein, shall be paid:
  - **21.3.1.1** After approval of the County by the Architect's Certificate of Payment,
  - 21.3.1.2 After the satisfaction of the conditions set forth herein, and
  - 21.3.1.3 After thirty-five (35) days after the recording of the Notice of Completion by County.

Highland Hospital – Koret Building Exhaust Fan and Waterproofing

**21.3.2** No interest shall be paid on any retention, or on any amounts withheld due to a failure of the Contractor to perform, in accordance with the terms and conditions of the Contract Documents, except as provided to the contrary in any Escrow Agreement between the County and the Contractor pursuant to Public Contract Code section 22300.

#### 21.4 Substitution of Securities

The County will permit the substitution of securities in accordance with the provisions of Public Contract Code section 22300.

#### 22. UNCOVERING OF WORK

If a portion of the Work is covered without Architect approval or not in compliance with the Contract Documents, it must, if required in writing, including by email, by the County or the Architect, be uncovered for the Architect's observation and be replaced at the Contractor's expense without change in the Contract Price or Contract Time.

If a portion of the Work has been covered, which the Project Inspector or the Architect has not specifically requested to observe prior to its being covered, the County, Project Inspector, or the Architect may request to see that Work, and it shall be uncovered by the Contractor. If that Work is in accordance with the Contract Documents, costs of uncover and replacement shall, by appropriate Change Order, be charged to the County. If that Work is not in accordance with Contract Documents, the Contractor shall pay these costs unless the condition was caused by the County or a separate contractor, in which event the County shall be responsible for payment of such costs to the Contractor.

# 23. NONCONFORMING WORK, CORRECTION OF WORK AND COUNTY'S RIGHT TO PERFORM WORK

#### 23.1 Nonconforming Work

- **23.1.1** Contractor shall promptly remove from Premises all Work identified by County as failing to conform to the Contract Documents whether incorporated or not. Contractor shall promptly replace and re-execute its own Work to comply with the Contract Documents without additional expense to the County and shall bear the expense of making good all work of other contractors destroyed or damaged by any removal or replacement pursuant hereto and/or any delays to the County or other Contractors caused thereby.
- 23.1.2 If Contractor does not remove Work that County has identified as failing to conform to the Contract Documents within a reasonable time, not to exceed FORTY-EIGHT (48) hours, County may remove it and may store any material at Contractor's expense. If Contractor does not pay expense(s) of that removal within ten (10) days' time thereafter, County may, upon ten (10) days' written notice, sell any material at auction or at private sale and shall deduct all costs and expenses incurred by the County and/or County may withhold those amounts from payment(s) to Contractor.

## 23.2 Correction of Work

# 23.2.1 Correction of Rejected Work

Pursuant to the notice provisions herein, the Contractor shall promptly correct the Work rejected by the County or the Architect as failing to conform to the requirements

of the Contract Documents, whether observed before or after Completion and whether or not fabricated, installed, or completed. The Contractor shall bear costs of correcting the rejected Work, including additional testing, inspections, and compensation for the Architect's services and expenses made necessary thereby.

# 23.2.2 Warranty Corrections

If, within the warranty period specified in 00 65 36 Warranty Form, after the date of Completion of the Work or a designated portion thereof, or after the date of commencement of warranties established hereunder, or by the terms of an applicable special warranty required by the Contract Documents, any of the Work is found to be not in accordance with the requirements of the Contract Documents, the Contractor shall correct it promptly after receipt of written notice from the County to do so. This period of two (2) years shall be extended with respect to portions of the Work first performed after Completion by the period of time between Completion and the actual performance of the Work. This obligation hereunder shall survive acceptance of the Work under the Contract and termination of the Contract. The County shall give such notice promptly after discovery of the condition.

# 23.3 County's Right to Perform Work

- **23.3.1** If the Contractor should neglect to prosecute the Work properly or fail to perform any provisions of this contract, the County, after FORTY-EIGHT (48) hours written notice to the Contractor, may, without prejudice to any other remedy it may have, make good such deficiencies and may deduct the cost thereof from the payment then or thereafter due the Contractor.
- **23.3.2** If it is found at any time, before or after completion of the Work, that Contractor has varied from the Drawings and/or Specifications, including, but not limited to, variation in material, quality, form, or finish, or in the amount or value of the materials and labor used, County may require at its option:
  - 23.3.2.1 That all such improper Work be removed, remade or replaced, and all work disturbed by these changes be made good by Contractor at no additional cost to the County;
  - 23.3.2.2 That the County deduct from any amount due Contractor the sum of money equivalent to the difference in value between the work performed and that called for by the Drawings and Specifications; or
  - 23.3.2.3 That the County exercise any other remedy it may have at law or under the Contract Documents, including but not limited to the County hiring its own forces or another contractor to replace the Contractor's nonconforming Work, in which case the County shall either issue a Deductive Change Order, a Construction Change Directive or invoice the Contractor for the cost of that work. Contractor shall pay any invoices within thirty (30) days of receipt of same or County may withhold those amounts from payment(s) to Contractor.

### 24. TERMINATION AND SUSPENSION

# 24.1 County's Right to Terminate Contractor for Cause

# **24.1.1** Grounds for Termination.

The County, in its sole discretion, may terminate the Contract and/or terminate the Contractor's right to perform the work of the Contract based upon the following:

- 24.1.1.1 Contractor refuses or fails to execute the Work or any separable part thereof with sufficient diligence as will ensure its completion within the time specified or any extension thereof, or
- **24.1.1.2** Contractor fails to complete said Work within the time specified or any extension thereof, or
- **24.1.1.3** Contractor persistently fails or refused to perform Work or provide material of sufficient quality as to be in compliance with Contract Documents; or
- 24.1.1.4 Contractor files a petition for relief as a debtor, or a petition is filed against the Contractor without its consent, and the petition not dismissed within sixty (60) days; or
- **24.1.1.5** Contractor makes a general assignment for the benefit of its creditors, or a receiver is appointed on account of its insolvency; or
- **24.1.1.6** Contractor persistently or repeatedly refuses or fails, except in cases for which extension of time is provided, to supply enough properly skilled workers or proper materials to complete the Work in the time specified; or
- **24.1.1.7** Contractor fails to make prompt payment to Subcontractors, or for material, or for labor; or
- **24.1.1.8** Contractor persistently disregards laws or ordinances, or instructions of County; or
- **24.1.1.9** Contractor fails to supply labor, including that of Subcontractors, that can work in harmony with all other elements of labor employed or to be employed on the Work; or
- **24.1.1.10** Contractor or its Subcontractor(s) is/are otherwise in breach, default, or in substantial violation of any provision of this Contract.

### 24.1.2 Notification of Termination

24.1.2.1 Upon the occurrence of County's sole determination of any of the above conditions, County may, without prejudice to any other right or remedy, serve written notice upon Contractor and its Surety of County's termination of this Contract and/or the Contractor's right to perform the work of the Contract. This notice will contain the reasons for termination. Unless, within three (3) days after the service of the notice, any and all condition(s) shall cease, and any and

all violation(s) shall cease, or arrangement satisfactory to County for the correction of the condition(s) and/or violation(s) be made, this Contract shall cease and terminate. Upon Determination, Contractor shall not be entitled to receive any further payment until the entire Work is finished.

- **24.1.2.2** Upon Termination, County may immediately serve written notice of tender upon Surety whereby Surety shall have the right to take over and perform this Contract only if Surety:
  - **24.1.2.2.1** Within three (3) days after service upon it of the notice of tender, gives County written notice of Surety's intention to take over and perform this Contract; and
  - **24.1.2.2.2** Commences performance of this Contract within (three (3) days from the date of serving of its notice to County.
- 24.1.2.3 If Surety fails to notify County or begin performance as indicated herein, County may take over the Work and execute the Work to completion by any method it may deem advisable at the expense of Contractor and/or its Surety. Contractor and/or its Surety shall be liable to County for any excess cost or other damages the County incurs thereby. Time is of the essence in this Contract. If the County takes over the Work as herein provided, County may, without liability for so doing, take possession of and utilize in completing the Work such materials, appliances, plan, and other property belonging to Contractor as may be on the Site of the Work, in bonded storage, or previously paid for.

### 24.1.3 Effect of Termination

- 24.1.3.1 Contractor shall, only if ordered to do so by the County, immediately remove from the Site all or any materials and personal property belonging to Contractor that have not been incorporated in the construction of the Work, or which are not in place in the Work. The County retains the right, but not the obligation, to keep and use any materials and personal property belonging to Contractor that have not been incorporated in the construction of the Work, or which are not in place in the Work. The Contractor and its Surety shall be liable upon the performance bond for all damages caused the County by reason of the Contractor's failure to complete the Contract.
- In the event that the County shall perform any portion of, or the whole of the Work, pursuant to the provisions of the General Conditions, the County shall not be liable nor account to the Contractor in any way for the time within which, or the manner in which, the Work is performed by the County or for any changes the County may make in the Work or for the money expended by the County in satisfying claims and/or suits and/or other obligations in connection with the Work.

- **24.1.3.3** In the event, that the Contract is terminated for any reason, no allowances or compensation will be granted for the loss of any anticipated profit by the Contractor.
- 24.1.3.4 If the expense to the County to finish the Work exceeds the unpaid Contract Price, Contractor and Surety shall pay the difference to County within twenty-one (21) days of County's request.
- 24.1.3.5 The County shall have the right (but shall have no obligation) to assume and/or assign to a general contractor or construction manager or other third party who is qualified and has sufficient resources to complete the Work, the rights of the Contractor under its subcontracts with any or all Subcontractors. In the event of an assumption or assignment by the County, no Subcontractor shall have any claim against the County or third party for Work performed by Subcontractor or other matters arising prior to termination of the Contract. The County or any third party, as the case may be, shall be liable only for obligations to the Subcontractor arising after assumption or assignment. Should the County so elect, the Contractor shall execute and deliver all documents and take all steps, including the legal assignment of its contractual rights, as the County may require, for the purpose of fully vesting in the County the rights and benefits of it Subcontractor under Subcontracts or other obligations or commitments. All payments due the Contractor hereunder shall be subject to a right of offset by the County for expenses and damages suffered by the County as a result of any default, acts, or omissions of the Contractor. Contractor must include this assignment provision in all of its contracts with its Subcontractors.
- **24.1.3.6** The foregoing provisions are in addition to and not in limitation of any other rights or remedies available to County.

# **24.2** Termination of Contractor for Convenience

- **24.2.1** County in its sole discretion may terminate the Contract upon three (3) days written notice to the Contractor. Under a termination for convenience, the County retains the right to all the options available to the County if there is a termination for cause. In case of a termination for convenience, the Contractor shall have no claims against the County except:
  - 24.2.1.1 The actual cost of labor, materials, and services performed that is unpaid and can be documented through timesheets, invoices, receipts, or otherwise, and
  - 24.2.1.2 Five percent (5%) of the total cost of work performed as of the date of termination, or five percent (5%) of the value of the Work yet to be performed, whichever is less. This five percent (5%) amount shall be full compensation for all Contractor's and its Subcontractor(s)' mobilization and/or demobilization costs and any anticipated loss profits resulting from termination of the Contractor for convenience.

# 24.3 Emergency Termination of Public Contracts Act of 1949

# **Alameda County, General Services Agency**

Highland Hospital – Koret Building Exhaust Fan and Waterproofing

**24.3.1** This Contract is subject to termination as provided by sections 4410 and 4411 of the Government Code of the State of California, being a portion of the Emergency Termination of Public Contracts Act of 1949.

### **24.3.1.1** Section 4410 of the Government Code states:

In the event a national emergency occurs, and public work, being performed by contract, is stopped, directly or indirectly, because of the freezing or diversion of materials, equipment or labor, as the result of an order or a proclamation of the President of the United States, or of an order of any federal authority, and the circumstances or conditions are such that it is impracticable within a reasonable time to proceed with a substantial portion of the work, then the public agency and the contractor may, by written agreement, terminate said contract.

### **24.3.1.2** Section 4411 of the Government Code states:

Such an agreement shall include the terms and conditions of the termination of the contract and provision for the payment of compensation or money, if any, which either party shall pay to the other or any other person, under the facts and circumstances in the case.

24.3.1.3 Compensation to the Contractor shall be determined at the sole discretion of County on the basis of the reasonable value of the Work done, including preparatory work. As an exception to the foregoing and at the County's discretion, in the case of any fully completed separate item or portion of the Work for which there is a separate previously submitted unit price or item on the accepted schedule of values, that price shall control. The County, at its sole discretion, may adopt the Contract Price as the reasonable value of the work done or any portion thereof.

# 25. CLAIMS AND DISPUTES

### **25.1** Performance During Claim Process

The Contractor shall continue to perform its Work under the Contract and shall not cause a delay in the Work during any dispute, claims definition, negotiation, mediation, or arbitration proceeding, except by written agreement by the County.

## 25.2 Definition of Claim

- **25.2.1** For purposes of this section, a "Claim" means a separate demand by the Contractor for:
  - **25.2.1.1** A time extension,
  - 25.2.1.2 Payment of money or damages arising from Work done by or on behalf of the Contractor pursuant to the Contract and payment of which is not otherwise expressly provided for or the claimant is not otherwise entitled to, or
  - **25.2.1.3** Payment of money that the County disputes is owing.

### 25.3 Claim Presentations

- **25.3.1** The attention of the Contractor is drawn to Government Code Section 12650, et seq. regarding penalties for false claims.
- **25.3.2** Contractor shall file with the County any written Claim, including the documents necessary to substantiate it, on or before the day of final payment on the Contract.
- **25.3.3** The Contractor shall not cause a delay in the Work during any dispute, claims definition, negotiation, mediation, or arbitration proceeding, except by written agreement by the County.
- **25.3.4** The Contractor shall bind all its Subcontractors, material persons, and suppliers to the provisions of this section on mediation and arbitration and will hold the County harmless against disputes and claims by Subcontractors, material persons, or suppliers.

### 25.4 Claim Resolution

- 25.4.1 In the event of a dispute between the parties as to performance of the Work, the interpretation of this Contract, or payment or nonpayment for Work performed or not performed, the parties shall attempt to resolve the dispute by those procedures set forth in Public Contract Code Section 9204, if applicable. Pending resolution of the dispute, if the dispute is not resolved, Contractor agrees it will neither rescind the Contract nor stop the progress of the Work but will allow determination by a court of the State of California having competent jurisdiction of the dispute, after the Project has been completed, and not before.
- **25.4.2** For all Claims which arise between a Contractor and a local agency, the procedure set forth in Public Contract Code Section 9204 shall apply:
  - 25.4.2.1 The County shall respond in writing within forty-five (45) days of receipt of the Claim identifying what portion of the Claim is disputed and what portion is undisputed.
- **25.4.2.1.1** Upon receipt of a Claim, County, and Contractor may, by mutual agreement, extend the time period for County to respond.
- 25.4.2.1.2 Any payment due on an undisputed portion of the Claim shall be processed and made within sixty (60) days after the County issues its written response to the Claim.
  - 25.4.3 If Contractor disputes County's written response, or if County fails to respond to a Claim issued pursuant to this Section 25 within the time prescribed, Contractor may demand in writing an informal conference to meet and confer for settlement of the issues in dispute.
    - **25.4.3.1** Upon receipt of a demand in writing sent by registered mail or certified mail, return receipt requested, County

shall schedule a meet and confer conference within thirty (30) days for settlement of the dispute.

- **25.4.4** Within ten (10) business days following the conclusion of the meet and confer conference, if the Claim or any portion of the Claim remains in dispute, the County shall provide Contractor a written statement identifying the portion of the Claim that remains in dispute and the portion that is undisputed.
  - **25.4.4.1** Any payment due on an undisputed portion of the Claim shall be processed and made within sixty (60) days after the County issues its written statement.
- **25.4.5** Any disputed portion of the Claim, as identified by the statement referenced in Section 25.4.4 shall be submitted to nonbinding mediation, with the County and Contractor sharing the associated costs equally.
  - **25.4.5.1** County and Contractor shall mutually agree to a mediator within ten (10) business days after the disputed portion of the claim has been identified in writing.
  - **25.4.5.2** If County and Contractor cannot agree to a mediator, each party shall select a mediator, and those mediators shall select a qualified neutral third party to mediate with regard to the disputed portion of the claim.
  - **25.4.5.3** County and Contractor shall each bear the fees and costs charged by its respective mediator in connection with the selection of the neutral mediator.
  - **25.4.5.4** If mediation is unsuccessful, the parts of the claim remaining in dispute shall be submitted to judicial arbitration pursuant to Chapter 2.5 (commencing with Section 1141.10) of the Title 3 of Part 3 of the Code of Civil Procedure, notwithstanding Section 1141.11 of that code. The Civil Discovery Act of 1986, (Article 3 (commencing with Section 2016) of Chapter 3 of Title 3 of part 4 of the Code of Civil Procedure) shall apply to any proceeding brought under this subdivision consistent with the rules pertaining to judicial arbitration.
- 25.4.6 The County shall not fail to pay money as to any portion of a claim which is undisputed except as otherwise provided in the Contract Documents. Unpaid claim amounts not paid in a timely manner will accrue interest at seven percent (7%) per annum. In any suit filed pursuant to this section, the County shall pay interest at the legal rate on any arbitration award or judgment. Interest shall begin to accrue on the date the suit is filed in a court of law.
- 26. LABOR, WAGE & HOUR, APPRENTICE, AND RELATED PROVISIONS

- **26.1 Wage Rates, Travel, and Subsistence (For Projects over \$1M** see also PROJECT STABILIZATION/ COMMUNITY BENEFITS AGREEMENT of the COUNTY OF ALAMEDA Document 00 73 49).
- **26.1.1** Pursuant to the provisions of Article 2 (commencing with Section 1770), chapter 1, part 7, division 2, of the Labor Code of California, the general prevailing rate of per diem wages and the general prevailing rate for holiday and overtime work in the locality in which this public work is to be performed for each craft, classification or type of worker needed to execute this Contract is on file at the County's principal office, and copies will be made available to any interested party on request. Contractor shall obtain and post a copy of these wage rates at the job site.
- **26.1.2** Holiday and overtime work, when permitted by law, shall be paid for at a rate of at least one and one-half times the above-specified rate of per diem wages unless otherwise specified. The holidays upon which those rates shall be paid need not be specified by the County but shall be all holidays recognized in the applicable collective bargaining agreement. If the prevailing rate is not based on a collectively bargained rate, the holidays upon which the prevailing rate shall be paid shall be as provided in Section 6700 of the Government Code.
- **26.1.3** Contractor shall pay and shall cause to be paid each worker engaged in Work on the Project not less than the general prevailing rate of per diem wages determined by the Director of the Department of Industrial Relations ("DIR") ("Director"), regardless of any contractual relationship which may be alleged to exist between Contractor or any Subcontractor and such workers.
- **26.1.4** Contractor shall pay and shall cause to be paid to each worker needed to execute the Work on the Project travel and subsistence payments, as such travel and subsistence payments are defined in the applicable Collective Bargaining Agreements filed with the Department of Industrial Relations in accordance with Labor Code Section 1773 et seq.
- **26.1.5** If during the period this bid is required to remain open, the Director determines that there has been a change in any prevailing rate of per diem wages in the locality in which the Work under the Contract is to be performed, such change shall not alter the wage rates in the Notice to Bidders or the Contract subsequently awarded.
- **26.1.6** Pursuant to Labor Code Section 1775, Contractor shall, as a penalty to County, forfeit the statutory amount for each calendar day, or portion thereof, for each worker paid less than the prevailing rates, determined by the County and/or the Director, for the work or craft in which that worker is employed for any public work done under Contract by Contractor or by any Subcontractor under it. The difference between such prevailing wage rates and the amount paid to each worker for each calendar day or portion thereof for which each worker was paid less than the prevailing wage rate, shall be paid to each worker by Contractor.
- **26.1.7** Any worker employed to perform Work on the Project, which Work is not covered by any classification listed in the general prevailing wage rate of per diem wages determined by the Director, shall be paid not less than the minimum rate of wages specified therein for the classification which most nearly corresponds to Work to be performed by him, and such minimum wage rate shall be retroactive to time of initial employment of such person in such classification.

- **26.1.8** Pursuant to Labor Code section 1773.1, per diem wages, are deemed to include employer payments for health and welfare, pension, vacation, travel time, subsistence pay, and apprenticeship or other training programs authorized by section 3093, and similar purposes.
- **26.1.9** Contractor shall post at appropriate conspicuous points on the Site of Project, a schedule showing all determined minimum wage rates and all authorized deductions if any, from unpaid wages, actually earned. In addition, Contractor shall post a sign-in log for all workers and visitors to the Site, a list of all subcontractors of any tier on the Site, and the required Equal Employment Opportunity poster(s).
- **26.1.10** Contractor stipulates that it shall comply with all requirements of PROJECT STABILIZATION/COMMUNITY BENEFITS AGREEMENT (**For projects over \$1M**) of the COUNTY OF ALAMEDA, and shall pay to persons performing labor in and about the Work provided for in the Contract an amount equal to or more than the following:
  - **26.1.10.1** Wage rate and fringe benefit payments and classification for that person's corresponding labor classification as required by the Department of Industrial Relations;
  - Wage rate and fringe benefit payments and classification for that person's corresponding labor classification as required under the PROJECT STABILIZATION/COMMUNITY BENEFITS AGREEMENT of the COUNTY OF ALAMEDA (For projects over \$1M) and California Labor Code.
- **26.1.11** If there are conflicts between the Wage rate and fringe benefit payments and classification between the Department of Industrial Relations and the PROJECT STABILIZATION/COMMUNITY BENEFITS AGREEMENT of the COUNTY OF ALAMEDA (**For projects over \$1M**), Contractor shall pay the higher wage rate and fringe benefits.

# 26.2 Hours of Work

- 26.2.1 As provided in Article 3 (commencing with Section 1810), chapter 1, part 7, division 2, of the Labor Code, eight (8) hours of labor shall constitute a legal days work. The time of service of any worker employed at any time by Contractor or by any Subcontractor on any subcontract under this Contract upon the Work or upon any part of the Work contemplated by this Contract shall be limited and restricted by Contractor to eight (8) hours per day, and forty (40) hours during any one week, except as hereinafter provided. Notwithstanding the provisions hereinabove set forth, Work performed by employees of Contractor in excess of eight (8) hours per day and forty (40) hours during any one week, shall be permitted upon this public work upon compensation for all hours worked in excess of eight (8) hours per day at not less than one and one-half times the basic rate of pay.
- **26.2.2** Contractor shall keep and shall cause each Subcontractor to keep an accurate record showing the name of and actual hours worked each calendar day and each calendar week by each worker employed by Contractor in connection with the Work or any part of the Work contemplated by this Contract. The record shall be kept open at all

reasonable hours to the inspection of County and to the Division of Labor Standards Enforcement of the DIR.

- **26.2.3** Pursuant to Labor Code Section 1813, Contractor shall as a penalty to the County forfeit the statutory amount for each worker employed in the execution of this Contract by Contractor or by any Subcontractor for each calendar day during which such worker is required or permitted to work more than eight (8) hours in any one calendar day and forty (40) hours in any one calendar week in violation of the provisions of Article 3 (commencing with Section 1810), chapter 1, part 7, division 2, of the Labor Code.
- **26.2.4** Any Work necessary to be performed after regular working hours, or on Sundays or other holidays shall be performed without additional expense to the County.

# 26.3 Payroll Records

- **26.3.1** County will use the Alameda County Contract Compliance System, including the Elation Systems, Inc. program, to monitor contract and labor compliance. Contractor shall use the Compliance System to meet County's requirements, and shall participate in training as directed by County in order to become and remain competent in the use of the Compliance System.
- **26.3.2** Pursuant to the provisions of section 1776 of the Labor Code, notice is hereby given that Contractor shall prepare and provide to the County and shall cause each Subcontractor performing any portion of the Work under this Contract to prepare and provide to the County an accurate and certified payroll record ("CPR(s)"), showing the name, address, social security number, work classification, straight time, and overtime hours worked each day and week, and the actual per diem wages paid to each journeyman, apprentice, worker, or other employee employed by the Contractor and/or each Subcontractor in connection with the Work.
- **26.3.3** The CPRs enumerated hereunder shall be certified and shall be provided to the County on a weekly basis. The CPRs from the Contractor and each Subcontractor for each week shall be provided on or before Wednesday of the week following the week covered by the CPRs. County shall not make any payment to Contractor until:
  - **26.3.3.1** Contractor and/or its Subcontractor(s) provide CPRs acceptable to the County, and
  - 26.3.3.2 The County is given sufficient time to review and/or audit the CPRs to determine their acceptability. Any delay in Contractor and/or its Subcontractor(s) providing CPRs to the County in a timely manner will directly delay the County's review and/or audit of the CPRs and Contractor's payment.
- **26.3.4** All CPRs shall be available for inspection at all reasonable hours at the principal office of Contractor on the following basis:
  - **26.3.4.1** A certified copy of an employee's CPR shall be made available for inspection or furnished to the employee or his/her authorized representative on request.

- 26.3.4.2 CPRs shall be made available for inspection or furnished upon request to a representative of County, Division of Labor Standards Enforcement, Division of Apprenticeship Standards, and/or the Department of Industrial Relations.
- 26.3.4.3 CPRs shall be made available upon request by the public for inspection or copies thereof made; provided, however, that a request by the public shall be made through either the County, Division of Apprenticeship Standards or the Division of Labor Standards Enforcement. If the requested CPRs have not been provided pursuant to the provisions herein, the requesting party shall, prior to being provided the records reimburse the costs of preparation by Contractor, Subcontractors, and the entity through which the request was made. The public shall not be given access to the records at the principal office of Contractor.
  - **26.3.5** The form of certification for the CPRs shall be as follows:
  - I, (Name-Print), the undersigned, am the (Position in business) with the authority to act for and on behalf of (Name of business and/or Contractor), certify under penalty of perjury that the records or copies thereof submitted and consisting of (Description, number of pages) are the originals or true, full, and correct copies of the originals which depict the payroll record(s) of actual disbursements by way of cash, check, or whatever form to the individual or individual named, and (b) we have complied with the requirements of sections 1771, 1811, and 1815 for any work performed by our employees on the Project.

Date: Signature:

(Section 16401 of the California Code of Regulations)

- **26.3.6** Each Contractor shall file a certified copy of the CPRs with the entity that requested the records within ten (10) days after receipt of a written request.
- **26.3.7** Any copy of records made available for inspection as copies and furnished upon request to the public or any public agency by County, Division of Apprenticeship Standards, or Division of Labor Standards Enforcement shall be marked or obliterated in such a manner as to prevent disclosure of an individual's name, address, and social security number. The name and address of Contractor awarded Contract or performing Contract shall not be marked or obliterated.
- **26.3.8** Contractor shall inform County of the location of the records enumerated hereunder, including the street address, city, and county, and shall, within five (5) working days, provide a notice of change of location and address.
- **26.3.9** In the event of noncompliance with the requirements of this section, Contractor shall have ten (10) days in which to comply subsequent to receipt of written notice specifying in what respects Contractor must comply with this section. Should noncompliance still be evident after the ten (10) day period, Contractor shall, as a penalty to County, forfeit twenty-five dollars (\$25) for each calendar day, or portion thereof, for each worker, until strict compliance is effectuated. Upon the request of Division of

Apprenticeship Standards or Division of Labor Standards Enforcement, these penalties shall be withheld from progress payments then due.

**26.3.10** It shall be the responsibility of Contractor to ensure compliance with the provisions of Labor Code section 1776.

# 26.4 Apprentices

- **26.4.1** Contractor acknowledges and agrees that, if this Contract involves a dollar amount greater than or a number of working days greater than that specified in Labor Code section 1777.5, then this Contract is governed by the provisions of Labor Code Section 1777.5. It shall be the responsibility of Contractor to ensure compliance with this Article and with Labor Code section 1777.5 for all apprenticeship occupations.
- **26.4.2** Apprentices of any crafts or trades may be employed and, when required by Labor Code section 1777.5, shall be employed provided they are properly registered in full compliance with the provisions of the Labor Code.
- **26.4.3** Every such apprentice shall be paid the standard wage paid to apprentices under the regulations of the craft or trade at which he/she is employed, and shall be employed only in the work of the craft or trade to which she/he is registered.
- **26.4.4** Only apprentices, as defined in section 3077 of the Labor Code, who are in training under apprenticeship standards and written apprentice agreements under Chapter 4 (commencing at Section 3070), Division 3, of the Labor Code, are eligible to be employed. The employment and training of each apprentice shall be in accordance with the provisions of the apprenticeship standards and apprentice agreements under which he/she is training.
- **26.4.5** Pursuant to Labor Code section 1777.5, if that section applies to this Contract as indicated above, Contractor and any Subcontractors employing workers in any apprentice able craft or trade in performing any Work under this Contract shall apply to the applicable joint apprenticeship committee for a certificate approving the Contractor or Subcontractor under the applicable apprenticeship standards and fixing the ratio of apprentices to journeymen employed in performing the Work.
- **26.4.6** Pursuant to Labor Code section 1777.5, if that section applies to this Contract as indicated above, Contractor and any Subcontractor may be required to make contributions to the apprenticeship program.
- **26.4.7** If Contractor or Subcontractor willfully fails to comply with Labor Code section 1777.5, then, upon a determination of noncompliance by the Administrator of Apprenticeship, it shall:
  - **26.4.7.1** Be denied the right to bid on any subsequent project for one (1) year from the date of such determination;
  - Forfeit as a penalty to County the full amount as stated in Labor Code section 1777.7. Interpretation and enforcement of these provisions shall be in accordance with the rules and procedures of the California Apprenticeship Council and under the authority of the Chief of the Division of Apprenticeship Standards.

- **26.4.8** Contractor and all Subcontractors shall comply with Labor Code section 1777.6, which section forbids certain discriminatory practices in the employment of apprentices.
- **26.4.9** Contractor shall become fully acquainted with the law regarding apprentices prior to commencement of the Work. Special attention is directed to Sections 1777.5, 1777.6, and 1777.7 of the Labor Code, and title 8, California Code of Regulations, section 200 et seq. Questions may be directed to the State Division of Apprenticeship Standards, 455 Golden Gate Avenue, San Francisco, California 94102.

### **26.5** Non-Discrimination

- **26.5.1** Contractor herein agrees not to discriminate in its recruiting, hiring, promotion, demotion, or termination practices on the basis of race, religious creed, national origin, ancestry, sex, age, or physical handicap in the performance of this Contract and to comply with the provisions of the California Fair Employment and Housing Act as set forth in part 2.8 of Division 3 of the California Government Code, commencing at section 12900; the Federal Civil Rights Act of 1964, as set forth in Public Law 88-352, and all amendments thereto; Executive Order 11246, and all administrative rules and regulations found to be applicable to Contractor and Subcontractor.
- **26.5.2** Special requirements for Federally Assisted Construction Contracts: During the performance of this Contract, Contractor agrees to incorporate in all subcontracts the provisions set forth in Chapter 60-1.4(b) of Title 41 published in Volume 33 No. 104 of the Federal Register dated May 28, 1968.

### 26.6 Labor First Aid

Contractor shall maintain emergency first aid treatment for Contractor's workers on the Project which complies with the Federal Occupational Safety and Health Act of 1970 (29 U.S.C. § 651 et seq.) and the California Occupational Safety and Health Act of 1973 (8 Cal. Code of Regs., §1 et seq.).

# 27. MISCELLANEOUS

# 27.1 Assignment of Antitrust Actions

**27.1.1** Section 7103.5(b) of the Public Contract Code states:

In entering into a public works contract or subcontract to supply goods, services, or materials pursuant to a public works contract, the Contractor or subcontractor offers and agrees to assign to the awarding body all rights, title, and interest in and to all causes of action it may have under Section 4 of the Clayton Act (15 U.S.C. Sec. 15) or under the Cartwright Act (Chapter 2 (commencing with Section 16700) of Part 2 of Division 7 of the Business and Professions Code), arising from purchases of goods, made and become effective at the time the awarding body tenders final payment to the Contractor, without further acknowledgment by the parties.

# **27.1.2** Section 4552 of the Government Code states:

In submitting a bid to a public purchasing body, the bidder offers and agrees that if the bid is accepted, it will assign to the purchasing body all rights, title, and interest in and to all causes of action it may have under Section 4 of the Clayton Act (15 U.S.C. Sec. 15) or under the Cartwright Act (Chapter 2 (commencing with Section 16700) of Part 2 of Division 7 of the Business and Professions Code), arising from purchases of goods,

# **Alameda County, General Services Agency**

Highland Hospital – Koret Building Exhaust Fan and Waterproofing

materials, or services by the bidder for sale to the purchasing body pursuant to the bid. Such assignment shall be made and become effective at the time the purchasing body tenders final payment to the bidder.

#### **27.1.3** Section 4553 of the Government Code states:

If an awarding body or public purchasing body receives, either through judgment or settlement, a monetary recovery for a cause of action assigned under this chapter, the assignor shall be entitled to receive reimbursement for actual legal costs incurred and may, upon demand, recover from the public body any portion of the recovery, including treble damages, attributable to overcharges that were paid by the assignor but were not paid by the public body as part of the bid price, less the expenses incurred in obtaining that portion of the recovery.

# **27.1.4** Section 4554 of the Government Code states:

Upon demand in writing by the assignor, the assignee shall, within one year from such demand, reassign the cause of action assigned under this part if the assignor has been or may have been injured by the violation of law for which the cause of action arose and (a) the assignee has not been injured thereby, or (b) the assignee declines to file a court action for the cause of action.

**27.1.5** Under this Article, "public purchasing body" is County and "bidder is Contractor.

#### 27.2 Excise Taxes

If, under Federal Excise Tax Law, any transaction hereunder constitutes a sale on which a Federal Excise Tax is imposed, and the sale is exempt from such Federal Excise Tax because it is a sale to a State or Local Government for its exclusive use, County, upon request, will execute documents necessary to show (1) that County is a political subdivision of the State for the purposes of such exemption, and (2) that the sale is for the exclusive use of County. No Federal Excise Tax for such materials shall be included in any Contract Price.

### **27.3** Taxes

Contract Price is to include any and all applicable sales taxes or other taxes that may be due in accordance with section 7051 of the Revenue and Taxation Code; Regulation 1521 of the State Board of Equalization or any other tax code that may be applicable.

# 27.4 Shipments

All shipments must be F.O.B. destination to Site or sites, as indicated in the Contract Documents. There must be no charge for containers, packing, unpacking, drayage or insurance. The total Contract Price shall be all inclusive (including sales tax), and no additional costs of any type will be considered.

### **END OF DOCUMENT**

## **DOCUMENT 00 73 13**

# **SPECIAL CONDITIONS**

# 1. <u>Substitution for Specified Items</u>

- 2.1 Requests for substitutions prior to award of the Contract shall be done within the time period indicated in the Instructions to Bidders.
- 2.2. Whenever in the Specifications any materials, process, or article is indicated or specified by grade, patent, or proprietary name, or by name of manufacturer, that Specification shall be deemed to be followed by the words "or equal." Contractor may, unless otherwise stated, offer any material, process, or article that shall be substantially equal or better in every respect to that so indicated or specified.
  - 2.2.1. If the material, process, or article offered by Contractor is not, in the opinion of the County, substantially equal or better in every respect to that specified, then Contractor shall furnish the material, process, or article specified in the Specifications without any additional compensation or change order.
  - 2.2.2. This provision shall not be applicable with respect to any material, product, thing or service for which County made findings and gave notice in accordance with Public Contract Code section 3400(b); therefore, Contractor shall not be entitled to request a substitution with respect to those materials, products or services.
- 2.3. A request for a substitution shall be in writing and shall include:
  - 2.3.1. All variations of the proposed substitute from the material specified including, but not limited to, principles of operation, materials, or construction finish, thickness or gauge of materials, dimensions, weight, and tolerances;
  - 2.3.2. Available maintenance, repair or replacement services;
  - 2.3.3. Increases or decreases in operating, maintenance, repair, replacement, and spare parts costs;
  - 2.3.4. Whether or not acceptance of the substitute will require other changes in the Work (or in work performed by the County or others under Contract with the County); and

- 2.3.5. The time impact on any part of the Work resulting directly or indirectly from acceptance of the proposed substitute.
- 2.4. No substitutions shall be made until approved, in writing, by the County. The burden of proof as to equality of any material, process, or article shall rest with Contractor. The Contractor warrants that if substitutes are approved:
  - 2.4.1. The proposed substitute is equal or superior in all respects to that specified, and that such proposed substitute is suitable and fit for the intended purpose and will perform adequately the function and achieve the results called for by the general design and the Contract Documents;
  - 2.4.2. The Contractor provides the same warranties and guarantees for the substitute that would be provided for that specified;
  - 2.4.3. The Contractor shall be fully responsible for the installation of the substitute and any changes in the Work required, either directly or indirectly, because of the acceptance of such substitute, with no increase in Contract Price or Contract Time. Incidental changes or extra component parts required to accommodate the substitute will be made by the Contractor without a change in the Contract Price or Contract Time;
  - 2.4.4. The Contractor shall be responsible for any re-design costs occasioned by County's acceptance and/or approval of any substitute; and
  - 2.4.5. The Contractor shall, in the event that a substitute is less costly than that specified, credit the County with one hundred percent (100%) of the net difference between the substitute and the originally specified material. In this event, the Contractor agrees to execute a deductive Change Order to reflect that credit.
- 2.5. In the event Contractor furnishes a material, process, or article more expensive than that specified, the difference in the cost of that material, process, or article so furnished shall be borne by Contractor.
- 2.6. In no event shall the County be liable for any increase in Contract Price or Contract Time due to any claimed delay in the evaluation of any proposed substitute or in the acceptance or rejection of any proposed substitute.

# 3. Weather Days

3.1 Delays due to adverse weather conditions will only be permitted in compliance with the provisions in the General Conditions and only if the number of days of adverse weather exceeds the following parameters and only if Contractor can verify that adverse weather caused delays exceeded the following number of days:

January	[11]	July	[0]
February	[10]	August	[0]
March	[10]	September	[1]
April	[6]	October	[4]
May	[3]	November	[7]
June	[1]	December	[10]

**Insurance Policy Limits**. All of Contractor's insurance shall be with insurance companies with an A.M. Best rating of no less than A: XI.

The limits of insurance shall not be less than:

Commercial General	Each Occurrence	[\$2,000,000]
Liability		
	General Aggregate	[\$2,000,000]
	Product Liability and Completed Operations	[\$1,000,000]
Automobile Liability  – Any Auto	Combined Single Limit	[\$2,000,000]
<b>Excess Liability</b>		[\$4,000,000]
Workers		Statutory limits
Compensation		pursuant to state law
Employers' Liability		[\$1,000,000]
<b>Builders Risk (Course</b>		Issued for the value of
of Construction)		the Contract

# 5. Permits, Certificates, Licenses, Fees, Approval

- 5.1 <u>Payment for Permits, Certificates, Licenses, and Fees</u>. As required in the General Conditions, the Contractor shall secure and pay for all permits, licenses and certificates necessary for the prosecution of the Work with the exception of the following:
  - 5.1.1 OSHPD Fees.

# 6. Work Restrictions

6.1 <u>Medical Safety and Infection Control Program</u>: Within fifteen (15) Days after Notice to Proceed with the first phase of Work, but not less than ten (10) Days

before gaining access to the site to start Work, Contractor shall submit its written program with detailed outline of procedures for complying with County's and Alameda Health System's requirements. The program shall be coordinated with the Contractor's Contract Schedule.

- 6.1.1 The Contractor shall revise and resubmit the Medical Safety and Infection Control Program to address changes requested by the County.
- 6.1.2 Enforcement of the protocols and procedures contained in the Medical Safety and Infection Control Program shall be implemented through measures incorporated into the Contractor's Quality Control Plan.
- 6.1.3 Later phases of Work shall be submitted for County review and approval no later that thirty (30) Days prior to the beginning of each phase of work.

# 7. <u>Communication, Coordination, and Planning</u>

7.1 The Contractor shall comply with supplemental instructions from the County concerning the facility's medical safety and infection control. When necessary to prevent unsafe incidents, supplemental instructions may include work stoppages to reschedule and/or redirect the Work.

# 8. Training

8.1 Provide training and orientation on infection control and Facility procedures for all personnel employed by the Contractor, subcontractors, and any other personnel entering the Facility in support of the Contractor.

# 9. Utility Interruptions and Notifications

9.1 Shutdown or interruption of water, chilled water, steam, electrical services, natural gas, compressed air, vacuum, oxygen, nitrous oxide, or any utility system requires written notice a minimum of fourteen (14) Business Days in advance, via a MOP. Contractor is not authorized to interrupt utility services without this advance notification and the prior approval of the County's Representative.

# 10. Environmental Controls

10.1 Noise: All work shall be performed with a minimum of noise or disruption to normal activities in the surrounding areas. In and around patient care areas, extraordinary care and concern must be exercised. If nursing department staff, medical staff, or administrative staff indicates a problem due to these conditions, the activity must be stopped until the County's Representative is contacted and satisfactory arrangements are determined. Contractor will give thirty (30) Day notice for any work to be done outside the hours of 7am and 7pm.

During the construction extraordinary care and concern must be exercised to not disrupt the patient population. The Contractor shall develop a Workplan, for the

County's approval, which demonstrates noise considerations for the patients' sleep period and the ongoing function of the facility. The following noise control procedures shall be employed:

- 10.1.1 Maximum increase in noise shall be limited to approximately 15db over ambient and shall not exceed regulatory standards for noise.
- 10.1.2 The on-site construction supervisor shall have the responsibility and authority to receive and resolve noise complaints. A clear appeal process shall be established prior to construction commencement that will allow for resolution of noise problems that cannot be immediately solved by the site supervisor.
- 10.1.3 All noise-producing project equipment and vehicles using internal combustion engines shall be equipped with mufflers, air-inlet silencers where appropriate, and any other shrouds, shields, or other noise-reducing features in good operating condition that meet or exceed original factory specification. Mobile or fixed "package" equipment (e.g., arc-welders, air compressors) shall be equipped with shrouds and noise control features that are readily available for that type of equipment.
- 10.1.4 All mobile or fixed noise-producing equipment used on the project, which is regulated for noise output by a local, state, or federal agency, shall comply with such regulation while in the course of project activity.
- 10.1.5 Electrically-powered equipment instead of pneumatic or internal combustion powered equipment shall be used, where feasible and needed to control excessive noise.
- 10.1.6 Material stockpiles and mobile equipment staging, parking, and maintenance areas shall be located as far as practicable from noise-sensitive receptors.
- 10.1.7 Construction site and access road speed limits shall be established and enforced during the construction period.
- 10.1.8 The hours of material transport shall be restricted to the periods and days permitted by both this contract and local noise or other applicable ordinance.
- 10.1.9 The use of noise producing signals, including horns, whistles, alarms, and bells shall be for safety warning purposes only.
- 10.1.10 No project related public address or music system shall be audible at any adjacent noise-sensitive receptor.
- 10.2 <u>Dust</u>: Dust control is a critical activity given the proximity of construction to Facility operations. The Contractor shall prepare a submittal that identifies source

air pollution and related pollution reduction measures. The following dust control measures shall be employed:

- 10.2.7 Implement fugitive dust control measures as provided in Bay Area Air Quality Management District (BAQMD),.
- 10.2.8 Develop a staging area, vehicle and truck routes, and a daily meeting to assure all applicable control measures are established for that particular workday.
- 10.2.9 Dust barriers shall be provided by the Contractor as necessary to contain dust within the construction site.
- 10.2.10 If necessary, install a water misting system along fence perimeter or any other necessary area to prevent fugitive dust from creating a nuisance to the public or to facility staff.
- 10.2.11 Reduce the use of diesel fuel powered equipment and use equipment with alternative fuel whenever practical to minimize diesel exhaust emissions in areas close proximity to the operating Facility.
- 10.2.12 Turn off equipment when not in use for long periods of time. No idling of diesel-fueled equipment for durations longer than five minutes when within 100 feet of any low level air intake or building opening (doors and windows).
- 10.2.13 Control fugitive dust at active soil grading/excavation areas using water in a manner that would not impact soil compaction. Continuous wet-down may be required in the area of construction activity.
- 10.2.14 Use ground-covering such as mulch, wood chips, straws, hydro-seeding, surfactants, or plastic sheeting to cover inactive exposed areas to minimize fugitive dust.
- 10.2.15 Provide drainage for erosion control measures.
- 10.2.16 Use sand bags, as necessary, along site perimeter to keep soil on site.
- 10.2.17 Provide gravel entry way into construction site entrance to reduce/eliminate mud and sediment carried off site by vehicles.
- 10.2.18 Cover top of haul trucks to eliminate wind-blown fugitive dust.
- 10.2.19 Schedule haul trucks and material delivery trucks to prevent traffic congestion and impede the normal operation of the Facility. Set up truck queuing area and have staff communicate via cell phone for efficiency and less traffic at the Facility.

- 10.2.20 As necessary, use street sweepers along travel routes in general vicinity of project area.
- 10.2.21 All vehicle routes are to be watered for dust control. All existing roadway and parking surfaces impacted by construction activity are to be swept and kept free of debris and dust. All areas within the construction site are to be broom swept as required to keep dust and debris to a minimum.
- 10.2.22 Limit the number of haul trucks on site and establish a haul route. Install a gravel or base road on site for loading trucks. Haul route shall be reviewed and approved by the Facility.
- 10.2.23 Place on-site portable toilets away from building air intakes, windows, and entryway.
- 10.2.24 All stockpiles shall be kept moist throughout the day to minimize particulate matter emissions. Wet down stockpiles on a regular basis including prior to end of work day.
- 10.2.25 Haul roads shall be paved, lined with gravel or base material, or kept moist to minimize particulate matter emissions.
- 10.2.26 Where practical, use paddle-wheel scrapers instead of traditional scrapers to minimize fugitive dust and reduce exhaust emissions.
- 10.2.27 Handling of soil shall be kept to a minimum.
- 10.2.28 Provide a boundary/zone where equipment shall not enter because of proximity to active facility operation, and if necessary, equipment shall operate on alternative fuel to reduce diesel particulate matter.
- 10.3 <u>Demolition</u>: The Contractor shall plan the demolition activities to minimize environmental impacts on facility operations in accordance with Sections 017320 Selective Demolition and 024116 Building Demolition. The Contractor has the option of recycling demolished building material, but must do so in compliance with these Contract Documents and regulations. In addition to the above mitigation measures, the following applies to demolition activities:
  - 10.3.1 Provide the County's Representative with fourteen (14) Day advance written notice of which building will be demolished next, so that they take the necessary steps to prevent airborne contaminants from entering nearby buildings.
  - 10.3.2 Identify the material that are recyclable and if possible, send to recycling facility.

- 10.3.3 All surface area of structures shall be saturated with water prior to start of demolition.
- 10.3.4 Waste and debris shall be segregated, processed, and recycled to minimize waste volume and number of trips.
- 10.3.5 If debris is to be separated for recycling purposes, a jaw-crusher type attachment shall be used instead of a concrete breaker to separate concrete and rebar (if this is not feasible, perhaps negotiate with local concrete and other recycling facilities to meet recycling requirements.
- 10.4 Odors: When odors are a concern, arrangements shall be made by the Contractor for their containment or control. Where this is not feasible, specific arrangements should be made to minimize the disturbance of normal Facility activities. Where controllable, fumes and odors shall not be allowed to migrate to occupied areas. The Contractor shall immediately notify the County's Representative. If necessary, Facilities Management personnel shall modify adjacent air circulation systems as deemed necessary during the construction period.
- 10.5 <u>Vibrations</u>: The impacts of vibration for facility replacement and renovation activities will be limited. The most sensitive area will be surgeries and MRIs within 200 feet of construction. If vibration becomes an impact to facility operations, the Contractor shall stop operations, reschedule and/or implement the following with the approval of the County Representative:
  - 10.5.1 Route heavily loaded trucks and equipment away from the most sensitive medical facilities if possible.
  - 10.5.2 Phase demolition, earthmoving and ground-impacting operations so as not to occur in the same time period, to the extent practicable. The total vibration level produced could be less when each vibration source operates separately.
  - 10.5.3 Select demolition methods not involving impact, where practicable
  - 10.5.4 Avoid vibratory rollers and packers near vibration-sensitive areas.
- 10.6 <u>Environmental Mitigation Measures</u>: Contractor shall become familiar with the full text of the project's Environmental Impact Report and take responsibility for implementation of applicable mitigation measures. Questions about which items are applicable to the Contractor shall be directed to the County's Representative.

# 11. Shipments and Materials

Equipment and materials shall not be shipped to the Facility unless specific arrangements are made for receipt and acceptance of these items. When such shipments are authorized, they are the total responsibility of the Contractor. The

placement of such materials shall be limited to the specific lay-down and staging areas as defined in the Construction Logistic Drawings unless approved in advance by the County Project Manager. The County accepts no responsibility for the receipt, storage, or protection of the Contractor's materials and equipment.

# 12. Salvage and Disposal

- 12.1 All existing property of the Facility that is removed from the construction site and has been identified to be salvaged by the County shall be delivered to a secure site as specified by the County's Representative.
- 12.2 Construction debris, or material that has no redeemable value, is to be placed in Contractor-furnished refuse bins for safe and legal removal from the premises. Facility refuse bins may not be utilized unless so authorized by the County.

# 13. Parking

- 13.1 The County's Representative will meet with the Contractor to determine parking requirements.
- 13.2 The primary parking and storage area shall be designated areas.
- 13.3 The County's Representative will notify Alameda County Engineering Department of parking area proposed to be used by construction personnel if at variance with this procedure.
- 13.4 Contractor and related personnel shall park in authorized areas only.
- 13.5 Parking in Garage parking spaces is prohibited.
- 13.6 Use of the Free Shuttle Service to the hospital from the Lake Merritt BART station is prohibited.

# 14. Sanitary

- 14.1 Contractor shall provide temporary toilet facilities adjacent to all projects. The Contractor will not be allowed to use the Facility restroom facilities whether in existing facilities or those being constructed.
- 14.2 Contractor shall submit proposed location of temporary toilet(s) to the County's Representative for approval.
- 14.3 Construction personnel will not be allowed to use restroom facilities for personal or equipment clean-up.
- 14.4 Sanitary Facilities shall be in accordance with OSHA regulations.

# 15. Cafeteria and Food

- 15.1 Construction personnel will not be allowed access to the Facility cafeteria.
- 15.2 Construction personnel shall police their own areas. All cups, cans, paper, wrappers, and discarded food must be placed in trash receptacles at the end of each break.
- 15.3 Contactors shall submit the proposed location of any break and eating areas, which shall be outside of the Project boundaries, to the County's Representative for approval.
- 15.4 Construction personnel are not allowed to have food within the facilities, whether those facilities exist or are under construction.

# 16. Badges

- 16.1 Badges shall be worn by all of the Contractor's personnel and all of their subcontractors' personnel.
- 16.2 Special badges shall be issued to construction personnel to identify management positions and when the term of construction exceeds six months.

# 17. Phones

- 17.1 No cellular telephones shall be operated in staff areas.
- 17.2 Construction personnel shall pay for separate phone services. Pay phones are not available for Contractor use. Pay phones on site are to be reserved for staff, patients and their families.

# 18. **Smoking and Tobacco**

- 18.1 Smoking is not permitted indoors.
- 18.2 Smoking is not permitted within the facilities under construction.
- 18.3 Smoking is permitted in designated areas as outlined by the Facility.
- 18.4 All ashes and cigarette butts must be deposited in approved receptors.
- 18.5 No chewing tobacco or spitting of tobacco is permitted.

# 19. Elevators

19.1 The County Project Manager will inform the Contractor as to which elevators will be available for use.

# 20. Security

- 20.1 All personnel must obey and act immediately upon any request by security.
- 20.2 A list of emergency phone numbers will be provided by the Engineering Department Service Center.

# 21. Safety

# 21.1 General

- 21.1.1 Watch for visitors and staff.
- 21.1.2 Work only where there is a positive barrier separation between construction activities and others.
- 21.1.3 Clean up all areas immediately in occupied areas.
- 21.1.4 Do not drape cords across corridors. All cords must be attached to the ceiling or taped to the floor (use tape with non-marring adhesive).
- 21.1.5 Maintain a minimum of 8'-0" clear within all corridors.
- 21.1.6 Do not leave materials or equipment in the corridor.
- 21.2 Safety equipment and consideration should include, but are not limited to:
  - 21.2.1 Anyone known to be under the influence of alcohol or drugs shall be dismissed from the Project at once and not be allowed to return.
  - 21.2.2 Offensive language is not permitted in any area where it may be overheard by staff or visitors.
  - 21.2.3 Provide adequate emergency first aid equipment.
  - 21.2.4 Post location and emergency phone numbers for local medical care.
  - 21.2.5 Monitor safe ladder usage.
  - 21.2.6 Provide exhaust controls for equipment.
  - 21.2.7 Monitor noise levels and establish safe limitations.
  - 21.2.8 Ensure adequate ventilation for air contaminants.
  - 21.2.9 Insist on personal protective equipments, such as hard hats, safety shoes, and eye, ear, and face protection equipment.

- 21.2.10 Safety nets, belts, and lifelines shall be used, as appropriate.
- 21.2.11 Provide adequate emergency fire protection equipment.
- 21.2.12 Post location and emergency phone numbers for local fire departments.
- 21.2.13 Provide safe storage for all flammable and combustible materials.
- 21.2.14 Insist on safe and proper use of hand power tools and electrical drop cords.
- 21.2.15 Operation of cranes, derricks, and hoists should be in accordance with manufacturer's recommendations and appropriate ANSI and Cal/OSHA regulations.
- 21.2.16 All construction operations and personnel are subject to CAL-OSHA and the Facility's Environmental Health & Safety regulations.
- 21.2.17 Provide adequate barricades and safety lighting at all open trenches adjacent to public access (must accommodate proper warning for blind students).
- 21.2.18 Properly fence entire confines of project site so as to avoid public access or unauthorized personnel.
- 21.2.19 All wall, floor, and ceiling penetrations shall be sealed to maintain fire and smoke ratings in accordance with CBC, NFPA 99 and *Life Safety Code*.
- 21.2.20 All emergency exit passages must be maintained free of obstructions.
- 21.2.21 Provide barricades and fencing in accordance with the Construction Logistic Drawings.
- 21.3 Fire Prevention During Welding, Cutting, and Other Hot Work
  - 21.3.1 All hot work shall be in accordance with the Facility Contracted Services Personnel Facility Handbook.
  - 21.3.2 Hot work includes welding, heat treating grinding, thawing pipe, powder-driven fasteners, hot riveting, and similar applications producing a spark, flame, or heat.
  - 21.3.3 Hot work shall be performed in a designated area that is approved for hot work by the County's Representative.

- 21.3.4 The Contractor shall ensure that only approved apparatus, such as torches, manifolds, regulators, or pressure-reducing valves, and acetylene generators, are used.
- 21.3.5 The Contractor shall ensure that all individuals involved in hot work are:
  - 21.3.5.1 Trained in the safe operation of their equipment and the safe use of the process.
  - 21.3.5.2 Have an awareness of the inherent risks involved and understand the emergency procedures in the event of a fire.
  - 21.3.5.3 Are aware if any special risks, such a flammable materials or hazardous conditions at the hot work site.

# 21.4 Emergency Codes Procedures

- 21.4.1 Emergency Codes are announced over the public address system. Any and all times that a Code is announced:
  - 21.4.1.1 Do not use the elevators.
  - 21.4.1.2 Quickly remove all equipment and obstructions from corridors and doorways.
  - 21.4.1.3 Maintain these conditions until "code clear" is announced.

### 21.5 Alarms

- 21.5.1 Fire alarm signals are initiated manually or automatically through smoke and heat sensing devices.
- 25.1.2 Construction activities often create dust or smoke, which will activate the fire alarm system. Prior to conducting any work, notify the County Project Manager of the scope of work, the duration and location of the work, and determine if the work will create dust or smoke. If the work will create dust or smoke, proceed as follows:
  - 25.1.2.1 For localized operations, cover the smoke detector with a dust cover approved by the County Project Manager and Regulatory Agency. Remove promptly when work is complete.
  - 25.2.1.2 For larger areas, the detection system must be disabled. This is to be done only by Facilities Management staff. Even though the system is disabled, dust covers must be installed on all area smoke detectors.

- Contractor shall provide a continuous fire-watch until the system is restored.
- Remove dust covers promptly when work is complete.
- Detection system shall be restored to proper working order prior to releasing the fire-watch.

# 21.6 Interim Life Safety Measures

- 21.6.1 Interim Life Safety Measures (ILSMs) are a series of 11 administrative actions required to temporarily compensate for the significant hazards posed by existing NFPA 101 2018 *Life Safety Code (LSC)* deficiencies or construction activities. Implementation of ILSM is required in or adjacent to all construction areas and throughout buildings with existing *LSC* deficiencies. ILSMs apply to all personnel (including construction workers), must be implemented upon project development, and must be continuously enforced through project completion. ILSMs are intended to provide a level of life-safety comparable to that described in Chapters 1-7, 31, and the applicable occupancy chapters of the *LSC*. Each ILSM action must be documented through written policies and procedures. The ILSMs are:
  - 21.6.1.1 Ensuring free and unobstructed exits. Personnel receive additional training when alternative exits are designated. Buildings or areas under construction must maintain escape routes for construction workers at all times. Means of exiting construction areas are inspected daily.
  - 21.6.1.2 Ensuring free and unobstructed access to emergency services and for fire, police, and other emergency forces.
  - 21.6.1.3 Ensuring fire alarm, detection, and suppression systems are in good working order. A temporary but equivalent system shall be provided when any fire system is impaired. Temporary systems must be inspected and tested monthly.
  - 21.6.1.4 Ensuring fire alarm, detection, and suppression systems are in good working order. A temporary but equivalent system shall be provided when any fire system is impaired. Temporary systems must be inspected and tested monthly.
  - 21.6.1.5 Ensuring temporary partitions are smoke tight and built of noncombustible or limited combustible materials that will not contribute to the development or spread of fire.
  - 21.6.1.6 Providing additional fire-fighting equipment and training personnel in its use.

- 21.6.1.7 Prohibiting smoking according to EC.5 throughout the organization's buildings, and in and adjacent to construction areas.
- 21.6.1.8 Developing and enforcing storage, housekeeping, and debris removal practices that reduce the building's flammable and combustible fire load to the lowest feasible level.
- 21.6.1.9 Conducting a minimum of two fire drills per shift per quarter.
- 21.6.1.10 Increasing hazard surveillance of buildings, grounds, and equipment with special attention to excavations, construction areas, construction storage, and field offices.
- 21.6.1.11 Training personnel to compensate for impaired structural or compartmentalization features of fire safety.
- 21.6.1.12 Conducting organization wide education programs to promote awareness of *LSC* deficiencies, construction hazards, and ILSMs.
- 21.6.1.13 The Contractor must maintain the ILSM Daily Monitoring Form on site for inspection by Environmental Health and Safety (EH&S) and submit it to the County and Alameda County Engineering Department monthly and provide a complete set complied as a report at each project close-out.
- 21.6.1.14 Fire Drills shall be conducted in the construction zones and areas adjacent to the construction zones.

# 22. Infection Control

- 22.1 Infection control is critical in all facility areas. Dust in ceilings and dust potentially entrained into buildings, as well as construction debris, can contain fungal spores or bacteria which, if inhaled by staff or visitors, can cause pneumonia and even death. Construction, demolition, remodeling and landscaping activities in and around hospitals have been implicated as a risk factor for certain nosocomial infections in immunocompromised patients. The most notable organism is aspergillus, a fungus ubiquitous in ceiling and wall spaces where dust has accumulated. Activities that disturb accumulation of dust may cause fungal spores to become airborne, inhaled by the susceptible individual, and cause disease.
- Patients most at risk include those with congenital or acquired immunodeficiency, premature neonates and those receiving immunosuppressive therapy.

- Activities that disturb the environment where settled dust is found may cause spores to become airborne and increase risk for nosocomial infection. Activities include:
  - 22.3.1 Demolition, construction activities, grading, excavation, landscaping or remodeling,
  - 22.3.2 Exposure of ceiling spaces, pipe chases, etc. and/or
  - 22.3.3 Storage and removal of uncovered or partially covered debris from construction areas.
- The Contractor shall prepare an Infection Control Plan for the entire Project that complies with, at a minimum, the Facility's Infection Control Risk Assessment Policy. (See 1.1 C.) The policy requires an Infection Control Risk Assessment (ICRA) to be completed for each OSHPD project. The Acute Care Tower and the Central Energy Plant are Type D Projects (Major Demolition & Construction). The Contractor must meet, at a minimum, the Infection Control requirements of CDC, JCAHO and AIA guidelines for this type of construction.
- 22.5 The Contractor shall implement the conditions of approval identified through the ICRA permit issued by the Infection Control Officer. The Contractor shall incorporate these requirements into their Quality Control Plan.
- 22.6 The Facility's Infection Control or Environmental Health & Safety may modify infection control requirements based on employee safety needs. Any modification does not relieve the Contractor of compliance with proper control procedures. Infection Control or Environmental Health & Safety have the authority to stop work immediately should a clear safety health violation be observed during performance of the Work.
- 22.7 Based on the degree of risk and the type of work to be performed, ICRA conditions of approval may include, but are not limited to, the following requirements:
  - 22.7.1 Before any construction on project site begins, all workers who will be working on site shall attend a mandatory meeting held by County Infection Control authorities, for training and instruction on precautions to be taken. Contractor is responsible for making sure that all workers receive the training.
  - 22.7.2 Develop and implement an appropriate airborne testing program for fungi and respirable dust including baseline, during construction, and post-construction measurements. All samples will be submitted to an independent laboratory accredited by the American Industrial Hygiene Association (AIHA) under the Environmental Microbiology Laboratory Accreditation Program (EMLAP)

- 22.7.3 Disturbed or removed materials shall be cautiously removed, contained and immediately removed to the Contractor's rubbish containment area.
- 22.7.4 In areas undergoing or potentially affected by construction, renovation, excavation, grading or landscaping activities:
  - 22.7.4.1 The existing supply ductwork serving the space shall be partially restricted to reduce the incoming air, as necessary. Temporary filter(s) shall be installed on the ducted returns to collect dust particles from the construction area(s). The overall construction space shall be placed under a negative pressure, when feasible, to minimize contaminating the adjacent areas. Filter media are to be changed by the Contractor as required to maintain negative pressure. Pressure relationships will be checked and logged daily by the Contractor.
  - 22.7.4.2 Maintain negative air pressure within work site utilizing HEPA-equipped air filtration units.
  - 22.7.4.3 Use wet method to control dust while cutting.
  - 22.7.4.4 Control moisture of materials during the construction process.
  - 22.7.4.5 Seal unused doors with non-marring duct tape.
  - 22.7.4.6 Isolate HVAC system in work area to prevent contamination of duct system.
  - 22.7.4.7 Complete all critical barriers before construction begins.
  - 22.7.4.8 Seal holes, pipes, conduits, and punctures prior to the start of work.
  - 22.7.4.9 Barriers must be installed whenever a worker must crawl into or place equipment through any ceiling space in staff or public areas.
  - 22.7.4.10 Barriers must be completely sealed from floor to ceiling to prevent dust from seeping into staff and public areas. The seals must be maintained throughout the construction period.
  - 22.7.4.11 When openings are made into existing ceilings, provide fire retardant, dust-tight polyethylene covering sealed at edges to enclose and dust. Provide thorough cleaning of existing surfaces that become exposed to dust before County's room occupation.

- 22.7.4.12 Whenever access panels are opened for Work above ceilings, provide a polyethylene shroud taped to floor and ceiling enclosing ladder and sealing off opening. All polyethylene shall be fire retardant.
- 22.7.4.13 Ceiling access panels must be closed when unattended.
- 22.7.4.14 Ceiling tiles that are removed must be covered over with plastic and sealed until replaced.
- 22.7.4.15 Construct an anteroom for cleaning equipment and clothing using a HEPA-equipped vacuum cleaner that all personnel are required to pass through as they enter and leave the work area. Alternatively, personnel can wear cloth or paper coveralls that are removed each time they leave the work area. Wet mop or HEPA vacuum the anteroom daily.
- 22.7.4.16 During demolition, dust-producing work, or work in the ceiling, disposable shoe covers and coveralls must be worn and removed in the anteroom when leaving the work area.
- 22.7.4.17 Dust will be kept to a minimum by frequent wet-mopping and placing door mats at entrances.
- 22.7.4.18 Carpets at barricade entrances shall be vacuumed daily using a HEPA filter equipped vacuum cleaner.
- 22.7.4.19 Dust tracked outside of barriers shall be removed immediately.
- 22.7.4.20 Direct construction and debris traffic away from staff and publicareas.
- 22.7.4.21 Debris shall be removed outside of normal work hours.

  Debris shall be transported in tightly covered containers to contain dust.
- 22.7.4.22 Removal of construction barriers and ceiling protection shall be done carefully, outside of normal work hours to minimize spreading of dirt and debris associated with construction. Vacuum and clean all surfaces free of dust after removal.
- 22.7.4.23 Wipe horizontal surfaces with a facility-approved disinfectant.
- 22.7.4.24 Remove blockage and filters from air vents.

- 22.7.5 Problems or questions regarding construction activities should be directed to the County's Representative.
- 22.7.6 If any pre-existing microbiological growth is noted during construction activities, the Facility's Infection Control Officer or the County's Representative must be notified immediately prior to any disturbance of the material.
- 22.7.7 During any of the construction activities described above, Design/Builder must implement an infection compliance monitoring program to ensure compliance with the Infection Control Plan.
- 22.7.8 If portable infectious material containment cubes are to be used, a monitoring, maintenance and servicing protocol must be included in the ICRA.
- 22.7.9 The Facility's Infection Control Officer or the County's Representative have authority to immediately stop work at any time if it is determined that activities or conditions or other emergency situations exist that could impact the health of staff and visitors in the facility.
- 22.7.10 The Facility's Infection Control Officer or the County's Representative have authority to stop work at any time if it is determined that work is not being performed according to these specifications or applicable regulations. The stoppage of work shall continue until conditions have been corrected and corrective steps have been taken to the satisfaction of the County. Standby time required for resolving deficiencies shall be at the Contractor's expense
- 22.8 The Contractor shall notify the County's Representative, Facilities Management, Environmental Health and Safety, and Infection Control before work begins on any construction projects adjacent to staff and visitor areas to establish all necessary and appropriate protective measures to minimize or eliminate risk of nosocomial infections.
- 22.9 The U.S. Department of Health and Human Services, Centers for Disease Control, has published *Guidelines for Environmental Infection Control in Health-Care Facilities*. The tables below, which are taken from the CDC guidelines, list the types of measures that are typically taken to implement infection control programs. These are guidelines to assist the Contractor in their submittal as noted in paragraph 1.2.

Table 2: Strategies to Reduce Dust & Moisture Intrusion During External Demolition & Construction (Adapted from CDC Guidelines Table 8)

Item Recommendation	
---------------------	--

**Alameda County, General Services Agency** Highland Hospital – Koret Building Exhaust Fan and Waterproofing

-	
Dust–generating	Prior to placing dust-generating equipment, evaluate the location to
Equipment	ensure that dust produced by the equipment will not enter any
	hospital building through open doorways or windows, or though
	ventilation air intakes.
Construction	Locate this storage away from the facility and ventilation air intakes.
	Locate this storage away from the facility and ventilation air intakes.
Materials Storage	
Adjacent Air Intakes	Seal off affected intakes, if possible. Take no action without the
	approval of County Representative.
Environment	Determine how environmental issues may affect the project such as
	prevailing winds and outdoor temperatures.
HVAC Systems	Determine location of air intakes for the hospital. Consult with the
Tivite Systems	Alameda County Engineering Department about pressure
	differentials and air recirculation options; Alameda County
	Engineering Department shall be responsible to keep the existing
	building air pressure positive to outside air. Contractor shall assist in
	determining pressure requirements. Pressure gradients between
	clean/dirty areas must meet current requirements. (e.g. 0.01 inches
	of water)
Filters	Alameda County Engineering Department shall ensure that existing
	building filters are properly installed;
	Design-Builder shall record status of existing building filters
	prior to beginning work.
	Design-Builder shall change temporary and roughing filters
	frequently to prevent dust build-up on high-efficiency filters.
Windows	Locate any potential infiltration points pre-construction such as
	windows and doors. Seal and caulk to prevent entry of airborne
	fungal spores.
Doors	When construction activities may impact a nearby building or when
2 0 0 1 2	working in an existing building, keep doors closed as much as
	possible; do not prop open; seal and caulk unused doors (i.e., those
	that are not designated as emergency exits). Use mats with tacky
	surfaces at outside entrances.
Water Utilities	Note location relative to construction area to prevent intrusion of
	dust into water systems. <sup>1</sup>
Medical Gas Piping	Ensure that these lines/pipes are insulated during periods of
	vibration. Isolate as much as practical.
Rooftops	Temporarily close off during active demolition/construction sites;
1	avoid rooftops.
Dust Generation	Provide methods (e.g., misting the area with water) to minimize
Dust Generation	dust.
T	
Immunocompromised	Design-Builder shall coordinate work with the County to allow staff
Patients	and patients to:
	• use walk-ways protected from demolition/ construction sites;
	avoid outside areas close to these sites;
	avoid rooftops.
Pedestrian Traffic	Close off entry ways as needed to minimize dust intrusion.
Truck Traffic	Reroute if possible, or arrange for frequent street cleaning.
Truck Traffic	refoure it possible, of affange for frequent street cleaning.

Education	Encourage reporting of hazardous or unsafe incidents associated
	with construction.
	Educate staff and construction workers about the importance of
	adhering to infection control measures during the project.

<sup>&</sup>lt;sup>1</sup> Contamination of water pipes during demolition activities has been associated with health-care-associated transmission of Legionella.

# 23. Project Inspector

- 23.1 Provision of inspectors by the County, if any, or by the Regulatory Agency pursuant to provisions of this section shall be subject to following:
  - 23.1.1 Contractor shall allow inspectors full access to project at all times Work is in progress.
  - 23.1.2 Contractor shall not take any direction, approvals or disapprovals from inspectors.
  - 23.1.3 Contractor shall not rely on inspectors to ensure Work is completed in accordance with Contract documents.
- 23.2 Acts or omissions of any inspector (including, without limitation, inspector's failure to observe or report deficiencies in Contractor's Work) shall not relieve Contractor from its responsibility to complete Work in accordance with Contract documents.

# 24. Construction Safety Protocols due to Covid-19

24.1 Small Construction Project Safety Protocol listed in Appendix B-1 of the County of Alameda Order of the Health Officer No. 20-10 dated April 29, 2020 will be required on this project. See attached Appendix B-1 immediately following this document.

END OF DOCUMENT

# **Alameda County, General Services Agency**

Highland Hospital – Koret Building Exhaust Fan and Waterproofing

# **Appendix B-2**

# **Large Construction Project Safety Protocol**

- 1. Any construction project meeting any of the following specifications is subject to this Large Construction Project Safety Protocol ("LCP Protocol"), including public works projects unless otherwise specified by the Health Officer:
  - a. For residential construction projects, any single-family, multi-family, senior, student, or other residential construction, renovation, or remodel project consisting of more than 10 units.
  - b. For commercial construction projects, any construction, renovation, or tenant improvement project consisting of more than 20,000 square feet of floor area.
  - c. For construction of Essential Infrastructure, as defined in section 16.c of the Order, any project that requires five or more workers at the jobsite at any one time.
- 2. The following restrictions and requirements must be in place at all construction job sites subject to this LCP Protocol:
  - a. Comply with all applicable and current laws and regulations including but not limited to OSHA and Cal-OSHA. If there is any conflict, difference or discrepancy between or among applicable laws and regulations and/or this LCP Protocol, the stricter standard will apply.
  - b. Prepare a new or updated Site-Specific Health and Safety Plan to address COVID-19-related issues, post the Plan on-site at all entrances and exits, and produce a copy of the Plan to County governmental authorities upon request. The Plan must be translated as necessary to ensure that all non-English speaking workers are able to understand the Plan.
  - c. Provide personal protective equipment (PPE) specifically for use in construction, including gloves, goggles, face shields, and face coverings as appropriate for the activity being performed. At no time may a contractor secure or use medical-grade PPE, unless required due to the medical nature of a job site. Face coverings must be worn in compliance with the April 17, 2020 Guidance from the County of Santa Clara Public Health Department, available at: https://www.sccgov.org/sites/covid19/Pages/learn-what-to-do.aspx#howto.
  - d. Ensure that employees are trained in the use of PPE. Maintain and make available a log of all PPE training provided to employees and monitor all employees to ensure proper use of the PPE.
  - e. Prohibit sharing of PPE.

# Appendix B-2

- f. Implement social distancing requirements including, at minimum:
  - i. Stagger stop- and start-times for shift schedules to reduce the quantity of workers at the jobsite at any one time to the extent feasible.
  - ii. Stagger trade-specific work to minimize the quantity of workers at the jobsite at any one time.
  - iii. Require social distancing by maintaining a minimum six-foot distance between workers at all times, except as strictly necessary to carry out a task associated with the project.
  - iv. Prohibit gatherings of any size on the jobsite, except for safety meetings or as strictly necessary to carry out a task associated with the project.
  - v. Strictly control "choke points" and "high-risk areas" where workers are unable to maintain minimum six-foot social distancing and prohibit or limit use to ensure that minimum six-foot distancing can easily be maintained between workers.
  - vi. Minimize interactions and maintain social distancing with all site visitors, including delivery workers, design professional and other project consultants, government agency representatives, including building and fire inspectors, and residents at residential construction sites.
  - vii. Prohibit workers from using others' phones or desks. Any work tools or equipment that must be used by more than one worker must be cleaned with disinfectants that are effective against COVID-19 before use by a new worker.
  - viii. Place wash stations or hand sanitizers that are effective against COVID-19 at entrances to the jobsite and in multiple locations dispersed throughout the jobsite as warranted.
    - ix. Maintain a daily attendance log of all workers and visitors that includes contact information, including name, address, phone number, and email.
    - x. Post a notice in an area visible to all workers and visitors instructing workers and visitors to do the following:
      - 1. Do not touch your face with unwashed hands or with gloves.
      - 2. Frequently wash your hands with soap and water for at least 20 seconds or use hand sanitizer with at least 60% alcohol.
      - 3. Clean and disinfect frequently touched objects and surfaces such as workstations, keyboards, telephones, handrails, machines, shared tools, elevator control buttons, and doorknobs.
      - 4. Cover your mouth and nose when coughing or sneezing or cough or sneeze into the crook of your arm at your elbow/sleeve.
      - 5. Do not enter the jobsite if you have a fever, cough, or other COVID-19 symptoms. If you feel sick, or have been exposed to anyone who is sick, stay at home.
      - 6. Constantly observe your work distances in relation to other staff. Maintain the recommended minimum six-feet distancing at all times when not wearing the necessary PPE for working in close proximity to another person.
      - 7. Do not share phones or PPE.

#### **Alameda County, General Services Agency**

Highland Hospital – Koret Building Exhaust Fan and Waterproofing

#### Appendix B-2

- xi. The notice in section 2.f.x must be translated as necessary to ensure that all non-English speaking workers are able to understand the notice.
- g. Implement cleaning and sanitization practices in accordance with the following:
  - i. Frequently clean and sanitize, in accordance with CDC guidelines, all high-traffic and high-touch areas including, at a minimum: meeting areas, jobsite lunch and break areas, entrances and exits to the jobsite, jobsite trailers, hand-washing areas, tools, equipment, jobsite restroom areas, stairs, elevators, and lifts.
  - ii. Establish a cleaning and decontamination protocol prior to entry and exit of the jobsite and post the protocol at entrances and exits of jobsite.
  - iii. Supply all personnel performing cleaning and sanitization with proper PPE to prevent them from contracting COVID-19. Employees must not share PPE.
  - iv. Establish adequate time in the workday to allow for proper cleaning and decontamination including prior to starting at or leaving the jobsite for the day.
- h. Implement a COVID-19 community spread reduction plan as part of the Site-Specific Health and Safety Plan that includes, at minimum, the following restrictions and requirements:
  - i. Prohibit all carpooling to and from the jobsite except by workers living within the same household unit, or as necessary for workers who have no alternative means of transportation.
  - ii. Cal-OSHA requires employers to provide water, which should be provided in singleserve containers. Prohibit any sharing of any food or beverage and if sharing is observed, the worker must be sent home for the day.
  - iii. Prohibit use of microwaves, water coolers, and other similar shared equipment.
- i. Assign a COVID-19 Safety Compliance Officer (SCO) to the jobsite and ensure the SCO's name is posted on the Site-Specific Health and Safety Plan. The SCO must:
  - i. Ensure implementation of all recommended safety and sanitation requirements regarding the COVID-19 virus at the jobsite.
  - ii. Compile daily written verification that each jobsite is compliant with the components of this LCP Protocol. Each written verification form must be copied, stored, and made immediately available upon request by any County official.
  - iii. Establish a daily screening protocol for arriving staff, to ensure that potentially infected staff do not enter the construction site. If workers leave the jobsite and return the same day, establish a cleaning and decontamination protocol prior to entry and exit of the jobsite. Post the daily screening protocol at all entrances and exit to the jobsite. More information on screening can be found online
    - at: https://www.cdc.gov/coronavirus/2019-ncov/community/index.html.
  - iv. Conduct daily briefings in person or by teleconference that must cover the following topics:
    - 1. New jobsite rules and pre-job site travel restrictions for the prevention of COVID-19 community spread.
    - 2. Review of sanitation and hygiene procedures.
    - 3. Solicitation of worker feedback on improving safety and sanitation.
    - 4. Coordination of construction site daily cleaning/sanitation requirements.
    - 5. Conveying updated information regarding COVID-19.

#### **Alameda County, General Services Agency**

Highland Hospital – Koret Building Exhaust Fan and Waterproofing

#### Appendix B-2

- 6. Emergency protocols in the event of an exposure or suspected exposure to COVID-19.
- v. Develop and ensure implementation of a remediation plan to address any non-compliance with this LCP Protocol and post remediation plan at entrance and exit of jobsite during remediation period. The remediation plan must be translated as necessary to ensure that all non-English speaking workers are able to understand the document.
- vi. The SCO must not permit any construction activity to continue without bringing such activity into compliance with these requirements.
- vii. Report repeated non-compliance with this LCP Protocol to the appropriate jobsite supervisors and a designated County official.
- j. Assign a COVID-19 Third-Party Jobsite Safety Accountability Supervisor (JSAS) for the jobsite, who at a minimum holds an OSHA-30 certificate and first-aid training within the past two years, who must be trained in the protocols herein and verify compliance, including by visual inspection and random interviews with workers, with this LCP Protocol.
  - i. Within seven calendar days of each jobsite visit, the JSAS must complete a written assessment identifying any failure to comply with this LCP Protocol. The written assessment must be copied, stored, and, upon request by the County, sent to a designated County official.
  - ii. If the JSAS discovers that a jobsite is not in compliance with this LCP Protocol, the JSAS must work with the SCO to develop and implement a remediation plan.
  - iii. The JSAS must coordinate with the SCO to prohibit continuation of any work activity not in compliance with rules stated herein until addressed and the continuing work is compliant.
  - iv. The remediation plan must be sent to a designated County official within five calendar days of the JSAS's discovery of the failure to comply.
- k. In the event of a confirmed case of COVID-19 at any jobsite, the following must take place:
  - i. Immediately remove the infected individual from the jobsite with directions to seek medical care.
  - ii. Each location the infected worker was at must be decontaminated and sanitized by an outside vendor certified in hazmat clean ups, and work in these locations must cease until decontamination and sanitization is complete.
  - iii. The County Public Health Department must be notified immediately and any additional requirements per the County health officials must be completed, including full compliance with any tracing efforts by the County.
- Where construction work occurs within an occupied residential unit, any separate work area must be sealed off from the remainder of the unit with physical barriers such as plastic sheeting or closed doors sealed with tape to the extent feasible. If possible, workers must access the work area from an alternative entry/exit door to the entry/exit door used by residents. Available windows and exhaust fans must be used to ventilate the work area. If residents have access to the work area between workdays, the work area must be cleaned and

#### **Appendix B-2**

sanitized at the beginning and at the end of workdays. Every effort must be taken to minimize contact between workers and residents, including maintaining a minimum of six feet of social distancing at all times.

m. Where construction work occurs within common areas of an occupied residential or commercial building or a mixed-use building in use by on-site employees or residents, any separate work area must be sealed off from the rest of the common areas with physical barriers such as plastic sheeting or closed doors sealed with tape to the extent feasible. If possible, workers must access the work area from an alternative building entry/exit door to the building entry/exit door used by residents or other users of the building. Every effort must be taken to minimize contact between worker and building residents and users, including maintaining a minimum of six feet of social distancing at all times.

#### **Alameda County, General Services Agency**

Highland Hospital – Koret Building Exhaust Fan and Waterproofing

#### **Appendix B-1**

#### **Small Construction Project Safety Protocol**

- 1. Any construction project meeting any of the following specifications is subject to this Small Construction Project Safety Protocol ("SCP Protocol"), including public works projects unless otherwise specified by the Health Officer:
  - a. For residential projects, any single-family, multi-family, senior, student, or other residential construction, renovation, or remodel project consisting of 10 units or less. This SCP Protocol does not apply to construction projects where a person is performing construction on their current residence either alone or solely with members of their own household.
  - b. For commercial projects, any construction, renovation, or tenant improvement project consisting of 20,000 square feet of floor area or less.
  - c. For mixed-use projects, any project that meets both of the specifications in subsection 1.a and 1.b.
  - d. All other construction projects not subject to the Large Construction Project Safety Protocol set forth in Appendix B-2.
- 2. The following restrictions and requirements must be in place at all construction job sites subject to this SCP Protocol:
  - a. Comply with all applicable and current laws and regulations including but not limited to OSHA and Cal-OSHA. If there is any conflict, difference, or discrepancy between or among applicable laws and regulations and/or this SCP Protocol, the stricter standard shall apply.
  - b. Designate a site-specific COVID-19 supervisor or supervisors to enforce this guidance. A designated COVID-19 supervisor must be present on the construction site at all times during construction activities. A COVID-19 supervisor may be an on-site worker who is designated to serve in this role.
  - c. The COVID-19 supervisor must review this SCP Protocol with all workers and visitors to the construction site.
  - d. Establish a daily screening protocol for arriving staff to ensure that potentially infected staff do not enter the construction site. If workers leave the jobsite and return the same day, establish a cleaning and decontamination protocol prior to entry and exit of the jobsite. Post the daily screening protocol at all entrances and exits to the jobsite. More information on screening can be found online at: https://www.cdc.gov/coronavirus/2019-ncov/community/index.html.
  - e. Practice social distancing by maintaining a minimum six-foot distance between workers at all times, except as strictly necessary to carry out a task associated with the construction project.

APPENDIX B-1 SMALL CONSTRUCTION PROJECT SAFETY PROTOCOL

#### **Appendix B-1**

- f. Where construction work occurs within an occupied residential unit, separate work areas must be sealed off from the remainder of the unit with physical barriers such as plastic sheeting or closed doors sealed with tape to the extent feasible. If possible, workers must access the work area from an alternative entry/exit door to the entry/exit door used by residents. Available windows and exhaust fans must be used to ventilate the work area. If residents have access to the work area between workdays, the work area must be cleaned and sanitized at the beginning and at the end of workdays. Every effort must be taken to minimize contact between workers and residents, including maintaining a minimum of six feet of social distancing at all times.
- g. Where construction work occurs within common areas of an occupied residential or commercial building or a mixed-use building in use by on-site employees or residents, separate work areas must be sealed off from the rest of the common areas with physical barriers such as plastic sheeting or closed doors sealed with tape to the extent feasible. If possible, workers must access the work area from an alternative building entry/exit door to the building entry/exit door used by residents or other users of the building. Every effort must be taken to minimize contact between worker and building residents and users, including maintaining a minimum of six feet of social distancing at all times.
- h. Prohibit gatherings of any size on the jobsite, including gatherings for breaks or eating, except for meetings regarding compliance with this protocol or as strictly necessary to carry out a task associated with the construction project.
- i. Cal-OSHA requires employers to provide water, which should be provided in single-serve containers. Sharing of any food or beverage is strictly prohibited and if sharing is observed, the worker must be sent home for the day. Use of microwaves, water coolers, and other similar shared equipment is prohibited.
- j. Provide personal protective equipment (PPE) specifically for use in construction, including gloves, goggles, face shields, and face coverings as appropriate for the activity being performed. At no time may a contractor secure or use medical-grade PPE unless required due to the medical nature of a jobsite. Face coverings must be worn in compliance with the April 17, 2020 Guidance from the County of Santa Clara Public Health Department, available at: https://www.sccgov.org/sites/covid19/Pages/learn-what-to-do.aspx#howto.
- k. Strictly control "choke points" and "high-risk areas" where workers are unable to maintain six-foot social distancing and prohibit or limit use to ensure that six-foot distance can easily be maintained between individuals.
- Minimize interactions and maintain social distancing with all site visitors, including delivery
  workers, design professional and other project consultants, government agency
  representatives, including building and fire inspectors, and residents at residential
  construction sites.
- m. Stagger trades as necessary to reduce density and allow for easy maintenance of minimum six-foot separation.

APPENDIX B-1 SMALL CONSTRUCTION PROJECT SAFETY PROTOCOL

#### Appendix B-1

- n. Discourage workers from using others' desks, work tools, and equipment. If more than one worker uses these items, the items must be cleaned and disinfected with disinfectants that are effective against COVID-19 in between use by each new worker. Prohibit sharing of PPE.
- o. If hand washing facilities are not available at the jobsite, place portable wash stations or hand sanitizers that are effective against COVID-19 at entrances to the jobsite and in multiple locations dispersed throughout the jobsite as warranted.
- p. Clean and sanitize any hand washing facilities, portable wash stations, jobsite restroom areas, or other enclosed spaces daily with disinfectants that are effective against COVID-19. Frequently clean and disinfect all high touch areas, including entry and exit areas, high traffic areas, rest rooms, hand washing areas, high touch surfaces, tools, and equipment
- q. Maintain a daily attendance log of all workers and visitors that includes contact information, including name, phone number, address, and email.
- r. Post a notice in an area visible to all workers and visitors instructing workers and visitors to do the following:
  - i. Do not touch your face with unwashed hands or with gloves.
  - ii. Frequently wash your hands with soap and water for at least 20 seconds or use hand sanitizer with at least 60% alcohol.
  - iii. Clean and disinfect frequently touched objects and surfaces such as work stations, keyboards, telephones, handrails, machines, shared tools, elevator control buttons, and doorknobs.
  - iv. Cover your mouth and nose when coughing or sneezing, or cough or sneeze into the crook of your arm at your elbow/sleeve.
  - v. Do not enter the jobsite if you have a fever, cough, or other COVID-19 symptoms. If you feel sick, or have been exposed to anyone who is sick, stay at home.
  - vi. Constantly observe your work distances in relation to other staff. Maintain the recommended minimum six feet at all times when not wearing the necessary PPE for working in close proximity to another person.
  - vii. Do not carpool to and from the jobsite with anyone except members of your own household unit, or as necessary for workers who have no alternative means of transportation.
  - viii. Do not share phones or PPE.
- s. In the event of a confirmed case of COVID-19 at any jobsite, the following must take place:
  - i. Immediately remove the infected individual from the jobsite with directions to seek medical care.
  - ii. Each location the infected worker was at must be decontaminated and sanitized by an outside vendor certified in hazmat clean ups, and work in these locations must cease until decontamination and sanitization is complete.
  - iii. The County Public Health Department must be notified immediately and any additional requirements per the County health officials must be completed, including full compliance with any tracing efforts by the County.

**DOCUMENT 00 73 49** 

## PROJECT STABILIZATION/COMMUNITY BENEFIT AGREEMENT of the COUNTY OF ALAMEDA

and

California Prevailing Wage Requirements

#### 1. Summary

1.1. In addition to Labor, Wage & Hour, Apprentice, and related provisions described in Document 00 72 13 Paragraph 26; the Work performed pursuant to this Contract is subject to the requirements of the "PROJECT STABILIZATION/COMMUNITY BENEFITS AGREEMENT for the COUNTY OF ALAMEDA" ("PSCBA"). The Contractor agrees to be party to and bound by the "PROJECT STABILIZATION/COMMUNITY BENEFITS AGREEMENT for the COUNTY OF ALAMEDA". Contractor agrees to execute the "PROJECT STABILIZATION/COMMUNITY BENEFITS AGREEMENT for the COUNTY OF ALAMEDA Letter of Assent" and shall require all of its subcontractors, of whatever tier, to become similarly bound for all work within the scope of this Contract by signing an identical Letter of Assent.

#### 2. PROJECT STABILIZATION/COMMUNITY BENEFIT AGREEMENT Of the COUNTY OF ALAMEDA

2.1. The PROJECT STABILIZATION/COMMUNITY BENEFITS AGREEMENT for the COUNTY OF ALAMEDA (PSCBA) is included for reference only in PROJECT STABILIZATION/COMMUNITY BENEFITS AGREEMENT for the COUNTY OF ALAMEDA Document 00 73 49B.

#### 2.1.1. ROLES AND RESPONSIBILITIES SUBCONTRACTS

- 2.1.1.1. Each Contractor, which includes all subcontractors of any tier, including trucking entities performing Covered Work of this Contract, agrees that neither it nor any of its subcontractors will subcontract any Work of this Contract except to a person, firm, or corporation who is or becomes party to the PSCBA by signing the Letter of Assent attached to the PSCBA as Exhibit "A". All Contractors performing Covered Work of this Contract shall, as a condition to performing Work of this Contract, become Signatory to and perform all work under the terms of the PSCBA.
- 2.1.1.2. Each Contractor, which includes all subcontractors of any tier performing Work of this Contract, shall give written notice to the Union(s) of any subcontract involving the performance of work covered by the PSCBA

within either five (5) business days of executing a contract with such subcontract or before the subcontractor commences work on the Project, whichever occurs first. Such notice shall specify the name and address of the subcontractor, the California State License Board license number of the Contractors and scope of work to be performed. Written notice at a Pre-Job Conference shall be deemed written notice under this provision only for those subcontractors listed at the Pre-Job Conference

2.1.1.3. The Contractor shall be responsible for PSCBA compliance by all subcontractor and lower tier subcontractor.

#### 2.1.2. WORK ASSIGNMENTS AND JUSIDICTIONAL DISPUTES

- 2.1.2.1. The assignment of the Work to subcontractors is solely the responsibility of the Contractor.
- 2.1.2.2. Each Contractor shall conduct a Pre-Job Conference with the Building and Construction Trades Council of Alameda County (Council) prior to commencing Work as specified in Paragraph 2.1.3 of this Document 00 73 49. The Contractor will notify the County in advance of all such conferences.
- 2.1.2.3. Any jurisdictional disputes regarding the assignment of the Work of this Contract will be resolved per the requirements of the PSCBA.

#### 2.1.3. PRE-JOB CONFERENCE

- 2.1.3.1. A mandatory Pre-Job Conference and/or Mark-Up Meeting will be held prior to the commencement of work to establish the scope of work in each Contractor and Subcontractor contract. All meeting shall be held at the offices of the Alameda County Building and Construction Trades Council.
- 2.1.3.2. The Contractor performing the work shall have the responsibility for making work assignments in accordance with the PSCBA, and will be required to bring relevant plans, specifications, and blueprints to the meeting, as requested by the Union
- 2.1.3.3. Contractor must submit written workforce projections at the Pre-Job Conference. The workforce projections shall include projected manhours on a craft-by-craft basis, consistent with the Contractor's bid proposal.
- 2.1.3.4. The County will schedule and attend all Pre-Job and Mark-Up Meetings and participate in discussions as they pertain to the terms and conditions of the PSCBA.

#### 2.1.4. JOINT ADMINISTRATIVE COMMITTEE MEETINGS

- 2.1.4.1. The Joint Administrative Committee (JAC) has been established to monitor compliance with the PSCBA. The JAC meets monthly and reviews monthly reporting by the Contractor.
- 2.1.4.2. The Contractors shall provide progress report as described in Paragraph 2.1.8 of this Document.

#### 2.1.5. COORDINATOR

2.1.5.1. The County will designate a Coordinator, who will be responsible for the administration and application of the PSCBA.

#### 2.1.6. LOCAL HIRING PROGRAM

- 2.1.6.1. The Contractor agrees to achieve the inclusion of Residents as defined in the PSCBA in the employment and apprenticeship opportunities created by the Work of this Contract, which will be known as the Local Hiring Program (LHP) as described in the PSCBA.
- 2.1.6.2. The Contractor agrees to a goal that Residents of the County will perform forty percent (40%) of all hours worked on the Work of this Contract, on a craft-by-craft basis, if such workers are available, capable and willing to work on the projects, together with the apprentice goals described in Paragraph 2.1.7 of this Document.
- 2.1.6.3. The Contractors and subcontractors shall make good faith efforts to reach these goals, as described in the PSCBA including but not limited to the following:
  - 2.1.6.3.1. Within one week of the issuance of the Notice to Proceed, the Contractors shall meet with the County to review and approve its compliance plan for reaching the Local Hiring Goals, using the required compliance plan form provided by the County.
  - 2.1.6.3.2. Submit copies of hiring hall dispatch requests and responses to the County within ten (10) days of County's request at any point during the execution of the Work of this Contract.
  - 2.1.6.3.3. Immediately contact the County if a union hiring hall dispatcher will not or cannot, upon request of the Contractor, dispatch local residents.

- 2.1.6.3.4. Use the "Name Call," "Rehire" or other available hiring hall procedures to reach goals and shall provide documentation of such requests to the County upon request.
- 2.1.6.3.5. Use community based organizations as a resource for local labor resources, if a union will not or cannot provide local Residents as requested
- 2.1.6.3.6. Sponsor local Residents for apprenticeship, when possible.
- 2.1.6.3.7. Maintain records for each Resident of Alameda County who was referred but not hired along with an explanation why the worker was not hired.
- 2.1.6.3.8. Document participation in any local employment training programs and submit documentation of such to the County within ten (10) days if requested by County.
- 2.1.6.3.9. To the extent possible, the parties agree to implement the Local Hiring Program while complying with the County's Local Vendor Preference and Enhanced Construction Outreach (ECOP) programs for the work of this Contract. To the extent that the County determines, in its sole discretion, that there is a conflict between the Local Hiring Program established in the PSCBA and the County's SLEB, ECOP, and/or Local Vendor Preference Programs, the conflict shall be resolved in favor of the Local Hiring Program of the PSCBA.
- 2.1.6.3.10. For the purpose of reaching the goal established in Paragraph 2.1.6.2 of this Document, a Contractor may qualify for full credit toward the goal by employing Alameda County Residents for other work the Contractor is performing in any of the nine Bay Area counties of: Alameda, Contra Costa, San Francisco, San Mateo, Santa Clara, Marin, Solano, Napa and Sonoma as outlined in the PSCBA.

#### 2.1.7. APPRENTICES

- 2.1.7.1. Although the PSCBA states that the County shall make available to the Unions a database of apprentices qualifying under the local hiring provision of the PSCBA, the County has not developed this database. Contractor is to contact the Unions for available apprentices.
  - 2.1.7.1.1. For each Covered Project, the Contractors will be responsible to ensure that it and/or its subcontractors hire at least one (1) new apprentice for the first \$1 million of construction value and for each succeeding \$5

million of construction contract value, the Contractors and/or their subcontractors will be required to hire at least one (1) additional new apprentice. All such apprentices may be graduates of pre apprenticeship programs with known and successful track record of apprentice placement into jobs. All the pre apprenticeship program graduates must be Residents of Alameda County and members of a Disadvantaged Population, as described in the PSCBA.

- 2.1.7.2. Contractors shall exercise their best efforts to recruit apprenticeship program applicants from Residents and who are members of a Disadvantaged Population as described in the PSCBA
- 2.1.7.3. The Contractor shall request dispatch of apprentices in writing from the local Unions and/or Joint Apprenticeship Training Committee in which the Contractor participates. Copies of the written requests shall be provided to the County within ten (10) days of request by the Coordinator.
- 2.1.7.4. For the purposes of meeting the goal established in Paragraph 2.1.6.1 of this Document, a Contractor may qualify for full credit toward the goal by employing Alameda County Residents as apprentices for other work the Contractor is performing in any of the nine Bay Area counties of: Alameda, Contra Costa, San Francisco, San Mateo, Santa Clara, Marin, Solano, Napa and Sonoma as described in the PSCBA

#### 2.1.8. DATA COLLECTION AND REPORTING

- 2.1.8.1. This Paragraph describes Contractor and data collection, reporting guidelines and responsibilities for the PSCBA.
- 2.1.8.2. On a monthly basis, Contractors must submit reports to the County on the status and progress of local hiring on a craft-by-craft basis, including utilization of apprentices as described in Document 00 73 49A "PSCBA Forms".

#### 2.1.9. HELMETS TO HARDHATS: VETERAN EMPLOYMENT

2.1.9.1. The Contractor agrees to utilize the series of the Center for Military Recruitment, Assessment and Veterans Employment (hereinafter "Center") and Center's "Helmets to Hardhats" program to serve as a resources for preliminary orientations, assessment of construction aptitude, referral to apprenticeship programs or hiring halls, counseling and mentoring, support network, employment opportunities and other needs as described in the PSCBA.

#### **Alameda County, General Services Agency**

Highland Hospital – Koret Building Exhaust Fan and Waterproofing

- 2.1.9.2. The Contractors may also utilize the services of the "Swords to Ploughshares" program.
- **3.** <u>California Labor Code</u>: In addition to complying with the PSCBA, Contractor shall also comply with the California Labor Code prevailing wage requirements.
  - 3.1. Pursuant to Labor Code Section 1770, *et seq.*, the Contractor shall pay to persons performing labor in and about the Work provided for in the Contract an amount equal to or more than the general prevailing rate of per diem wages for work of a similar character in the locality in which the Work is performed, and not less than the general prevailing rate of per diem wages for legal holiday and overtime work in said locality, which per diem wages shall be equal to or more than the stipulated rates contained in a schedule thereof which has been ascertained and determined by the Director of the State Department of Industrial Relations to be the general prevailing rate of per diem wages for each craft or type of workman or mechanic needed to execute this contract. The Contractor shall also cause a copy of this determination of the prevailing rate of per diem wages to be posted at each Site.
  - 3.2. The Contractor shall forfeit, as a penalty to the County, fifty dollars (\$50.00) for each laborer, workman, or mechanic employed in performing labor in and about the work provided in the Contract Documents for each day, or portion thereof, on which such laborer, workman or mechanic is paid less than the said stipulated rates for any work done under these Contract Documents by him or her or by any Subcontractor or designer under him or her, in violation of Articles 1 and 2 of Chapter 1 of Part 7 of Division II of the Labor Code. The sums and amounts which shall be forfeited pursuant to this paragraph 3.2 and the terms of the Labor Code shall be withheld and retained from payments due or to become due to the Contractor under this Contract and the terms of the Labor Code, but no sum shall be so withheld, retained or forfeited except from the final payment without a full investigation by either the State Department of Industrial Relations or by the County. The final amount of forfeiture shall be determined by the Labor Commissioner pursuant to Labor Code § 1775.
  - 3.3. The Contractor shall insert in every subcontract or other arrangement which Contractor may make for performance of work or labor on the Work provided for in the Contract Documents, a provision that the Subcontractor shall pay persons performing labor or rendering service under subcontract or other arrangement not less than the general prevailing rate of per diem wages for work of a similar character in the locality in which the Work is performed, and not less than the general prevailing rate of per diem wages for holiday and overtime work fixed as provided in the Labor Code.
  - 3.4. The Contractor stipulates that it shall comply with all applicable wage and hour laws, including without limitation Labor Code § 1813.
- 4. <u>Project Stabilization/Community Benefits Agreement/ Labor Compliance Program</u> Monitoring.

- 4.1. The County has elected to retain the services of a third party to monitor compliance with the PSCBA and California Labor Code Requirement.
- 4.2. The PSCBA/Labor Compliance Program ("PSCBA/LCP") will enforce PSCBA, prevailing wage, apprentice employment and local hiring requirements consistent with California Labor Code and the PSCBA. PSCBA/LCP services do not limit the scope of Work and do not relieve the Contractor of any responsibility for coordination of the Work with California Labor Code or the PSCBA.
- 4.3. The Contractor shall be responsible for any costs that the County incurs as the result of any actions taken by DIR, or by the County when exercising its enforcement duties, to address Contractor and/or Subcontractor violations related to California Labor Code or the PSCBA. If the Contractor or any of its Subcontractor are notified that they should take certain actions to be in compliance with the PSCBA or applicable state law and those actions are not taken or not taken in a timely manner, then the County shall have the right to recover the cost of all work performed by or for the County or its contractors from the date of such notice and the County shall have the right to back charge the Contractor for any and all costs associated with such work.
- 4.4. Certified payroll reports for the duration of the Project shall be maintained by the Contractor and submitted electronically, and are subject to all of the following conditions:
  - 4.4.1. Certified Payroll Reports (CPR) shall be submitted to the County electronically on the web-based software system, described in Document 00 45 46.01 "Prevailing Wage and Related Labor Requirements Certification", to be utilized for collection and verification of payroll reports for the Project.
  - CPR must contain all of information required by California Labor Code section 4.4.2. 1776 and must be organized in a manner that is similar or identical to the format in which the information is reported on the DIR "Public Works Payroll Reporting Form" (Form A-1-131);
  - Statement of Compliance. CPR shall be accompanied by a signed "Statement of 4.4.3. Compliance" certifying that the payroll reports are correct and complete and that each laborer or mechanic has been paid not less than the proper prevailing wage rate for the work performed. The wording of the certification shall comply with California Labor Code section 1776 and 29 C.F.R. § 5.5(a)(3)(ii)(B)-(D).
  - Electronic CPR submitted to the County, the DIR Division of Labor Standards 4.4.4. Enforcement (DLSE), or other entity within the DIR, must be in the form of a non-modifiable image or record that bears an electronic signature or includes a copy of any original certification made on paper. Printed reports submitted on paper with an original signature will be accepted as supplemental information to electronic reports, and will not relieve the Contractor or its Subcontractor from their obligation to submit electronic reports.

#### **Alameda County, General Services Agency**

**PROJECT # 2043** 

Highland Hospital – Koret Building Exhaust Fan and Waterproofing

4.4.5. <u>Apprenticeship Program</u>. Reference is made to General Conditions Document 00 72 13, Paragraph 26 and the PSCBA for the Contractor and its Subcontractors obligation to comply, and be responsible for ensuring compliance, with the requirements of the California Labor Code provisions concerning the employment of apprentices, including Labor Code sections 1776, 1777.5, and 1777.6.

END OF DOCUMENT

**DOCUMENT 00 73 49A** 

# PROJECT STABILIZATION/COMMUNITY BENEFIT AGREEMENT of the COUNTY OF ALAMEDA FORMS

#### 1. Summary

1.1. The Contractor and each subcontractor at all tiers must complete and submit all forms required by the Labor Compliance Program Guidebook included in this DOCUMENT 00 73 49A "PSCBA FORMS".

END OF DOCUMENT



www.davillier-sloan.com

#### **Labor Compliance Program Guidebook**

Please provide this packet to all subcontractors with instruction that they provide it to all lower tier subcontractors. The contractor is ultimately responsible for labor compliance on the entire project.

Labor Compliance

Certified Payroll

Elvira Didic (510) 385-1255

Elation Systems, Inc. (925) 924 - 0340

elvira@davillier-sloan.com

support@elationsystems.com

#### **TABLE OF CONTENTS**

#### A. INTRODUCTION

- A. Overview
- B. Prevailing Wage Determinations
- C. Site Visits
- D. Required Forms
- E. Project/Program Closeout
- F. Apprenticeship Utilization

#### B. PRIOR TO CONSTRUCTION FORMS

- A. Checklist of Labor Law Requirements
- B. Public Works Contract Award Information (DAS 140 Form)
- C. Request for Dispatch of an Apprentice (DAS 142 Form)

#### C. DURING CONSTRUCTION SUBMITTALS

- A. Statement of Employer Payments
- B. CAC Training Fund Contributions (CAC-2 Form)
- C. Statement of Non-Performance
- D. Public Works Payroll Reporting From (A-1-131 Form)
- E. Verification of Apprenticeship Status

#### D. PROJECT/PROGRAM CLOSEOUT

A. Contractor Affidavit



# Section 1: Introduction

- A. Overview
- B. Prevailing Wage Determinations
- C. Site Visits
- D. Required Forms
- E. Project/Program Closeout
- F. Apprenticeship Utilization

#### Overview

Davillier-Sloan, Inc. (DSI) is a third party Labor Compliance Program (LCP) administrator. This LCP Guidebook has been developed for your reference and highlights the requirements, submittals and timelines necessary to be compliant with the program.

Certified Payrolls for this project will be submitted electronically into the Elation web based program. Related forms will be available online and should be accessed and downloaded according to the instructions provided by Elation Systems once you have registered on the site.

The law requires that all workers including employees, independent contractors, owneroperators etc. on a public works project must be paid the prevailing wage of the area in which the project is located. Detailed information pertaining to labor compliance may be found in the contract specifications.

#### **Prevailing Wage Determinations**

The California prevailing wage determinations are based on the first bid advertisement/publication date. For design build projects, the construction contract signing/construction contract award date shall be considered the bid advertisement date.

The California prevailing wage determinations and rates are published twice each year, in February and August. All determinations are effective ten (10) days after issuance. Some trades are issued regionally (Northern and Southern California) and other sub trades are by the county in which the project is located. There are separate determinations for apprentices on public works.

The prevailing wage determination by craft can be found on the Department of Industrial Relations (DIR) web site: <a href="www.dir.ca.gov">www.dir.ca.gov</a> (Labor Law/Public Works). Prevailing wage determinations and any rate changes must be posted at the job site available for workers to view.

#### Asterisk (\*) clarifications:

- i Prevailing wage determinations with a single asterisk (\*) after the expiration date, which are in effect on the date of advertisement of bids, remain in effect for the life of the project.
- i Interested parties should contact the DIR at (415) 703-4774 for the new rates after ten (10) days from the expiration date (if no subsequent determination is required).
- i Prevailing wage determinations with double asterisks (\*\*) after the expiration date indicate that the basic hourly rate, overtime, holiday pay and employers' payments for work performed after this date have been predetermined. If work is to extend past this date, the new rates must be paid and should be incorporated into contracts entered now.

#### **Site Visits**

Site visits will be conducted weekly pursuant to Labor Code 16432(d). Information on certified payrolls will be verified by visual inspection and random in-person worker interviews.

#### **Required Forms**

The required forms are available for download in the Elations system. Copies of completed, signed forms should be forwarded to the appropriate agency and uploaded into Elation for verification.

#### **Prior to Construction Forms**

#### 1. Checklist of Labor Law Requirements

The contractor and each subcontractor at all tiers must complete and submit this form acknowledging the California Labor Codes Regulations governing public works projects.

#### 2. Division of Apprenticeship Standards Form DAS 140

Public Works Contract Award Information

- The contractor and each subcontractor at all tiers must complete and submit this form to the local Apprenticeship Committee to inform them of the award of your contract.
- Submit the DAS 140 to the Joint Apprenticeship Training Committee (JATC) for each apprentice able craft or trade within the area of the project site. The contractor and each subcontractor at all tiers must submit this form within ten (10) days of the date of the execution of the contract but no later than the first day the contractor has workers employed on-site.

#### 3. <u>Division of Apprenticeship Standards Form DAS 142</u>

Request for Dispatch of an Apprentice

- The contractor and each subcontractor at all tiers must complete and submit a Request for Dispatch of an Apprentice in writing at least 72 business hours prior to the date apprentices are needed.
- Submit the DAS 142 to each of the JATCs in the area of the project for each apprenticeable craft, until the required number of apprentices has been provided. If the required number of apprentices is not provided and a request has been submitted to all of the Committee's in the area of the project, then the contractor shall be considered in compliance.

#### **During Construction Forms**

#### 1. Statement of Employer Payments

- i Must be submitted with the first certified payroll, when prevailing wage rates are updated, and when there is a change in fringe benefits.
- i Additional annuity payments can be indicated in the notes section of the CPR and a union dispatch slip should be uploaded into the Elation System.

#### 2. California Apprentice Council Training Fund Contribution (CAC –2)

The training fund contributions to the CAC are due on the 15<sup>th</sup> of each month for work performed during the preceding month. Refer to the DIR applicable prevailing wage

determinations for the amount owed for each hour of work performed for journeymen and apprentices.

#### 3. <u>Statement of Non-Performance</u> (when applicable)

- This form is submitted when the contractor is not working on the job site for a period of more than one week but has not completed their work. Does not need to be submitted until after the first certified payroll report is received.
- i One form may be submitted for consecutive non-performing weeks.

#### 4. Certified Payroll Reporting Form

- i Any person employed upon the project that is working with tools must be listed on the certified payroll including but not limited to owners, operators, surveyors, and foremen.
- i The certified payroll records shall be submitted and maintained electronically subject to the following conditions:
  - i. The certified payroll reports contain all of the information required by California Labor Code Section 1776. The information must include name, address, social security number, craft, classification, wages, and hours worked.
  - ii. The reports shall be in a format and/or use software that is readily accessible to Contractors, Awarding Bodies, LCPs, the DIR, and the DOL.
- i Certified Payroll submitted to DSI, the DLSE, or another entity within the DIR must be in the form of a non-modifiable image or record that bears an electronic signature or includes a copy of any original certification made on paper.
- i The requirements for redacting information shall be followed when certified payroll records are disclosed to the public pursuant to California Labor Code Section 1776(e). This requirement will apply whether the records are provided electronically or as hard copies.
- i No contractor or subcontractor shall be mandated to submit or receive electronic reports when it otherwise lacks the resources or capacity to do so, nor shall any contractor or subcontractor be required to purchase or use proprietary software that is not generally available to the public.

#### 5. Statement of Compliance Certificate

A Statement of Compliance shall accompany each certified payroll record.

Verification of Apprenticeship Status (DAS)
 Verification is available on the DIR website at <a href="http://www.dir.ca.gov/DAS/appcertpw/AppCertSearch.asp">http://www.dir.ca.gov/DAS/appcertpw/AppCertSearch.asp</a>.

#### **Project/Program Closeout**

#### Contractor Affidavit

Verifies the contractor's information, work classifications used, type of work completed, first payroll report date to final payroll report date, and how the apprenticeship utilization requirement was reached.

#### **Apprenticeship Utilization**

California Labor Codes require Contractors to hire apprentices unless the total construction contract for the project is less than \$30,000 or it is not an apprenticeable craft.

Contractors, including Design-Build, General or Specialty subcontractors shall employ registered apprentices during the performance of public works in accordance with the required one (1) hour of work performed by an apprentice for every five (5) hours of work performed by a journeyman. Unless an exemption has been granted, the contractor shall employ apprentices for the number computed above, before the end of the contract or provide good faith effort documentation.



# Section 2: Prior to Construction Forms

- A. Checklist of Labor Law Requirements
- B. Public Works Contract Award Information (DAS 140 Form)
- C. Request for Dispatch of an Apprentice (DAS 142 Form)

#### **Checklist of Labor Law Requirements**

(CCR Title 8, Section 16421)

Ultimately the prime contractor is liable for their sub and specialty contractors. This checklist is a useful tool for the prime contractor to ensure that their sub and specialty contractors know their responsibilities on public works projects. Contractors who understand and comply with the law are more likely to deliver the job on time, on budget and done right the first time. We suggest the prime contractor encourage completion of this checklist by their sub and specialty contractors.

NAME (PRINT)		DATE	
COMPANY		PHONE	
ADDRESS		FAX	
CITY		State	ZIP CODE
Project manager		Superintendent/Foreman	
CERTIFIED PAYROLL		PHONE/EX	Г
CONTRACTOR LICENSE NO	EXP. DATE	SPECIALTY LICENSE N	NO
SELF-INSURED CERTIFICATE NO.			
Project name			
AWARDING BODY			
IF SUB-CONTRACTING, LIST YOUR PRIME/GENERAL CONTRACTOR _			
THE FEDERAL AND STATE LABOR LAW REQUIREM	ENTS APPLICA	ABLE TO THE CONTRACT ARE (	COMPOSED OF, BUT NOT

LIMITED TO, THE FOLLOWING:

#### □ Payment of Prevailing Wage Rates

The contractor to whom the contract is awarded and its subcontractors hired for the public works project are required to pay not less than the specified general prevailing wage rates to all workers employed in the execution of the contract. Labor Code Section 1770 et seq.

The contractor is responsible for ascertaining and complying with all current general prevailing wage rates for crafts and any rate changes that occur during the life of the contract. Information on all prevailing wage rates and all rate changes are to be posted at the job site for all workers to view. Additionally, current wage rate information can be found at the DLSR web site, www.dir.ca.gov/dlsr/statistics\_research.html.

#### □ Apprentices

It is the duty of the contractor and subcontractors to employ registered apprentices on the public works project and to comply with all aspects of Labor Code Section 1777.5, relating to Apprentices on Public Works. (1) Notify approved apprenticeship programs of contract award; (2) employ apprentices; (3) pay training fund contributions.

#### □ Penalties

There are penalties required for contractor's/subcontractor's failure to pay prevailing wages and for failure to employ apprentices, including forfeitures and debarment under Labor Code Sections 1775; 1776; 1777.1; 1777.7 and 1813.

#### ☐ Certified Payroll Reports

Under Labor Code Section 1776, contractors and subcontractors are required to keep accurate payroll records showing the name, address, social security number and work classification of each employee and owner performing work; also the straight time and overtime hours worked each day for each week, the fringe benefits, and, the actual per diem wage paid to each owner, journey person, apprentice worker or other employee hired in connection with the public works project.

This requirement includes and applies to all subcontractors performing work on Awarding Body projects even if their portion of the work is less than one half of one percent (0.05%) of the total amount of the contract.

The certified payroll records shall contain the same data fields listed on the Public Works Payroll Reporting Form (A-1-131) and contain or is accompanied by a declaration made under penalty of perjury. (California Code of Regulations, Section 16401).

Prime Contractors are responsible for submittal of their payrolls and those of their respective subcontractors as one package. Any payroll not submitted in the proper form will be rejected. In the event that there has been no work performed during a

#### **Checklist of Labor Law Requirements, continued**

given week, the Certified Payroll Report shall be annotated: "No work" for that week or a Non-Performance Statement must be submitted.

Employee payroll records shall be certified and shall be made available for inspection at all reasonable hours at the principal office of the contractor/subcontractor, or shall be furnished to any employee, or his/her authorized representative on request, pursuant to Labor Code Section 1776.

	Under <i>Labor Code Section 1776(g)</i> there are penalties required for contractor's/subcontractor's failure to maintain and submit copies of certified payroll records on request.
	Nondiscrimination in Employment
	There exist prohibitions against employment discrimination under <i>Labor Code Sections 1735 and 1777.6,</i> the <i>Government Code,</i> the <i>Public Contracts Code,</i> and <i>Title VII of the Civil Rights Act of 1964.</i>
	Kickbacks Prohibited
	Contractors and subcontractors are prohibited from recapturing wages illegally by accepting or extracting "kickbacks" from employee wages under <i>Labor Code Section 1778</i> .
	Acceptance of Fees Prohibited
	There exists a prohibition against contractor/subcontractor acceptance of fees for registering any person for public work under <i>Labor Code Section 1779</i> ; or for filling work orders on public works contracts pursuant to <i>Labor Code Section 1780</i> .
	Listing of Subcontractors
	All prime contractors are required to list properly all subcontractors hired to perform work on the public works projects covering more than one-half of one percent, pursuant to <i>Government Code Section 4104</i> .
	Proper Licensing
	Contractors are required to be licensed properly and to require that all subcontractors be properly licensed. Penalties are required for employing workers while unlicensed under <i>Labor Code Section 1021</i> and under the California Contractor License Law found at <i>Business and Professions Code Section 7000 et seq</i> .
	Unfair Competition Prohibited
	Contractors and sub-contractors are prohibited from engaging in unfair competition as specified under <i>Business and Professions Code Sections 17200 to 17208.</i>
	Workers Compensation Insurance
	Labor Code Section 1861 requires that contractors and subcontractors be insured properly for Workers Compensation.
	OSHA
	Contractors and subcontractors are required to abide by the Occupational, Safety and Health laws and regulations that apply to the particular construction project.
	Proof of Eligibility/Citizenship
	The federal prohibition against hiring undocumented workers, and the requirement to secure proof of eligibility/citizenship from all workers, is required.
	Itemized Wage Statement
	Labor Code Section 226 requires that employees be provided with itemized wage statements.
CE	RTIFICATION
Lac	knowledge that I have been informed and am aware of the foregoing requirements and that I am authorized to make this
cert	tification on behalf of(COMPANY NAME)
l fui	lly understand that failure to comply with any of the above requirements may subject me, or my company, to penalties as
	vided above.

(SIGNATURE)

(SIGNATURE)

Awarding Agency /Labor Compliance Program

(DATE)

(DATE)

#### **PUBLIC WORKS CONTRACT AWARD INFORMATION**

Contract award information must be sent to your Apprenticeship Committee if you are approved to train. If you are not approved to train, you must send the information (which may be this form) to ALL applicable Apprenticeship Committees in your craft or trade in the area of the site of the public work. Go to: http://www.dir.ca.gov/das/PublicWorksForms.htmfor information about programs in your area and trade. You may also consult your local Division of Apprenticeship Standards (DAS) office whose telephone number may be found in your local directory under California, State of, Industrial Relations, Division of Apprenticeship Standards.

#### Do not send this form to the Division of Apprenticeship Standards.

	Do not send this form to the Division (	or whhi curing suith standards.
NAME OF	YOUR COMPANY	CONTRACTOR'S STATE LICENSE NO
MAILING ADI	DRESS- NUMBER & STREET, CITY, ZIP CODE	AREA CODE & TELEPHONE NO.
DDA & AME	DRESS OF PUBLIC WORKS PROJECT	DATE YOUR CONTRACT EXECUTED
		DATE OF EXPECTED OR ACTUAL START OF PROJECT
DDA & AMAN	DRESS OF PUBLIC AGENCY AWARDING CONTRACT	ESTIMATED NUMBER OF JOURNEYMEN HOURS
		OCCUPATION OF APPRENTICE
THIS FORM	M IS BEING SENT TO: (NAME & ADDRESS OF APPRENTICESHIP PROGRAM(S))	ESTIMATED NUMBER OF APPRENTICE HOURS
		APPROXIMATE DATES TO BE EMPLOYED
Contr  1  2	This is not a request for disperactors must make a separate request for actual dispatch, in accordance Check One Of The Bowler and train apprentices by the Apprenticeship Committee. We will employ and train under the world will comply with the standards of	dance with Section 230.1(a) California Code of Regulations  OXES Below
<sup>2.</sup> []	Apprenticeship Committee for the duration of this job or	nly. Enter name of the Committee
3.	We will employ and train apprentices in accordance wit including § 230.1 (c) which requires that apprentices en perform work of the craft or trade to which the apprentic times work with or under the direct supervision of journ	nployed on public projects can only be assigned to ce is registered and that the apprentices must at all
	Signature	Date
	Typed Name	
	Title	

State of California - Department of Industrial Relations DIVISION OF APPRENTICESHIP STANDARDS



### REQUEST FOR DISPATCH OF AN APPRENTICE – DAS 142 FORM DO NOT SEND THIS FORM TO DAS

You may use this form to request dispatch of an apprentice from the Apprenticeship Committee in the craft or trade in the area of the public work. Go to: <a href="http://www.dir.ca.gov/databases/das/pwaddrstart.asp">http://www.dir.ca.gov/databases/das/pwaddrstart.asp</a> for information about programs in your area and trade. You may also consult your local Division Apprenticeship Standards (DAS) office whose telephone number may be found in your local directory under California, State of, Industrial Relations, Division of Apprenticeship Standards. <a href="mailto:Except for projects with less than 40 hours of journeyman work">Except for projects with less than 40 hours of journeyman work</a>, you must request and employ apprentices in no less than 8 hour increments.

Date:	Contractor Requesting Dispatch:
To Applicable Apprenticeship Committee:	Name:
Name:	Address:
Address:	
	License No
Tel. No Fax No	Tel. No Fax No
Project Information:	
Contract No	
Name of the Project:	
Address:	
Dispatch Request Information:	
Number of Apprentice(s) Needed: Craft	or Trade:
Date Apprentice(s) to Report: (72 hrs. notic	e required) Time to Report:
Name of Person to Report to:	
Address to Report to:	

You may use this form to make your written request for the dispatch of an apprentice. Requests for dispatch must be in writing and submitted at least 72 hours in advance (excluding weekends and holidays) via first class mail, fax or email. **Proof** of submission may be required. Please take note of California Code of Regulations, Title 8, § 230.1 (a) for all applicable requirements regarding apprenticeship requests and/or visit

http://www.dir.ca.gov/DAS/DASApprenticesOnPublicWorksSummaryOfRequirements.htm

DAS 142 (Revised 04/14)



# Section 3: During Construction Submittals

- A. Statement of Employer Payments
- B. CAC Training Fund Contributions (CAC-2 Form)
- C. Statement of Non-Performance
- D. Public Works Payroll Reporting From (A-1-131 Form)
- E. Verification of Apprenticeship Status



#### **Statement of Employer Payments**

Date:		In Reply, Refer to Case No	):	SEAL OF THE
Prime:				
Subcontractor:				
PROJECT NAME:				
PROJECT CONTRACT NO.:		County/location:		CALIFORNIA
	HEALT	TH AND WELFARE		
NAME OF PLAN		Address, City and Zip		
ADMINISTRATOR		Address, City and Zip		
CLASSIFICATION(S) USED		CONTRIBUTION PER CL	ASSIFICATION PER I	HOUR
CONTRIBUTIONS:	WEEKLY_	MONTHLY	QUARTERLY	ANNUALLY
		PENSION		
NAME OF PLAN		Address, City and Zip		
ADMINISTRATOR		Address, City and Zip		
CLASSIFICATION(S) USED		CONTRIBUTION PER CL	ASSIFICATION PER I	HOUR
CONTRIBUTIONS:	WEEKLY_	MONTHLY	QUARTERLY_	ANNUALLY
		VACATION/HOLIDAY		
NAME OF PLAN		Address, City and Zip		
ADMINISTRATOR		Address, City and Zip		
CLASSIFICATION(S) USED		CONTRIBUTION PER CL	ASSIFICATION PER F	HOUR
CONTRIBUTIONS:	WEEKLY_	MONTHLY	QUARTERLY	ANNUALLY
		TRAINING		
NAME OF PLAN		Address, City and Zip		
ADMINISTRATOR		Address, City and Zip		
CLASSIFICATION(S) USED		CONTRIBUTION PER CL	ASSIFICATION PER I	HOUR
CONTRIBUTIONS:	WEEKLY	MONTHLY	OUARTERLY	ANNUALLY

IF YOU USE OTHER PLANS NOT LISTED ABOVE, YOU MAY USE THE BACK OF THIS FORM TO PROVIDE THIS ADDITIONAL INFORMATION



CA.gov | Contact DIR | Press Room

Go to Search

Home Labor Law Cal/OSHA - Safety & Health Workers' Comp Self Insurance Apprenticeship Director's Office Boards

Division of Apprenticeship Standards (DAS)

#### CAC - Training Fund Contributions

You must enter all requested infomation in order to ensure successful submission and processing of your payment. Training Fund Contributions are due on the 15th of each month.

All fields with \* are required.

You must use the **BUTTON** on the bottom of the page to submit for an invoice coupon.

TO NAVIGATE BETWEEN FIELDS, DO NOT HIT RETURN OR ENTER KEY AFTER EACH ENTRY. USE THE TAB KEY INSTEAD.

You need to have a working printer currently connected to your computer in order to print the complete paper form in the end of this session so that you can mail it with your payment.

#### Training Fund Contributions Form CAC2

Date: 9/25/2014

#### **CLEAR FORM**

Contractor/Sub Contractor making contributions	Contractor	Period covered by contribution (from –to)	Jobsite Location (including County)
* Name:	* License Number:	* Period Start:	If applicable, give name of school, hospital, building, etc.
* Address:	* Contract/Project Number	* Period End:	Comments:
* City:		(MM/DD/YYYY)	
* State: * ZIP:			

* Name of the submitting party:	* Submitter's Title:	* Submitter's Email:	'Submitter's Phone: e.g., (999) 999-9999

Instructions: You may want to use the keyboard TAB key to navigate the fields and the Up  $\land$  | Down  $\lor$  ARROW keys to select a list item.

	*County of Work	* Classification 1	' Hours (min.: 0.5; max: 9 999.99)	' Rate (min.: 0.01; max: \$9.99)	Amount
*1)	Select a county	Select an occupation	**	\$ **	\$ 0.00
2)	Select a county	Select an occupation		\$	\$ 0.00
3)	Select a county	Select an occupation		\$	\$ 0.00
4)	Select a county	Select an occupation		\$	\$ 0.00
5)	Select a county	Select an occupation	-	\$	\$ 0.00
6)	Select a county	Select an occupation		\$	\$ 0.00
7)	Select a county	Select an occupation		\$	\$ 0.00
8)	Select a county	Select an occupation		\$	\$ 0.00

9)	Select a county	Select an occupation	\$	\$ 0.00
10)	Select a county	Select an occupation	\$	\$ 0.00
11)	Select a county	Select an occupation	\$	\$ 0.00
12)	Select a county	Select an occupation	\$	\$ 0.00
13)	Select a county	Select an occupation	\$	\$ 0.00
14)	Select a county	Select an occupation	\$	\$ 0.00
15)	Select a county	Select an occupation	\$	\$ 0.00
16)	Select a county	Select an occupation	\$	\$ 0.00
17)	Select a county	Select an occupation	\$	\$ 0.00
18)	Select a county	Select an occupation	\$	\$ 0.00
19)	Select a county	Select an occupation	\$	\$ 0.00
20)	Select a county	Select an occupation	\$	\$ 0.00

Footnote 1 – If you are unable to locate the occupation in the pull down menu, please click on this link: http://www.dir.ca.gov/databases/das/aigstart.asp for specific information assistance.

**TOTAL AMOUNT: \$ 0.00** 

When done with some or all the entries above, <u>please carefully review</u> and then enter the green code you see below:

22749

Calculate Total Amount

August 2014

About DIR

Who we are DIR Divisions, Boards & Commissions

Contact DIR

Work with Us

Licensing, registrations, certifications & permits
Notification of activies
Public Records Act

Learn Mo

Site Map Frequently Asked Questions

Jobs at DIR

Conditions of Use | Privacy Policy | Disclaimer | Disability accommodation | Site Help

Copyright © 2014 State of California

#### TO BE TYPED ON COMPANY LETTERHEAD

#### STATEMENT OF NON-PERFORMANCE

Payroll #	Date
I do hereby state that no	persons employed on the construction of the
	Project, for
(Project Name)	Project, for (Awarding Body)
Company, Contract No	during the payroll period commencing on the
th day of	, 2007 and ending on theth day of
, 2007.	
(Company Name)	
(Authorized Signer)	

# California Department of Industrial Relations

# PUBLIC WORKS PAYROLL REPORTING FORM

	OR SUI	OR SUBCONTRACTOR:	JR:			S	SPECIALITY LICENSE NO.:	CONTRACTOR'S LICENSE NO.: SPECIALITY LICENSE NO.:			ADDRESS:	ESS:				
	PAYROLL NO.:	, NO.:		FOR WEEK ENDI	OING:		SELF-INSU	SELF-INSURED CERTIFICATE NO.:	TE NO.:			PROJECT	PROJECT OR CONTRACT NO.:	T NO.:		
	L_		(4)	DAY	(5)	(9)	WORKERS	WORKER'S COMPENSATION POLICY NO.:	ON POLICY NO	9		PROJECT	PROJECT AND LOCATION:	:NC		
(1)	(2) (3)	(1	M	W TH F S	S			(7)				(8)				(6)
NAME, ADDRESS AND		RK		TE		HOURLY AL RATE		3 AMOUNT							NET WGS	
SOCIAL SECURITY NUMBER OF EMPLOYEE	F WITH CLASSIFICATION	ICATION			HOURS			EARNED		DEDUCT	DEDUCTIONS, CONTRIBUTIONS AND PAYMENTS	BUTIONS AN	ID PAYMENT	iS.	PAID FOR WEEK	CN
	EXEM HOFI NO' O		нот	HOURS WORKED EACH DAY	Ā											
							THIS	ALL PROJECTS	FED. TAX	FICA (SOC. SEC.)	STATE	SDI HC	VAC/ HEA HOLIDAY & W	HEALTH PENSION & WELF.	7	
	-		ν.													
									TRAING.	FUND	DUES S	TRAV/ SUBS. SA	SAVINGS OTF	OTHER* DEDUC-		
	· <b>-</b> -		0												<u> </u>	
							THIS	ALL PROJECTS	FED. TAX	FICA (SOC. SEC.)	STATE	SDI HC	VAC/ HEA	HEALTH PENSION & WELF.	7	
			ν.													
		I					I		TRAING.	FUND ADMIN	DUES S	TRAV/ SA	SAVINGS OTF	OTHER* DEDUC-		
			0												Ī	
							THIS PROJECT	ALL PROJECTS	FED. TAX	FICA (SOC. SEC.)	STATE	SDI HC	VAC/ HEA	HEALTH PENSION & WELF.	7	
			S				Ī									
									TRAING.	FUND	DUES S	TRAV/ SUBS. SA	SAVINGS OTF	OTHER* TOTAL DEDUC-TIONS		
			0													
							THIS	ALL PROJECTS	FED. TAX	FICA (SOC. SEC.)	STATE	SDI HC	VAC/ HEA HOLIDAY & W	HEALTH RENSION & WELF.	7	
			ς.													
		1							TRAING.	FUND ADMIN	DUES S	TRAV/ SUBS. SA'	SAVINGS OTF	OTHER* TOTAL DEDUCTIONS		
			0													

#### NOTICE TO PUBLIC ENTITY

#### For Privacy Considerations

Fold back along dotted line prior to copying for release to general public (private persons).

(Paper Size then 8-1/2 x 11 inches)
I the undersigned am the
I,, the undersigned, am the (Name – print)
with the authority to act for and on behalf of
(Position in business)
(Name of business and/or contractor) , certify under penalty of perjury
that the records or copies thereof submitted and consisting of
are the originals or true, full, and correct copies of the originals which depict the payroll record(s
of the actual disbursements by way of cash, check, or whatever form to the individual or
individuals named.
Date: Signature:

A public entity may require a stricter and/or more extensive form of certification.

CA.gov | Contact DIR | Press Room This Site California Cal/OSHA - Safety & Health Workers' Comp Self Insurance Apprenticeship Director's Office Boards Labor Law Division of Apprenticeship Standards (DAS) Apprenticeship certification for public works Become an apprentice Find an apprenticeship program Enter search string ( LLLLF9999 ) here Search Find a registered apprentice Train employees through How to compile the search string: apprenticeship The search string is a total of nine letters and numbers (no characters ', - , etc): the first four letters of the last name (use spaces to make four letters if the last name is shorter than four letters), the first letter of the Program sponsors first name and the last four digits of the social security number (LLLLF9999). Letters can be entered as Use apprentices on public works lower or upper case. projects Examples: Uncle Sam ssn 123-45-6789 would be entered as Sam U6789 Apprenticeship Council meetings Goddess Minerva ssn 123-45-5555 would be entered as MineG5555 **DIR Laws and Regulations** Richard Al-Ham ssn 111-44-1111 would be entered as AlhaR1111 Veterans Robert O'Brian ssn 111-22-3333 would be entered as OBriR3333 James McHenry ssn 555-66-1234 might be entered as McHeJ1234 or Mc HJ1234 If you cannot find the individual you are looking for, complete the certifications you have and see the About Us (Overview of DAS) notes below. Locations If a search string that was entered does not match with any apprentices in the Division of Apprenticeship **DAS Home** Standards (DAS) database, this could be due to any of the following:

1. Not a registered apprentice.

Who we are

Contact DIR

DIR Divisions, Boards & Commissions

- The submitted search string does not match to DAS records (either the database has the wrong search criteria or you have the wrong search criteria).
- The apprentice agreement has not been submitted to DAS or has not been entered into the database (agreements must be submitted within 30 days of the date that the apprentice signed the agreement).

If you believe that the apprentice should be reported as registered and is not; please contact your local office of the Division of Apprenticeship Standards.

ork with Us Learn Mor

Licensing, registrations, certifications & permits

Site Map

Notification of activies

Frequently Asked Questions

Public Records Act

Jobs at DIR

Conditions of Use | Privacy Policy | Disclaimer | Disability accommodation | Site Help

Copyright © 2014 State of California



# Section 4: Closeout

# This form must be submitted to DSI during project/program closeout

A. Contractor Affidavit



## Contractor Affidavit Contract # \_\_\_\_\_

	1.	I am the	(owner, officer, partner) of		
			(Company) who performed work on the		
			(Project) in the classification (s) of		
		The labor performed by these workers can best be described by			
	2.	ending on _ this project	payroll periods commencing on and all persons employed by my company on have been paid the specified prevailing rate of per diem wages for d craft or classification pursuant to Labor Code Section 1771 <sup>1</sup> .		
	3.	3. The apprenticeship committee (s) either denied or failed to respond to a request for the dispatch of apprentices, and therefore all workers were classified as journeyman			
Or					
	4.	journeyman	(s) worked a total of hours and worked a total of hours establishing an apprentice \ ratio in hours of to		
Or					
	5.		were employed in accordance with the DAS exemption that apprentice for every five journeyman employed on each day of		
Ex	ecu	ted this	day of, california.		
Sig	natu	ire			
0					

 $<sup>^1</sup>$  Except for public works project of one thousand dollars (\$1000) or less, not less than the general prevailing rate of per diem wages for work of a similar character in the locality in which the public work is performed, and not less than the general prevailing rate of per diem wages for holiday and overtime work fixed as provided in this chapter, shall be paid to all workers employed on public works.

# PROJECT STABILIZATION/COMMUNITY BENEFITS AGREEMENT for the COUNTY OF ALAMEDA

## Table of Contents

PREAMBLE	3
DEFINITIONS	4
PURPOSE	
SCOPE OF AGREEMENT	7
RELATIONSHIP BETWEEN PARTIES	10
ROLES AND RESPONSIBILITIES	10
SUBCONTRACTS	10
WORK ASSIGNMENTS AND JURISDICTIONAL DISPUTES	12
PRE-JOB CONFERENCE	13
JOINT ADMINISTRATNE COMMITTEE MEETINGS	
COORDINATOR	15
UNION RECOGNITION AND REPRESENTATION	15
NO STRIKES-NO LOCKOUTS	16
MANAGEMENT RIGHTS	19
WORK RULES	
WAGE SCALES AND FRINGE BENEFITS	20
HOURS OF WORK., OVERTIME, SHIFT AND HOLIDAYS	21
HEALTH AND SAFETY	21
LOCAL HIRING PROGRAM	21
APPRENTICES	
REFERRAL PROCESS	
REPORTING AND DATA COLLECTION	
HELMETS TO HARDHATS: VETERAN EMPLOYMENT	28
NON-DISCRIMINATION	29
GRIEVANCE PROCEDURE	
MISCELLANEOUS PROVISIONS	32
ENTIRE AGREEMENT	32
GENERAL SAVINGS CLAUSE	
DURATION OF AGREEMENT	33

## **PREAMBLE**

This Agreement is made and entered into on this day of <u>September 18. 2018</u>, by and between the County of Alameda ("County") together with Contractors and/or subcontractors, who shall subsequently become signatory to this Agreement by signing the "Contractor Agreement To Be Bound" (Exhibit A), ("Contractors"), the Building and Construction Trades Council of Alameda County, AFL-CIO ("Council") and the Local Unions signatory hereto, all in their behalf and in behalf of the various Local Unions involved, ("Union(s)") for the construction of all Covered Projects ("Covered Projects").

## Recitals

WHEREAS, the Projects described in this Agreement have been identified by the County as those in which a Project Stabilization/Community Benefits Agreement would benefit the County; and

WHEREAS, the Contractors will be engaged in construction of the project; and

WHEREAS, a skilled labor pool represented by Building Trades Unions will be required to complete the work involved; and

WHEREAS, the Building Trades Unions agree to cooperate in every way possible with employees of the Contractors; and

WHEREAS, the parties to this Agreement mutually agree that safety, quality, productivity and labor harmony are primary goals; and

WHEREAS, the County desires to provide, enhance and encourage construction training and employment opportunities for Alameda County residents and small business enterprises within the County through apprentice and pre-apprentice programs; and

WHEREAS, the County also desires to use this Agreement as a vehicle for building the capacity of Alameda County residents and businesses and to maximize their potential to successfully participate in other large-scale projects; and

WHEREAS, the parties recognize the need for safe, efficient and speedy construction in order to reduce unnecessary delays and result in timely completion of the project; and

WHEREAS, the parties desire to mutually establish and stabilize wages, hours and working conditions for the employees employed on the project by the Contractors, and further to encourage close cooperation to achieve a satisfactory, continuous and harmonious relationship between the parties to this Agreement;

WHEREAS, the County of Alameda's mission is to enrich the lives of all residents through visionary policies and accessible, responsible and effective services and historically the County has supported contracting outreach programs that recognize the economic and workforce development potential of capital construction projects on government owned facilities; and

WHEREAS, the Parties recognize that disadvantaged individuals, families, and communities within the county experience high unemployment and are also often recipients of County services, and that these disadvantaged populations may economically benefit through participation in local hire, apprenticeship and pre-apprenticeship programs; and

WHEREAS, the Union(s), Contractors, subcontractors, and the County wish to insure labor peace at the Covered Project sites devoid of any disruption that could jeopardize the schedule and timeliness of the construction process, where both Contractors that are signatory to collective bargaining agreements of the Union(s) are supervising employees that are members of the Union(s) and where Contractors that are not Signatory to collective bargaining agreements are supervising employees;

NOW THEREFORE, the parties, in consideration of the mutual promises and covenants herein contained, mutually agree as follows:

#### ARTICLE 1

## **DEFINITIONS**

- 1. 1 For purposes of this Agreement, the following terms will have the following meanings:
- "Acceptance" shall mean action by the County notifying Contractor and other entities of Completion, as required by and in accordance with contract terms and relevant applicable statutes.
- "Agreement" shall mean this Project Stabilization/Community Benefit Agreement.
- "Alternative Employees" shall mean an employee whose services have been obtained from other than the Union referral facilities as permitted in Section 19.6 of this Agreement.
- "Apprentice" shall mean a person enrolled in a State approved apprenticeship training program administered by a Joint Labor-Management Apprenticeship Training Committee (JATC).
- "Completion" means that the work of Contractors' is completed, as follows:
  - The occupation, beneficial use, and enjoyment of a work of improvement, excluding any operation only for testing, startup, or commissioning, by the public agency, or its agent, accompanied by a cessation of labor on the work of improvement.
  - 2. The acceptance by the public agency, or its agent, of the work of improvement.

"Contractors" means all contractors and subcontractors at all tiers, any individual, firm, Partnership or corporation, or combination thereof; including joint ventures, which is an independent business enterprise and has entered into a contract with the County or any of its contractors or subcontractors at any tier, with respect to the construction work covered by this Agreement and necessary for the project or any part thereof, including construction building material delivery (if the material is for direct incorporation) and removal truckers, trucking companies and trucking brokers, including the operating of construction equipment,

performance of labor and/or installation of materials.

"Coordinator" shall mean the company or individual designated or retained by the County to administer this Agreement.

"Core Employee" shall mean an individual meeting the criteria listed in Section 19.1.1-19.1.5.

"Council" shall mean the Building and Construction Trades Council of Alameda County.

"County" shall mean the County of Alameda acting by and through its Board of Supervisors, Agency and Department heads and administrative staff.

"Covered Projects" and "Projects" means projects covered by the Agreement.

"Covered Work" means work done on the project and subject to the provisions of this Agreement.

"Disadvantaged Population" shall mean those Residents of Alameda County who meet at least one of the following criteria: household income below 50% of the Alameda County median, non-minor dependent youth (AB-12 youth-emancipated foster youth), homeless, and welfare recipients, have a history of involvement with the criminal justice system, are unemployed, or a single parent.

"Emergency Work" shall mean those projects undertaken when an immediate or imminent critical impact to a facility or to the ability to provide essential services is likely within thirty (30) days should no further action be taken, or in circumstances where mandatory environmental, health and/or safety requirements will be violated without said project.

"General Prevailing Wage Determination" shall mean the decisions made by the Director of the California Department of Industrial Relations (DIR) establishing a journeyman craft or classification's prevailing wage determination, holiday, advisory scope of work, or travel and subsistence provision.

"Local Hiring Goals" shall mean the Resident and Apprentice hiring goals set forth in Article 17 and Article 18 of this Agreement.

"Local Hiring Program" shall mean the program set forth in Article 17 and Article 18 of this Agreement intended to achieve the inclusion of County Residents in the employment and apprenticeship opportunities created by the Covered Work.

"Master Labor Agreement" or "MLA" shall mean the collective bargaining agreement of each craft Union that is Signatory to this Agreement.

"New Apprentice" shall mean an Alameda County Resident who on the date that such individual is hired or assigned to perform the applicable work, is newly enrolled (less than one year) in a labor-management apprenticeship program that is currently registered with the State of California's Division of Apprenticeship Standards.

"Owner Operator" shall mean a sole individual that owns and drives/operates a maximum of one unit and who is employed in the movement or transportation of materials or goods of another. The owner operator shall be carried on the payroll of the entity that employs or otherwise uses the Owner/Operator. The Owner/Operator shall direct a maximum of one unit which he or she shall drive themselves. In addition, the owner operator must provide documentation of insurance, a business license, and a valid motor carrier permit issued solely in their name.

"Post Disaster Work" shall mean County approved construction projects consistent with Post Disaster response and recovery efforts per the California Government Code Section 20168 where the public interest and necessity demand immediate expenditure of public funds to safeguard life, health, or property following a local state or federally declared disaster per the Stafford Act.

"Project Manager" shall mean the person or persons designated by the County of Alameda Board of Supervisors to act on behalf of the County in all matters involving or related to individual Covered Projects.

"Resident" shall mean an individual who has lived or resided in Alameda County for a period of not less than thirty (30) calendar days prior to the date of dispatch/referral of that individual by the Union to a Contractor performing work on the project or for a period of not less than thirty (30) calendar days prior to applying for work or inclusion in the Local Hire Program if the individual is an Alternative Employee, a Core Employee, a member of a Disadvantaged Population, or a Local Hire Program applicant.

"Signatory" shall mean those Unions who have through their officers and or agents executed this Agreement.

"Sole Proprietor" shall mean an owner who will self-perform the designated Covered Project Work without hiring field support staff for the Project.

"Trust Agreements" shall mean the agreements between Unions and employers and or employer associations to govern trust funds contributed on behalf of covered workers for benefits for said workers.

"Union" or "Unions" shall mean the Building and Construction Trades Council of Alameda County and its affiliated local unions Signatory to the Agreement, acting on their own behalf or on behalf of their respective affiliates and member organizations.

## **PURPOSE**

- 2.1 The purposes of this Agreement are to promote efficient construction operations on the Projects, to insure an adequate supply of skilled craftspeople and to provide for peaceful efficient and binding procedures for settling labor disputes. In so doing, the parties to this Agreement establish the foundation to promote the public interest, to provide a safe work place, to assure high quality construction, to ensure uninterrupted construction Projects, and to secure optimum productivity, on-schedule performance and County satisfaction.
- 2.2 It is the intent of the parties to set out uniform and fair working conditions for the efficient completion of the Projects, maintain harmonious labor/management relations and eliminate strikes, lockouts and other delays.
- 2.3 The parties agree that one of the primary purposes of this Agreement is to avoid the tensions that might arise on the Projects if union and nonunion workers of different employers were to work side by side on the Projects thereby leading to labor disputes that could delay completion of the Projects.
- 2.4 This Agreement is entered into pursuant to and consistent with California Public Contract Code ("PCC") Sections 2500 through 2502. PCC Section 2500(a)(3) requires a public entity PLA to include an agreed-upon protocol concerning drug testing for workers employed on the Projects, as set forth in Article 16.3.

#### **ARTICLE 3**

## SCOPE OF AGREEMENT

- 3.1 The parties agree that this Agreement will cover all projects undertaken by the County of Alameda with a construction value of \$1 million or more. In addition, the Agreement will cover all projects with a construction value of \$1 million or more which are undertaken on behalf of the County or in circumstances where County is executing projects for Special Districts. The parties further agree that the Board of Supervisors may at their discretion elect to include any project with a value less than \$1 million under the terms of this Agreement.
- 3.2 This Agreement covers all on-site construction, fabrication, demolition, alteration, painting or repair of buildings, structures, landscaping, temporary fencing and other work and related activities that are within the craft jurisdiction of one of the Union(s) and that is part of the work, including site preparation, survey work, and all construction, demolition or improvements required to be performed as a condition of approval by the County.
- 3.3 This Agreement shall apply only to construction/craft employees, performing work on projects represented by the Signatory Unions, and shall not apply to Contractors' supervisors, technical or non-manual employees including, but not limited to, executives, engineers, office and clerical employees, drafters, architects, supervisors, timekeepers, messengers, guards, other employees above the classification of general foreman, inspectors, material testers, and/or x-ray technicians, except to the extent that such inspectors, material testers, and/or x-ray technicians are customarily covered by the

Page 7 of 40 County of Alameda

Countywide Project Stabilization/ Community Benefits Agreement

MLA and as to which classification prevailing wage determination has been published.

- 3.4 There shall be no limitation or restriction upon the choice of materials or upon the full use and installation of equipment, machinery, package units, factory pre-cast, prefabricated or preassembled materials, tools or other labor-saving devices. The lawful fabrication provisions of the appropriate national or local agreements shall be applicable. The covered projects include work necessary for the covered projects and/or in temporary yards or areas adjacent to and dedicated to the covered projects, and at any batch plant(s) constructed or used solely to supply materials to the Covered Projects, when those sites or processes are dedicated exclusively to the covered projects.
- 3.5 This Agreement covers all on-site fabrication work over which the County or Contractors possess the right of control (including work done for the covered projects in any temporary yard or area established for the Covered Projects). Additionally, any offset work, including fabrication, necessary for the Covered Projects defined herein, that is lawfully covered by a current MLA or local addenda to a National Agreement of the applicable Union(s) that is in effect as of the execution of this Agreement shall be considered covered work under this Agreement.
- 3.6 This Agreement shall apply to any start-up, calibration, performance testing, repair, maintenance, operational revisions to systems and/or subsystems performed up to nine (9) months after Completion by the Contractors. It is understood the County reserves the right to perform any start-up, operation. repair, maintenance or revision of equipment or systems with employees of the County. If required, Contractor's personnel may make a final check and may direct their staff on site to make any necessary repairs to protect the terms of a manufacturer's guarantee or warranty of a piece of equipment.
- 3.7 The on-site installation or application of all items shall be performed by the craft having jurisdiction over such work as set forth under the provisions of this Agreement; provided, however, it is recognized that installation of specialty items which may be furnished by the County or a Contractor shall be performed by construction persons of the vendor or other companies where necessary to protect a manufacturer's warranty. The issue of whether it is necessary to use construction persons of the vendor or other companies to protect the manufacturer's warranty shall be subject to the grievance and arbitration clause of this Agreement.
- 3.8 It is recognized by the parties to this Agreement that the Coordinator designated in Article 9 below, and Contractors are acting only on behalf of said Coordinator and Contractors, and said Coordinator and Contractors have no authority, either expressed, implied, actual, apparent or ostensible, to speak for or bind the County.
- 3.9 It is expressly agreed and understood that the County retains the right and ability to meet all competitive bidding requirements of public contracting law and to select the lowest responsive and responsible bidder who provides the County with best value within a stipulated sum regardless of union signatory status. Further, the County may, at its sole discretion, end, delay, and/or suspend any or all portions of the work and may combine, consolidate, modify and/or not build any one or more portions of work covered by this Agreement at any time

- 3.10 It is expressly agreed and understood by the parties hereto that the County shall retain the right at all times to perform and/or subcontract all portions of the construction and related work on project sites not covered by this Agreement.
- 3.11 It is expressly agreed and understood by the parties hereto that the County shall have the right to purchase material and equipment from any source and the craftsperson will handle and install such material and equipment, subject to the requirements of Section 3.6.
- 3.12 Without limiting the foregoing, items specifically excluded from the scope of this Agreement include the following:
  - 3.12.1 The operation of equipment and machinery owned or controlled by the County and its subcontractors and not directly related to construction of covered projects;
  - 3.12,2 All employees of any Contractor or any other consultant of the County not performing construction craft labor within the scope of this Agreement;
  - 3.12.3 Any work performed on or near or leading to or on to the site of work covered by this Agreement and undertaken by state, county, city or other governmental bodies, or their Contractors, or by public utilities or their Contractors, and/or by the County or its Contractors (for work which is not part of the scope of this Agreement);
  - 3.12.4 Off-site maintenance of leased equipment and on-site supervision of such work;
  - 3.12.5 Non-construction support services contracted by the County or any Contractor in connection with covered projects;
  - 3.12.6 All work by employees of the County;
  - 3.12.7 Operations or maintenance work executed by the County;
  - 3.12.8 All work on covered projects under any contract entered into prior to the date of this agreement;
  - 3.12.9 All warranty functions, warranty work, corrective work, repair and maintenance work on purchased equipment performed by manufacturers' representatives or vendors after completion and acceptance of any covered projects by the County; and
  - 3.12.10 All Post Disaster and Emergency Work as defined in Article 1.
- 3.13 The Council shall assist the County and its contractors in encouraging and soliciting subcontractors in bidding on all covered projects.

## RELATIONSHIP BETWEEN PARTIES

- 4.1 This Agreement shall only be binding on the Signatory parties hereto, and shall not apply to parents, affiliates, subsidiaries, or other divisions of the Coordinator and Signatory Contractors unless signed by such parent, affiliate, subsidiary, or other division of such company.
- 4.2 Each Contractor shall alone be liable and responsible for its own individual acts and conduct and for any breach or alleged breach of this Agreement. Any alleged breach of this Agreement by a Contractor or any dispute between the Signatory Union and the Contractor respecting compliance with the terms of this Agreement, shall not affect the rights, liabilities, obligations and duties between the signatory Union and each other Contractor party to this Agreement.
- 4.3 It is mutually agreed by the parties that any liability by a Signatory Union(s) to this Agreement shall be several and not joint. Any alleged breach of this Agreement by a signatory Union shall not affect the rights, liabilities, obligations and duties between the Signatory Contractors and the other Unions party to this Agreement.

#### ARTICLE 5

## ROLES AND RESPONSIBILITIES

#### SUBCONTRACTS

- 5.1 Each Contractor, which includes all subcontractors of any tier, including trucking entities performing Covered Work on the Projects, agrees that neither it nor any of its subcontractors will subcontract any work to be done on the Project except to a person, firm, or corporation who is or becomes party to this Agreement by signing the Agreement to be bound attached to this Agreement as Exhibit "A." All Contractors performing Covered Work on the Project shall, as a condition to performing work on the Project, become Signatory to and perform all work under the terms of this Agreement.
- 5.2 A Contractor includes any person, firm or corporation who agrees under contract with another Contractor of any tier, to perform on the Project any part or portion of the construction work covered by the prime contract, including the operating of construction equipment, performance of labor and/or installation of materials.
- 5.3 Notwithstanding any other provisions of this Agreement, the Contractor, as appropriate, in conformance with paragraph 3.7 of this Agreement shall have the absolute right to award contracts or subcontracts for this Project notwithstanding the existence or nonexistence of any collective bargaining agreements between the prospective Contractor and any Union party, and provided that such Contractor is willing, ready and able to comply with this Project Stabilization/Community Benefits Agreement and shall execute a Letter of Assent (in the

form attached as Exhibit A), should such Contractor be awarded work covered by this Agreement.

- 5.4 The furnishing of supplies, equipment or materials which are stockpiled for later use shall in no case be considered subcontracting and shall be covered to the extent permitted by law. The delivery of ready-mix, asphalt, aggregate, sand or other fill material which are directly incorporated into the construction process as well as the off-hauling of debris and excess fill material and/or mud, shall be covered by the terms and conditions of this Agreement.
- 5.5 Each Contractor with a contract directly with the County has the primary obligation for performance of all conditions of this Agreement, including the performance of all of that Contractor's subcontractors. This obligation cannot be relieved, evaded or diminished by subcontracting. Should a Contractor elect to subcontract, that Contractor shall continue to have such primary obligation.
- 5.6 Each Contractor, which includes all subcontractors of any tier performing work on the Project, shall give written notice to the Union(s) of any subcontract involving the performance of work covered by this Agreement within either five (5) business days of entering such subcontract or before the subcontractor commences work on the Project, whichever occurs first. Such notice shall specify the name and address of the subcontractor, the California State License Board license number of the Contractors and the scope of work to be performed. Written notice at a Pre-Job Conference shall be deemed written notice under this provision only for those subcontractors listed at the Pre-Job Conference.

#### 5.7 Signatory Contractors:

- 5.7.1 With regard to any Contractor that is independently signed to any Master Labor Agreement, this Agreement shall in no way supersede or prevent the enforcement of any subcontracting clause contained in such MLA, except as specifically set forth in Section 5.7.2 below. Any such subcontracting clause in an MLA shall remain and be fully enforceable between each craft union and its signatory Contractors, and no provision of this Agreement shall be interpreted and/or applied in any manner that would give this Agreement precedence over subcontracting obligations and restrictions that exist between craft unions and their respective signatory Contractors under an MLA, except as specifically set forth in subsection 5.7.2 below.
- 5.7.2 If a craft union ("aggrieved union") believes that an assignment of work for this Project has been made improperly by a Contractor or subcontractor, even if that assignment was as a result of another craft union's successful enforcement of the subcontracting clause in its MLA, as permitted by subsection 5.7.1 above, the aggrieved union may submit a claim under the jurisdictional resolution procedure contained in Article 6 of this Agreement, and the decision rendered as part of that process shall be enforceable to require the Contractor or subcontractor that made the work assignment to assign that work prospectively to the aggrieved union. An award made to a craft union under the subcontracting clause of its MLA, as permitted pursuant to subsection 5.7.1 above, shall be valid and fully enforceable by that craft union unless it conflicts with a jurisdictional award made pursuant to this Agreement. If the award made under the MLA conflicts with the jurisdictional award, the award of damages under the former shall be null and void ab initio.

## WORK ASSIGNMENTS AND JURISDICTIONAL DISPUTES

- 6.1 The following language is specifically agreed to for the resolution of any Jurisdictional Disputes which may arise during the construction which is specifically covered by this Agreement. This agreement regarding resolution of jurisdictional disputes shall apply only to such disputes arising on Covered Projects.
- 6.2 There will be no strikes, no work stoppages, no picketing, sympathy strikes, slowdowns or other interferences with the work because of jurisdictional disputes between signatory Unions. Individuals violating this section shall be subject to immediate discharge.
- 6.3 The assignment of Covered Work will be solely the responsibility of the Contractor performing the work involved; and such work assignments will be in accordance with the Plan for the Settlement of Jurisdictional Disputes in the Construction Industry (the "Plan") or any successor Plan.
- 6.4 All jurisdictional disputes on this Project between or among the Building and Construction Trades Unions and the Contractors, parties to this Agreement, shall be settled and adjusted according to the present Plan established by the Building and Construction Trades Department, or any other plan or method of procedure that may be adopted in the future by the Building and Construction Trades Department. Decisions rendered shall be final, binding and conclusive on the Contractor and Union parties to this Agreement. 6.4.1 For the convenience of the parties, and in recognition of the expense of travel between Northern California and Washington, D.C., at the request of any party to a jurisdictional dispute under this Agreement, an Arbitrator shall be chosen by the procedures specified in Article V, Section 5, of the Plan from a list composed of John Kegel, Thomas Angelo, Robert Hirsch and Thomas Pagan, and the Arbitrator's hearing on the dispute shall be held at the applicable Building and Construction Trades Council. All other procedures shall be as specified in the Plan.
- 6.5 All jurisdictional disputes shall be resolved without the occurrence of any strike, work stoppage, or slow-down of any nature, and the Contractor's assignment shall be adhered to until the dispute is resolved. Individuals violating this section shall be subject to immediate discharge.
- 6.6 Each Contractor shall conduct a Pre-Job Conference with the Council prior to commencing Covered Work. The Prime Contractor, the County and the Coordinator will be advised in advance of all such conferences and may participate if they wish. Prejob conferences for different Contractors may be held together.

## PRE-JOB CONFERENCE

- 7.1 A mandatory Pre-Job Conference with each Contractor will be held prior to the commencement of work to establish the scope of work in each Contractor's contract. When a contract has been let to Contractors covered by this Agreement, a Pre-Job Conference and/or Markup Meeting shall be requited and shall be held. The parties may mutually agree to waive the requirement to hold a Pre-Job Conference and/or Mark-up Meeting for any particular contract or contractor. All meetings shall be held at the offices of the Alameda County Building and Construction Trades Council.
- 7.2 The Contractor performing the work shall have the responsibility for making work assignments in accordance with Section 6.3 of this Agreement, and will be required to bring relevant plans, specifications, and blueprints to the meeting, as requested by Union.
- 7.3 The Coordinator will schedule and attend all Pre-Job and Mark-Up Meetings and participate in discussions as they pertain to the terms and conditions of this Agreement

## JOINT ADMINISTRATIVE COMMITTEE MEETINGS

- The parties to this Agreement will form a five (5) person Joint Administrative Committee (JAC). The Committee will be comprised of two (2) representatives selected by the Council, two (2) representatives selected by the County, and one (1) community representative, nominated by the Board of Supervisors and agreeable to the Council. The parties shall appoint an alternate. The JAC meetings will be convened by the Coordinator and chaired jointly by a representative of the Council and the County, and a quorum shall be three (3) members, including at least one (1) from the County and one (1) from the Council. The purpose of these meetings is to promote harmonious labor/management relations, ensure adequate communications and advance the proficiency and efficiency of the employees and the Contractors for the Covered Projects. The Committee shall also monitor compliance with Article 17 and Article 18. These meetings will also include discussion of the scheduling, productivity and safety of work performed for the Covered Projects.
- 8.2 The JAC shall appoint a Joint Administrative Subcommittee, comprised of one (1) representative of the County and, one (1) representative of the Council for the purpose of convening to confer in an attempt to resolve any grievance that has been filed consistent with Article 23. This Subcommittee shall meet as required to resolve grievances by consensus vote. If no resolution can be mutually agreed upon, the grievance shall proceed to the grievance procedure outlined in Article 23, Step 4.
- 8.3 The JAC shall appoint a Joint Administrative Subcommittee, comprised of one (1) representative of the County, one (1) representative of the Council and one (1) representative of a community-based organization to resolve any grievance filed consistent with Article 17 or Article 18.
- 8.4 The JAC shall have the initial authority to investigate and resolve by consensus vote any allegation of violations of Articles 19 and 20. If the JAC cannot resolve the allegations, then any signatory party may take the matter directly to final and binding arbitration as described in Article 23.
- 8.5 JAC Meetings
  - 8.5.1 The JAC will meet monthly at the call of either chair.
  - 8.5.2 The Coordinator will establish agenda topics with input from the Committee and send notices of meetings with the agenda in advance of the meetings.
  - 8.5.3 3 The JAC will receive reports and consider work progress and practices, local hire utilization. Disadvantaged Population utilization, pre-apprentice recruitment, training and referral, and apprentice development and utilization.
  - 8.5.4 The Coordinator and the Contractors shall report progress on these issues and provide ongoing workforce projections for their work.

- 8.6 Joint Administrative Subcommittee Meetings
  - 8.6.1 Both Joint Administrative Subcommittees will meet as required to address grievances/disputes.
  - 8.6.2 The Coordinator will establish agenda topics with input from the Subcommittee and send notices of meetings with the agenda in advance of the meetings.

## COORDINATOR

- 9.1 The County will designate a Coordinator, who will be responsible for the administration and application of this Agreement.
- 9.2 The Coordinator shall endeavor to facilitate harmonious relations between the Contractors and Unions Signatory hereto and will conduct the Joint Administrative Committee meeting at the request of either joint chair referred to in Article 8 above. The Coordinator shall not be responsible for the acts of the Contractors or Unions Signatory hereto and will not be a party to any arbitration or litigation arising out of this Agreement.

#### ARTICLE 10

## UNION RECOGNITION AND REPRESENTATION

- 10.1 The Contractors recognize the Union(s) Signatory hereto as the sole and exclusive collective bargaining representatives for all craft employees on the Project.
- 10.2 All employees who are employed by the Contractors shall, as a condition of employment, on or before the eighth (8th) day of consecutive or cumulative employment for a construction contract subject to this Agreement, be responsible for the payment of the applicable monthly working dues and any associated fees uniformly required for membership in the Union(s). However, there is nothing in this Agreement that would prevent non-union employees from joining the Union(s).
- 10.3 Authorized representatives of the Union(s) shall have access to the Project site at all times when work is being, has been or will be performed. Such representatives shall comply with the reasonable visitor safety and security rules established for the Project. Access for Union(s) representatives will not be unduly restricted.
- 10.4 The treatment and payment of stewards shall be in accordance with the applicable MLA

## NO STRIKES-NO LOCKOUTS

- 11.1 During the life of this Agreement, the Unions and their members, agents, representatives and employees shall not incite, encourage, condone or participate in any strike, walkout, slowdown, sit-down, stay-in, boycott, wobble, sympathy strike, picketing or other work stoppage or hand-billing on the Covered Projects for any cause whatsoever, or any other type of interference of any kind, coercive or otherwise, and it is expressly agreed that any such action is a violation of this Agreement.
  - 1 1.1.1 Withholding of employees for failure of a Contractor to meet its weekly payroll is not a violation of this Article 11; however, the Union shall submit documentation of the failure to pay to the Coordinator and shall give the affected Contractor and the Coordinator written notice seventy-two (72) hours prior to the withholding of employees.
  - 11.1.2 Should a Contractor performing work for this Project be delinquent in the payment of Trust Fund contributions required under this Agreement with respect to employees represented by the Union, the Union may request, that the Contractor issue joint checks payable to the Contractor and the appropriate employee benefit Trust Fund until such delinquencies are satisfied. Any Trust Fund claiming that a Contractor is delinquent in its fringe benefit contributions to the funds will provide written notice of the alleged delinquency to the affected Contractor, with copies to the General Contractor, the Coordinator and the County. The notice will indicate the amount of delinquency asserted and the period that the delinquency covers. It is agreed, however, with respect to Contractors delinquent in trust or benefit contribution payments, that nothing in this Agreement shall affect normal contract remedies available under the local collective bargaining agreements. If the Contractor is delinquent in the payment of Trust Fund contributions for covered work performed for this Project, the Contractor agrees that the affected Trust Fund may place the County on notice of such delinquencies and the Contractor further agrees that the County may issue joint checks to the Contractor and the Trust Fund until the delinquency is satisfied.
- 11.2 Expiration of Local and Other Applicable Agreements. It is specifically agreed that there shall be no strike, sympathy strike, picketing, lockout, slowdown, withholding of work, refusal to work, walk-off, sick-out, sit-down:, stand-in, wobble, boycott or other work stoppage of any kind as a result of the expiration of any local, regional or other applicable labor agreement having application on the Project and/or failure of the parties to that agreement to reach a new contract. If a Master Labor Agreement between a Contractor and the Union expires before the Contractor completes the performance of a construction contract and the Union or Contractor gives notice of demands for a new or modified Master Labor Agreement, the Union agrees that it will not strike or withhold labor from the Contractor for said contract for work covered under this Agreement and the Union and the Contractor agree that the expired collective bargaining agreement shall continue

in full force and effect for work covered under this Agreement until a new or modified Master Labor Agreement is reached between the Union and Contractor. If the Union and Contractors agree to an interim agreement that will apply until a new Master Labor Agreement is reached, then, the Contractor may work under the terms of the interim agreement until a new or modified Master Labor Agreement is reached between the Union and Contractor. If the new or modified Master Labor Agreement reached between the Union and Contractor provides that any terms of compensation of the Master Agreement shall be retroactive, the Contractor agrees to comply with any retroactive terms of the new or modified Master Labor Agreement to its effective date which is applicable to employees who performed work for the project during the interim period. Such compliance shall occur within seven (7) days after notification by the Union.

- 11.3 In consideration of the foregoing, the Contractor shall not incite, encourage or participate in any lockout or cause to be locked out any employee covered under the provisions of this Agreement. The term "lockout" does not refer to the discharge, termination or layoff of employees by the Contractor for any reasons in the exercise of its rights as set forth in any provision of this Agreement, nor does "lockout" include the County's or Contractor's decision to terminate or suspend work for the site or any portion thereof for any reason.
- 11.4 Any employee or employees inciting, encouraging or participating in any strike, slowdown, picketing, sympathy strike or other activity in violation of this Agreement may be subject to immediate discharge and the procedure under this Article 11, if invoked.
- 11.5 Upon written or electronic mail notice of a violation to the Local and/or International Union offices, the Union and its officers shall take immediate action and will use its (their) best efforts to prevent, end or avert any such aforementioned activity or the threat thereof by any of its officers, members, representatives or employees, either individually or collectively, including but not limited to, publicly disavowing any such action and ordering all such officers, representatives, employees or members who participate in such unauthorized activity to cease and desist from same immediately and to return to work and comply with its orders. The Contractor shall have the right, in the event of a work stoppage by the Union to replace the employees represented by the Union in violation of this Agreement. Nothing in this Agreement shall be construed to limit or restrict the right of any of the parties to this Agreement to pursue fully any and all remedies available under law in the event of violation of this Article 11.
- 11.6 Any party to this Agreement may institute the following binding arbitration procedure when such a breach is alleged. In the event a party institutes this procedure, arbitration shall be mandatory.
  - 11.6.1 The party invoking this procedure shall immediately notify Robert Hirsch, who the parties agree shall be the permanent Arbitrator under this procedure. Thomas Angelo shall serve as alternate in the event that the permanent Arbitrator is unavailable at any time. Notice to the Arbitrator shall be by the most expeditious means available, with written notice by email or similar means to the party alleged to be in violation and the involved Union General President.

- 11.6.2 Upon receipt of said notice the Arbitrator named above or the alternate shall designate a place for, schedule and hold a hearing within twenty-four (24) hours.
- 11.6.3 The Arbitrator shall notify the parties by electronic mail or similar means of the place and time chosen for the session. A failure of any party or parties to attend said hearing shall not delay the hearing of evidence or issuance of an award by the Arbitrator.
- 11.6.4 The sole issue at the hearing shall be whether or not a violation of this Article has in fact occurred, and the Arbitrator shall have no authority to consider any matter in justification, explanation or mitigation of such violation or to award damages, which issue is reserved for court or other arbitration proceedings, if any. The award shall be issued in writing within three (3) hours after the close of the hearing and may be issued without a written opinion. If any party desires a written opinion, one shall be issued within fifteen (15) days, but its issuance shall not delay compliance with, or enforcement to the award. The Arbitrator shall order cessation of the violation of this Article and other appropriate relief and such award shall be served on all parties by hand or registered mail upon issuance.
- 11.6.5 Liquidated Damages. A party found to have violated the provisions of the No Strike-No Lockout section in this Article 11 shall cease such violation within eight (8) hours of the award of the Arbitrator. Should the violation continue past eight (8) hours, the party in violation shall pay to the affected party as liquidated damages either the actual damages incurred or the sum of ten thousand dollars (\$10,000.00) per shift, or portion thereof, whichever is greater, until such violation is ceased. The Arbitrator shall retain jurisdiction to resolve any disputes regarding the liquidated damages claimed under this section.
- 11.6.4 The award shall be final, binding and non-reviewable as to the merits. A judgment of any court of competent jurisdiction shall be entered upon the award, which may be enforced by any such court, upon the filing of this Agreement and all other relevant documents referred to herein in the following manner. Electronic mail or similar notice of the filing of such enforcement proceedings shall be given to the other party. In the proceeding to obtain a temporary order enforcing the Arbitrator's award as issued under subsection 11.6.4 of this Article, all parties waive the right to a hearing and agree that such proceedings may be ex parte. Such agreement does not waive any party's right to participate in a hearing for a final order of enforcement. The Court's order or orders enforcing the Arbitrator's award shall be served on all parties by hand or by delivery to their last known address or by registered mail. Any rights created by statute or law governing arbitration or injunction proceedings inconsistent with the above procedure, or which interfere with compliance therewith, are hereby waived by the parties to whom they accrued.
- 1 t.6.8 The costs of the arbitration, including the fee and expenses of the Arbitrator, shall be borne equally by the affected Union(s) and the affected Contractors.
- 11.6.9 The procedures contained in this Section 11.6 shall be applicable only to alleged violations of this Article. Discharge or discipline of employees for violation of this Article shall be subject to the grievance and arbitration procedures of Article 23.

## **MANAGEMENT RIGHTS**

- 1 2.1 The Contractor retains full and exclusive authority for the management of their work forces for all work performed under this Agreement. This authority includes, but is not limited to, the right to:
  - 12.1.1 Plan, direct and control the operation of all the work.
  - 12.1.2 Decide the number and types of employees required to perform the work safely and efficiently. The lawful manning provisions of the applicable Master Collective Bargaining Agreement shall be recognized.
  - 12.1.3 Hire, promote and lay off employees as deemed appropriate to meet work requirements and/or skills required.
  - 12.1.4 Require all employees to observe the County's Project Rules, the Contractor's Project Rules, Security and Safety Regulations, consistent with the provisions of this Agreement. The Contractor's and County's Project Rules and Regulations shall be reviewed and mutually agreed upon at the Pre-Job meeting and supplied to all employees and/or posted on the jobsite.
  - 12.1.5 Discharge, suspend or discipline employees under the applicable ML.A.
  - 12.1.6 Assign and schedule work at its sole discretion and determine when overtime will be worked consistent with this Agreement and the applicable MLA.
  - 12.1.7 Utilize any work methods, procedures or techniques and select and use any type or kind of materials, apparatus or equipment regardless of source, manufacturer or designator and in accordance with this Agreement, which covers the fabrication provisions and any other conflicts that are addressed in this Agreement.
- 12.2 The foregoing listing of management rights shall not be deemed to exclude other functions not specifically set forth herein. The Contractors, therefore, retain all legal rights not specifically enumerated in this Agreement.

## WORK RULES

13.1 Work Rules shall be governed by the applicable MLA for each craft.

#### **ARTICLE 14**

## WAGE SCALES AND FRINGE BENEFITS

- 14.1 All employees covered by this Agreement shall be classified and paid in accordance with the classification and wage scales contained in the appropriate MLAs which have been negotiated by the historically recognized bargaining parties and in compliance with the applicable general prevailing wage determination made by the Director of Industrial Relations pursuant to the California Labor Code.
- 14.2 For the duration of its work on this Project, the Contractors agree to recognize and put into effect such increases in wages and recognized fringe benefits as shall be negotiated between the various Union(s) and the historically recognized local bargaining parties on the effective date as set forth in the applicable MLA. The Union(s) shall notify the Contractors in writing of the specific increases in wages and recognized fringe benefits and the date on which they become effective.
- 14.3 The Contractors hereby adopt and agree to be bound by the written terms of the legally established Trust Agreements specifying the detailed basis on which payments are to be made into, and benefits paid out of, such appropriately qualified employee fringe benefit funds established by such appropriate Trust Agreements. The Contractors authorize the parties to such Trust Agreements to appoint Trustees and successor Trustees to administer the trust funds, and hereby ratify and accept the Trustees so appointed as if made by the Contractors.
- 14.4 If a Contractor fails to pay wages or benefits, the County agrees to honor a properly submitted, legally enforceable Stop Payment Notice.

## HOURS OF WORK, OVERTIME, SHIFTS AND HOLIDAYS

- 15.1 The hours of work, establishment of overtime and the establishment of shifts and shift pay shall be governed by the applicable MLA for each craft and in accordance with the current General Prevailing Wage Determination made by the Director of Industrial Relations pursuant to the California Labor Code. It is understood that the County may, at its discretion, establish a uniform starting time and or ending time.
- 15.2 Holidays and designated days off will be incompliance with the applicable General Prevailing Wage Determination made by the Director of Industrial Relations pursuant to the California Labor Code, unless otherwise set forth in the MLA.

#### **ARTICLE 16**

## HEALTH AND SAFETY

- 16.1 The employees covered by the terms of this Agreement shall at all times, while in the employ of the Contractor, be bound by the safety rules and regulations as established by the County and Contractors and in accordance with OSHA Cal-OSHA. These rules and regulations will be published and posted at conspicuous places throughout the Project site.
- 16.2 In accordance with the requirements of OSHA/Cal-OSHA, it shall be the exclusive responsibility of each Contractors working on the Project to assure safe working conditions for its employees and compliance by them with any safety rules contained herein or established by the Contractors or the County. Nothing in this Agreement shall in any way be construed to make the Union(s), the County, liable for safety violations on the Project.
- 16.3 The parties agree to abide by the substance abuse policies contained in the applicable MLA, subject to the Article 12. Should the County decide that there is a need for an OCIP on a Covered Project, the parties mutually agree to the side letter attached.

#### **ARTICLE 17**

## LOCAL HIRING PROGRAM

17.1 The Parties agree to achieve the inclusion of Residents in the employment and apprenticeship opportunities created by the Covered Work, which will be known as the Local Hiring Program (LHP). With day-to-day support from the Coordinator, the Joint Administrative Committee (JAC) formed pursuant to the provisions of Article 8 shall

Page 21 of 40

County of Alameda

Countywide Project Stabilization/ Community Benefits Agreement

monitor the progress of the LHP and will serve as the central forum for representatives of all interested or affected parties to exchange information and ideas and to advise the County staff and the Coordinator concerning the operation and results of the LHP and the ongoing role of this Project Stabilization/Community Benefits Agreement as an integral component of LHP. As part of these responsibilities, the JAC will assess the obstacles to success of achieving inclusion of local Residents in the construction opportunities and shall make recommendations for a program to overcome some of those obstacles.

- The parties agree to a goal that Residents of the County will perform up to 40 percent (40%) of all hours worked on all covered projects, on a craft-by-craft basis, if such workers are available, capable and willing to work on the projects, together with the apprentice goals established in Article 18, below.
- 17.4 below and to reach these goals working through the normal hiring hall procedures listed in the MLA and the procedures identified in Article 18.4 and the County and Unions shall make good faith efforts to assist the Contractor in reaching this goal. In cases of alleged noncompliance, the issue may be referred to the Coordinator and then to the JAC for resolution. If the JAC can make no resolution, the issue may then be referred to Step 4 of Section 23.2.2 of the grievance procedure described in Article 14 for submission to an arbitrator for a final and binding determination. For purposes of resolution of any dispute arising under this Section or Article 18.4, the County shall be considered a party-in-interest with full right of participation in the arbitration proceeding.
- 17.4 The Contractors must take, and require their subcontractors to take, the following good faith steps to demonstrate that they have made every effort to reach the Local Hiring Goals:
  - 17.4.1 The Contractors shall attend the scheduled pre-job meetings identified in Article 7. At this meeting, the Contractor must submit written workforce projections and projected man-hours on a craft-by-craft basis, consistent with the Contractor's bid proposal. In the event the pre-job meeting is waived, the Contractor must submit written workforce projections to the Coordinator within five (5) days.
  - 17.4.2 Within one (1) week of the issuance of the Notice to Proceed, the Contractors shall meet with the Coordinator to review and approve its compliance plan for reaching the Local Hiring Goals, using the required compliance plan form provided by the County.
  - 17.4.3 The Contractors shall submit copies of hiring hall dispatch requests and responses to the Coordinator within ten (10) days of Coordinator's request at any point during the execution of the Project.
  - 17.4.4 The Contractors shall immediately contact the Coordinator if a union hiring hall dispatcher will not or cannot, upon request of the Contractor, dispatch local Residents.
  - 17.4.5 The Contractor shall use the "Name Call," "Rehire" or other available hiring hall procedures to reach goals and shall provide documentation of such requests to the Coordinator upon request per subsection 17.4.3.

- 17.4.6 The Contractors shall use community-based organizations as a resource for local labor resources, if a union will not or cannot provide local Residents as requested, and in conformity with the collectively bargained union hiring hall agreement.
- 17.4.7 The Contractors shall sponsor local Residents as defined herein for apprenticeship, when possible.
- 17.4.8 The Contractors shall maintain records for each Resident of Alameda County who was referred but, not hired along with an explanation why the worker was not hired. Upon request, such records shall be made available for review by the County, Coordinator, and JAC for the duration of the Covered Projects.
- I 7.4.9 The Contractors shall document participation in any local employment training programs and submit documentation of such to the Coordinator within ten (10) days if requested by Coordinator.
- 17.5 The Unions will exert their utmost efforts to recruit sufficient numbers of skilled craft persons who are Residents to fulfill the requirements of the Contractors. The parties to this Agreement support the development and placement of increased numbers of skilled construction workers from the Residents within the County to meet the needs of the covered project and the requirements of the industry generally.
- 17.6 To the extent possible, the parties agree to implement the Local Hiring Program while complying with the County's Local Vendor Preference and Enhanced Construction Outreach (ECOP) programs for the covered project. To the extent that the County determines, in its sole discretion, that there is a conflict between the Local Hiring Program established in this Agreement and the County's SLEB, ECOP, and/or Local Vendor Preference Programs, the conflict shall be resolved in favor of the Local Hiring Program on the construction work covered by this Agreement.
- 17.7 For the purposes of reaching the goal established in Article 17.2, a Contractor may qualify for full credit toward the goal by employing Alameda County Residents for other work the Contractor is performing in any of the nine (9) Bay Area counties of: Alameda, Contra Costa, San Francisco, San Mateo, Santa Clara, Marin, Solano, Napa and Sonoma. Credit will only be given for work performed during the life of the Covered Project. In order to receive such credit, the Contractor must submit certified payrolls as documentation to the Coordinator. No credit for off-site work will be allowed until the Contractor has demonstrated a good faith effort to reach the goal on the Covered Projects and has received approval from the JA

## APPRENTICES

- 18.1 Recognizing the need to maintain continuing support of programs designed to develop adequate numbers of competent apprentice workers in the construction industry, the Contractors will make a good faith effort to employ apprentices in the respective crafts to perform such work as is within their capabilities and which is customarily performed by the craft in which they are indentured.
- 18.2 For the purpose of meeting the goals of this Article 18, the parties recognize Stateapproved apprenticeship training programs administered by Joint
  Labor/Management Apprenticeship Training Committees (JATC) as the sole source
  for an eighteen (18) month trial period. If after the eighteen-month trial period, it has
  been demonstrated that the JATCs are unable to provide sufficient Residents to meet
  the established goals, the parties to this agreement will meet to negotiate
  implementation strategies to meet the established goals. If resolution is not reached
  within six (6) months, then the parties may refer the item to the grievance procedure
  as outlined in Article 23.
- 18.3 The Signatory parties agree that the County shall make available to the Unions a database of apprentices qualifying under the local hiring provisions of this Agreement. The Signatory Unions agree to report in accordance within any limits set by applicable labor law, the availability and dispatch/placement of qualifying apprentices. The reports will be submitted to the Coordinator on at least a quarterly basis and more often, if requested and possible.
- 18.4 For each Covered Project, the Contractors will be responsible to ensure that it and/or its subcontractors hire at least one (I) new apprentice for the first \$1 million of construction value and for each succeeding \$5 million of construction contract value, the Contractors and/or their subcontractors will be required to hire at least one (1) additional new apprentice. All such apprentices may be graduates of pre-apprenticeship programs with a known and successful track record of apprentice placement into jobs. All the pre-apprenticeship program graduates must be Residents of Alameda County and members of a Disadvantaged Population, as described in Article 1.
  - 18.4.1 Contractors will make a good faith effort to maximize the project work hours for the new hire apprentices, and shall report those hours to the JAC, which will evaluate those good faith efforts.
  - 18.4.2 Each Signatory Union will be responsible for dispatching/referring such County Residents to the contractor if they are available, capable and willing to work on the Covered Projects. No one trade can be used to satisfy the goal by the provision of more than two (2) such first stage apprentices, unless required by the nature of the work and or agreed upon by the JAC.

- 18.4.3 The Signatory Unions and Contractors shall exercise, to the extent of their authority, their best efforts to recruit apprenticeship program applicants from Residents who are members of a Disadvantaged Population, as defined in Article I. Further, for apprentices hired to comply with Article 18.4, there will be no limitation on where such apprentices will work subsequent to being hired for the Covered Projects. Contractors will be allowed to receive credit for Article 18.4 when utilizing apprentices for non-Project work during the life of the Covered projects, regardless of the location of the work as long as it is in the nine (9) Bay Area counties described in Article 18.4.5.
- 18.4.4 The Contractor shall request dispatch of apprentices in writing from the local Unions and/or Joint Apprenticeship Training Committee in which the Contractor participates. Copies of the written requests shall be provided to the Coordinator within ten (10) days of request by the Coordinator. The Unions shall honor all Contractor dispatch requests for such Apprentices.
- 18.4.5 For the purposes of meeting the goal established in. Section 18.4, a Contractor may qualify for full credit toward the goal by employing Alameda County Residents as apprentices for other work the Contractor is performing in any of the nine Bay Area counties of: Alameda, Contra Costa, San Francisco, San Mateo, Santa Clara, Marin; Solano, Napa and Sonoma. Credit will only be given for work performed during the life of the Covered projects. In order to receive such credit, the Contractor must submit certified payrolls as documentation to the Coordinator. No credit for non-Covered Projects work will be allowed until the Contractor has demonstrated a good faith effort to reach the goal on the Covered projects and has received approval from the JAC.
- 18.5 The Unions will cooperate with the County, the Contractors, and the Coordinator in conducting outreach activities to recruit and refer qualified Alameda County Resident applicants to apprenticeship programs. In addition, the Unions will work with designated pre-apprenticeship programs to promote graduates and enhance their entry into the Apprenticeship programs.
- 18.6 To the extent permitted by law and the JATC requirements, the Unions will give credit for bona fide, provable past experience to applicants, including work for non-union Contractors who become signatory to the PS/CBA. The experience and practical knowledge of applicants will be reviewed and tested by the applicable Joint Apprenticeship Training Committee. Applicants will be placed at the appropriate stage of apprenticeship or journey level as the case may be. Final decisions will be the responsibility of the applicable Joint Apprenticeship Training Committee.

## REFERRAL PROCESS

- 19.1 The Union(s) shall be the primary source of all craft labor employed on the Project. However, in the event that a Contractor has its own core workforce, the Contractors may request by name, and the Union(s) shall honor, referral of persons who have applied to the local union for Project work and who demonstrate the following qualifications ("Core Employees"):
  - 19.1.1 Possess any license and/or certification required by state or federal law for the Project work to be performed;
  - 19.1.2 Have worked a total of at least one-thousand (1,000) hours in the construction craft during the prior three (3) years;
  - 19.1.3 Were on the Contractors' active payroll for at least sixty (60) out of the one-hundred forty (140) calendar days prior to the contract award; and
  - 19.1.4 Have the ability to perform safely the basic functions of the applicable trade.
  - 19.1.5 Be a resident of Alameda County at least six months prior to the hire date.
- 19.2 In the case of a Sole Proprietor/Owner Operator that is self-performing work, this Sole Proprietor/Owner Operator is not required to request a dispatch from the union hall. Sole Proprietors/Operators must be certified as such by some public agency acceptable to the County and the affected signatory unions. If the Sole Proprietor/Owner Operator hires employees subsequent to starting work on Covered Projects, all such employees would need to be requested from the union hall as described in subsection 19.3 below.
- 19.3 The Union(s) will first refer to such Contractors one journeyman employee from the hiring hall out-of-work list for the affected trade or craft, and will thereafter refer one of such Contractors' "core" employees as a journeyman and shall repeat the process, one and one, until such Contractors crew requirements are met or until such Contractors have hired no more than five (5) Core Employees, whichever occurs first. Thereafter, all additional employees shall be hired exclusively from the Union(s)' hiring hall out of work list(s). For the duration of the Contractors' work the ratio shall be maintained and when the Contractors' workforce is reduced, Employees shall be laid off in the same ratio of core employees to hiring hall referrals as was applied in the initial hiring. Contractors signatory to a Local, Regional, and/or National collective bargaining agreement(s) with Signatory Union(s) hereto shall be bound to use the hiring hall provisions contained in the relevant MLA of the affected Union(s), and nothing in the referral provisions of this Agreement shall be construed to supersede the local hiring hall provisions of the MLAs as they relate to such Contractors.

- 19.4 For purposes relating to "Owner/Operators" used for the hauling of workers and materials, including water or oil it is agreed that the Owner/Operator doing such hauling work may be dispatched to the job first (as a core employee) provided that such Owner/Operator has complied with and completed all registration requirements with the Union prior to dispatch.
- 19.5 All Contractors shall be bound by and utilize the registration facilities and referral systems established or authorized by the Signatory Union(s) so long as such procedures are in compliance with applicable federal, state or local law. The Contractor shall have the right to determine the competency of all employees and may reject any referral for any reason, provided that the Contractor complies with Article 22, Non-Discrimination, and in accordance with the applicable MLA.
- 19.6 In accordance with the Master Labor Agreement and in the event that referral facilities maintained by the Union(s) are unable, despite good faith efforts, to fill the request of a Contractor for employees within a forty-eight (48) hour period after such request is made by the Contractor, Saturdays, Sundays and Holidays excluded, the Contractor shall be free to obtain work persons from any source ("Alternative Employees"). Upon hiring Alternative Employees, the Contractor shall immediately notify the appropriate Union(s) of the name and address of the Alternative Employees hired, which Alternative Employees shall be bound by the provisions of this Article and the Union(s)' hiring hall rules.
- 19.7 The Union(s) will exert their utmost efforts to recruit sufficient numbers of skilled craft persons to fulfill the requirements of the Contractors. The parties to this Agreement support the development of increased numbers of skilled construction workers from the Residents of Alameda County to meet the needs of the Project and the requirements of the industry generally. Accordingly, contingent upon request by the Contractor, the Unions agree to encourage the referral and utilization of Residents as journeyman and apprentices on the Project and the entrance of Residents into apprenticeships and training programs, as long such Residents possess the requisite skills and qualifications.

## REPORTING AND DATA COLLECTION

- 20. 1 This article describes data collection, reporting guidelines and responsibilities for parties signatory to the PSCBA.
- 20.2 The County shall be responsible for collecting and maintaining accurate data on the availability of the Disadvantaged Population Residents who are available, capable and willing to work on Projects. This data will be made available and accessible to the Union.
- 20.3 On a monthly basis, Contractors must submit reports on the status and progress of local hiring on a craft by craft basis, including utilization of apprentices.

- 20.4 The signatory Unions agree to report in accordance with any limits set by applicable labor law, the availability and dispatch/placement of apprentices. These reports will be submitted to the Coordinator on at least a quarterly basis and more often, if requested and possible.
  - 20.4.1 In advance of pre-job meetings, the Union shall assist the Coordinator with developing a current list of Disadvantaged Population first stage apprentices available to work on the project.
  - 20.4.2 At pre-job meetings, the Coordinator shall supply contractors with a current list of Disadvantaged Population apprentices and their status of completion of their apprenticeship.
  - 20.4.3 On a quarterly basis, the Union shall provide the County and Coordinator a report on the status of Disadvantaged Population apprentices, including but not limited to their placement and advancement.
- 20.5 On an annual basis, the Union, County and Coordinator shall provide a report for the Board of Supervisor's review as described and required in Article 27.1. This report shall include but not be limited to the local hiring and apprentice goal performance, as well as challenges and benefits of the PSCBA.

## **HELMETS TO HARDHATS: VETERAN EMPLOYMENT**

- 21.1 The Contractors and the Unions recognize a desire to facilitate the entry into the building and construction trades of veterans who are interested in careers in the building and construction industry. The Contractors and Unions agree to utilize the services of the Center for Military Recruitment, Assessment and Veterans Employment (hereinafter "Center) and Center's "Helmets to Hardhats" program to serve as a resource for preliminary orientation, assessment of construction aptitude, referral to apprenticeship programs or hiring halls, counseling and mentoring, support network, employment opportunities and other needs as identified by the Contractors and the Unions.
- 21.2 The Unions and Contractors agree to coordinate with the Center to create and maintain an integrated database of veterans interested in working on the Project and of apprenticeship and employment opportunities for this Project. To the extent permitted by law, the Unions will give credit to such veterans for bona fide, provable past experience.
- 21.3 The Contractors may also utilize the services of the "Swords to Ploughshares" program.

## NON-DISCRIMINATION

22.1 The Unions and Contractors shall not discriminate against any employee or applicant for employment because of race, creed, color, sex, actual or perceived sexual orientation, national origin, age, religion, political affiliation, or membership or non-membership in labor organization union activity, military veteran status, and disability as identified in the Americans With Disabilities Act, or any other basis recognized by law.

#### **ARTICLE 23**

#### GRIEVANCE PROCEDURE

- 23.1 All disputes concerning the interpretation and/or application of this Agreement that do not fall within the Article 11 No-Strike/No-Lockout procedure, Article 6 Work Assignments and Jurisdictional Disputes, Article 17 Local Hiring Program, or Article 18 Apprentices, shall be governed by the following grievance and arbitration procedure.
- 23.2 Grievances between one or more Union(s) and one or more Contractor regarding interpretation and/or application of this Agreement shall be pursued according to the following provisions:
  - A grievance shall be considered null and void if not brought to the attention of the Contractors or the Union(s) within ten (10) working days after the grievance is alleged to have occurred but in no event more than thirty (30) days after the charging party became aware of the event giving rise to the dispute.
  - 2322 Grievances between one or more Union(s) and one or more Contractors regarding provisions of this Agreement shall be settled or otherwise resolved according to the following Steps and provisions:
  - Step 1: The Contractors or the Union(s)' representative and the grievant shall attempt to resolve the grievance with the craft supervisor or Steward.
  - Step 2: In the event the matter remains unresolved in Step I above, within five (5) working days, the grievance shall be reduced to writing and may then be referred by the Contractors or Union(s) to the grievant for discussion and resolution.
  - Step 3: In the event that the representatives are unable to resolve the dispute within the five (5) working days after its referral to Step 2, either involved party may submit it within five (5) working days to the Joint Administrative Sub-committee, established in Section 10.2, which shall meet within five (5) working days after such referral (or such longer time as is mutually agreed upon by the representatives on the Joint Administrative Subcommittee) to confer in an attempt to resolve the grievance. Regardless of which party has initiated the grievance proceeding, prior to the meeting of the Joint Administrative Subcommittee, the Union shall notify its international union representative(s), which shall advise both parties if it intends on participating in the meeting. The participation by the International Union Representative in this Step 3 meeting shall not delay the time set herein for the meeting,

unless otherwise mutually agreed by the parties. Decisions by the Joint Administrative Subcommittee shall be by majority vote with such resolutions to be final and binding on all signatories of the Agreement. If the dispute is not resolved by the Joint Administrative Subcommittee, it may be referred within five (5) working days by either party to Step 4.

- Step 4: In the event the matter remains unresolved in Step 3, either Party may request, within five (5) working days, that the dispute be submitted to arbitration. The time limits set out in this procedure may, upon mutual agreement, be extended. Any request for arbitration, request for extension of time limits, and agreement to extend such time limits shall be in writing.
- Step 5: Within seven (7) calendar days after referral of dispute in Step 4, the parties shall choose an arbitrator for final and binding arbitration. The parties agree that an arbitrator shall be selected by the alternate striking method from the following list. The party who shall strike the first name shall be selected by the toss of a coin (1) Carol Ibsen (2) Barbara Kong-Brown (3) Thomas Angelo (4) Robert Hirsch (5) William Ricker. Should a Party to the procedure fail or refuse to participate in the hearing, if the Arbitrator determines that proper notice of the hearing has been given, said hearing shall proceed to a default award. The Arbitrator's award shall be final and binding on all Parties to the arbitration. The costs of the arbitration, including the arbitrator's fee and expenses, shall be borne equally by the Parties. The Arbitrator's decision shall be confined to the question(s) posed by the grievance and the Arbitrator shall not have authority to modify amend, alter, add to, or subtract from, any provisions of this Agreement.
- 23.3 Grievances raised by County against one or more Union(s) and/or the Building Trades Council, or against the County by one or more Union(s) and/or the Building Trades Council, regarding provisions of this Agreement, shall be settled or otherwise resolved according to the following Steps and provisions:
  - 23.3.1 A grievance shall be considered null and void if not brought to the attention of the County or the Union(s) within ten (10) working days after the grievance is alleged to have occurred but in no event more than ten (10) days after the charging party became aware of the event giving rise to the dispute.
  - Step 1: The County/Union(s) Joint Administrative Subcommittee shall attempt to resolve the grievance. The County/Union(s) Joint Administrative Subcommittee shall meet within five (5) working days after receipt of the grievance (or such longer time as is mutually agreed upon by the representatives on this Joint Administrative Subcommittee) to confer with regard to the grievance. Decisions by the Joint Administrative Subcommittee shall be by majority vote with such resolutions to be final and binding on all signatories of the Agreement. If the dispute is not resolved by the Joint Administrative Subcommittee, within the five (5) working days after meeting on the grievance, either involved party may proceed to Step 2.

- Step 2: In the event the matter remains unresolved pursuant to Step 2, either Party may request that the dispute be submitted to arbitration. The time limits set out in this procedure may, upon mutual agreement, be extended. Any request for arbitration, request for extension of time limits, and agreement to extend such time limits shall be in writing.
- Step 3: Within seven (7) calendar days after referral of dispute in Step 2, the parties shall choose an arbitrator for final and binding arbitration. The parties agree that an arbitrator shall be selected by the alternate striking method from the following list. The party who shall strike the first name shall be selected by the toss of a coin (1) Carol Ibsen (2) Barbara Kong-Brown (3) Thomas Angelo (4) Robert Hirsch (5) William Ricker. Should a Party to the procedure fail or refuse to participate in the hearing, if the Arbitrator determines that proper notice of the hearing has been given, said hearing shall proceed to a default award. The Arbitrator's award shall be final and binding on all Parties to the arbitration. The costs of the arbitration, including the arbitrator's fee and expenses, shall be borne equally by the Parties. The Arbitrator's decision shall be confined to the question(s) posed by the grievance and the Arbitrator shall not have authority to modify amend, alter, add to, or subtract from, any provisions of this Agreement.
- Where an issue is addressed in this Agreement and an MLA, this Agreement shall prevail.
  Where an issue is addressed in an MLA and not in this Agreement, the MLA shall control.
- 23.5 Grievances between a Union(s) and a Union(s) signatory contractor involving interpretation or application of the Master Agreement shall be governed by the grievance procedures contained in the faster Agreement.

## MISCELLANEOUS PROVISIONS

- 24.1 <u>Counterparts.</u> This Agreement may be executed in counterparts, such that original signatures may appear on separate pages, and when bound together all necessary signatures shall constitute an original. Facsimile signature pages transmitted to other parties to this Agreement shall be deemed equivalent to original signatures.
- 24.2 Warranty of Authority. Each of the persons signing this Agreement represents and warrants that such person has been duly authorized to sign this Agreement on behalf of the party indicated, and each of the pallies by signing this Agreement warrants and represents that such party is legally authorized and entitled to enter into this Agreement.
- 24.3 <u>Ratification by Governing Board</u>. This Agreement shall not be binding on the County until it is ratified by the Board of Supervisors.

#### **ARTICLE 25**

## **ENTIRE AGREEMENT**

- 25.1 This Agreement represents the complete understanding of the parties. The provisions of this Agreement, including the MLAs, shall in every instance exclusively apply to and control work performed on the Project. The provisions of this Agreement shall take precedence over provisions of local, area, regional or national labor agreements. Nothing contained in the MLAs, working rules, by-laws, constitution and other similar documents of the Unions, shall in any way affect, modify or add to this Agreement unless otherwise specifically indicated in this Agreement or mutually agreed to in writing and executed by the parties. Practices not part of the terms and conditions of this Agreement shall not be recognized.
- 25.2 The Unions agree that this Agreement covers all matters affecting wages, hours and other terms and conditions of employment, and that during the terms of this Agreement, neither the Contractors, nor the Unions will be required to negotiate on any further matters affecting these or any other subject not specifically set forth in this Agreement except by mutual agreement of the Unions involved and the County.
- 25.3 The parties to this Agreement understand and agree that nothing in this Agreement shall supersede or take precedence over any Board policy or requirement including, but not limited to, the construction contract, contract documents, project manual, and general conditions for the Project.
- 25.4 Provisions negotiated into any new or modified MLA which are less favorable to the Contractor shall not apply to work covered by this Agreement. Any disagreement between the parties regarding the application of the provisions of any new or modified MLA shall be resolved under the dispute and grievance arbitration procedures set forth in Article 23.

25.5 This Agreement may be executed in counterparts, such that the original signatures may appear on separate pages and when bound together all necessary signatures shall constitute an original. Facsimile signature pages transmitted to other parties to this Agreement shall be deemed the equivalent to original signatures.

#### **ARTICLE 26**

## GENERAL SAVINGS CLAUSE

26.1 It is not the intention of the parties to violate any laws governing the subject matter of this Agreement. If any Article or provision of this Agreement shall be declared invalid, inoperative, or unenforceable by any competent authority of the executive, legislative, judicial or administrative branch of the federal state or local government, the parties shall suspend the operation of each such article or provision during the period of invalidity. Such suspension shall not affect the operation of any provision covered in this Agreement to which the law or regulation is not applicable. Further, parties agree that if and when any or all provisions of this Agreement are finally held or determined to be illegal or void by a Court of competent jurisdiction, the parties will promptly enter into negotiations concerning the substance affected by such decision for the purpose of achieving conformity with the requirements of an applicable law and the intent of the parties hereto.

#### **ARTICLE 27**

## **DURATION OF AGREEMENTENT**

27.1 This Agreement shall become effective on the day the County Board of Supervisors ratifies this Agreement and shall continue in full force and effect for 1 year. At any time after the effective date, if parties so desire they may enter negotiations for a new Agreement or an extension to be determined. There shall be an annual report with a presentation to the County Board of Supervisors prepared by the General Services Agency in collaboration with other applicable County Departments.

#### SIGNATURES

 BUILDING AND CONSTRUCTION TRADES
COUNCIL OF ALAMEDA COUNTY

By: Andreas Cluver, Secretary Treasurer

ADDITIONAL SIGNATURES OF THE SIGNATORY UNION REPRESENTATIVES ARE CONTAINED ON THE FOLLOWING PAGES

## Signatory Unions

Asbestos Workers Lacal 116	Ву:
Boilermakers Locat 549	Ву:
Bricklayers & Allied Craftsmen Local 3	Ву:
Northern California Carpenters Regional Council (on behalf of Carpenters, Local 713, Carpenters, Local 2236, Lathers, Local 68L, Millwrights, Local 182, Pile Drivers, Local 34)	Ву:
Pile Drivers, Local 34)	
Cement Masons Local 300	Ву:
Electrical Workers, Local 595	Ву:
Elevator Constructors, Local 8	Ву:
Hod Carriers, Local 166	Ву:
Iron Workers, Local 378	Ву:
Laborers, Local 67	Ву:
Laborers, Local 304	Ву:
Operating Engineers, Local 3	By: Retrit Miss
Plasterers, Local 66	By: 1 Colact ///che
Roofers, Local 81	Ву:
Sheet Metal Workers, Local 104	Ву:
Sign Display, Local 510 Sprinkler Fitters, Local 483	By:
Teamsters, Local 853	Бу:
United Association of Journeymen and Apprentices Fitting Industry, Underground Utility & Landscape, Local 355	Ву:

Asbestos Workers Local 116	Ву:
Boilermakers Local 549	Ву:
Bricklayers & Allied Craftsmen Local 3	Ву:
Northern California Carpenters Regional Council (on behalf of Carpenters, Local 713, Carpenters, Local 2236, Lathers, Local 68L, Milwrights, Local 102,	
Pile Drivers, Local 34)	Ву:
Cement Masons Local 300	Ву:
Electrical Workers, Local 595	Ву:
Elevator Constructors, Local 8	Ву:
Hod Carriers, Local 166	Ву:
Iron Workers, Local 378	Ву:
Laborers, Local 67	Ву;
Laborers, Local 304	Ву:
Operating Engineers, Local 3	Ву:
Plasterers, Local 66	Ву:
Roofers, Local 81	Ву:
Sheet Metal Workers, Local 104	By: Put When
Sign Display, Local 510 Sprinkler Fitters, Local 483	By:
Teamsters, Local 853	Ву:
United Association of Journeymen and Apprentices Fitting Industry, Underground Utility & Landscape, Local 355	By:

Asbestos Workers Local 116	Ву:
Boilermakers Local 549	Ву:
Bricklayers & Alfied Craftsmen Local 3	Ву:
Northern California Carpenters	
Regional Council (on behalf of	
Carpenters, Local 713, Carpenters,	
Local 2236, Lathers, Local 68L,	11/11/2
Millwrights, Local 102,	Addition
Plie Drivers, Local 34)	ву:
Cement Masons Local 300	Ву:
Electrical Workers, Local 595	Ву:
Elevator Constructors, Local 8	Ву:
Hod Carriers, Local 166	Ву:
tron Workers, Local 378	Ву:
Laborers, Local 67	Ву:
Laborers, Local 304	Ву:
Operating Engineers, Local 3	Ву:
Plasterers, Local 66	Ву:
Roofers, Local 81	Ву:
Sheet Metal Workers, Local 104	Ву:
Sign Display, Local 510	Ву:
Sprinkler Fitters, Local 483	Ву:
Teamsters, Local 853	Ву:
United Association of Journeymen	
and Apprentices Fitting Industry,	
Underground Utility & Landscape,	Bv:
Local SEE	

Asbestos Workers Local 116	Ву:
Boilermakers Local 549	Ву:
Bricklayers & Allied Craftsmen Local 3	Ву:
Northern California Carpenters Regional Council (on behalf of Carpenters, Local 713, Carpenters, Local 2236, Lathers, Local 68L, Millwrights, Local 102, Pile Drivers, Local 34)	Ву:
Cement Masons Local 300	Ву:
Electrical Workers, Local 595	Ву:
Elevator Constructors, Local 8	Ву:
Hod Carriers, Local 166	Ву:
Iron Workers, Local 376	Ву:
Laborers, Local 67	Ву:
Laborers, Local 304	Ву:
Operating Engineers, Local 3	Ву:
Plasterers, Local 66	Бу:
Roofers, Local 81	By: Usualas Brager Ksign
Sheet Metal Workers, Local 104	By:
Sign Display, Local 510 Sprinkler Fitters, Local 483	By:
Teamsters, Local 853	Ву:
United Association of Journeymen and Apprentices Fitting Industry, Underground Utility & Landscape, Local 355	Ву:

Asbestos Workers Local 116	Ву:
Boilermakers Local 549	Ву:
Bricklayers & Aliled Craftsmen Local 3	Ву:
Northern California Carpenters Regional Council (on behalf of Carpenters, Local 713, Carpenters, Local 2236, Lathers, Local 68L, Millwrights, Local 102,	
Pile Drivers, Local 34)	Ву:
Cement Masons Local 300	By:
Electrical Workers, Local 595	By: Daniel & Chwello-
Elevator Constructors, Local 8	Ву:
Hod Carriers, Local 166	Ву:
Iron Workers, Local 378	Ву:
Laborers, Local 67	Ву:
Laborers, Local 304	Ву:
Operating Engineers, Local 3	Ву:
Plasterers, Local 66	Ву:
Roofers, Local 81	Ву:
Sheet Metal Workers, Local 104	Ву:
Sign Display, Local 510 Sprinkler Fitters, Local 483	Ву: Ву:
Teamsters, Local 853	Ву:
United Association of Journeymen and Apprentices Fitting Industry, Underground Utility & Landscape,	of the const
Local 266	Dan

Asbestos Workers Local 116	Ву:
Boilermakers Local 549	Ву:
Bricklayers & Allied Craftsmen Local 3	By: Ohon Salde
Northern California Carpenters Regional Council (on behalf of Carpenters, Local 713, Carpenters, Local 2236, Lathers, Local 68L, Millwrights, Local 102, Pile Drivers, Local 34)	Ву:
Cement Masons Local 300	Ву:
Electrical Workers, Local 595	Ву:
Elevator Constructors, Local 8	Ву:
Hod Carriers, Local 166	Ву:
Iron Workers, Local 378	Ву:
Laborers, Local 67	Ву:
Laboreris, Local 304	Ву:
Operating Engineers, Local 3	Ву:
Plasterers, Local 66	Ву:
Roofers, Local 81	Ву:
Sheet Metal Workers, Local 104	Ву:
Sign Display, Local 510 Sprinkler Fitters, Local 483	By:
Teamsters, Lotal 1353	Ву:
United Association of Journeymen and Apprentices Fitting Industry, Underground Utility & Landscape, Local 365	Ву:

# Signatory Unions By: Mary Plubell Asbestos Workers Local 116 **Bollermakers Local 549** By: \_\_\_\_\_ Bricklayers & Allied Craftsmen Local 3 Northern California Carpenters Regional Council (on behalf of Carpenters, Local 713, Carpenters, Local 2236, Lathers, Local 68L, Millwrights, Local 102, Pile Drivers, Local 34) **Cement Masons Local 300** By: \_\_\_\_\_ Electrical Workers, Local 595 By: \_\_\_ Elevator Constructors, Local 8 Ву: \_\_\_\_\_ Hod Carriers, Local 166 By: \_\_\_\_ Iron Workers, Local 378 Ву: \_\_\_\_ Laborers, Local 67 Laborers, Local 304 Operating Engineers, Local 3 Plasterers, Local 66 Roofers, Local 81 Sheet Metal Workers, Local 104 Sign Display, Local 510 Sprinkler Fitters, Local 483 Teamsters, Local 853

United Association of Journeymen and Apprentices Fitting Industry, Underground Utility & Landscape,

Local 355

Asbestos Workers Logal 116	By: M
	Malla la
Bollermakers Local 549	By: The State
Brickieyers & Allied Craftsmen Local 3	Ву:
Northern California Corpenters Regional Council (on behalf of Carpenters, Local 713, Carpenters, Local 2238, Lathers, Local 68L, Millwrights, Local 102,	
Pile Drivers, Local 34)	Ву:
Cement Masons Local 300	Ву:
Electrical Workers, Local 595	Ву:
Elevator Constructors, Local B	Ву:
Hod Carriers, Local 166	Ву:
Iron Workers, Local 378	Ву:
Laborers, Local 67	Ву:
Laborera, Local 304	Ву:
Operating Engineers, Local 3	Ву:
Plasterers, Local 66	Ву:
Receiers, Local 81	Ву:
Sheet Metal Workers, Local 104	Ву:
Sign Display, Local 510 Sprinkler Fitters, Local 483	By:
Teamsters, Local 853	By:
United Association of Journeymen and Approvinces Fitting Industry, Underground Utility & Landscape,	Bur

Asbestos Workers Local 116	Ву:
Bollermakers Local 549	Ву:
Bricklayers & Allied Craftsmen Local 3	Ву:
Northern California Carpenters Regional Council (on behalf of Carpenters, Local 713, Carpenters, Local 2236, Lathers, Local 68L, Millwrights, Local 102,	
Pile Drivers, Local 34)	By:
Cement Masons Local 300	By: Keith flits
Electrical Workers, Local 595	Ву:
Elevator Constructors, Local 8	Ву:
Hod Carriers, Local 166	Ву:
Iron Workers, Local 378	Ву:
Laborers, Local 67	Ву:
Laborers, Local 304	Ву:
Operating Engineers, Local 3	Ву:
Plasterers, Local 66	Ву:
Roofers, Local 81	Ву:
Sheet Metal Workers, Local 104	Ву:
Sign Display, Local 510 Sprinkler Fitters, Local 483	By:
Teamsters, Local 853	Ву:
United Association of Journeymen and Apprentices Fitting Industry, Underground Utility & Landscape,	Rus

Asbestos Workers Local 116	Ву:
Bollermakers Local 549	Ву:
Bricklayers & Alited Craftsmen Local 3	Ву:
Northern California Carpenters Regional Council (on behalf of Carpenters, Local 713, Carpenters, Local 2236, Lathers, Local 68L, Milwrights, Local 102, Pile Drivers, Local 34)	By:
Cement Masons Local 300	Ву:
Electrical Workers, Local 595	Ву:
Elevator Constructors, Local 8	Ву:
Hod Carriers, Local 166	By: HI wall
Iron Workers, Local 378	By: Att
Laborers, Local 67	Ву:
Laborers, Local 304	Ву:
Operating Engineers, Local 3	Ву:
Plasterers, Local 66	Ву:
Roofers, Local 81	Ву:
Sheet Metal Workers, Local 104	Ву:
Sign Display, Local 510 Sprinkler Fitters, Local 483	By:
Teamsters, Local 853	Ву:
United Association of Journeymen and Apprentices Fitting Industry, Underground Utility & Landscape, Local 355	By:

Asbestos Workers Local 116	Ву:
Boilermakers Local 549	Ву:
Bricklayers & Aliled Craftsmen Local 3	Ву:
Northern California Carpenters Regional Council (on behalf of Carpenters, Local 713, Carpenters, Local 2236, Lathers, Local 68L, Millwrights, Local 102, Pile Drivers, Local 34)	Ву:
Cement Masons Local 300	Ву:
Electrical Workers, Local 595	Ву:
Elevator Constructors, Local 8	Ву:
Hod Carriers, Local 168	Ву:
Iron Workers, Local 378	Ву:
Laborers, Local 67	By: Vish
Laborers, Local 304	Ву:
Operating Engineers, Local 3	Ву:
Plasterers, Local 66	Ву:
Roofers, Local 81	Ву:
Sheet Metal Workers, Local 104	Ву:
Sign Display, Local 510 Sprinkler Fitters, Local 483	By:
Teamsters, Local 853	Ву:
United Association of Journeymen and Apprentices Fitting Industry, Underground Utility & Landscape, Local 356	Ву:

Cement Masons Local 300	Ву:
Electrical Workers, Local 595	Ву:
Elevator Constructors Local 8	Ву:
Hod Carriers Local 166	ву:
Iron Worker, Local 378	Ву:
Laborers, Local 67	Ву:
Labors, Local 304	Ву:
Operating Engineers, Local 3	Ву:
Plasterers, Local 66	Ву:
Roofers, Local 81	Ву:
Sheet Metal Workers, Local 104	By:
Sign Display, Local 510	Ву:
Sprinkler Fitters, Local 483 Teamsters, Local 853	By:
United Association of Journeymen and Apprentices, Fitting Industry, Underground Utility and Landscape Local 355	Ву:
United Association of Steamfitters Pipelitters, Plumbers, & Gas Fitters, Local 342	By: Che Timm
District Council No. 16, Northern California International Union of Painters and Allied Trades (on behalf of Auto and Marine Painters Local 1176, Carpet and Linoleum Layers Local 12, Glaziers, Architectural Metal and Glassworkers Local 169, Painters	
and Tapers Local 3)	Ву:

Cement Masons Local 300	Ву:
Electrical Workers, Local 595	Ву:
Elevator Constructors Local 8	Ву:
Hod Carriers Local 166	Ву:
Iron Worker, Local 378	Ву:
Laborers, Local 67	Ву:
Labors, Local 304	Ву:
Operating Engineers, Local 3	Ву:
Plasterers, Local 66	Ву:
Roofers, Local 81	Ву:
Sheet Metal Workers, Local 104	Ву:
Sign Display, Local 510	Ву:
Sprinkler Fitters, Local 483 Teamsters, Local 853	By: Straly W. Smith
United Association of Journeymen and Apprentices, Fitting industry, Underground Utility and Landscape Local 355	Ву:
United Association of Steamfitters Pipefitters, Plumbers, & Gas Fitters, Local 342	Ву:
District Council No. 16, Northern California International Union of Painters and Allied Trades (on behalf of Auto and Marine Painters Local 1176, Carpet and Linoleum Layers Local 12, Glaziers, Architectural Metal and Glassworkers Local 169, Painters	
and Tapers Local 3)	Ву:

Cement Masons Local 300	Ву:
Electrical Workers, Local 595	Ву:
Elevator Constructors Local 8	Ву:
Hod Carriers Local 166	Ву:
Iron Worker, Local 378	Ву:
Laborers, Local 67	Ву:
Labors, Local 304	Ву:
Operating Engineers, Local 3	By: Gussel & Quin
Plasterers, Local 66	Ву:
Roofers, Local 81	Ву:
Sheet Metal Workers, Local 104	Ву:
Sign Display, Local 510	Ву:
Sprinkler Fitters, Local 483 Teamsters, Local 853	By:
United Association of Journeymen and Apprentices, Fitting Industry, Underground Utility and Landscape Local 355	Ву:
United Association of Steamfitters Pipefitters, Plumbers, & Gas Fitters, Local 342	Ву:
District Council No. 16, Northern California International Union of Painters and Allied Trades (on behalf of Auto and Marine Painters Local 1176, Carpet and Linoleum Layers Local 12, Glaziers, Architectural Metal and Glassworkers Local 169, Painters and Tapers Local 3)	Ву:

Cement Masons Local 300	Ву:
Electrical Workers, Local 595	Ву:
Elevator Constructors Local 8	Ву:
Hod Carriers Local 166	Ву:
Iron Worker, Local 378	Ву:
Laborers, Local 67	Ву:
Labors, Local 304	Ву:
Operating Engineers, Local 3	Ву:
Plasterers, Local 66	Ву:
Roofers, Local 81	Ву:
Sheet Metal Workers, Local 104	By:
Sign Display, Local 510	By: Joseph B. Tobach
Sprinkler Fitters, Local 483 Teamsters, Local 853	By:
United Association of Journeymen and Apprentices, Fitting Industry, Underground Utility and Landscape Local 355	By:
United Association of Steamfitters Pipefitters, Plumbers, & Gas Fitters, Local 342	By:
District Council No. 16, Northern California International Union of Painters and Allied Trades (on behalf of Auto and Marine Painters Local 1176, Carpet and Linoleum Layers Local 12, Glaziers, Architectural Metal and Glaziers Local 169, Painters	

Page 36 of 40
County of Alamoda
Countywide Project Stabilization/ Community Benefits Agreement

Cement Masons Local 300	By:
Electrical Workers, Local	Ву:
595 Elevator Constructors	By:
Local 8 Hod Carriers Local	D
166	Ву:
Iron Worker, Local 378	
· Laborers, Local 67	By: By:
Labors, Local 304	By:
Operating Engineers, Local 3	
Plasterers, Local 66	By:
Roofers, Local 81	By.
Sheet Metal Workers, Local 104	Den
Sign Display, Local 510	By:
Sprinkler Fitters, Local 483 Teamsters, Local 853	I The state of the
United Association of Journeymen and Apprentices, Fitting Industry, Underground Utility and Landscape Local 355	By: Migueldia
United Association of Steamfitters Pipefitters, Plumbers, & Gas Fitters, Local 342	Ву:
District Council No. 16, Northern California International Union of Painters and Allied Trades (on behalf of Auto and Marine Painters Local 1176, Carpet and Linoleum Layers Local 12, Glaziers, Architectural Metal and Glassworkers Local 169, Painters and Tapers Local 3)	Ву:
Ву:	
By:	
Ву:	Page 36 of 40 County of Alameda

Countywide Project Stabilization/ Community Benefits Agreement

## EXHIBIT A (Letter of Assent)

#### PROJECT STABILIZATION/COMMUNITY BENEFITS AGREEMENT

#### For the

# COUNTY OF ALAMEDA CONTRACTOR AGREEMENT TO BE BOUND

The undersigned, as a Contractor or Subcontractor (CONTRACTOR) on the County of Alameda, (hereinafter PROJECTS), for and in consideration of the award to it of a contract to perform work on said PROJECTS, and in further consideration of the mutual promises made in the "Project Stabilization/Community Benefits Agreement for the County of Alameda Project" (hereinafter AGREEMENT), a copy of which was received and is acknowledged, hereby:

- (l) Accepts and agrees to be bound by the terms and conditions of the AGREEMENT, together with any and all amendments and supplements now existing or which are later made thereto:
- (2) The CONTRACTOR agrees to be bound by the legally established local trust agreements as set

  Forth in Article 14 of this AGREEMENT.
- (3) The CONTRACTOR authorizes the parties to such local trust agreements to appoint trustees and successor trustees to administer the trust funds and hereby ratifies and accepts the trustees so appointed as if made by the CONTRACTOR;
- (4) Certifies that it has no commitments or agreements which would preclude its full and complete compliance with the terms and conditions of said AGREEMENT.
- (5) Agrees to secure from any CONTRACTORS (as defined in said AGREEMENT) which is or becomes a Subcontractor (of any tier) to it, a duly executed Agreement to be bound in form identical to this document.

Dated:	
	(Name of Contractor)
(Name of Prime Contractor or Higher Level Subcontractor)	(Authorized Officer & Title)
CA Number	(Address)
Contract or Project#	(Phone) (Fax)

# MEMORANDUM OF UNDERSTANDING

# COUNTY OF ALAMEDA PROJECT STABILIZATION/COMMUNITY BENEFIT AGREEMENT

Notwithstanding any provision to the contrary in the County of Alameda Project Stabilization/Community Benefit Agreement ("Project Stabilization Agreement"), this memorandum will confirm that work covered by the Project Stabilization/Community Benefits Agreement within the craft jurisdiction of the Elevator Constructors will be performed under the terms of the National Agreement of the International Union of Elevator Constructors, except that Articles 6,11 and 23 of the Project Stabilization Agreement will apply to such work.

COUNTY OF ALAMEDA	BUILDING AND CONSTRUCTION COUNCIL OF ALAMEDA COUNCIL
By: Board of Supervisors	By: Andreas Quiver, Secretary
Date:JUL 1 6 2019	
APPROVED AS TO FORM: DONNA R. ZIEGLER, COUNTY COUNSEL  Decublinged by:  By  Rulry Bramsin	
Name: Audrey Beaman	
Title; Deputy County Counsel	

reasurer

#### MEMORANDUM OF UNDERSTANDING

# COUNTY OF ALAMEDA PROJECT STABILIZATION/COMMUNITY BENEFIT AGREEMENT

In the event the County decides to cover any given Project with an Owner Controlled Insurance Program (OCIP) during the life of the PS/CBA, the following language will apply:

The County intends to implement an OCIP, or wrap up insurance, on the Project. All Contractors and employees performing work on the Project, and not otherwise excluded from the OCIP; will be bound by the requirement of the OOP Safety Manual; provided however, discipline imposed for alleged violations of the OCIP Safety Manual is subject to the Grievance procedures in Article 23. Any drug testing protocol established by the Contractor for the Project shall satisfy the requirements of the OCIP Safety Manual and be consistent with the MLAs. In the event that there is a conflict between the MLAs and the OCIP requirements, the OCIP requirements shall prevail.

COUNTY OF ALAMEDA	BUILDING AND CONSTRUCTION TRADES
By: President, Board of Supervisors	By: Andreas Clover, Secretary Treasurer
Date:JUL 1 6 2019	
APPROVED AS TO FORM: DONNA R. ZIEGLER, COUNTY COUNSEL  Docublighed by: By:  Rudry Braman	
Name: Audrey Beaman Title: Deputy County Counsel	

#### MEMORANDUM OF UNDERSTANDING

# COUNTY OF ALAMEDA PROJECT STABILIZATION/COMMUNITY BENEFIT AGREEMENT

The parties agree that the PLA to which this letter is addended shall not apply to any contracts entered into by the Alameda County Public Works Agency for a period of one year from the effective date of the PLA, except that any and all trucking, as described in Article 5.4 of said PLA, shall be covered by the PLA.

Any disputes concerning the interpretation and or application of this side letter shall be subject to the dispute resolution process set forth in Article 23 of the PLA.

COUNTY OF ALAMEDA	BUILDING AND CONSTRUCTION TRADE
By: Board of Supervisors	By: Andreas Cluyer, Secretary Treasurer
Date: JUL 1 6 2019	Date: 8/30/18
APPROVED AS TO FORM: DONNA R. ZIEGLER, COUNTY COUNSEL	WH I H 2019
By: Crumy Byman	
Deputy County Counsel	

## **DOCUMENT 00 73 56**

# HAZARDOUS MATERIALS PROCEDURES & REOUIREMENTS

## 1. Summary

This document includes information applicable to hazardous materials and hazard waste abatement.

# 2. Notice of Hazardous Waste or Materials Conditions

- 2.1. Contractor shall give notice in writing, including by e-mail, to the County, the Construction Manager, and the Architect promptly, before any of the following conditions are disturbed, and in no event later than twenty-four (24) hours after first observance, of any:
  - 2.1.1. Material that Contractor believes may be material that is hazardous waste or hazardous material, as defined in section 25117 or 25260 of the Health and Safety Code, that is required to be removed to a Class I, Class II, or Class III disposal site in accordance with provisions of existing law;
  - 2.1.2. Other material that may present a substantial danger to persons or property exposed thereto in connection with Work at the site.
- 2.2. Contractor's written notice shall indicate whether the hazardous waste or material was shown or indicated in the Contract Documents to be within the scope of Work, and whether the materials were brought to the site by Contractor, its Subcontractors, suppliers, or anyone else for whom Contractor is responsible. As used in this section the term "hazardous materials" shall include, without limitation, asbestos, lead, mercury, Polychlorinated biphenyl (PCB), petroleum and related hydrocarbons, and radioactive material.
- 2.3. In response to Contractor's written notice, the County shall investigate the identified conditions.
- 2.4. If the County determines that conditions do not involve hazardous materials or that no change in terms of Contract is justified, the County shall so notify Contractor in writing, stating reasons. If the County and Contractor cannot agree on whether conditions justify an adjustment in Contract Price or Contract Times, or on the extent of any adjustment, Contractor shall proceed with the Work as directed by the County.
- 2.5. If after receipt of notice from the County, Contractor does not agree to resume Work based on a reasonable belief it is unsafe, or does not agree to resume Work under special

ALAMEDA COUNTY GSA-CP

HAZARDOUS MATERIALS PROCEDURES & REQUIREMENTS DOCUMENT 00 73 56

conditions, then County may order such portion of Work that is in connection with such hazardous condition or such affected area to be deleted from the Work, or performed by others, or County may invoke its rights to terminate the Contract in whole or in part. County will determine entitlement to or the amount or extent of an adjustment, if any, in Contract Price or Contract Times as a result of deleting such portion of Work, or performing the Work by others.

2.6. If Contractor stops Work in connection with any hazardous condition and in any area affected thereby, Contractor shall immediately redeploy its workers, equipment, and materials, as necessary, to other portions of the Work to minimize delay and disruption.

# 3. Additional Warranties and Representations

- 3.1. Contractor represents and warrants that it, its employees, and its subcontractors and their employees, shall at all times have the required levels of familiarity with the Site and the Work, training, and ability to comply fully with all applicable law and contract requirements for safe and expeditious performance of the Work, including whatever training is or may be required regarding the activities to be performed (including, but not limited to, all training required to address adequately the actual or potential dangers of Contract performance).
- 3.2. Contractor represents and warrants that it, its employees, and its subcontractors and their employees, shall at all times have and maintain in good standing any and all certifications and licenses required by applicable federal, state, and other governmental and quasi-governmental requirements applicable to the Work.
- 3.3. Contractor represents and warrants that it has studied carefully all requirements of the Specifications regarding procedures for demolition, hazardous waste abatement, or safety practices, specified in the Contract, and prior to submitting its bid, has either (a) verified to its satisfaction that the specified procedures are adequate and sufficient to achieve the results intended by the Contract Documents, or (b) by way of approved "or equal" request or request for clarification and written Addenda, secured changes to the specified procedures sufficient to achieve the results intended by the Contract Documents. Contractor accepts the risk that any specified procedure will result in a completed Project in full compliance with the Contract Documents.

# 4. Monitoring and Testing

4.1. County reserves the right, in its sole discretion, to conduct air monitoring, earth monitoring, Work monitoring, and any other tests (in addition to testing required under the agreement or applicable law), to monitor Contract requirements of safe and statutorily compliant work methods and (where applicable) safe re-entry level air standards under state and federal law upon completion of the job, and compliance of the

ALAMEDA COUNTY GSA-CP

- work with periodic and final inspection by public and quasi-public entities having jurisdiction.
- 4.2. Contractor acknowledges that County has the right to perform, or cause to be performed, various activities and tests including, but not limited to, pre-abatement, during abatement, and post-abatement air monitoring, that County shall have no obligation to perform said activities and tests, and that a portion of said activities and tests may take place prior to the completion of the Work by Contractor. In the event County elects to perform these activities and tests, Contractor shall afford County ample access to the Site and all areas of the Work as may be necessary for the performance of these activities and tests. Contractor will include the potential impact of these activities or tests by County in the Contract Price and the Scheduled Completion Date.
- 4.3. Notwithstanding County's rights granted by this paragraph, Contractor may retain its own industrial hygiene consultant at Contractor's own expense and may collect samples and may perform tests including, but not limited to, pre-abatement, during abatement, and post-abatement personal air monitoring, and County reserves the right to request documentation of all such activities and tests performed by Contractor relating to the Work and Contractor shall immediately provide that documentation upon request.

# 5. Compliance with Laws

- 5.1. Contractor shall perform safe, expeditious, and orderly work in accordance with the best practices and the highest standards in the hazardous waste abatement, removal, and disposal industry, the applicable law, and the Contract Documents, including, but not limited to, all responsibilities relating to the preparation and return of waste shipment records, all requirements of the law, delivering of all requisite notices, and obtaining all necessary governmental and quasi-governmental approvals.
- 5.2. Contractor represents that it is familiar with and shall comply with all laws applicable to the Work or completed Work including, but not limited to, all federal, state, and local laws, statutes, standards, rules, regulations, and ordinances applicable to the Work relating to:
  - 5.2.1. The protection of the public health, welfare and environment;
  - 5.2.2. Storage, handling, or use of asbestos, PCB, lead, petroleum based products or other hazardous materials;
  - 5.2.3. The generation, processing, treatment, storage, transport, disposal, destruction, or other management of asbestos, PCB, lead, petroleum, or hazardous waste materials or other waste materials of any kind; and

5.2.4. The protection of environmentally sensitive areas such as wetlands and coastal areas

## 6. Disposal

- 6.1. Contractor has the sole responsibility for determining current waste storage, handling, transportation, and disposal regulations for the job Site and for each waste disposal facility. Contractor must comply fully at its sole cost and expense with these regulations and any applicable law. County may, but is not obligated to, require submittals with this information for it to review consistent with the Contract Documents.
- 6.2. Contractor shall develop and implement a system acceptable to County to track hazardous waste from the Site to disposal, including appropriate "Hazardous Waste Manifests" on the EPA form, so that County may track the volume of waste it put in each landfill and receive from each landfill a certificate of receipt.
- 6.3. Contractor shall provide County with the name and address of each waste disposal facility prior to any disposal, and County shall have the express right to reject any proposed disposal facility. Contractor shall not use any disposal facility to which County has objected. Contractor shall document actual disposal or destruction of waste at a designated facility by completing a disposal certificate or certificate of destruction forwarding the original to the County.

# 7. Permits

- 7.1. Before performing any of the Work, and at such other times as may be required by applicable law, Contractor shall deliver all requisite notices and obtain the approval of all governmental and quasi-governmental authorities having jurisdiction over the Work. Contractor shall submit evidence satisfactory to County that it and any disposal facility
  - 7.1.1. have obtained all required permits, approvals, and the like in a timely manner both prior to commencement of the Work and thereafter as and when required by applicable law, and
  - 7.1.2. are in compliance with all such permits, approvals and the regulations.

For example, before commencing any work in connection with the Work involving asbestos-containing materials, or PCBs, or other hazardous materials subject to regulation, Contractor agrees to provide the required notice of intent to renovate or demolish to the appropriate state or federal agency having jurisdiction, by certified mail, return receipt requested, or by some other method of transmittal for which a return receipt is obtained, and to send a copy of that notice to County. Contractor shall not conduct any Work involving asbestos-containing materials or PCBs unless Contractor has first confirmed that the

ALAMEDA COUNTY GSA-CP

appropriate agency having jurisdiction is in receipt of the required notification. All permits, licenses, and bonds that are required by governmental or quasi-governmental authorities, and all fees, deposits, tap fees, offsite easements, and asbestos and PCB disposal facilities expenses necessary for the prosecution of the Work, shall be procured and paid for by Contractor. Contractor shall give all notices and comply with the all applicable laws bearing on the conduct of the Work as drawn and specified. If Contractor observes or reasonably should have observed that Plans and Specifications and other Contract Documents are at variance therewith, it shall be responsible for promptly notifying County in writing, including by e-mail, of such fact. If Contractor performs any Work contrary to applicable laws, it shall bear all costs arising therefrom.

7.2. In the case of any permits or notices held in County's name or of necessity to be made in County's name, County shall cooperate with Contractor in securing the permit or giving the notice, but the Contractor shall prepare for County review and execution upon approval, all necessary applications, notices, and other materials.

#### 8. Indemnification

8.1. To the extent permitted by law, the indemnities and limitations of liability expressed throughout the Contract Documents apply with equal force and effect to any claims or liabilities imposed or existing by virtue of the removal, abatement, and disposal of hazardous waste. This includes, but is not limited to, liabilities connected to the selection and use of a waste disposal facility, personal injury, property damage, loss of use of property, damage to the environment or natural resources, or "disposal" and "release" of materials associated with the Work (as defined in 42 U.S.C. § 960l et seq.).

## 9. Termination

9.1. County shall have an absolute right to terminate for default immediately without notice and without an opportunity to cure should Contractor knowingly or recklessly commit a material breach of the terms of the Contract Documents, or any applicable law, on any matter involving the exposure of persons or property to hazardous waste. However, if the breach of contract exposing persons or property to hazardous waste is due solely to an ordinary, unintentional, and non-reckless failure to exercise reasonable care, then the procedures for termination for cause shall apply without modification.

END OF DOCUMENT



## **ECOP GOALS**

#### **COMPLIANCE TABLE**

	ECOP Form 101A			
General Contract	tor Name:			
Project #		Project Name:		
_ :				

Project #						
Total Contract Amount (BI	D):					
ECOP GOALS: 60% Local (LBE FIRMS)						
Name (Prime or Subcontractor)	LBE Location (City)	License/Certification by	Lic Expiration Date mm/dd/yyyy	Work To Be Performed	Dollar Value of Contract	Auto Percent
				TOTAL LBE		

This form is required to provide LBE information for the project specified above.

LBE must have current and valid certification from acceptable agencies identified in Section 3.5 of the Supplemental Instruction to Bidders

DOCUMENT 00 22 19 on the date bids are opened OR meet the Local definition in Section 3.1. Attach Copy of all certification and current and valid

business license issued by the County or a city within the County.

THIS FORM MUST BE SUBMITTED BY 2:00 p.m., TWO BUSINESS DAYS FOLLOWING THE BID OPENING TO THE GENERAL SERVICES **AGENCY, 1401 LAKESIDE DRIVE,** 

**SUITE 907, OAKLAND, CALIFORNIA 94612** 

# **ECOP GOALS**

**COMPLIANCE TABLE Continued** ECOP Form 101B

ECOP GOALS: 20% Small (SLBE Firms)						
Name (Prime or Subcontractor)	SLBE Location (City)	Certified By*	SLBE Cert Expiration Date	Work To Be Performed	Dollar Value of Contract	Auto Percent
				TOTAL SLBE		

This form is required to provide SLBE information for the project specified above.

SLBE must have current and valid certification from acceptable agencies identified in Section 3.5 of the Supplemental Instruction to Bidders

DOCUMENT 00 22 19 on the date bids are opened. SLBEs must meet the Local definition in Section 3.1. Attach Copy of all certification and current and valid business license issued by the County or a city with the County.

THIS FORM MUST BE SUBMITTED BY 2:00 p.m., TWO BUSINESS DAYS FOLLOWING THE BID OPENING TO THE GENERAL SERVICES **AGENCY, 1401 LAKESIDE DRIVE, SUITE 907, OAKLAND, CALIFORNIA 94612** 

ECOP Form 102A

ECOP GOALS: 15% Minority (MBE FIRMS)						
Name (Must be Subcontractor)	MBE Location (City)	Certified By*	MBE Cert Expiration Date	Work To Be Performed	Dollar Value of Contract	Auto Percent
				TOTAL MBE		

This form is required to provide MBE information for the project specified above.

MBE must have current and valid certification from acceptable agencies identified in Section 3.2 of the Supplemental Instruction to Bidders

DOCUMENT 00 22 19 as of the date bids are opened. Attach MBE certifications to this form.

Subcontractor participation to include manufacturers, suppliers and truckers.

THIS FORM MUST BE SUBMITTED BY 2:00 p.m., TWO BUSINESS DAYS FOLLOWING THE REQUEST FROM GSA TO THE GENERAL **SERVICES AGENCY,** 

1401 LAKESIDE DRIVE, SUITE 907, OAKLAND, CALIFORNIA 94612

ECOP Form 102B

COMPLIANCE TABLE Continued ECOP Form 102B						
		ECOP GOALS: 5	5% Woman (WBE FIRI	MS)		
Name (Must be Subcontractor)	WBE Location (City)	Certified By*	WBE Cert Expiration Date	Work To Be Performed	Dollar Value of Contract	Auto Percent
				TOTAL WBE		
				. –		

This form is required to provide WBE information for the project specified above.

WBE must have current and valid certification from acceptable agencies identified in Section 3.2 of the Supplemental Instruction to Bidders

DOCUMENT 00 22 19 as of the date bids are opened. Attach WBE certifications to this form.

Subcontractor participation to include manufacturers, suppliers and truckers.

THIS FORM MUST BE SUBMITTED BY 2:00 p.m., TWO BUSINESS DAYS FOLLOWING THE REQUEST FROM GSA TO THE GENERAL **SERVICES AGENCY,** 

1401 LAKESIDE DRIVE, SUITE 907, OAKLAND, CALIFORNIA 94612

COOD EVITH EEEODT INSTRUCTIONS CHECKLIST			ECOP FORM 102C
GOOD FAITH EFFORT INSTRUCTIONS CHECKLIST  See County of Alameda Supplementary Instructions to Bidders	Section 00 22 19	for complete Go	ood Faith Effort requirements summarized below. See Required Good Faith Efforts for acceptable
documentation to support that GFEs (1-9 below) were made. In			
	Yes	No	
Attended pre-solicitation/pre-bid meetings held by GSA			If yes, note documentation:
2. Identified and selected items for performance by MWSLBE's			If yes, note documents submitted to support Items identified and selected:
3. Advertised not <10 calendar days before bids opened in publication specified by GSA per GSA specification			If yes, note publications and attach advertisement:
Written notice to number of MWSLBEs specified on bid specs			Note documentation and attach written notice(s) -
5. Follow-up initial solicitations of interest with MWSLBEs			If Yes, note documentation and attach proof of follow-up:
Provided interested MWSLBEs with information about plans, specs and requirements			If Yes, note documentation and attach proof of response -
7. Requested assistance from MWSLBE community organizations that provide assistance in recruitment and placement of MWSLBEs			If Yes, note documents and attach proof of requests (date, organization name, contact name, phone number, summary) -
Negotiated in good faith w/MWSLBEs and did not unjustifiably reject MWSBLE bids			If Yes, note documentation and attach proof of negotiations/acceptance -
Advised and made efforts to assist interested MWSLBEs in obtaining bonds, lines of credit or insurance			If Yes, note documents and attach proof of assistance -
	ACCEPTA	BLE DOCUMENT	ATION TO SUPPORT GFE'S
REQUIRED GOOD FAITH EFFORTS			EXAMPLES OF ACCEPTABLE DOCUMENTATION
<ol> <li>The bidder attended mandatory pre-solicitation or pre-bid meetings that were scheduled by the local agency to inform all bidders of the ECOP requirements for the project for which the contract will be awarded.</li> </ol>			Copy of pre-bid meeting sign in sheet (which is e-mailed to attendees and available on County Current Contracting Opportunities website listed below). The name of the firm must be listed. <a href="http://www.acgov.org/gsa_app/gsa/purchasing/bid_content/contractopportunities.jsp">http://www.acgov.org/gsa_app/gsa/purchasing/bid_content/contractopportunities.jsp</a>
The bidder identified and selected specific items of the particle.	oroject for which	the contract	Copy of advertisements, certified letters, successfully completed faxes and/or other notices.
will be awarded to be performed by M/W/S/LBEs to provide a by those enterprises.	n opportunity fo	r participation	to M/W/S/LBEs with selected specific items identified.
3. The bidder advertised, not less than ten (10) calendar days before the date the bids are opened, in one or more local daily or weekly newspapers, trade association publications, minority or trade oriented publications, trade journals, or other media specified by the local agency for M/W/S/LBEs that are interested in participating in the project.		<ul> <li>Copy of advertisements placed showing publication name and date, and dated receipts.</li> <li>Dated receipt with ad copy.</li> </ul>	
4. The bidder provided written notice of his or her interest in bidding on the contract to the number of M/W/S/LBEs required to be notified by the project specifications not less than ten (10) calendar days prior to the opening of bids. The minimum number of M/W/S/LBE firms required to be notified is three (3) for each item of the project selected to be performed by a M/W/S/LBE, where a M/W/S/LBE subcontractor has not been secured for that item. The bidder may utilize the list of certified minority or women business enterprises prepared by the Department of Transportation pursuant to Section 14030.5 of the Government Code and the list of local business enterprises in the on-line County Small Local Emerging Business (SLEB) Vendor Query System located at:  http://www.acgov.org/sleb_query_app/gsa/sleb/query/slebmenu.jsp		Copy of dated notice, complete distribution list(s) and evidence of distribution (proof of faxes, e-mails sent etc.)     Undelivered faxes do not count toward the effort to meet the minimum requirement     Trades and specialties, in addition to M/W/S/LBE designation, must be clearly identified to meet the minimum requirement by using certification letter or source documentation	
5. The bidder followed up initial solicitations of interest by condetermine with certainty whether the enterprises were interest of the project.	-		Successfully completed telephone log containing specific dates, name of caller, person contacted and comments (i.e., why not bidding, information sent to/date, etc.)
The bidder provided interested M/W/S/LBEs with information about the plans, specifications, and requirements for the selected subcontracting or material supply work.			<ul> <li>Copy of published advertisements, letters, successfully completed faxes, etc. with M/W/S/LBE name/contact information including the required information or directions on how to obtain it and the date the information was provided</li> <li>Agenda, meeting notes, etc. including specific topics discussed, M/W/S/LBE firm names and contact persons in attendance that received information, and the location and date information was provided</li> </ul>
7. The bidder requested assistance from local and small business and minority and women community organizations; local and small, minority and women contractor groups, local, state, or federal M/W/S/LBE assistance offices, or other organizations that provide assistance in recruitment and placement of M/W/S/LBEs.			<ul> <li>Copy of dated written request and response (letter, successfully completed fax, e-mail, etc.) or 2nd written request to follow-up, if needed. Contractor must attempt to contact at least two organizations.</li> <li>Phone log is not acceptable.</li> </ul>
8. The bidder negotiated in good faith with the M/W/S/LBEs and did not unjustifiably reject as unsatisfactory bids prepared by any M/W/S/LBEs as determined by GSA.			<ul> <li>Copies or list of all bids and a spreadsheet listing all bids with firm name, contact person, bid items(s), bid price, M/W/S/LBE classification, and comments re: selection or rejection</li> <li>M/W/S/LBE bids accepted and included in bid response</li> </ul>
in obtaining bonds, lines of credit, or insurance required by either the GSA or the contractor.			Copy of advertisements or other notices with specifics referencing willingness to assist M/W/S/LBEs Agenda, meeting notes including presenter's name and title, specific topics discussed, hand outs etc., name of M/W/S/LBE firms in attendance, contact persons who received advice, location and date advice was provided  Revised 10/6/10





Report for:

Eric Dyer, Ms. Karin Schroeter Terracon Consultants, Inc. - Emeryville 1466 66th Street Emeryville, CA 94608

Regarding: Project: R1197G03; Highland Hospital/1411 East 31st St./Oakland, CA., K. Bldg.

EML ID: 2403934

Approved by:

Dates of Analysis: Asbestos PLM: 05-14-2020

Approved Signatory Amin Suliman

Service SOPs: Asbestos PLM (EPA 40CFR App E to Sub E of Part 763 & EPA METHOD 600/R-93-116, SOP EM-AS-S-1267) NVLAP Lab Code 200728-0

All samples were received in acceptable condition unless noted in the Report Comments portion in the body of the report. The results relate only to the samples as received. The results include an inherent uncertainty of measurement associated with estimating percentages by polarized light microscopy. Measurement uncertainty data for sample results with >1% asbestos concentration can be provided when requested.

Eurofins EMLab P&K ("the Company") shall have no liability to the client or the client's customer with respect to decisions or recommendations made, actions taken or courses of conduct implemented by either the client or the client's customer as a result of or based upon the Test Results. In no event shall the Company be liable to the client with respect to the Test Results except for the Company's own willful misconduct or gross negligence nor shall the Company be liable for incidental or consequential damages or lost profits or revenues to the fullest extent such liability may be disclaimed by law, even if the Company has been advised of the possibility of such damages, lost profits or lost revenues. In no event shall the Company's liability with respect to the Test Results exceed the amount paid to the Company by the client therefor.

Lab ID-Version †: 11464497-1

6000 Shoreline Ct, Ste 205, So. San Francisco, CA 94080

(866) 888-6653 Fax (623) 780-7695 www.emlab.com

Date of Sampling: 05-11-2020 Date of Receipt: 05-11-2020 Date of Report: 05-14-2020

Client: Terracon Consultants, Inc. - Emeryville C/O: Eric Dyer, Ms. Karin Schroeter

Re: R1197G03; Highland Hospital/1411 East 31st St./ Oakland, CA., K. Bldg.

### ASBESTOS PLM REPORT

**Total Samples Submitted:** 22

**Total Samples Analyzed:** 22

**Total Samples with Layer Asbestos Content > 1%:** 0

Location: 1A, Stucco; level 5 G5 column	Lab ID-Version‡: 11464496-1
Sample Layers	Asbestos Content
Gray Stucco	ND
Sample Composite Homogeneity:	Poor

Location: 1B. Stucco: level 5 GE column

Escation: 1B, Stuces, level 5 GE column	Eur IP Version 4. 11 10 1157 1
Sample Layers	Asbestos Content
Gray Stucco	ND
Sample Composite Homogeneity:	Poor

Location: 1C, Stucco; level 4 E1 wall	Lab ID-Version‡: 11464498-1
Sample Layers	Asbestos Content
Gray Stucco	ND
Sample Composite Homogeneity:	Poor

Location: 1D Stucco: level 4 A1 7 wall

Location: 1D, Stucco; level 4 A1.7 wall	Lab ID-Version‡: 11464499-1
Sample Layers	Asbestos Content
Gray Stucco	ND
Sample Composite Homogeneity:	Poor

The test report shall not be reproduced except in full, without written approval of the laboratory. The report must not be used by the client to claim product certification, approval, or endorsement by any agency of the federal government. Eurofins EMLab P&K reserves the right to dispose of all samples after a period of thirty (30) days, according to all state and federal guidelines, unless otherwise specified.

Inhomogeneous samples are separated into homogeneous subsamples and analyzed individually. ND means no fibers were detected. When detected, the minimum detection and reporting limit is less than 1% unless point counting is performed. Floor tile samples may contain large amounts of interference material and it is recommended that the sample be analyzed by gravimetric point count analysis to lower the detection limit and to aid in asbestos identification.

 $\ddagger$  A "Version" indicated by -"x" after the Lab ID# with a value greater than 1 indicates a sample with amended data. The revision number is reflected by the value of "x".

EMLab P&K, LLC EMLab ID: 2403934, Page 2 of 7

Lab ID-Version †: 11464500-1

Lab ID-Version : 11464501-1

Lab ID-Version : 11464502-1

6000 Shoreline Ct, Ste 205, So. San Francisco, CA 94080

(866) 888-6653 Fax (623) 780-7695 www.emlab.com

Date of Sampling: 05-11-2020 Date of Receipt: 05-11-2020

Date of Report: 05-14-2020

Client: Terracon Consultants, Inc. - Emeryville C/O: Eric Dyer, Ms. Karin Schroeter

Re: R1197G03; Highland Hospital/1411 East 31st St./

Oakland, CA., K. Bldg.

### ASBESTOS PLM REPORT

Location: 1E. Stucco: level 6 C1.6.7 wall

Location: 12, Staces, level o Clist, wan	
Sample Layers	Asbestos Content
Gray Stucco	ND
Sample Composite Homogeneity: Poor	

Location: 1F. Stucco: level 6 F5 wall

Sample Layers	Asbestos Content
Gray Stucco	ND
Sample Composite Homogeneity:	Poor

Location: 1G, Stucco; level 8 roof B1 5.7 wall

Sample Layers	Asbestos Content
Gray Stucco	ND
Sample Composite Homogeneity:	Poor

Location: 1H, Stucco; level 8 roof C3 waLL	Lab ID-Version‡: 11464503-1
Sample Layers	Asbestos Content
Gray Stucco	ND
Sample Composite Homogeneity:	Poor

The test report shall not be reproduced except in full, without written approval of the laboratory. The report must not be used by the client to claim product certification, approval, or endorsement by any agency of the federal government. Eurofins EMLab P&K reserves the right to dispose of all samples after a period of thirty (30) days, according to all state and federal guidelines, unless otherwise specified.

Inhomogeneous samples are separated into homogeneous subsamples and analyzed individually. ND means no fibers were detected. When detected, the minimum detection and reporting limit is less than 1% unless point counting is performed. Floor tile samples may contain large amounts of interference material and it is recommended that the sample be analyzed by gravimetric point count analysis to lower the detection limit and to aid in asbestos identification.

6000 Shoreline Ct, Ste 205, So. San Francisco, CA 94080

(866) 888-6653 Fax (623) 780-7695 www.emlab.com

Date of Sampling: 05-11-2020 Date of Receipt: 05-11-2020 Date of Report: 05-14-2020

Client: Terracon Consultants, Inc. - Emeryville C/O: Eric Dyer, Ms. Karin Schroeter

Re: R1197G03; Highland Hospital/1411 East 31st St./

Oakland, CA., K. Bldg.

## ASBESTOS PLM REPORT

Location: 1I, Stucco; level 8 roof F2 wall

Location: 1I, Stucco; level 8 roof F2 wall	Lab ID-Version‡: 11464504-1
Sample Layers	Asbestos Content
Gray Stucco	ND
Sample Composite Homogeneity: Poor	

Location: 1J, Stucco; level 8 roof E5 wall

Sample Layers	Asbestos Content
Gray Stucco	ND
Sample Composite Homogeneity: Poor	

Location: 2A, Exterior white window sealant (frame to building); level 5 G5 column

Lab ID-Version : 11464506-1

Lab ID-Version : 11464505-1

Sample Layers	Asbestos Content
White Sealant	ND
Sample Composite Homogeneity: Moderate	

Location: 2B, Exterior white window sealant (frame to building); level 4 A.5.1 wall

Lab ID-Version : 11464507-1

Sample Layers	Asbestos Content
White Sealant	ND
Sample Composite Homogeneity: Moderate	

The test report shall not be reproduced except in full, without written approval of the laboratory. The report must not be used by the client to claim product certification, approval, or endorsement by any agency of the federal government. Eurofins EMLab P&K reserves the right to dispose of all samples after a period of thirty (30) days, according to all state and federal guidelines, unless otherwise specified.

Inhomogeneous samples are separated into homogeneous subsamples and analyzed individually. ND means no fibers were detected. When detected, the minimum detection and reporting limit is less than 1% unless point counting is performed. Floor tile samples may contain large amounts of interference material and it is recommended that the sample be analyzed by gravimetric point count analysis to lower the detection limit and to aid in asbestos identification.

6000 Shoreline Ct, Ste 205, So. San Francisco, CA 94080

(866) 888-6653 Fax (623) 780-7695 www.emlab.com

Client: Terracon Consultants, Inc. - Emeryville C/O: Eric Dyer, Ms. Karin Schroeter

Re: R1197G03; Highland Hospital/1411 East 31st St./

Oakland, CA., K. Bldg.

Date of Sampling: 05-11-2020 Date of Receipt: 05-11-2020 Date of Report: 05-14-2020

#### ASBESTOS PLM REPORT

Location: 2C, Exterior white window sealant (frame to building); level 6 F5 wall

Lab ID-Version : 11464508-1

Sample Layers	Asbestos Content
White Sealant	ND
Sample Composite Homogeneity: Moderate	

Location: 3A, Exterior black window sealant (frame to glass); level 5 G5 column

Lab ID-Version 11464509-1

Sample Layers	Asbestos Content
Black Sealant	ND
Sample Composite Homogeneity:	Poor

Location: 3B, Exterior black window sealant (frame to glass); level 5 A.5 1 wall

Lab ID-Version : 11464510-1

Sample Layers	Asbestos Content
Black Sealant	ND
Sample Composite Homogeneity: Poor	

## Location: 3C, Exterior black window sealant (frame to glass); level 6 f5 WALL

Lab ID-Version‡: 11464511-1

Sample Layers	Asbestos Content
Black Sealant	ND
Sample Composite Homogeneity:	Poor

The test report shall not be reproduced except in full, without written approval of the laboratory. The report must not be used by the client to claim product certification, approval, or endorsement by any agency of the federal government. Eurofins EMLab P&K reserves the right to dispose of all samples after a period of thirty (30) days, according to all state and federal guidelines, unless otherwise specified.

Inhomogeneous samples are separated into homogeneous subsamples and analyzed individually. ND means no fibers were detected. When detected, the minimum detection and reporting limit is less than 1% unless point counting is performed. Floor tile samples may contain large amounts of interference material and it is recommended that the sample be analyzed by gravimetric point count analysis to lower the detection limit and to aid in asbestos identification.

Lab ID-Version : 11464512-1

Lab ID-Version 1: 11464514-1

Lab ID-Version 1: 11464515-1

6000 Shoreline Ct, Ste 205, So. San Francisco, CA 94080

(866) 888-6653 Fax (623) 780-7695 www.emlab.com

Client: Terracon Consultants, Inc. - Emeryville C/O: Eric Dyer, Ms. Karin Schroeter

Re: R1197G03; Highland Hospital/1411 East 31st St./

Oakland, CA., K. Bldg.

Date of Sampling: 05-11-2020 Date of Receipt: 05-11-2020 Date of Report: 05-14-2020

#### ASBESTOS PLM REPORT

Location: 4A, Beige elastomeric paint; vallecito place wall

Sample Layers	Asbestos Content
Beige Paint	ND
Sample Composite Homogeneity:	Poor

Location: 4B, Beige elastomeric paint; vallecito place w	all Lab ID-Version‡: 11464513-1
Sample Layers	Asbestos Content
Beige Paint	ND
Sample Composite Homogeneity:	Poor

Location: 4C, Beige elastomeric paint; 14th avenue wall

Sample Layers	Asbestos Content
Beige Paint	ND
Sample Composite Homogeneity:	Poor

Location: 5A. Decorative plaster with grout; vallectito/14th column

Sample Layers	Asbestos Content
Yellow Plaster	ND
Grout	ND
Sample Composite Homogeneity: Poor	

The test report shall not be reproduced except in full, without written approval of the laboratory. The report must not be used by the client to claim product certification, approval, or endorsement by any agency of the federal government. Eurofins EMLab P&K reserves the right to dispose of all samples after a period of thirty (30) days, according to all state and federal guidelines, unless otherwise specified.

Inhomogeneous samples are separated into homogeneous subsamples and analyzed individually. ND means no fibers were detected. When detected, the minimum detection and reporting limit is less than 1% unless point counting is performed. Floor tile samples may contain large amounts of interference material and it is recommended that the sample be analyzed by gravimetric point count analysis to lower the detection limit and to aid in asbestos identification.

Lab ID-Version‡: 11464516-1

Lab ID-Version 1: 11464517-1

6000 Shoreline Ct, Ste 205, So. San Francisco, CA 94080

(866) 888-6653 Fax (623) 780-7695 www.emlab.com

Date of Sampling: 05-11-2020 Date of Receipt: 05-11-2020 Date of Report: 05-14-2020

Client: Terracon Consultants, Inc. - Emeryville C/O: Eric Dyer, Ms. Karin Schroeter

Re: R1197G03; Highland Hospital/1411 East 31st St./

Oakland, CA., K. Bldg.

### ASBESTOS PLM REPORT

Location: 5B, Decorative plaster with grout; vallectito/14th column

Sample Layers	Asbestos Content
Yellow Plaster	ND
Grout	ND
Sample Composite Homogeneity: Poor	

Location: 5C. Decorative plaster with grout: vallectito/14th column

200mion 0 0, 2 000 mix o primately (10m ground, 1 min outside)	
Sample Layers	Asbestos Content
Yellow Plaster	ND
Grout	ND
Sample Composite Homogeneity: Poor	

The test report shall not be reproduced except in full, without written approval of the laboratory. The report must not be used by the client to claim product certification, approval, or endorsement by any agency of the federal government. Eurofins EMLab P&K reserves the right to dispose of all samples after a period of thirty (30) days, according to all state and federal guidelines, unless otherwise specified.

Inhomogeneous samples are separated into homogeneous subsamples and analyzed individually. ND means no fibers were detected. When detected, the minimum detection and reporting limit is less than 1% unless point counting is performed. Floor tile samples may contain large amounts of interference material and it is recommended that the sample be analyzed by gravimetric point count analysis to lower the detection limit and to aid in asbestos identification.

 $\ddagger$  A "Version" indicated by -"x" after the Lab ID# with a value greater than 1 indicates a sample with amended data. The revision number is reflected by the value of "x".

EMLab P&K, LLC

## Terracon

***E-MAIL R  PM - S. Stein spsteiner@terracon	.com kmschroeter@terracon.com kmpilgrim@terracon.co	om PLM Analysis (Analyze all samples)
PM- M. Benefic		
□PM- D. Block  David block@terrace		nt d thill d
roject Name/ Add	ress/ Building No. Tigh and tospita	East 3/st Speel Convent
roject# R119	7603 Sampled By: John ARKANO	Sampling Date: 5/11/20 K Blo
ample(s) sent to:	□ MAL □ ASB TEM □ EMLAB □ Other	
AT Rush	☐ 24HRS ☐ 48HR ☐ 3-5 days	
HM#	Material Description Styce a	
Sample ID	Sample Location & Material Location	Quantity:
IA	Level 5 65 Colymn	
IR	Level 5 El Colymn	
10	Level 4 E. Wall	
HM#	Material Description:	
Sample ID	Sample Location & Material Location	Quantity:
17	Level 4 Alo7 Wall	
	Later to the contract of the c	
16	Level 6 F3 Wall	
HM#	Material Description: Sample Location & Material Description	Quantity:
Sample ID	1 0 0 F DI [7 1/4]	
16	11 1 2 4 5 6 2 1.///	
1	Level 8 Roof C3 Wally	
-	Level 8 Kut +2 Wall	
HM#	Material Description:	Quantity:
Sample ID	Sample Location & Material Location	0
7	- Level 8 Rut = 5 Wall	024
		0.3
		2403934
HM#	Material Description:	
Sample ID	Sample Location & Material Location	Quantity:
		1/2
Relinquished By:	T. Alexaul Signature:	Date/Time: 5/1//20
Received By:	Signature:	Date/ Time: 5/1/20 /2
Relinquished By:	Signature:	Date/Time: /
Received By:	Signature:	Date/Time:



	·
***E-MAIL R  PM - S. Stein spsteiner@terracon  PM- M. Benefi msbenefield@terraco  PM- D. Block David block@terraco  Project Name/ Add Project# R	Manalysis (Analyze all samples)   Mana
<b>НМ#</b> 1	Material Description Exchien Whik Window Sealant Frame to Building
Sample ID	Sample Location & Material Location Quantity:
2A 2B 2C	Level 5 65 Column Level 4 At. A.51 Wall Level 6 F5 Wall
HM# 2	Material Description: Exemon Black Window Sealant (Frame to
Sample ID	Sample Location & Material Location Quantity: Gland
3A 3B 3C	Level 5 G5 Column Level 5 A 5   Vall Level 6 F5 Wall  Material Description: Refue Elostomen's Paint
HM# +	Material Description: Beige Elastomenic Paint Sample Location & Material Location Quantity:
HM# S Sample ID	Vallecito Place Wall  Vallecito Place Wall  Vallecito Place Wall  Why Avenue Wall  Material Description: Decorative Plasten with Grount  Sample Location & Material Location Quantity:
5A 51 5c	Vallecto Ath Colynn  Material Description:  Semple Leasting & Material Leasting Oughtity:
HM#	Material Description:
Sample ID	Sample Location & Material Location Quantity:
Relinquished By: Received By: Relinquished By:	Signature: Signature: Signature: Date/Time: Date/Time: Date/Time:
Received By:	Signature: Date/Time:





Report for:

Eric Dyer, Ms. Karin Schroeter Terracon Consultants, Inc. - Emeryville 1466 66th Street Emeryville, CA 94608

Regarding: Project: R1207159; Highland Hospital/1411 East 31st St./Oakland, CA.

EML ID: 2403936

Approved by:

Dates of Analysis: Asbestos PLM: 05-13-2020

Approved Signatory Amin Suliman

Service SOPs: Asbestos PLM (EPA 40CFR App E to Sub E of Part 763 & EPA METHOD 600/R-93-116, SOP EM-AS-S-1267) NVLAP Lab Code 200728-0

All samples were received in acceptable condition unless noted in the Report Comments portion in the body of the report. The results relate only to the samples as received. The results include an inherent uncertainty of measurement associated with estimating percentages by polarized light microscopy. Measurement uncertainty data for sample results with >1% asbestos concentration can be provided when requested.

Eurofins EMLab P&K ("the Company") shall have no liability to the client or the client's customer with respect to decisions or recommendations made, actions taken or courses of conduct implemented by either the client or the client's customer as a result of or based upon the Test Results. In no event shall the Company be liable to the client with respect to the Test Results except for the Company's own willful misconduct or gross negligence nor shall the Company be liable for incidental or consequential damages or lost profits or revenues to the fullest extent such liability may be disclaimed by law, even if the Company has been advised of the possibility of such damages, lost profits or lost revenues. In no event shall the Company's liability with respect to the Test Results exceed the amount paid to the Company by the client therefor.

#### **Eurofins EMLab P&K**

Lab ID-Version 1: 11464620-1

Lab ID-Version‡: 11464622-1

6000 Shoreline Ct, Ste 205, So. San Francisco, CA 94080

Client: Terracon Consultants, Inc. - Emeryville

(866) 888-6653 Fax (623) 780-7695 www.emlab.com

C/O: Eric Dyer, Ms. Karin Schroeter Date of Sampling: 05-11-2020 Re: R1207159; Highland Hospital/1411 East 31st St./ Date of Receipt: 05-11-2020 Oakland, CA. Date of Report: 05-13-2020

#### ASBESTOS PLM REPORT

Total Samples Submitted: 6
Total Samples Analyzed: 6
Total Samples with Layer Asbestos Content > 1%: 0

#### Location: 1A, Wallboard with taping mud; ED lobby 4th floor 40002 wall

Sample Layers	Asbestos Content
White Compound	ND
Cream Tape	ND
White Joint Compound	ND
White Drywall with Brown Paper	ND
Composite Non-Asbestos Content:	15% Cellulose
Sample Composite Homogeneity:	Poor

## Location: 1B, Wallboard with taping mud; ED lobby 4th floor 40002 back side wall

Location: 1B, wallboard with taping mud; ED lobby 4t	I HOOF 40002 back side wall Lab ID- version 1: 11464621-1
Sample Layers	Asbestos Content
White Joint Compound	ND
White Drywall with Brown Paper	ND
Composite Non-Asbestos Content:	15% Cellulose
Sample Composite Homogeneity:	Poor

#### Location: 1C, Wallboard with taping mud; ED lobby 4th floor 40002 back side wall

Sample Layers	Asbestos Content
White Joint Compound	ND
White Drywall with Brown Paper	ND
Composite Non-Asbestos Content:	15% Cellulose
Sample Composite Homogeneity:	Poor

The test report shall not be reproduced except in full, without written approval of the laboratory. The report must not be used by the client to claim product certification, approval, or endorsement by any agency of the federal government. Eurofins EMLab P&K reserves the right to dispose of all samples after a period of thirty (30) days, according to all state and federal guidelines, unless otherwise specified.

Inhomogeneous samples are separated into homogeneous subsamples and analyzed individually. ND means no fibers were detected. When detected, the minimum detection and reporting limit is less than 1% unless point counting is performed. Floor tile samples may contain large amounts of interference material and it is recommended that the sample be analyzed by gravimetric point count analysis to lower the detection limit and to aid in asbestos identification.

 $\ddagger$  A "Version" indicated by -"x" after the Lab ID# with a value greater than 1 indicates a sample with amended data. The revision number is reflected by the value of "x".

EMLab P&K, LLC

#### **Eurofins EMLab P&K**

Lab ID-Version‡: 11464623-1

Lab ID-Version 1: 11464625-1

6000 Shoreline Ct, Ste 205, So. San Francisco, CA 94080

Client: Terracon Consultants, Inc. - Emeryville

(866) 888-6653 Fax (623) 780-7695 www.emlab.com

C/O: Eric Dyer, Ms. Karin Schroeter Date of Sampling: 05-11-2020 Re: R1207159; Highland Hospital/1411 East 31st St./ Date of Receipt: 05-11-2020 Oakland, CA. Date of Report: 05-13-2020

#### ASBESTOS PLM REPORT

Location: 2A, Stucco; 4th floor staff courtvard wall

Sample Layers	Asbestos Content
White Skim Coat	ND
Gray Stucco	ND
Sample Composite Homogeneity:	Moderate

## Location: 2B, Stucco; 4th floor staff courtyard wall

Lab ID-Version 1: 11464624-1 Sample Layers **Asbestos Content** White Skim Coat ND Gray Stucco ND Sample Composite Homogeneity: | Moderate

#### Location: 2C, Stucco; 4th floor staff courtvard wall

Sample Layers	Asbestos Content
White Skim Coat	ND
Gray Stucco	ND
Sample Composite Homogeneity:	Moderate

The test report shall not be reproduced except in full, without written approval of the laboratory. The report must not be used by the client to claim product certification, approval, or endorsement by any agency of the federal government. Eurofins EMLab P&K reserves the right to dispose of all samples after a period of thirty (30) days, according to all state and federal guidelines, unless otherwise specified.

Inhomogeneous samples are separated into homogeneous subsamples and analyzed individually. ND means no fibers were detected. When detected, the minimum detection and reporting limit is less than 1% unless point counting is performed. Floor tile samples may contain large amounts of interference material and it is recommended that the sample be analyzed by gravimetric point count analysis to lower the detection limit and to aid in asbestos identification.

 $\ddagger$  A "Version" indicated by -"x" after the Lab ID# with a value greater than 1 indicates a sample with amended data. The revision number is reflected by the value of "x".

EMLab P&K, LLC

## Terracon

***E-MAIL  PM - S. Ste spsteiner@terraci  PM- M. Bene msbenefield@terra  PM- D. Bkc David.block@terra	con.com kmschroeter@terracon.com kmplkgrim@terracon.com  efield	PLM Analysis (Analyze all samples) Stop Analysis at First Positive Point Count Analysis (400-point)
	dress/ Building No. High and Hospital 1911 195	- 3/st Street / Oakland, CA
roject# R	-07 59 Sampled By: Akrander	Sampling Date: 5/11/20
ample(s) sent to:	☐ MAL ☐ ASB TEM ☐ EMLAB ☐ Other _	
AT Rush	□ 24HRS □ 48HR □ 3-5 days	
HM#	Material Description Wallboam with Tap	the Myd
Sample ID	Sample Location & Material Location	Quantity:
IA	ED Lobby 4th Floor 4002	Wall
IR	EN Lohou 4th Floor 4002	Back Side, Wall,
IC.	EN Luby 4th Floor 4002	Buck Side Wall
HM# 1	Material Description: Stucco	
Sample ID	Sample Location & Material Location	Quantity:
2.4	All of of Company	1 Wall
28	the Floor State Countyard	1 , 1
20	The Floor Part Countygro	Wall
HM#		
Sample ID	Material Description: Sample Location & Material Location	Quantity:
HM#	Material Description:	Quantity:
Sample ID	Sample Location & Material Location	Quantity.
		002403
HM#	Material Description:	Quantity:
Sample ID	Sample Location & Material Location	Quantity:
Relinquished By	: J. Alexander Signature: Jelle U.	Date/Time: 5/1/20
Received By:	Signature:	Date/ Time: 3/11/20
Relinquished By	: Signature:	Date/Time:
Deceived Rv	Signature:	Date/Time:





Report for:

Eric Dyer, Ms. Karin Schroeter Terracon Consultants, Inc. - Emeryville 1466 66th Street Emeryville, CA 94608

Regarding: Project: R1197G03; Highland Hospital, 1411 East 31st Street, Oakland, CA

EML ID: 2403937

Approved by:

Dates of Analysis:

Lead - Flame AA: 05-12-2020

**Technical Manager** Andrew Ikeda

Industida

Service SOPs: Lead - Flame AA (EM-BC-S-8443) AIHA-LAP, LLC accredited service, Lab ID #178697

All samples were received in acceptable condition unless noted in the Report Comments portion in the body of the report. Due to the nature of the analyses performed, field blank correction of results is not applied. The results relate only to the samples as received. Sample size, as it relates to Wipe samples only, is supplied by the client.

Eurofins EMLab P&K ("the Company") shall have no liability to the client or the client's customer with respect to decisions or recommendations made, actions taken or courses of conduct implemented by either the client or the client's customer as a result of or based upon the Test Results. In no event shall the Company be liable to the client with respect to the Test Results except for the Company's own willful misconduct or gross negligence nor shall the Company be liable for incidental or consequential damages or lost profits or revenues to the fullest extent such liability may be disclaimed by law, even if the Company has been advised of the possibility of such damages, lost profits or lost revenues. In no event shall the Company's liability with respect to the Test Results exceed the amount paid to the Company by the client therefor.

Eurofins EMLab P&K's LabServe® reporting system includes automated fail-safes to ensure that all AIHA-LAP, LLC quality requirements are met and notifications are added to reports when any quality steps remain pending.

## **Eurofins EMLab P&K**

17461 Derian Ave, Suite 100, Irvine, CA 92614

Client: Terracon Consultants, Inc. - Emeryville

C/O: Eric Dyer, Ms. Karin Schroeter

Re: R1197G03; Highland Hospital, 1411 East 31st Street, Oakland, CA

(866) 888-6653 Fax (623) 780-7695 www.emlab.com

Date of Sampling: 05-11-2020 Date of Receipt: 05-11-2020 Date of Report: 05-13-2020

#### LEAD: FLAME ATOMIC ABSORPTION SPECTROMETRY

Location:	Pb-1: Beige concrete wall, Vallecito Place wall
Comments (see below)	A
Lab ID-Version‡:	11464282-1
Analysis Date:	05/12/2020
Sample type	Paint Chip sample
Method*	NIOSH 7082 & EPA 7000B modified
† Method Reporting Limit	39 ppm
Sample size	0.2556 grams
§Total Lead Result	< 39 ppm

**Comments:** A) The relative percent difference of the matrix duplicate pair was above control limits. The laboratory control sample and matrix blank were both within control limits and validated the batch.

Sample results have not been corrected for blank values.

Bulk samples are not covered under the AIHA-LAP, LLC service accreditation.

Wipe samples must meet ASTM E1792 criteria. Method Reporting Limits may not be valid for non-ASTM E1792 wipe samples.

- † The Method Reporting Limit is the minimum concentration of Lead that the laboratory can confidently detect in the sample.
- § Total Lead Result has been rounded to two significant figures to reflect analytical precision.
- ‡ A "Version" indicated by -"x" after the Lab ID# with a value greater than 1 indicates a sample with amended data. The revision number is reflected by the value of "x".

<sup>\*</sup>Sample preparation and analytical methods are based upon NIOSH 7082 and EPA 7000B.

# Terracon

***E-MAIL RE	PORT TO: PROJEC	T MANAGER (PM)***	I FAD DAINT SAME	4 40 2 700
denise waller Engineerin	@terracon.com g Assistant	eric.dver@terracon.com Engineering Assistant	* Lead Analysis  Flame AA (EPA 7420)	TTLC
□PM = \$ spsteinen@tr	S. Steiner steiner serracon.com	kmschroeter@terracon.com		AGE
PM – K. Pilgrir kmpilgrim@terracon.	n DPM- M. Ber com msbenefield@tem	efield DPM - W. Frieszel acon.com wmfrieszell@terracon.com		PM – D. Block d.block@terracon.com
	ddress/Building No.		05pfa/14/1 East 3/st 3	Treet Oakh
Sample(s) sent to:		Sampled By: John  MSL □ Aerobiology □ (	Quantem Other Fully	: 5/1//20
TAT R		☑ 48HRS ☐ 3-5 Day	PETITAL Y	
Sample ID		Paint Description and	Sample Location	Condition (I/F/P)
0 1	Paint Beige	Substrate: Conc	were Component: Wall	
Pb-1	Sample Location: Bl	dg #	Unit # Room Valle cito	P
	Paint Color:	Substrate:	Component:	
	Sample Location: Bl	dg #	Unit # Room	
	Paint Color:	Substrate:	Component:	
	Sample Location: Bl	dg #	Unit # Room	
	Paint Color:	Substrate:	Component:	
	Sample Location: Blo	dg # U	Unit # Room	00240
	Paint Color:	Substrate:	Component:	02403937
	Sample Location: Blo	lg# U	Jnit # Room	
Relinquished By:	J. Aleta	Signature:	Date/Time:	slilho
Received By:	2	Signature:	Date/ Time:	5/1/20 123
Received By:	J. Timmony	Signature:	Date/Time:	112/20709:3





Report for:

Eric Dyer, Ms. Karin Schroeter Terracon Consultants, Inc. - Emeryville 1466 66th Street Emeryville, CA 94608

Regarding: Project: R1207159; Highland Hospital, 1411 East 31st Street, Oakland, CA

EML ID: 2403939

Approved by:

Dates of Analysis:

Lead - Flame AA: 05-12-2020

**Technical Manager** Andrew Ikeda

Industida

Service SOPs: Lead - Flame AA (EM-BC-S-8443) AIHA-LAP, LLC accredited service, Lab ID #178697

All samples were received in acceptable condition unless noted in the Report Comments portion in the body of the report. Due to the nature of the analyses performed, field blank correction of results is not applied. The results relate only to the samples as received. Sample size, as it relates to Wipe samples only, is supplied by the client.

Eurofins EMLab P&K ("the Company") shall have no liability to the client or the client's customer with respect to decisions or recommendations made, actions taken or courses of conduct implemented by either the client or the client's customer as a result of or based upon the Test Results. In no event shall the Company be liable to the client with respect to the Test Results except for the Company's own willful misconduct or gross negligence nor shall the Company be liable for incidental or consequential damages or lost profits or revenues to the fullest extent such liability may be disclaimed by law, even if the Company has been advised of the possibility of such damages, lost profits or lost revenues. In no event shall the Company's liability with respect to the Test Results exceed the amount paid to the Company by the client therefor.

Eurofins EMLab P&K's LabServe® reporting system includes automated fail-safes to ensure that all AIHA-LAP, LLC quality requirements are met and notifications are added to reports when any quality steps remain pending.

## **Eurofins EMLab P&K**

17461 Derian Ave, Suite 100, Irvine, CA 92614

(866) 888-6653 Fax (623) 780-7695 www.emlab.com

Client: Terracon Consultants, Inc. - Emeryville

C/O: Eric Dyer, Ms. Karin Schroeter

Re: R1207159; Highland Hospital, 1411 East 31st Street, Oakland, CA

Date of Sampling: 05-11-2020 Date of Receipt: 05-11-2020 Date of Report: 05-13-2020

## LEAD: FLAME ATOMIC ABSORPTION SPECTROMETRY

Location:	Pb-1:	Pb-2:
	Off-white wallboard wall, ED lobby,	Beige stucco wall, 4th floor staff
	4th	courtyard
	floor	
Comments (see below)	A	A
Lab ID-Version‡:	11464351-1	11464352-1
Analysis Date:	05/12/2020	05/12/2020
Sample type	Paint Chip sample	Paint Chip sample
Method*	NIOSH 7082 & EPA 7000B modified	NIOSH 7082 & EPA 7000B modified
† Method Reporting Limit	47 ppm	39 ppm
Sample size	0.2133 grams	0.2577 grams
§Total Lead Result	< 47 ppm	41 ppm

**Comments:** A) The relative percent difference of the matrix duplicate pair was above control limits. The laboratory control sample and matrix blank were both within control limits and validated the batch.

Sample results have not been corrected for blank values.

Bulk samples are not covered under the AIHA-LAP, LLC service accreditation.

Wipe samples must meet ASTM E1792 criteria. Method Reporting Limits may not be valid for non-ASTM E1792 wipe samples.

- † The Method Reporting Limit is the minimum concentration of Lead that the laboratory can confidently detect in the sample.
- § Total Lead Result has been rounded to two significant figures to reflect analytical precision.
- ‡ A "Version" indicated by -"x" after the Lab ID# with a value greater than 1 indicates a sample with amended data. The revision number is reflected by the value of "x".

EMLab P&K, LLC

<sup>\*</sup>Sample preparation and analytical methods are based upon NIOSH 7082 and EPA 7000B.

## **Tierracon**

**E-MAIL REPO	ORT TO: PROJECT M	ANAGER (PM)***	LEAD PAINT S.	AMPLE DATA SHEET
denise.wallen@ Engineering A		eric.dyer@terracon.com Engineering Assistant	* Lead Analysis Flame AA (EPA 7420)	)TTLC
□PM - S. S spsteiner@terra		DPM - K. Schroeter kmschroeter@terracon.com		PAGEOF
PM – K. Pilgrim kmpilgrim@terracon.com	PM- M. Benefield msbenefield@terracon.		□ PM – T. Kattchee takattchee@terracon.com	PM – D. Block david.block@terracon.com
010	dress/Building No.	11 / A	19/14/1 East 3/5)	- Street/ OgKlano
Project# R Sample(s) sent to:	Jan Sai	npled By: Qu	samp Samp	ling Date: 5///20
TAT Rus	1	8HRS 3-5 Day		
Sample ID	1	Paint Description and Sa	ample Location	Condition (I/F/P)
0 1	Paint OF-While Color:	Substrate: Wall 00	Component: Wq	
Pb	Sample Location: Bldg	# U	Jnit# Room	Plan Han
21 0	Paint Belge	Substrate: Stuc	Co Component: W	
16	Sample Location: Bldg	# U	Init # Room	to Floor Countyand
	Paint Color:	Substrate:	Component:	
	Sample Location: Bldg	# t	Unit # Room	
	Paint Color:	Substrate:	Component:	- 8
	Sample Location: Bldg	# t	Jnit # Room	N
	Paint Color:	Substrate:	Component:	403939
	Sample Location: Bldg	#	Jnit # Room	
Relinquished By	J. Aktander	Signature:	Date	Time: 5/1/20 /721
Received By:	- Maria	Signature:		Time: 5/1/27 175
Received By:	J. Tilmmon	Signature:		Time: 5/10/1000

## **DOCUMENT 01 10 00**

## **SUMMARY OF WORK**

## **PART 1 - GENERAL**

## 1.01 RELATED DOCUMENTS AND PROVISIONS:

All Contract Documents must be reviewed for applicable provisions related to the provisions in this document, including without limitation:

- A. General Conditions
- B. Special Conditions.
- C. Construction Waste Management Document 01 74 19
- D. General Commissioning Requirements 01 91 13

## 1.02 SUMMARY OF WORK COVERED BY CONTRACT DOCUMENTS

A. The Work of this Contract may consist of the following:

**Koret Building Emergency Department Exhaust Fan -** Provide code-required airflows (supply and exhaust) in Emergency Department Waiting Rooms 4000, 4002, 4003, 4007, and 4017. A new ground-mounted exhaust fan will be installed in the maintenance courtyard adjacent to the Emergency Department Waiting Room and ducted into the space.

**Koret Building Exterior Maintenance -** Provide sealant replacement and installation of elastomeric coating over cement plaster assemblies at the Koret Building exterior walls.

#### 1.03 CONTRACTS

- A. Perform the Work under a single, fixed-price Contract.
- B. Any bid item may be deleted in total or in part prior to or after award of Contract without compensation in any form or adjustment of other bid items or prices.

#### 1.04 WORK BY OTHERS

- A. Work on the Project that will be performed and completed prior to the start of the Work of this Contract:
  - (1) None.

- B. Work on the Project that will be performed by others concurrent with the Work of this Contract:
  - (1) Facility maintenance and modernization projects. As projects are known and funded, the work will be identified so that necessary coordination can be performed.

## 1.05 CODES, REGULATIONS, AND STANDARDS

- A. The codes, regulations, and standards adopted by the state and federal agencies having jurisdiction shall govern minimum requirements for this project. Where codes, regulations, and standards conflict with the Contract Documents, these conflicts shall be brought to the immediate attention of the County and the Architect/Engineer.
- B. Codes, regulations, and standards shall be as published effective as of date of bid opening, unless otherwise specified or indicated.

## 1.06 PROJECT RECORD DOCUMENTS:

- A. Contractor shall maintain on Site one set of the following record documents; Contractor shall record actual revisions to the Work:
  - (1) Contract Drawings.
  - (2) Specifications.
  - (3) Addenda.
  - (4) Change Orders and other modifications to the Contract.
  - (5) Reviewed shop drawings, product data, and samples.
  - (6) Field test records.
  - (7) Inspection certificates.
  - (8) Manufacturer's certificates.
- B. Contractor shall store Record Documents separate from documents used for construction. Provide files, racks, and secure storage for Record Documents and samples.
- C. Contractor shall record information concurrent with construction progress.

- D. Specifications: Contractor shall legibly mark and record at each product section of the Specifications the description of the actual product(s) installed, including the following:
  - (1) Manufacturer's name and product model and number.
  - (2) Product substitutions or alternates utilized.
  - (3) Changes made by Addenda and Change Orders and written directives.

## 1.07 EXAMINATION OF EXISTING CONDITIONS

- A. The Contractor shall be held to have examined the Project Site and acquainted itself with the conditions of the Site or of the streets or roads approaching the Site.
- B. Prior to commencement of Work, Contractor shall survey the Site and existing buildings and improvements to observe existing damage and defects such as cracks, sags, broken, missing or damaged glazing, other building elements and Site improvements, and other damage.
- C. Should Contractor observe cracks, sags, and other damage to and defects of the Site and adjacent buildings, paving, and other items not indicated in the Contract Documents, Contractor shall immediately report same to the County and the Architect/Engineer.

## 1.08 CONTRACTOR'S USE OF PREMISES

- A. If unoccupied and only with County's prior written approval, Contractor may use the building(s) at the Project Site without limitation for its operations, storage, and office facilities for the performance of the Work. If the County chooses to beneficially occupy any building(s), Contractor must obtain the County's written approval for Contractor's use of spaces and types of operations to be performed within the building(s) while so occupied. Contractor's access to the building(s) shall be limited to the areas indicated.
- B. If the space at the Project Site is not sufficient for Contractor's operations, storage, office facilities and/or parking, Contractor shall arrange and pay for any additional facilities needed by Contractor.
- C. Contractor shall not interfere with use of or access to occupied portions of the building(s) or adjacent property.

- D. Contractor shall maintain corridors, stairs, halls, and other exit-ways of building clear and free of debris and obstructions at all times.
- E. No one other than those directly involved in the demolition and construction, or specifically designated by the County or the Architect/Engineer shall be permitted in the areas of work during demolition and construction activities.
- F. The Contractor shall install the construction security fence and maintain that it will be locked when not in use. Keys to this fencing will be provided to the County.
- G. Any work to be performed outside of the Contractor's work area or outside of the normal work hours will require coordination with the facility via the Method of Procedure (MOP) process. MOPs must be submitted at least 72 hours prior to proposed start of work.

## 1.09 PROTECTION OF EXISTING STRUCTURES AND UTILITIES

- A. The Drawings show above-grade and below-grade structures, utility lines, and other installations that are known or believed to exist in the area of the Work. Contractor shall locate these existing installations before proceeding with excavation and other operations that could damage same; maintain them in service, where appropriate; and repair damage to them caused by the performance of the Work. Should damage occur to these existing installations, the costs of repair shall be at the Contractor's expense and made to the County's satisfaction.
- B. Contractor shall be alert to the possibility of the existence of additional structures and utilities. If Contractor encounters additional structures and utilities, Contractor will immediately report to the County for disposition of same as indicated in the General Conditions.
- C. Contract shall employ utility locating services to locate any underground utilities within the limits of work prior to the start of any excavating work. Contractor shall subsequently be responsible for repairs of any located underground utilities damaged by construction activities.

## 1.10 UTILITY SHUTDOWNS AND INTERRUPTIONS

A. Contractor shall give the County a minimum of fourteen (14) days written notice in advance, via a MOP, of any need to shut off existing utility services or to effect equipment interruptions. The facility provides essential services to the County of Alameda and any interruption to services will require coordination with facility Engineering and Infection Control Departments, at a minimum. The County, in conjunction with the facility, will set exact time and duration for shutdown, and

- will assist Contractor with shutdown. Work required to re-establish utility services shall be performed by the Contractor.
- B. Contractor shall obtain County's written approval as indicated in the General Conditions in advance of deliveries of material or equipment or other activities that may conflict with County's use of the building(s) or adjacent facilities.

## 1.11 STRUCTURAL INTEGRITY

- A. Contractor shall be responsible for and supervise each operation and work that could affect structural integrity of various building elements, both permanent and temporary.
- B. Contractor shall include structural connections and fastenings as indicated or required for complete performance of the Work.

## 1.12 WORK SEQUENCE

- A. Contractor shall be responsible for compliance with all requirements outlined in the hazardous materials sections of the Contract Documents.
- B. Construct Work in stages and at times to accommodate County operation requirements during the construction period; coordinate construction schedule and operations with the County.
- C. Phasing Schedule included in item 1.13 must be adhered to. Moving dates are included in the Phasing Schedule to show planned windows for relocation of building occupants. Liquidated damages may be assessed as described in the Contract Documents for failure to achieve milestone dates.
- D. This facility will be occupied 24 hours a day/7 days a week during construction. County will provide contractor with schedule of uses; at the site during the construction period; Contractor is to coordinate work with the County and maintain safe access to all buildings at all times and to not disrupt ongoing uses. Contractor must comply with the following requirements:
  - (1) If Contractor must shut down power to any part of the site, Contractor must provide temporary power for that section of the site.
  - (2) Scope of work includes utility and systems upgrade and replacement that may impact the entire site. Work must be coordinated so that site-wide systems remain functional at all times until new systems work is complete and tested.

(3) If Contractor is working in an occupied portion of the building, Contractor must complete at Infection Control Risk Assessment (ICRA) for review and approval by the facility. Contractor must provide any infection control measures required by the ICRA, including, but not limited to plastic barriers, ante rooms, negative air machines, and walk-off mats.

PART 2 – PRODUCTS Not Used.

**PART 3 – EXECUTION** Not Used.

END OF DOCUMENT

## **DOCUMENT 01 22 00**

## **UNIT PRICES AND ALTERNATES**

## PART I – ALTERNATES

## 1.01 RELATED DOCUMENTS AND PROVISIONS:

All Contract Documents should be reviewed for applicable provisions related to the provisions in this document, including without limitation:

- A. General Conditions;
- B. Special Conditions;
- C. Bid Form;
- D. Instruction to Bidders.

## 1.02 DESCRIPTION

The items of work indicated below propose modifications to, substitutions for, additions to and/or deletions from the various parts of the Work specified in other Sections of the Specifications. The acceptance or rejection of any of the alternates is strictly at the option of the County subject to County's acceptance of Contractor's stated prices contained in this Proposal.

## 1.03 GENERAL

Where an item is omitted, or scope of Work is decreased, all Work pertaining to the item whether specifically stated or not, shall be omitted and where an item is added or modified or where scope of Work is increased, all Work pertaining to that required to render same ready for use on the Project in accordance with the intention of the Drawings and Specifications shall be included in an agreed upon price amount.

## **1.04 BASE BID**

The Base Bid includes all work required to construct the Project completely and in accordance with the Contract Documents.

#### 1.05 ALTERNATES

Not Applicable.

#### **PART 2 - UNIT PRICING**

## **Alameda County, General Services Agency**

Highland Hospital – Koret Building Exhaust Fan and Waterproofing

### 2.01 GENERAL

Contractor shall completely state all required figures based on Unit Prices listed below. Where scope of Work is decreased, all Work pertaining to the item, whether specifically stated or not, shall be omitted and where scope of Work is increased, all work pertaining to that item required to render same ready for use on the Project in accordance with intention of Drawings and Specifications shall be included in an agreed upon price amount.

## 2.02 UNIT PRICES

Furnish unit prices for each of the named items on a square foot, lineal foot, or per each basis, as applies. Unit prices shall include all labor, materials, services, profit, overhead, insurance, bonds, taxes, and all other incidental costs of Contractor, subcontractors, and supplier(s).

- A. Exterior Maintenance Project: glass lite caulking/LF. Unit price to include cut back existing gaskets, remove existing wet seals, and install fillet bead of sealant at all glass lites.
- B. Exterior Maintenance Project: sealant at window perimeters/LF. Unit price to include remove and replace sealant over backer rod at window perimeters.
- C. Exterior Maintenance Project: window and metal spandrel panel joints. Unit price to include install sealant and extruded silicone tape over all window and metal spandrel panel joints.
- D. Exterior Maintenance Project: Elastomeric paint/SF. Unit price to include preparation of substrate and install elastomeric coating over cement plaster walls.

END OF DOCUMENT

#### **DOCUMENT 01 26 00**

## **CONTRACT MODIFICATION PROCEDURES**

#### **PART 1 - GENERAL**

#### 1.01 GENERAL

- A. Only Contractor or the County may initiate changes in scope of Work or deviation from Contract Documents.
  - (1) Contractor may initiate changes by submitting a Request for Information (RFI), Notice of Concealed or Unknown Conditions, Notice of Hazardous Waste Conditions or Notice of Regulatory Agency Field Change.
    - a. RFIs shall be submitted to seek clarification of or request changes in the Contract Documents.
    - b. Notices of Concealed or Unknown Conditions shall be submitted in accordance with Document 007200 (General Conditions).
    - c. Notices of Hazardous Waste Conditions shall be submitted in accordance with Document 007200 (General Conditions).
    - d. Notices of Regulatory Agency Field Change shall be submitted when the permitting regulatory agency directs in the field a change in design previously approved by the regulatory agency, which change is not the result of design error or omission or defective Work.
  - (2) The County may initiate changes by issuing a Supplemental Instruction, which may revise, add to or subtract from the Work.
  - (3) The County may initiate changes in the Work or Contract Time by issuing RFPs to Contractor. Such RFPs will detail all proposed changes in the Work and request a quotation of changes in Contract Sum and Contract Time from Contractor.
  - (4) The County may also, by Construction Change Directive ("CCD"), order changes in the Work within the general scope of the Contract consisting of additions, deletions, or other revisions, the Contract Sum and Contract Time being adjusted accordingly. A CCD shall be used in the absence of total agreement on the terms of a Change Order and may, upon notice, consist of a Change Order executed by the County only.

#### 1.02 PROCEDURES

- A. <u>Cost Proposal and Procedures</u>: Whenever Contractor is required in this Section to prepare a Cost Proposal, and whenever Contractor is entitled to submit a Cost Proposal and elects to do so, Contractor shall prepare and submit to the County for consideration a Cost Proposal using the form attached to this Section. All Cost Proposals must contain a complete breakdown of actual, current costs of credits, deducts, and extras; and itemizing materials, labor, taxes, overhead and profit. All Subcontractor Work shall be so indicated. Individual entries on the Cost Proposal form shall be determined as provided in paragraphs 1.3 and 1.4 of this Section. After receipt of a Cost Proposal with a detailed breakdown, the County will act promptly thereon.
  - (1) If the County accepts a Cost Proposal, the County will prepare Change Order for the County and Contractor signatures.
  - (2) If Cost Proposal is not acceptable to the County because it does not agree with cost and/or time included in Cost Proposal, the County will submit in a response what it believes to be a reasonable cost and/or adjustment, if any. Except as otherwise provided in this Section, Contractor shall have seven (7) Days in which to respond to the County with a revised Cost Proposal.
  - When necessity to proceed with a change does not allow the County sufficient time to conduct a proper check of a Cost Proposal (or revised Cost Proposal), the County may order Contractor to proceed on basis to be determined at earliest practical date. In this event, value of change, with corresponding equitable adjustment to Contract, shall not be more than increase or less than decrease proposed.
- B. Request for Information: Whenever Contractor requires information regarding the Project or Contract Documents, or receives a request for information from a Subcontractor, Contractor may prepare and deliver a RFI to the County. Contractor shall use RFI format provided by the County. Contractor must submit time critical RFIs at least thirty (30) Days before scheduled start date of the affected Work activity. Contractor shall reference each RFI to an activity of Progress Schedule and shall note time criticality of the RFI, indicating time within which a response is required. Contractor's failure to reference RFI to an activity on the Progress Schedule and note time criticality on the RFI shall constitute Contractor's waiver of any claim for time delay or interruption to the Work resulting from any delay in responding to the RFI.

- (1) The County will respond within seven (7) Days from receipt of RFI with a written response to Contractor. Contractor shall distribute response to all appropriate Subcontractors.
- (2) If Contractor is satisfied with the response and does not request change in Contract Sum or Contract Time, then the response shall be executed without a change.
- (3) If Contractor believes the response is incomplete, Contractor shall issue another RFI (with the same RFI number with the letter "A" indicating if it is a follow-up RFI) to the County clarifying original RFI. Additionally, the County may return RFI requesting additional information should original RFI be inadequate in describing condition.
- (4) If Contractor believes that the response results in change in Contract Sum or Contract Time, Contractor shall notify the County in writing within seven (7) Days after receiving the response, or within seven (7) Days of becoming aware of such changes in Contract Sum or Contract Time. If the County disagrees with Contractor, then Contractor may give notice of intent to submit a Claim as described in Article 12 (Claims by Contractor) of Document 007200 (General Conditions), and submit its Claim within thirty (30) Days. If the County agrees with Contractor, then Contractor must submit a Cost Proposal within seven (7) Days of receiving the County's agreement. Contractor's failure to deliver either the foregoing notice and Claim or Cost Proposal by the respective deadlines stated in the foregoing sentences shall result in waiver of the right to file a Cost Proposal or Claim.
- C. <u>Supplemental Instruction</u>: the County may issue Supplemental Instruction to Contractor.
  - (1) If Contractor is satisfied with Supplemental Instruction and does not request change in Contract Sum or Contract Time, then Supplemental Instruction shall be executed without a Change Order.
  - (2) If Contractor believes that Supplemental Instruction results in change in Contract Sum or Contract Time, then Contractor must submit a Cost Proposal to the County within twenty-one (21) Days of receiving the Supplemental Instruction.
- D. <u>Construction Change Directives</u>: If at any time the County believes in good faith that a timely Change Order will not be agreed upon using the foregoing procedures, the County may issue a CCD with its recommended cost and/or time adjustment. Upon receipt of CCD, Contractor shall promptly proceed with the

change of Work involved and concurrently respond to the County's CCD within ten (10) Days.

- (1) Contractor's response must be any one of following:
  - a. Return CCD signed, thereby accepting the County's response, time, and cost.
  - b. Submit a (revised if applicable) Cost Proposal with supporting documentation (if applicable, reference original Cost Proposal number followed by letter A, B, etc. for each revision), if the County so requests.
  - c. Give notice of intent to submit a Claim as described in Article 12 (Claims by Contractor) of Document 007200 (General Conditions), and submit its Claim within thirty (30) Days.
- (2) If the CCD provides for an adjustment to the Contract Sum, the adjustment shall be based on one of the following methods:
  - a. Mutual acceptance of a lump sum properly itemized and supported by sufficient substantiating data to permit evaluation.
  - b. Unit prices stated in the Contract Documents or subsequently agreed upon.
  - c. Cost to be determined in a manner agreed.
  - d. County directs the Contractor to perform the work as "Force Account". Upon receiving a "Force Account' directive Contractor shall diligently perform the work by utilizing the time and material method. Daily time and verification on all man hours and materials shall be submitted and approved by the County.
- (3) CCD signed by Contractor indicates the agreement of Contractor therewith, including adjustment in Contract Sum and Contract Time or the method for determining them. Such agreement shall be effective immediately and shall be recorded as a Change Order.
- (4) If Contractor does not respond promptly or disagrees with the method for adjustment in the Contract Sum, the method and the adjustment shall be determined by the County on the basis of reasonable

expenditures and savings of those performing the Work attributable to the change including, in case of an increase in the Contract Sum, a reasonable allowance for overhead and profit. If the parties still do not agree on the price for a CCD, Contractor may file a Claim per Article 12 (Claims by Contractor) of Document 007200 (General Conditions). Contractor shall keep and present, in such form as the County may prescribe, an itemized accounting together with appropriate supporting data. Unless otherwise provided in the Contract Documents, costs for the purposes of this paragraph shall be limited to those provided in paragraphs 1.3 and 1.4 of this Section 012600.

- (5) Pending final determination of cost to the County, amounts not in dispute may be included in Applications for Payment. The amount of credit to be allowed by Contractor to the County for a deletion or change that results in a net decrease in the Contract Sum shall be actual net cost as confirmed by the County. When both additions and credits covering related Work or substitutions are involved in a change, the allowance for overhead and profit shall be figured on the basis of net increase, if any, with respect to that change.
- E. <u>County Requested RFP</u>: Contractor shall furnish a Cost Proposal within seven (7) Business Days of the County's RFP. Upon approval of RFP, the County will issue a Change Order directing Contractor to proceed with extra Work. If the parties do not agree on the price for an RFP, the County may either issue a CCD or decide the issue per Article 12 (Claims by Contractor) of Document 007200 (General Conditions). Contractor shall perform the changed Work notwithstanding any claims or disagreements of any nature.
- F. <u>Differing Site Conditions</u>: Contractor shall submit Notices of Differing Site Conditions to resolve problems regarding differing underground Site conditions encountered in the execution of the Work pursuant to paragraph 13.4 (Notice of Concealed or Unknown Conditions) of Document 007200 (General Conditions), which shall govern. If the County determines that a change in Contract Sum or Contract Time is justified, the County will issue RFP or CCD.
- G. <u>Hazardous Waste Conditions</u>: Contractor shall submit Notices of Hazardous Waste Conditions to resolve problems regarding hazardous materials encountered in the execution of the Work pursuant to paragraph 13.5 (Notice of Hazardous Waste or Materials Conditions) of Document 007200 (General Conditions), which shall govern. If the County determines that a change in Contract Sum or Contract Time is justified, the County will issue RFP or CCD.

#### 1.03 COST DETERMINATION

- A. Total cost of extra Work or of Work omitted shall be the sum of labor costs, material costs, equipment rental costs, and specialist costsas defined herein plus overhead and profit as allowed herein. This limit applies in all cases of claims for extra Work, whether calculating Cost Proposals, Change Orders or CCDs, or calculating contract claims of all types. Contractor may recover no other costs arising out of or connected with the performance of extra Work, of any nature. No special, incidental or consequential damages may be claimed or recovered against the County, its representatives or agents, whether arising from breach of Contract, negligence, or strict liability, unless specifically authorized in the Contract Documents.
- B. Overhead and Profit: (Overhead shall be as defined in paragraph 1.7 of this Section 012600)
  - Overhead and profit on labor for extra Work shall be fifteen percent (15%).
  - Overhead and profit on materials for extra Work shall be fifteen percent (15%).
  - Overhead and profit on equipment rental for extra Work shall be ten percent (10%).
  - When extra Work is performed by a first tier Subcontractor, Contractor shall receive a ten percent (10%) markup on Subcontractor's total costs of extra Work. First tier Subcontractor's markup on its Work shall not exceed fifteen percent (15%).
  - (5) When extra Work is performed by a lower tier Subcontractor, Contractor shall receive a total of ten percent (10%) markup on the lower tier Subcontractor's total costs of extra Work. Contractor and first tier Subcontractors and lower tier Subcontractors shall divide the fifteen percent (15%) markup as mutually agreed.
  - (6) Notwithstanding the foregoing, in no case shall the total markup on any extra Work exceed twenty-six and a half percent (26.5%) of the direct cost, notwithstanding the actual number of Contract tiers.
  - On proposals covering both increases and decreases in Contract Sum, overhead and profit shall be allowed on the net increase only as determined in this paragraph 1.3. When the net difference is a deletion, no percentage for overhead profit and commission shall be allowed, but rather a deduction shall apply.

(8) The markup shall include profit, small tools, cleanup, supervision, warranties, cost of preparing the cost proposal, jobsite overhead, and home office overhead. No markup will be allowed on taxes, insurance, and bonds.

## C. Taxes:

- (1) All State sales tax, use tax, and Alameda County taxes shall be included.
- (2) Federal and Excise tax shall not be included.
- D. <u>The County-Operated Equipment</u>: When owner-operated equipment is used to perform extra Work, Contractor will be paid for operator as follows:
  - (1) Payment for equipment will be made in accordance with paragraph 1.4.3 below.
  - Payment for cost of labor will be made at no more than rates of such labor established by collective bargaining agreements for type of worker and location of Work, whether or not owner-operator is actually covered by such an agreement.
- E. Accord and Satisfaction: Every Change Order and accepted CCD shall constitute a full accord and satisfaction, and release, of all Contractor (and if applicable, Subcontractor) claims for additional time, money or other relief arising from or relating to the subject matter of the change including, without limitation, impacts of all types, cumulative impacts, inefficiency, overtime, delay, and any other type of claim. Contractor may elect to reserve its rights to disputed claims arising from or relating to the changed Work at the time it signs a Change Order or approves a CCD, but must do so expressly in a writing delivered concurrently with the executed Change Order or approved CCD, and must also submit a Claim for the reserved disputed items pursuant to Article 12 (Claims by Contractor) of Document 007200 (General Conditions) no later than thirty (30) Days after Contractor's first written notice of its intent to reserve rights.

## 1.04 COST BREAKDOWN

A. <u>Labor</u>: Contractor will be paid cost of labor for workers (including forepersons when authorized by the County) used in actual and direct performance of extra Work. Contractor shall provide hourly wage rate breakdowns for all forces prior to acceptance by the County of any change orders and/or adjustments. Labor rate, whether employer is Contractor, Subcontractor or other forces, will be sum of following:

- (1) <u>Actual Wages</u>: Actual wages paid shall include any employer payments to or on behalf of workers for health and welfare, pension, vacation, and similar purposes.
- (2) <u>Labor Surcharge</u>: Payments imposed by local, county, state, and federal laws and ordinances, and other payments made to, or on behalf of, workers, other than actual wages as defined in paragraph 1.4.1.1 above, such as taxes and insurance.
- B. <u>Material</u>: Only materials furnished by Contractor and necessarily used in performance of extra Work will be paid for. Cost of such materials will be cost, including sales tax and delivery charges, to purchaser (Contractor, Subcontractor or other forces) from supplier thereof, except as the following are applicable:
  - (1) If cash or trade discount by actual supplier is offered or available to purchaser, it shall be credited to the County notwithstanding fact that such discount may not have been taken.
  - (2) For materials salvaged upon completion of extra Work, salvage value of materials shall be deducted from cost, less discounts, of materials.
  - (3) If cost of a material is excessive, then cost of material shall be deemed to be lowest reasonably available wholesale price at which material is available in quantities concerned delivered to Site, less any discounts as provided in paragraph 1.4.2.1 above.
- C. Equipment Rental: For Contractor- or Subcontractor-owned equipment, payment will be made at rental rates listed for equipment in California Department of Transportation official equipment rental rate schedule which is in effect on date upon which extra Work is accomplished and which schedule is incorporated herein by reference as though fully set forth herein. If there is no applicable rate for an item of equipment, then payment shall be made for Contractor- or Subcontractor-owned equipment at rental rate listed in the most recent edition of the Association of Equipment Distributors (AED) book. For rented equipment, payment will be made based on actual rental invoices. Equipment used on extra Work shall be of proper size and type. If, however, equipment of unwarranted size or type and cost is used, cost of use of equipment shall be calculated at rental rate for equipment of proper size and type, as determined by the County. Rental rates paid shall be deemed to cover cost of fuel, oil, lubrication, supplies, small tools, necessary attachments, repairs and maintenance of any kind, depreciation, storage, insurance, and all incidentals. Unless otherwise specified, manufacturer's ratings, and manufacturer-approved modifications, shall be used to classify equipment for determination of applicable rental rates. Individual pieces of equipment or tools not listed in said publication and having a

replacement value of \$100 or less, whether or not consumed by use, shall be considered to be small tools. Rental time will not be allowed while equipment is inoperative due to breakdowns.

- (1) For equipment on Site, rental time to be paid for equipment shall be time equipment is in operation on extra Work being performed or on standby as approved by the County. The following shall be used in computing rental time of equipment:
  - a. When hourly rates are listed, less than thirty (30) minutes of operation shall be considered to be ½ hour of operation.
  - b. When daily rates are listed, less than four (4) hours of operation shall be considered to be one-half (1/2) day of operation.
- (2) For equipment that must be brought to Site to be used exclusively on extra Work, cost of transporting equipment to Site and its return to its original location shall be determined as follows:
  - a. The County will pay for costs of loading and unloading equipment.
  - b. Cost of transporting equipment in low bed trailers shall not exceed hourly rates charged by established haulers.
  - c. Cost of transporting equipment shall not exceed applicable minimum established rates of California Public Utilities Commission.
  - d. The County will not make any payment for transporting and loading and unloading equipment if equipment is used on Work in any other way than upon extra Work.
- (3) Rental period may begin at time equipment is unloaded at Site of extra Work and terminate at end of the performance of the extra Work or day on which the County directs Contractor to discontinue use of equipment, whichever first occurs. Excluding Saturdays, Sundays, and the County's legal holidays, unless equipment is used to perform extra Work on such days, rental time to be paid per day shall be four (4) hours for zero (0) hours of operation, six (6) hours for four (4) hours of operation and eight (8) hours for eight (8) hours of operation, time being prorated between these parameters. Hours to be paid for equipment that is operated less than eight hours due to breakdowns,

shall not exceed eight less number of hours equipment is inoperative due to breakdowns.

D. Work Performed by Special Forces or Other Special Services: When the County and Contractor, by agreement, determine that special service or item of extra Work cannot be performed by forces of Contractor or those of any Subcontractors, service or extra Work item may be performed by specialist. Invoices for service or item of extra Work on basis of current market price thereof may be accepted without complete itemization of labor, material, and equipment rental costs when it is impracticable and not in accordance with established practice of special service industry to provide complete itemization. In those instances wherein Contractor is required to perform extra Work necessitating a fabrication or machining process in a fabrication or machine shop facility away from Site, charges for that portion of extra Work performed in such facility may, by agreement, be accepted as a specialist billing. The County must be notified in advance of all off-Site Work. In lieu of overhead and profit provided in paragraph 1.3.2 above, fifteen percent (15%) will be added to specialist invoice price, after deduction of any cash or trade discount offered or available, whether or not such discount may have been taken.

## 1.05 FORCE-ACCOUNT WORK

- A. If it is impracticable because of nature of Work, or for any other reason, to fix an increase or decrease in price definitely in advance, the Contractor may be directed to proceed at a not-to-exceed ("NTE") maximum price. Subject to such limitation, such extra Work shall be paid for at actual necessary cost for Force-Account Work or at the negotiated cost, as determined by the County. The cost for Force-Account Work shall be determined pursuant to paragraphs 1.3 and 1.4 above.
- B. Force-Account Work shall be used when it is not possible or practical to price out the changed Work prior to the start of that Work. In these cases, Force-Account Work will be utilized during the pricing and negotiation phase of the change. Once negotiations have been concluded and a bilateral agreement has been reached, the tracking of the Work under Force-Account is no longer necessary. Force-Account Work shall also be used when negotiations between the County and Contractor have broken apart and a bilateral agreement on the value of the changed Work cannot be reached. The County may approve other uses of Force-Account Work.
- C. Whenever any Force-Account Work is in progress, definite price for which has not been agreed on in advance, Contractor shall report to the County each Business Day in writing in detail amount and cost of labor and material used, and any other expense incurred in Force-Account Work on preceding day, by using

- the Cost Proposal form attached hereto. No claim for compensation for Force-Account Work will be allowed unless report shall have been made.
- D. Whenever Force-Account Work is in progress, definite price for which has not been agreed on in advance, Contractor shall report to the County when seventy-five percent (75%) of the NTE amount has been expended.
- E. Force-Account Work shall be paid as extra Work under this Section 012600. Methods of determining payment for Work and materials provided in this paragraph 1.5 shall not apply to performance of Work or furnishings of material that, in judgment of the County, may properly be classified under items for which prices are otherwise established in Contract Documents.

#### 1.06 COUNTY-FURNISHED MATERIALS

A. The County reserves right to furnish materials as it deems advisable, and Contractor shall have no claims for costs and overhead and profit on such materials.

#### 1.07 OVERHEAD DEFINED

- A. The following constitutes charges that are deemed included in overhead for all Contract Modifications, including Force-Account Work or CCD Work, whether incurred by Contractor, Subcontractors, or suppliers, and Contractor shall not invoice or receive payment for these costs separately:
  - (1) Drawings: field drawings, Shop Drawings, etc., including submissions of drawings
  - (2) Routine field inspection of Work proposed
  - (3) General Superintendence
  - (4) General administration and preparation of cost proposals, schedule analysis, Change Orders, and other supporting documentation as necessary
  - (5) Computer services
  - (6) Reproduction services
  - (7) Salaries of project engineer, superintendent, timekeeper, storekeeper, and secretaries
  - (8) Janitorial services

- (9) Small tools, incidentals and consumables
- (10) Temporary on-Site facilities:
  - a. Offices
  - b. Telephones
  - c. Plumbing
  - d. Electrical: Power, lighting
  - e. Platforms
  - f. Fencing, etc.
  - g. Water
- (11) Home office expenses
- (12) Procurement and use of vehicles and fuel used coincidentally in Work otherwise included in the Contract Documents
- (13) Surveying
- (14) Estimating
- (15) Protection of Work
- (16) Handling and disposal fees
- (17) Final cleanup
- (18) Other incidental Work
- (19) Related warranties

## 1.08 RECORDS AND CERTIFICATION

A. Force-Account (cost reimbursement) charges shall be recorded daily and summarized in Cost Proposal form attached hereto. Contractor or authorized representative shall complete and sign form each day. Contractor shall also provide with the form: the names and classifications of workers and hours worked by each; an itemization of all materials used; a list by size, type, and identification number of equipment and hours operated; and an indication of all Work performed by specialists.

- B. No payment for Force-Account Work shall be made until Contractor submits original invoices substantiating materials and specialists charges.
- C. The County shall have the right to audit all records in possession of Contractor relating to activities covered by Contractor's claims for Modification of Contract, including Force-Account Work and CCD Work.
- D. Further, the County will have right to audit, inspect, or copy all records maintained in connection with this Contract, including financial records, bidding records, in possession of Contractor relating to any transaction or activity occurring or arising out of, or by virtue of, the Contract. If Contractor is a joint venture, right of the County shall apply collaterally to same extent to records of joint venture sponsor, and of each individual joint venture member. This right shall be specifically enforceable, and any failure of Contractor to voluntarily comply shall be deemed an irrevocable waiver and release of all claims then pending that were or could have been subject to the Article 12 (Claims by Contractor) of Document 007200 (General Conditions).

#### 1.09 REGULATORY AGENCY APPROVAL OF CHANGE ORDERS

- A. The Architect shall be responsible for the submission and coordination of all Change Orders to the regulatory agency having jurisdiction. Changes in the approved plans and specifications for the Project shall be approved by the regulatory agency before proceeding with the Work. Submission to, and approval of regulatory agency for Change Orders to the Project shall not in any way invalidate or modify Document 007200 (General Conditions).
  - (1) Change Orders submitted to regulatory agency for review and approval shall state the reason for the change, show the related addition to or deduction from the current contract price, and shall be accompanied by supplementary plans, specifications, and other supporting documents, when necessary.
  - (2) All Change Orders submitted to regulatory agency shall be signed by the County's Representative and the Contractor, and approved by regulatory agency prior to performance of the work described in the Change Order, except under the following circumstances:
    - a. Emergency changes to the Work relating to the safety of persons at the construction site. These changes may be made immediately and shall be documented by subsequent change orders.

- b. To prevent undue delay, changes may be commenced following preliminary approval of an instruction bulletin by regulatory agency. Such changes shall be confirmed immediately by Change Order.
- B. <u>Impact of regulatory agency Review on Project Schedule</u>. Regulatory agency review of Change Orders is estimated to require from 40 to 80 days. During this review period, the Contractor cannot proceed with the work that is the subject of the Change Order, except as noted above in paragraphs 1.9.1.2.1 and 1.9.1.2.2. The responsibility for impacts to the Contractor's Project Schedule that are attributable to regulatory agency review of Change Orders shall be assigned as follows:
  - (1) Contractor Change Order Requests. Impacts to the Project Schedule caused by regulatory agency review of Contractor requested Change Orders, or Change Orders that result from any lack of quality, completeness, or coordination of any submittal by Contractor to regulatory agency, shall be the responsibility of the Contractor. The Contractor shall not be entitled to any extension of the Contract Time or additional compensation (including Compensable Delay) due to delays in the Contract Schedule caused by regulatory agency review.
  - County-Initiated Change Orders. Impacts to the Project Schedule caused by regulatory agency review of County-initiated Change Orders shall be the responsibility of the County. The Contractor shall incorporate the time and cost of regulatory agency's review in preparing its Cost Proposal to the County's Representative. The Contractor shall be entitled to adjustments to the Contract Sum and Contract Time that may be allowed under Article 15, Section 007200 (General Conditions).
  - Regulatory Agency Change Orders. The responsibility for impacts to the Project Schedule caused solely by regulatory agency review of Change Orders that result from Governmental or Regulatory Agency requirements shall be shared unless it is the result of unforeseen conditions in accordance with paragraph 13.4 (Notice of Concealed or Unknown Conditions) of Document 007200 (General Conditions). The Contractor shall incorporate the time and cost of regulatory agency's review in preparing its Cost Proposal to the County's Representative. Unless otherwise mutually agreed to by the parties, the Contractor shall be entitled to an adjustment to the Contract Time and fifty percent (50%) of any cost adjustments to the Contract Sum attributable to the adjustment in Contract Time that may be allowed under Document 007200.

## **Alameda County General Services Agency**

**PROJECT #2043** 

Highland Hospital – Koret Building Exhaust Fan and Waterproofing

**PART 2 - PRODUCTS** 

Not used.

**PART 3 - EXECUTION** 

Not used.

**END OF SECTION** 

COST PROPOSAL FORM FOLLOWS ON NEXT PAGE

## COST PROPOSAL (CP)

[Contract Number]	CP Number:
	Date:
	In Response To
	RFP #, etc.
To: []	
Attention: Contract Administration/Inspection	
[]	
	Subject Ref. No:
	(for Project Manager use only)
Phone:	Fax: ()
From: [Insert Contractor's Name/Address]	
This Cost Proposal is in response to the above-referenced applicable].	[insert RFP, etc. as
Brief description of change(s):	

ITEM DESCRIPTION	PRIME CONT R.	SUB 1	SUB 2	SUB 3	SUB 4	TOTAL
Material						
<b>Direct Labor Cost</b>						
Equipment						
Other (Specify) Extended Overhead						
Total Cost						
Subcontractor's Overhead & Profit (15 %)						
Contractor's Overhead & Profit (15%)						
Overhead & Profit to Contractor for Subcontractor's Work (5%)						
(percent of Total Cost above not including any Overhead & Profit)						
GRAND TOTAL						
REQU	ESTED CH	HANGE IN	N CONTRA	ACT TIME	E (DAYS)	

By Contractor:	Signature:	Date:

Project Name		
Change Order Title		
PCO No:		
Date		
PCO Desciption		
DESC	Contractor	Note
Design	\$0	
Subtotal Design	\$0.00	
Subcontractor #1	\$0	
Subcontractor #2	\$ -	
Subtotal Subcontractors	\$ -	
GC Costs		
Subtotal GC	\$0	
Subtotal Change Cost	\$0	
Design Fee, sliding scale %	\$0	
Subtotal	\$0	
Markup, 5%	\$0	
GC's Insurance, CCIP, 1.950%		
GC's Insurance, Offsite CCIP, .0075%	\$0	
Bond, 1.80625% - including MU	\$0	
Total	\$0	
Estimate Notes:		

#### **SECTION 01 29 00**

#### PAYMENT PROCEDURES

#### **PART 1 - GENERAL**

#### 1.1 SUMMARY

1.1.1 This section describes requirements and procedures for determining amount of work done and for obtaining payment for work done.

# 1.1.2 Related Sections.

- 1.1.2.1 Section 01 11 00 (Summary of Work)
- 1.1.2.2 Section 01 26 00 (Contract Modification Procedures)
- 1.1.2.3 Section 01 32 16 (Schedules and Reports)
- 1.1.2.4 Section 01 33 00 (Submittal Procedures)
- 1.1.2.5 Section 01 77 00 (Construction Closeout)

#### 1.2 REFERENCES

California Public Contract Code

#### 1.3 SCOPE OF WORK

Work under the Contract Documents, or under any Bid item, allowance or alternate, shall include all labor, materials, transport, handling, storage, supervision, administration and all other items necessary for the satisfactory completion of Work, whether or not expressly specified or shown.

#### 1.4 DETERMINATION OF QUANTITIES

Quantity of work to be paid for under any item for which a unit price is fixed in the Contract Documents shall be the number, as determined by County, of units of work satisfactorily completed in accordance with Plans and Specifications and as directed pursuant to Plans and Specifications. Unless otherwise provided, determination of number of units of work so completed will be based, so far as practicable, on actual measurement or count within prescribed or ordered limits, and no payment will be made for work done outside of limits. Measurements and computations will be made by methods as County may consider appropriate for class of work measured.

# 1.5 SCOPE OF PAYMENT

1.5.1 Except as otherwise expressly stipulated in Section 01 11 00 (Summary of Work), payment to Contractor at the unit price or other price fixed in the Contract for performing the work required under any item, or (if the Contract is on a lump sum price basis) at the lump sum price fixed in the Contract for performing all work required under the Contract, and as either may be adjusted pursuant to any approved change order, shall be full compensation for completing, in accordance with the Contract Documents, all work

required under the item or under the Contract, and for all expense incurred by Contractor for any purpose in connection with the performance and completion of said work, including all incidental work necessary for completion of the Work.

- 1.5.2 The Contract Sum, whether lump sum, unit price or otherwise, shall be deemed to include all costs necessary to complete required Work.
- 1.5.3 Whenever it is specified herein that Contractor is to do work or furnish materials of any class for which no price is fixed in the Contract Documents, it shall be understood that Contractor is to do such work or furnish such materials without extra charge or allowance or direct payment of any sort, and that cost of doing work or furnishing materials is to be included in its Proposal price, unless it is expressly specified herein, in particular cases, that work or material is to be paid for as extra work.
- 1.5.4 For the materials and equipment referenced in Section 01 11 00 (Summary of Work) as subject to payment prior to incorporation into the work, where Contractor requests payment on the basis of such materials and equipment not incorporated in the Work, Contractor must satisfy the following conditions:
  - 1.5.4.1 The materials and/or equipment shall be delivered and suitably stored at the Site or at another local location agreed to in writing, for example, a mutually acceptable warehouse;
  - 1.5.4.2 Full title to the materials and/or equipment shall vest in County at the time of delivery to the Site, warehouse or other storage location;
  - 1.5.4.3 Contractor shall obtain a negotiable warehouse receipt, endorsed over to County for materials and/or equipment stored in an off-site warehouse. No payment shall be made until such endorsed receipts are delivered to County;
  - 1.5.4.4 Stockpiled materials and/or equipment shall be available for County's inspection, but County shall have no obligation to inspect them and its inspection or failure to inspect shall not relieve Contractor of any obligations under the Contract Documents. Materials and/or equipment shall be segregated and labeled or tagged to specifically identify these specific Contract Documents;
  - 1.5.4.5 After delivery of materials and/or equipment, if any inherent or acquired defects are discovered, defective materials and/or equipment shall be removed and replaced with suitable materials and/or equipment at Contractor's expense;
  - 1.5.4.6 At its expense, Contractor shall insure the materials and/or equipment against theft, fire, vandalism, and malicious mischief, as well as any other coverages required under the Contract Documents;
  - 1.5.4.7 Contractor's application for payment shall be accompanied by a bill of sale, invoice or other documentation warranting that County has received the

materials and equipment free and clear of all liens and evidence that the materials and equipment are covered by appropriate property insurance and other arrangements to protect County's interest therein, all of which must be satisfactory to County. This documentation shall include, but not be limited to, conditional releases of mechanics' liens and stop notices from all those providing materials and equipment as to which the application for payment relates, as well as unconditional releases of the same from the same as to the previous applications for payment for which they have not already been provided.

#### 1.6 BASIS OF PAYMENT

- 1.6.1 <u>Unit Pay Quantities</u>: Not applicable.
- 1.6.2 <u>Lump Sum</u>: When estimated quantity for specific portion of Work is not indicated and unit is designated as Lump Sum, payment will be on a Lump Sum basis for Work satisfactorily completed in accordance with Plans and Specifications.

#### 1.7 PROGRESS PAYMENTS

1.7.1 If requested by Contractor, progress payments will be made monthly.

#### 1.7.2 Schedule of Values.

- 1.7.2.1 Within the time frame set in Document 007200 (General Conditions), Contractor shall submit a detailed breakdown of its Bid by scheduled Work items and/or activities, including coordination responsibilities and project record document responsibilities. The Project Schedule (see Section 01 31 26 (Schedules and Reports)) shall serve as the basis for developing the Schedule of Values. Contractor shall furnish such breakdown, of the total Contract Sum, by assigning dollar values (cost estimates) to each applicable Progress Schedule network activity, which cumulative sum equals the total Contract Sum. The format and detail of the breakdown shall correspond directly with the Project Schedule and as directed by County to facilitate and clarify future progress payments to Contractor for direct Work under the Contract Documents. This breakdown shall be referred to as the Schedule of Values.
- 1.7.2.2 Contractor's overhead, profit, insurance, cost of bonds and/or other financing, as well as "general conditions costs," (e.g., site cleanup and maintenance, temporary power and lighting, security and the like), shall be prorated through Project duration.
- 1.7.2.3 County will review the breakdown in conjunction with the Progress Schedule to ensure that the dollar amounts of this Schedule of Values are, in fact, fair market cost allocations for the Work items listed. Upon favorable review by County, this Schedule of Values will be accepted for use by County. County shall be the sole judges of fair market cost allocations.

1.7.2.4 Any attempt to increase the cost of early activities, i.e., "front loading," will be rejected by County, resulting in a complete reallocation of monies until such "front loading" is corrected.

#### 1.7.3 Payment Requests

- 1.7.3.1 Unless otherwise agreed, Contractor shall submit to the County, during the final week of each payment period, a draft request for payment projected out through the end of the payment period. The County and Contractor shall meet to review the draft request for payment to reach consensus on percentage completion. The Contractor shall revise and resubmit request for payment to include agreed upon percentages complete.
- 1.7.3.2 Unless otherwise agreed, Contractor shall submit to County, on or before the first (1st) day of each payment period five (5) copies of the revised request for payment for the cost of the Work put in place during the previous onemonth payment period. Such requests for progress payments shall be based upon Schedule of Values prices of all labor and materials incorporated in the Work up until midnight of the last day of that one month period, less the aggregate of previous payments. If Contractor is late submitting its payment request, that payment request may be processed at any time during the succeeding one month period, resulting in processing of Contractor's payment request being delayed for more than a day for day basis.
- 1.7.3.3 Payment requests may include, but are not necessarily limited to the following:
  - 1.7.3.3.1 Material, equipment and labor incorporated into the Work, less any previous payments for the same;
  - 1.7.3.3.2 Up to one hundred percent (100%) of the cost of major equipment identified in paragraph 1.5.4 above, if purchased and delivered to the Site or stored off site, as may be approved by County.
  - 1.7.3.3.3 Up to one hundred percent (100%) of the cost of materials identified in paragraph 1.5.4 above specifically fabricated for the Project that are not yet incorporated into the Work.
- 1.7.3.4 Contractor shall, at the time any payment request is submitted, certify in writing the accuracy of the payment request and that Contractor has fulfilled all scheduling requirements of Document 00 72 00 (General Conditions) and Section 01 32 16 (Schedules and Reports), including updates and revisions. The certification shall be executed by a responsible officer of Contractor.
- 1.7.3.5 No progress payment will be processed prior to County receiving all requested, acceptable schedule update information.

- 1.7.3.6 Each payment request shall be organized to provide amounts due for Work related to each major phase of work including but not limited to: (a) Exhaust Fan Project, (b) Waterproofing Project.
- 1.7.3.7 Each payment request shall list each Change Order executed prior to date of submission, including the Change Order Number, and a description of the work activities, consistent with the descriptions of original work activities. Contractor shall submit a monthly Change Order status log to County with their payment request.
- 1.7.3.8 If County requires substantiating data, Contractor shall submit information requested by County, with cover letter identifying Project, payment request number and date, and detailed list of enclosures. Contractor shall submit one copy of substantiating data and cover letter for each copy Payment request submitted.
- 1.7.3.9 Monthly progress payments shall be made, based on total value of activities completed or partially completed, as determined by County with participation of Contractor, and based upon approved activity costs. Accumulated retainage will be shown as separate item in payment summary. If Contractor fails or refuses to participate in construction progress evaluation with County, Contractor shall not receive current payment until Contractor has participated fully in providing construction progress information and schedule update information for County.
- 1.7.3.10 Legal title to all Work shall pass to and vest in County as Work is performed, and title to all materials and equipment shall pass to and vest in County when such materials and equipment are delivered to the Site or as soon as title passes from the vendor or supplier thereof. Contractor shall keep the Site and all materials and equipment free and clear of all liens, stop notices and charges arising out of performance of the Contract Documents, and shall indemnify, defend and hold harmless all those indemnified pursuant to Document 007200 (General Conditions) from the claims, suits, actions, losses and liabilities described therein, including those which are a result of any breach of this responsibility and shall defend any claim or suit brought against any party required to be indemnified hereunder based upon any such claim of title or lien.
- 1.7.3.11 Contractor shall promptly pay each Subcontractor the amount to which such Subcontractor is entitled, and shall, by an appropriate agreement with each Subcontractor, require each Subcontractor to make payments to its sub-Subcontractors in a similar manner. Contractor shall submit on its behalf and on behalf of each Subcontractor or consultant for which payment is being requested a conditional release of stop notice in statutory form for the Work which is the subject of each progress payment request and an unconditional release of stop notice in statutory form for the immediately preceding progress payment as to the Work of each. If the unconditional release of stop notice for the amount paid for the immediately preceding month is not

reasonably available, it shall be submitted with the next progress payment request, such that it is submitted no more than two months after the date on which the conditional release of stop notice was submitted for the amount paid.

Contractor shall submit Document 012900A Conditional Waiver and Release on Progress Payment (see Attachment A) stating the "Amount of Check" due Contractor in the current Application for Progress Payment; and Document 012900B Unconditional Waiver and Release on Progress Payment (see Attachment B) stating the amount of the previous progress payment received by Contractor.

#### 1.7.4 Progress Payments

- 1.7.4.1 Upon receiving Contractor's payment request, County will review the payment request and make necessary adjustments to percent of completion of each activity. One copy will be returned to Contractor with description of adjustments made. All parties will update percentage of completion values in the same manner, i.e., express value of an accumulated percentage of completion to date.
- 1.7.4.2 The payment request may be reviewed by County for the purpose of determining that the payment request is a proper payment request, and shall be rejected, revised or approved by County pursuant to the cost breakdown prepared in accordance with this Section.
- 1.7.4.3 If it is determined that the payment request is not a proper payment request suitable for payment, County shall return it to Contractor as soon as practicable, but no later than seven (7) Days after receipt, together with a document setting forth in writing the reasons why the payment request is not proper. If County determines that portions of the payment request is not proper or not due under the Contract Documents, then County may approve the other portions of the payment request, and in the case of disputed items or defective work not remedied, may withhold up to one hundred fifty percent (150%) of the disputed amount from the progress payment.
- 1.7.4.4 Pursuant to Public Contract Code § 20104.50, if County fails to make any progress payment within thirty (30) Days after receipt of an undisputed and properly submitted payment request from a Contractor, County shall pay interest to Contractor equivalent to the legal rates set forth in subdivision (a) of Code of Civil Procedure § 685.010. The thirty (30) Day period shall be reduced by the number of days by which County exceeds the seven (7) Day return requirement set forth herein.
  - 1.7.4.4.1 If County fails to make a payment as described in paragraph 1.7.4.4 above, Contractor may give County written notice of Contractor's intent to suspend work for lack of payment.

- 1.7.4.4.2 Only after having given such notice, and no earlier than thirty (30) Days after serving such notice on the County, Contractor may suspend any or all work required of it by the Contract Documents for thirty (30) Days or for such other time as the County and Contractor mutually agree.
- 1.7.4.4.3 Following such noticed suspension of work, the Contractor may terminate this Agreement. Contractor's compensation and damages related to a termination pursuant to this paragraph 1.7.4.4.1 shall be limited to that described in Document 007200 (General Conditions).
- 1.7.4.5 As soon as practicable after approval of each request for progress payment, County will pay to Contractor in manner provided by law, an amount equal to ninety-five percent (95%) of County's estimate, or a lesser amount if so provided in Contract Documents and County shall retain the amount so withheld as retention. Provided that payments may at any time be withheld if Work is not proceeding in accordance with the Contract Documents, or Contractor is not complying with requirements of the Contract Documents, or to comply with stop notices or to offset liquidated damages accruing or expected.
- 1.7.4.6 Retention will not be reduced if Contractor is behind schedule. If retention is reduced at any point during performance of the Work and Contractor subsequently falls behind schedule, retention may be raised back to original percentage specified in paragraphs 1.7.4.5 and 1.7.4.6 of this Section.
- 1.7.4.7 Before any progress payment or final payment is made, Contractor is required to submit satisfactory evidence that Contractor is not delinquent in payments to employees, Subcontractors, suppliers, or creditors for labor and materials incorporated into Work.
- 1.7.4.8 County reserves and shall have the right to withhold payment for any equipment and/or specifically fabricated materials that, in the sole judgment of County, is not adequately and properly protected against weather and/or damage, prior to or following incorporation into the Work.
- 1.7.4.9 Granting of progress payment or payments by County, or receipt thereof by Contractor, shall not be understood as constituting in any sense acceptance of Work or of any portion thereof, and shall in no way lessen liability of Contractor to replace unsatisfactory work or material, though unsatisfactory character of work or material may have been apparent or detected at time payment was made.
- 1.7.4.10 When County shall charge a sum of money against Contractor under any provision of the Contract Documents, amount of charge shall be deducted and retained by County from amount of next succeeding progress payment or from any other moneys due or that may become due Contractor under the

Contract Documents. If, on completion or termination of the Contract Documents, such moneys due Contractor are found insufficient to cover County's charges against it, County shall have right to recover balance from Contractor or Sureties.

# 1.8 SUBSTITUTION OF SECURITIES IN LIEU OF RETENTION

Pursuant to provisions of Public Contract Code § 22300, incorporated herein by reference, substitution of securities for any monies withheld under the Contract Documents to insure performance is permitted under following conditions:

- 1.8.1 At request and expense of Contractor, securities listed in Government Code § 16430, bank or savings and loan certificates of deposit, interest bearing demand deposit accounts, standby letters of credit, or any other security mutually agreed to by Contractor and County which are equivalent to the amount withheld under retention provisions of the Contract Documents shall be deposited with County or with a state or federally chartered bank in California, as the escrow agent, who shall then pay such monies to Contractor. Upon satisfactory completion of the Contract Documents, securities shall be returned to Contractor.
- 1.8.2 Alternatively, Contractor may request and County shall make payment of retentions earned directly to the escrow agent at the expense of Contractor. At the expense of Contractor, Contractor may direct the investment of the payments into securities and Contractor shall receive the interest earned on the investments upon the same terms provided for in this section for securities deposited by Contractor. Upon satisfactory completion of the Contract Documents, Contractor shall receive from escrow agent all securities, interest, and payments received by the escrow agent from County, pursuant to the terms of this section. Contractor shall pay to each Subcontractor, not later than twenty (20) days after receipt of the payment, the respective amount of interest earned, net of costs attributed to retention withheld from each Subcontractor, on the amount of retention withheld to insure the performance of Contractor.
- 1.8.3 Contractor shall be beneficial owner of securities substituted for monies withheld and shall receive any interest thereon. Contractor shall submit any form W-9 or other required tax form.
- 1.8.4 Contractor shall enter into escrow agreement according to Document 006140 (Escrow Agreement for Security Deposits in Lieu of Retention) as authorized under Public Contract Code § 22300, specifying amount of securities to be deposited, terms and conditions of conversion to cash in case of default of Contractor, and termination of escrow upon completion of the Contract Documents.

#### 1.9 FINAL PAYMENT

1.9.1 As soon as practicable after all required Work is completed in accordance with the Contract Documents, including Contractor's maintenance after Final Acceptance, County will pay to Contractor, in manner provided by law, unpaid balance of Contract price of Work, or whole Contract price of Work if no progress payment has been made,

determined in accordance with terms of the Contract Documents, less sums as may be lawfully retained under any provisions of the Contract Documents or by law.

- 1.9.2 Prior progress payments shall be subject to correction in the final payment. County's determination of amount due as final payment shall be final and conclusive evidence of amount of Work performed by Contractor under the Contract Documents, and shall be full measure of compensation to be received by Contractor.
- 1.9.3 Contractor and each assignee under an assignment during the course of the Work shall execute and deliver at time of final payment and as a condition precedent to final payment, Documents 00 65 19.26 (Final Settlement Certificate Form), discharging County, its officers, agents, employees and consultants, of and from liabilities, obligations, and claims arising under the Contract Documents.
- 1.9.4 Contractor shall execute and deliver at time of final payment and as a condition precedent to final payment, Document 012900C Conditional Waiver and Release on Final Payment (see Attachment C) and Document 012900D Unconditional Waiver and Release on Final Payment (see Attachment D).

#### 1.10 EFFECT OF PAYMENT

Payment will be made by County, based on County's observations at the Site and the data comprising the application for payment. Payment will not be a representation that County has:

- 1.10.1 Made exhaustive or continuous on-site inspections to check the quality or quantity of Work;
- 1.10.2 Reviewed construction means, methods, techniques, sequences or procedures;
- 1.10.3 Reviewed copies of requisitions received from Subcontractors and material suppliers and other data requested by County to substantiate Contractor's right to payment; or
- 1.10.4 Made an examination to ascertain how or for what purpose Contractor has used money previously paid on account of the Contract Sum.

#### 1.11 COUNTY PROJECT FINANCING AND PERFORMANCE OF PAYMENT OBLIGATIONS

Upon written request of the Contractor, the County shall provide to Contractor, not more frequently than quarterly, information concerning the status of its project financing. If Contractor at any time reasonably believes that the County is or may be unable to make payment as payment becomes due under the Contract, Contractor may notify the County of this fact and request written assurances from the County of performance of its payment obligations. The County shall, within ten (10) Days of Contractor's request, provide written adequate assurances of performance of its payment obligations.

#### **PART 2 - PRODUCTS**

Not used.

ALAMEDA COUNTY GSA-CP Page 9 of 10

PROJECT #2043

**Alameda County General Services Agency** Highland Hospital – Koret Building Exhaust Fan and Waterproofing

**PART 3 - EXECUTION** 

Not used.

END OF SECTION

# **Alameda County, General Services Agency**

**Identifying Information** 

Highland Hospital – Koret Building Exhaust Fan and Waterproofing ATTACHMENT A

#### CONDITIONAL WAIVER AND RELEASE ON PROGRESS PAYMENT

NOTICE: THIS DOCUMENT WAIVES THE CLAIMANT'S LIEN, STOP PAYMENT NOTICE, AND PAYMENT BOND RIGHTS EFFECTIVE ON RECEIPT OF PAYMENT. A PERSON SHOULD NOT RELY ON THIS DOCUMENT UNLESS SATISFIED THAT THE CLAIMANT HAS RECEIVED PAYMENT.

Name of Claimant: Name of Customer: Job Location: Owner: Through Date:	
Conditional Waiver a	and Release
claimant has for labor customer on this job to service provided, or ec- been fully executed by are waived and release	s and releases lien, stop payment notice, and payment bond rights the or and service provided, and equipment and material delivered, to the through the Through Date of this document. Rights based upon labor or quipment or material delivered, pursuant to a written change order that has the parties prior to the date that this document is signed by the claimant, and by this document, unless listed as an Exception below. This document is claimant's receipt of payment from the financial institution on which the wn:
Maker of Check: Amount of Check: Check Payable to:	\$
<ol> <li>Retentions.</li> <li>Extras for which th</li> <li>The following prog waiver and release but Date(s) of waiver and a Amount(s) of unpaid p</li> <li>Contract rights, inc</li> </ol>	e claimant has not received payment. gress payments for which the claimant has previously given a conditional has not received payment: release: progress payment(s): \$ luding (A) a right based on rescission, abandonment, or breach of ght to recover compensation for work not compensated by the payment.
Claimant's Signature: Claimant's Title: Date of Signature:	

ALAMEDA COUNTY GSA-CP

**Identifying Information** 

# ATTACHMENT B

#### UNCONDITIONAL WAIVER AND RELEASE ON PROGRESS PAYMENT

NOTICE TO CLAIMANT: THIS DOCUMENT WAIVES AND RELEASES LIEN, STOP PAYMENT NOTICE, AND PAYMENT BOND RIGHTS UNCONDITIONALLY AND STATES THAT YOU HAVE BEEN PAID FOR GIVING UP THOSE RIGHTS. THIS DOCUMENT IS ENFORCEABLE AGAINST YOU IF YOU SIGN IT, EVEN IF YOU HAVE NOT BEEN PAID. IF YOU HAVE NOT BEEN PAID, USE A CONDITIONAL WAIVER AND RELEASE FORM.

Name of Claimant:  Name of Customer:  Job Location:  Owner:  Through Date:	
<b>Unconditional Waiver</b>	and Release
claimant has for labor customer on this job the service provided, or equipment fully executed by are waived and released received the following p	and releases lien, stop payment notice, and payment bond rights the and service provided, and equipment and material delivered, to the rough the Through Date of this document. Rights based upon labor or ipment or material delivered, pursuant to a written change order that has he parties prior to the date that this document is signed by the claimant, by this document, unless listed as an Exception below. The claimant has rogress payment:
Exceptions	
	affect any of the following:
(1) Retentions.	alaiment has not received normant
(3) Contract rights, incl	claimant has not received payment.  Iding (A) a right based on rescission, abandonment, or breach of to recover compensation for work not compensated by the payment.
Signature	
Claimant's Signature: _ Claimant's Title: _ Date of Signature: _	

**Identifying Information** 

#### ATTACHMENT C

# CONDITIONAL WAIVER AND RELEASE ON FINAL PAYMENT

NOTICE: THIS DOCUMENT WAIVES THE CLAIMANT'S LIEN, STOP PAYMENT NOTICE, AND PAYMENT BOND RIGHTS EFFECTIVE ON RECEIPT OF PAYMENT. A PERSON SHOULD NOT RELY ON THIS DOCUMENT UNLESS SATISFIED THAT THE CLAIMANT HAS RECEIVED PAYMENT.

Name of Claimant: Name of Customer: Job Location: Owner:		
Conditional Waiver	r and Release	
claimant has for lal customer on this job delivered, pursuant the date that this docunless listed as an Experience.	bor and service provided, and ob. Rights based upon labor or so a written change order that has cument is signed by the claimant,	nent notice, and payment bond rights the equipment and material delivered, to the ervice provided, or equipment or material been fully executed by the parties prior to are waived and released by this document, as effective only on the claimant's receipt of llowing check is drawn:
Maker of Check: Amount of Check: Check Payable to:	\$	<u> </u>
Exceptions		
	not affect any of the following: extras in the amount of: \$	
Signature		
Claimant's Signature Claimant's Title: Date of Signature:	::	<u></u>

**REVISED 10/04/17** 

# **Alameda County, General Services Agency**

**Identifying Information** 

Highland Hospital – Koret Building Exhaust Fan and Waterproofing

#### ATTACHMENT D

# UNCONDITIONAL WAIVER AND RELEASE ON FINAL PAYMENT

NOTICE TO CLAIMANT: THIS DOCUMENT WAIVES AND RELEASES LIEN, STOP PAYMENT NOTICE, AND PAYMENT BOND RIGHTS UNCONDITIONALLY AND STATES THAT YOU HAVE BEEN PAID FOR GIVING UP THOSE RIGHTS. THIS DOCUMENT IS ENFORCEABLE AGAINST YOU IF YOU SIGN IT, EVEN IF YOU HAVE NOT BEEN PAID. IF YOU HAVE NOT BEEN PAID, USE A CONDITIONAL WAIVER AND RELEASE FORM.

Name of Claimant:		
Job Location:		
Owner:		_
Unconditional Waive	er and Release	
claimant has for all l customer on this job. delivered, pursuant to the date that this docu	es and releases lien, stop payment notice abor and service provided, and equipment Rights based upon labor or service prov a written change order that has been fully ment is signed by the claimant, are waived ception below. The claimant has been paid i	t and material delivered, to the ided, or equipment or material executed by the parties prior to and released by this document,
Exceptions		
This document does no	ot affect the following:	
Disputed claims for ex	stras in the amount of: \$	
Signature		
Claimant's Signature:		
Date of Signature:		

Highland Hospital – Koret Building Exhaust Fan and Waterproofing DOCUMENT 01 31 00

# PROJECT MANAGEMENT AND COORDINATION

#### PART 1 - GENERAL

#### 1.01 SUMMARY

- A. This section describes requirements for job site administration, including:
  - (1) County's Project Manager
  - (2) Contractor's Project Management Team.

# B. Related Sections.

- (1) Section 011100 (Summary of Work)
- (2) Section 013216 (Schedules and Reports)
- (3) Section 013300 (Submittal Procedures)
- (4) Section 017839 (Project Record Documents)

#### 2.01 COUNTY'S MANAGEMENT TEAM

- A. County shall be represented on this Contract by Hadi Hadjarzadeh, serving as County's Project Manager, who will act personally or through designated County Representative(s). The County has designated the Project Manager to represent County in carrying out the duties of County. The County may delegate all or a portion of the Project Manager's duties to a Construction Manager or other County Representative, which shall then perform all or a portion of the Project Manager's duties specified herein.
- B. Functions of the Project Manager include, but are not limited to, the following:
  - (1) The Project Manager functions as the primary County representative with the Contractor in all matters concerning the Contract, monitoring the Contractor's performance in all respects to ascertain that the Work is performed in accordance with all the requirements of the Contract.
  - (2) The Project Manager is the focal point of contact with the Contractor regarding clarification of discrepancies and resolution of questions of fact that arise during performance of the Work under the Contract.

    The Project Manager also performs this role with regard to all agency and utility construction interfaces with the Work under this Contract.
  - (3) The Contractor is required by the Contract to provide formal notice of any and all potential claims arising during the performance of the Work. The Project Manager will administer the processing and resolution of any such claims in accordance with the requirements of the Contract.

(4) All contractual correspondence, including submittals, shall be directed and processed through the Project Manager unless otherwise specifically directed in the Contract. Any required or requested interface between the Contractor and County, the Project Manager, or any other representative of County, will be coordinated by the Project Manager.

#### 3.01 CONTRACTOR'S PROJECT MANAGEMENT TEAM

- A. The Contractor shall staff the Project with a management team qualified and experienced in construction of a public works project of this value, nature and complexity. This team shall possess the competency, skills and authority specified in Article 6 and 7 of Document 007200 (General Conditions).
  - (1) The Contractor shall submit to County prior to Notice to Proceed the names, detailed project experience, references, and proposed project position for each team member. Key team members shall have appropriate experience in the proposed position.
  - The Contractor shall not replace members of the Contractor's management team without prior written approval of the County. If, during the course of the Project, the Contractor finds it necessary to replace a member of the Project Management Team, the name, qualifications, and experience of the proposed replacement shall be submitted to County for approval. The County reserves the right in its discretion not to pay for staff to be trained and oriented to become familiar with the project.
  - Any staff deemed by the County to be unsuitable will be replaced with staff that is acceptable to the County.
  - (4) The Contractor shall have a dedicated safety manager responsible for maintaining a safe working place as described in the approved Safety Plan required by Section 013300 (Submittal Procedures).
- B. The Project Management Team shall be composed of members with the necessary skills and be sufficient in number to handle all duties normal to a project of this scale and complexity. Special attention shall be given to the responsibility for coordination, scheduling, and working in an occupied hospital facility.

# **PART 2 - PRODUCTS**

Not used.

**PART 3 - EXECUTION** 

Not used.

END OF SECTION

# **DOCUMENT 01 31 19**

# **PROJECT MEETINGS**

#### PART I – GENERAL

#### 1.01 RELATED DOCUMENTS AND PROVISIONS:

All Contract Documents should be reviewed for applicable provisions related to the provisions in this document, including without limitation:

- A. General Conditions
- B. Special Conditions

# 1.02 PRECONSTRUCTION CONFERENCE:

The contractor shall attend a conference a the Project Site prior to the start of construction for the purpose of determining Contractor's access to, and use of the site, verifying utilities, review construction administrative procedures, and such other items as may be pertinent to the start of construction.

# 1.03 PROGRESS MEETINGS:

- A. Construction Manager shall schedule and hold regular weekly progress meetings after a minimum of one week's prior written notice of the meeting date and time to all Invitees as indicated below. Progress Meetings shall review items such as the status of the work, Methods of Procedures (MOPs), Look Ahead Schedule, Monthly Update Schedule, Proposed Change Orders, Change Orders, OSHPD inspections, Quality Control, and Commissioning.
- B. Location: Contractor's field office.
- C. The Contractor shall notify and invite the following entities ("Invitees"):
  - (1) County Representative(s).
  - (2) Contractor.
  - (3) Contractor's Project Manager.
  - (4) Contractor's Superintendent.
  - (5) Subcontractors/suppliers, as appropriate to the agenda of the meeting.
  - (6) Inspector of record.

- (7) Architect and/or Engineer(s), if any and as appropriate to the agenda of the meeting.
- (8) Facility Representative
- (9) Others, as appropriate to the agenda of the meeting.
- D. The County's, the Architect's, and/or an engineer's Consultants will attend at their discretion, in response to the agenda.
- E. The County representative, the Construction Manager, and/or another County Agent shall take and distribute meeting notes to attendees and other concerned parties. If exceptions are taken to anything in the meeting notes, those exceptions shall be stated in writing to the County within five (5) working days following County's distribution of the meeting notes.

# 1.04 PRE-INSTALLATION/PERFORMANCE MEETING:

- A. Contractor shall schedule a meeting prior to the start of each of the following portions of the Work: cutting and patching of plaster and roofing, other weather-exposed and moisture-resistant products, and other pre-installation meetings outlined in the technical specifications. Contractor shall invite all Invitees to this meeting, and others whose work may affect or be affected by the quality of the cutting and patching work.
- B. Contractor shall review in detail prior to this meeting, the manufacturer's requirements and specifications, applicable portions of the Contract Documents, Shop Drawings, and other submittals, and other related work. At this meeting, invitees shall review and resolve conflicts, incompatibilities, or inadequacies discovered or anticipated.
- C. Contractor shall review in detail Project conditions, schedule, requirements for performance, application, installation, and quality of completed Work, and protection of adjacent Work and property.
- D. Contractor shall review in detail means of protecting the completed Work during the remainder of the construction period.

#### 1.05 SPECIAL MEETINGS:

Special meetings may be requested by the County. Contractor, subcontractors, material suppliers and any other members of the project team may be required to attend.

END OF DOCUMENT

#### DOCUMENT 013216

# SCHEDULES AND REPORTS

#### **PART 1 - GENERAL**

#### 1.1 CONSTRUCTION SCHEDULE

- 1.1.1 The Construction Schedule shall adhere to times in Document 005200 (Agreement). The Construction Schedule (original and updates) shall indicate all separate fabrication, procurement and field construction activities required for completion of the Work. Failure to include any work item required for performance of this Contract on the Schedule shall not excuse Contractor from completing all work within applicable completion dates, regardless of County's approval of the schedule.
- 1.1.2 Activities: All Contractor and Subcontractor work shall be shown in a logical work sequence that demonstrates a coordinated plan of work for all Subcontractors. The intent is to provide a common basis of acceptance, understanding, and communication, as well as interface among all Subcontractors. Indicate the estimated time duration, sequence requirements, and relationship of each activity in relation to other activities. Show dependencies and logic between activities so that the effect of progress (or lack of progress) on related activities and the overall schedule can be monitored. The list of activities shall include, but not be limited to, the following:
  - 1.1.2.1 Submittal Preparation and Review: Include review and resubmittal times indicated in Section 013000 (Submittals) in schedule. Coordinate submittal review times in Contractor's Construction Schedule with Submittals Schedule. Phase the submittal process to ensure that items are submitted in order of their importance to the construction process. Implement a system that staggers submittals by "start no earlier than" date in a number of waves.
  - 1.1.2.2 Include procurement process activities for long lead items and major items, requiring a cycle of more than sixty (60) Days, as separate activities in schedule. Procurement cycle activities include, but are not limited to, submittals, approvals, purchasing, fabrication, and delivery. Delivery dates indicated stipulate the earliest possible delivery date.
  - 1.1.2.3 Significant construction milestones including but not limited to:
    - 1.1.2.3.1 Mobilization.
    - 1.1.2.3.2 Earthwork and underground utility site work completion.
    - 1.1.2.3.3 Foundation completion.
    - 1.1.2.3.4 HVAC installation.
    - 1.1.2.3.5 Electrical installation.
    - 1.1.2.3.6 Caulking completion.
    - 1.1.2.3.7 Elastomeric Covering installation.
    - 1.1.2.3.8 Completion.
    - 1.1.2.3.9 Demobilization.

- 1.1.2.4 Dates for decision from County on designated items or orders affecting schedule.
- 1.1.2.5 Utility interruptions, relocation, and connections.
- 1.1.2.6 Connecting to or penetrating existing structures.
- 1.1.2.7 Show the effect of the following factors on the construction schedule:
  - 1.1.2.7.1 Limitations of continued occupancies.
  - 1.1.2.7.2 Uninterruptible services.
  - 1.1.2.7.3 Use of premises restrictions.
  - 1.1.2.7.4 Environmental control.
- 1.1.2.8 Punch list preparation.
- 1.1.2.9 Work by County and/or by other contractors that may affect or be affected by Contractor's activities. Include a separate activity for each contract, which may include, but are not limited to, utility companies, communications systems providers, equipment providers, and others.
- 1.1.2.10 Testing and commissioning. Include sufficient time to comply with the requirements of the General Commissioning Requirements; and any regulatory requirements; and assure completion of the Work within the Contract Time.
- 1.1.2.11 Licensing: allow time for County's administrative procedures necessary for certification.
- 1.1.2.12 All regulatory agency approvals (e.g., OSHPD, Fire Marshal, Public Works Agency, City, others).
- 1.1.3 All activities shall be identified through codes or other identification to indicate the scope of work (i.e. ED Fan, Waterproofing) and Contractor/Subcontractor responsibility to which they pertain.
- 1.1.4 Break up the Work schedule into activities of durations of fourteen (14) calendar days [ten (10) working days] or less each, except for non-field design and activities as otherwise deemed acceptable by County.
- 1.1.5 <u>Critical Path Activities</u>: Identify critical path activities in red. For each activity, show early start, late start, early finish, late finish, durations measured in Days, total and available float, resources, predecessor and successor activities, planned workday/week for the activity and scheduled/actual progress payments. "Critical path" shall mean all activities with zero float. A path with three (3) work days or less of float shall be considered a "near critical path" and shown in a lighter shade of red. No more than

twenty percent (20%) of the schedule activities are to be considered critical or near critical.

- 1.1.6 Seasonal weather conditions (which do not constitute a delay as defined herein) shall be considered in the planning and scheduling of all work influenced by high or low ambient temperatures or presence of high moisture for the completion of the Work within the allotted Contract Time. The number of allowable days of adverse weather is outlined in Section 007313. The Contractor shall adhere to procedures as specified in Section 007200 (General Conditions) for claiming delays resulting from adverse weather.
- 1.1.7 The Contractor shall meet with the County to review and discuss each Construction Schedule update (i.e., original Construction Schedule for each phase and each monthly update) within seven (7) Days after each Schedule has been submitted to County.
  - 1.1.7.1 County's review and comment on any Schedule shall be limited to Contract conformance (with sequencing, coordination, and milestone requirements).
  - 1.1.7.2 Contractor shall make corrections to Schedule necessary to comply with Contract requirements and shall adjust Schedule to incorporate any missing information requested by County. Contractor shall resubmit any schedules if requested by County.
- 1.1.8 If Contractor is of the opinion that any of the Work included on its Schedule has been impacted such that there will be a delay to completion of any phase of the Project, submit to County a written Time Impact Analysis ("TIA") in accordance with paragraph 1.10 below. The TIA shall be based on the most current update of the Schedule.
- 1.1.9 A three (3) week "Look Ahead Schedule," detailed daily bar chart schedule shall be updated and issued weekly.
  - 1.1.9.1 Look Ahead Schedule shall cover a eighteen (18) Day period, beginning with the week preceding the 3-week detailed look ahead.
  - 1.1.9.2 Use the Construction Schedule as the basis for generating the 3-week detailed schedule.
  - 1.1.9.3 Format:
    - 1.1.9.3.1 Provide bar chart using same logic as Construction Schedule, with maximum fourteen (14) Day construction activity duration. Provide activity identification used on the accepted Construction Schedule.
    - 1.1.9.3.2 Provide daily resource allocation for each trade.
    - 1.1.9.3.3 Provide exact activity location for scheduled Work.
  - 1.1.9.4 Provide information for each significant activity, with special care taken to describe scheduling and coordination with other contracts, Work by the

- County, and Facility operations/functions, including but not limited to utility shutdowns, road closures, etc.
- 1.1.9.5 Show the status of all outstanding and pending submittals including scheduled and actual submittal dates, the durations and expiration of submittal review periods, etc.

#### 1.2 MONTHLY SCHEDULE UPDATE SUBMITTALS

- 1.2.1 Following acceptance of Contractor's Construction Schedule, Contractor shall monitor progress of Work and update the Construction Schedule each month to reflect actual progress on each activity and any anticipated changes to planned activities. Monthly Schedule Updates shall include the following:
  - 1.2.1.1 Contractor's estimated percentage complete for each activity in progress.
  - 1.2.1.2 Actual start/finish dates for all activities shown on Construction Schedule with all subsequent approved additions.
  - 1.2.1.3 List of materials and/or equipment delivered for which Contractor is requesting payment and original invoice verifying cost.
  - 1.2.1.4 Identification of processing errors, if any, on the previous update reports.
  - 1.2.1.5 Resolution of any conflicts between actual Work progress and schedule logic when out-of-sequence activities develop because of actual construction progress. Contractor shall submit revisions to schedule logic to conform to current job status and directions, without changing original activity identification.
  - 1.2.1.6 Each update shall include a written narrative report with the updated progress analysis.
  - 1.2.1.7 CPM Reports: Concurrent with CPM schedule, submit the following computer-generated reports. Format for each activity in reports shall contain activity number, activity description, original duration, remaining duration, early start date, early finish date, late start date, late finish date, and total float.
    - 1.2.1.7.1 Provide printed report indicating actual versus planned resource loading for each trade and each activity. This report shall be provided on a monthly basis. Actual resource shall be accumulated in field by Contractor, and shall be as noted on Contractor's daily reports. These reports will be basis for information provided in monthly printed reports. Explain all variances and mitigation measures.

- 1.2.1.7.2 A Predecessor / Successor Report: List of all activities showing associated predecessor / successor activities, their logical relationships, free float, total float, early start/early finish date.
- 1.2.1.7.3 Total Float Report: List of all activities sorted in ascending order of total float, and then early start/early finish date.
- 1.2.1.7.4 The Contractor shall submit any other type of report as deemed necessary by the County.
- 1.2.1.8 The updated Construction Schedule shall accurately represent the as-built condition of all completed Work and the percentage remaining of all inprogress Work activities as of the date of the updated Construction Schedule.
- 1.2.1.9 The updated Construction Schedule shall incorporate all changes mutually agreed upon by Contractor and County during preceding periodic reviews, all changes resulting from Change Orders and Field Orders, and all remaining days of the inclement weather and regulatory review durations.
- 1.2.1.10 Contractor shall perform the Work in accordance with the updated Construction Schedule. Contractor may change the Construction Schedule to modify the order or sequence of accomplishing the Work only with the County's prior agreement.
- 1.2.1.11 Procedures for review and acceptance of the monthly updated schedule are outlined in paragraph 1.3 below.
- 1.2.1.12 No Application for Payment will be processed, nor shall any progress payments become due, until updated Construction Schedules are accepted by the County's Representative.
- 1.2.1.13 The accepted, updated Construction Schedule shall be the contractual Construction Schedule of record for the period it is current and shall be the basis for payment during that period.
- 1.2.1.14 The electronic version of the Construction Schedule update and reports will be furnished to the County's Representative for processing the application for payment. The Contractor shall provide full access to electronic schedule files for the County.

#### 1.3 MONTHLY SCHEDULE REVISIONS/UPDATES

- 1.3.1 The County will administer a monthly schedule update meeting to review and discuss each monthly updated schedule submittal.
  - 1.3.1.1 Included with each monthly updated schedule submittal, the Contractor shall submit any proposed schedule revisions to the County's Representative, including, but not limited to, the following:

- 1.3.1.1.1 Actual Start and/or Actual Finish Dates of activities that have started and/or have been completed.
- 1.3.1.1.2 Actual and anticipated duration changes including revisions due to inclement weather or regulatory agency review delays;
- 1.3.1.1.3 TIAs for Change Orders and Time Extension Requests;
- 1.3.1.1.4 Submit schedule diagrams showing resolution of conflicts between actual Work progress and schedule logic when out-of-sequence activities develop because of actual construction progress. Contractor shall submit revisions to schedule logic to conform to current job status and directions, without changing original activity identification;
- 1.3.1.1.5 Actual and anticipated Contractor delays;
- 1.3.1.1.6 A narrative report with the updated progress analysis, which shall include, but shall not be limited to, a description of problem areas, current and anticipated delaying factors and their impacts, and explanations of corrective action taken and any proposed revisions for a Recovery Plan.
- 1.3.1.2 These meetings are considered a critical component of overall monthly schedule update submittal; accordingly, Contractor shall ensure that appropriate personnel from its organization attend. At a minimum, Contractor's General Superintendent shall attend these meetings.
- 1.3.1.3 Monthly schedule update meetings will be scheduled for no less than four hours duration.
- 1.3.2 Within seven (7) Days after the monthly schedule update meeting, the County will either accept the proposed schedule update as the new Construction Schedule, or reject the proposed schedule update and provide comments, suggested changes, and revisions that must be addressed by the Contractor to the satisfaction of the County. Contractor shall correct and resubmit the schedule update within seven (7) Days.
- 1.3.3 The updating, changing, or revising of any report, curve, schedule or narrative submitted to County by Contractor under this Contract, nor County's review or acceptance of any such report, curve, schedule, or narrative shall not have the effect of amending or modifying, in any way, Contract Time or milestone dates or of modifying or limiting, in any way, Contractor's obligations under this Contract.
- 1.3.4 For rejected schedule update resubmittals that the County considers close to resolution, the County may request further information and justification and Contractor shall, within three (3) Days, provide County with a complete written narrative response to the County's request discussing how the resubmittal addresses each of the remaining contract conformance deficiencies noted in the schedule update resubmittal.
- 1.3.5 If the County does not accept Contractor's schedule update resubmittal, and Contractor disagrees with County's position, Contractor has seven (7) Days from receipt of County's letter rejecting the revision to provide a written narrative providing full justification and explanation for the revision. Contractor's failure to respond in writing within seven (7) Days of County's written rejection of a schedule revision shall constitute Contractor's

acceptance of County's position, and Contractor thereby waives its rights to subsequently dispute or file a claim regarding the County's position. If Contractor files a timely response as provided in this paragraph, and the parties are still unable to agree, Contractor's sole right shall be to file a Claim as provided in Document 007200 (General Conditions).

#### 1.4 RECOVERY SCHEDULE

- If the monthly schedule update or Look Ahead Schedule shows Project completion date fourteen (14) Days beyond the Contract Time, or any individual milestone completion dates, Contractor shall within seven (7) Days, submit to County a Recovery Plan to recover the lost time. As part of this Recovery Plan submittal, Contractor shall provide a written narrative and a Recovery Schedule to recapture the lost time. The Recovery Plan consisting of a Recovery Schedule and written narrative shall propose revisions to the Contract Schedule for the next 60-day period to show how the Contractor intends to bring the Work back on schedule. If the Recovery Schedule includes sequence changes, Contractor shall provide a schedule diagram comparing the original Contractor sequence to the revised sequence of the Work. The Recovery Schedule shall show the intended critical path; secure appropriate Subcontractor and supplier consent to the Recovery Schedule; the narrative shall explain trade flow and construction flow changes, duration changes, added/deleted activities, critical path changes and identify all near critical paths and resource loading assumptions for major Subcontractors. The Recovery Plan shall also describe how the measures that the Contractor intends to take to regain schedule compliance will be accomplished without additional cost to the County.
- 1.4.2 The Recovery Schedule shall not be incorporated into any monthly schedule update until County has reviewed the Recovery Schedule.
- 1.4.3 If County does not accept Contractor's Recovery Schedule, County and Contractor shall follow the procedures in paragraphs 1.3.4 and 1.3.5 above.
- 1.4.4 At County's discretion, Contractor can be required to provide Subcontractor certifications for any Recovery Schedule affecting said Subcontractors.
- 1.4.5 Contractor shall provide supervision, labor, equipment and materials, as necessary, to recover the lost time.
- 1.4.6 If Contractor believes that any portion of the delay addressed in the Recovery Schedule is due to circumstances entitling Contractor to additional time or money, it may seek a modification of the Contract Documents under Document 007200 (General Conditions), or make a Claim for the same pursuant to Document 007200, and other applicable provisions of the Contract Documents.

#### 1.5 TIME IMPACT ANALYSIS FOR CHANGE ORDERS, AND OTHER DELAYS

1.5.1 Any request for an adjustment of the Contract Time for completion submitted by Contractor for changes or alleged delays shall be accompanied by a complete Time Impact Analysis ("TIA") which includes both a written narrative and a hard and fully operational electronic copy of a schedule diagram depicting how the changed work

affects other schedule activities. The schedule diagram shall show how Contractor proposes to incorporate the changed work in the schedule, and how it impacts the current schedule update critical path. Contractor is also responsible for requesting time extensions based on the TIA's impact on the critical path. The diagram must be tied to the main sequence of schedule activities to enable County to evaluate the impact of changed work to the scheduled critical path.

- 1.5.2 Contractor shall comply with the requirements of Paragraph 1.10.1 for all types of delays such as, but not limited to, Contractor/Subcontractor delays, asserted County or third party caused delays, adverse weather delays, strikes, procurement delays, fabrication delays, etc.
- 1.5.3 Contractor shall be responsible for all costs associated with the preparation of TIAs, and the process of incorporating them into the current schedule update. Contractor shall provide County with an electronic copy of each TIA. Contractor's TIAs must be based on the as-built critical path. The TIA shall also show the as-planned critical path.
- 1.5.4 Once agreement has been reached on a TIA, the Contract Time will be adjusted accordingly. If agreement is not reached on a TIA, the Contract Time may be extended in an amount County allows, and Contractor may submit a Claim for additional time claimed by Contractor as provided in Document 007200 (General Conditions).

#### 1.6 TIME EXTENSIONS

- 1.6.1 Contractor is responsible for requesting Contract Time extensions for time impacts that, in the opinion of Contractor, impact the critical path of the current schedule update. Notice of time impacts shall be given in accord with Document 007200 (General Conditions).
- 1.6.2 Where an event for which either Contractor or County is responsible impacts the projected Contract Time, Contractor shall provide a written mitigation plan, including a schedule diagram, which explains how (e.g., increase crew size, overtime, etc.) the impact will be mitigated. Contractor shall also include a detailed cost breakdown of the labor, equipment and material Contractor would expend to mitigate the delay. Contractor shall submit its mitigation plan to County within seven (7) Days from the date of discovery of said impact. Contractor is responsible for the cost to prepare the mitigation plan.
- 1.6.3 Contractor's failure to request time, provide TIA, or provide the required mitigation plan will result in Contractor waiving its right to a time extension and cost to mitigate the delay.
- 1.6.4 Contractor shall be responsible to provide timely and proper notice to the County of all impacts that could result in Contract Time extensions and shall comply with requirements as specified in Section 007200 (General Conditions).
- 1.6.5 No time will be granted under the Contract Documents for cumulative effect of impacts or changes.

- 1.6.6 County will not be obligated to consider any time extension request unless requirements of Contract Documents are complied with.
- 1.6.7 Failure of Contractor to perform in accordance with the current schedule update shall not be excused by submittal of time extension requests.

#### 1.7 PROJECT STATUS REPORTING

- 1.7.1 Monthly. In addition to submittal requirements for scheduling identified in this Section 013216, provide a monthly project status report (i.e., written narrative report) to be submitted in conjunction with each Schedule as specified herein in electronic and hard copy. Written status reports shall include:
  - 1.7.1.1 Status of major Project components (percent complete, amount of time ahead or behind schedule) and an explanation of how Project will be brought back on schedule if delays have occurred.
  - 1.7.1.2 Progress made on critical activities indicated on each Schedule, including inspections.
  - 1.7.1.3 Explanations for any lack of work on critical path activities planned to be performed during last month.
  - 1.7.1.4 Explanations for any schedule changes, including changes to logic or to activity durations.
  - 1.7.1.5 List of critical activities scheduled to be performed during the next month.
  - 1.7.1.6 Status of major material and equipment procurement.
  - 1.7.1.7 Description of problem areas, current and anticipated delaying factors and their impacts, and an explanation of corrective action taken.
  - 1.7.1.8 Any proposed revisions for a Recovery Plan.
  - 1.7.1.9 Contractor shall produce additional status reports as requested by County at no additional cost.
  - 1.7.1.10 Status reports, and the information contained therein, shall not be construed as claims, notice of claims, notice of delay, or requests for changes or compensation.
- 1.7.2 Reporting Unusual Events: When an event of an unusual and significant nature occurs at Project site, whether or not related directly to the Work, prepare and submit an Incident Report. List chain of events, date and time of initiating event as well as subsequent activities, persons participating, response by Contractor's personnel, evaluation of results or effects, and similar pertinent information. Advise County in advance when these events are known or predictable. Incident Reports shall be numbered sequentially and

# **Alameda County General Services Agency**

**PROJECT #2043** 

Highland Hospital – Koret Building Exhaust Fan and Waterproofing

shall be submitted within three (3) days of occurrence and be signed by the Superintendent and Project Manager.

PART 2 - PRODUCTS

Not used.

**PART 3 - EXECUTION** 

Not used.

**END OF SECTION** 

# **DOCUMENT 01 33 00**

# SUBMITTAL PROCEDURES

#### **PART 1 - GENERAL**

#### 1.01 RELATED DOCUMENTS AND PROVISIONS:

All Contract Documents should be reviewed for applicable provisions related to the provisions in this document, including without limitation:

- A. General Conditions, including, without limitation, Contractor's Submittals and Schedules, Drawings and Specifications;
- B. Special Conditions.

#### 1.02 SECTION INCLUDES:

#### A. Definitions:

- (1) Shop Drawings and Product Data are as indicated in the General Conditions and include, but are not limited to, fabrication, erection, layout and setting drawings, formwork and falsework drawings, manufacturers' standard drawings, descriptive literature, catalogues, brochures, performance and test data, wiring and control diagrams. In addition, there are other drawings and descriptive data pertaining to materials, equipment, piping, duct and conduit systems, and methods of construction as may be required to show that the materials, equipment or systems and all positions conform to the requirement of the Contract Documents, including, without limitation, the Drawings.
- "Manufactured" applies to standard units usually mass-produced;
  "fabricated" means specifically assembled or made out of selected
  materials to meet design requirements. Shop Drawings shall establish the
  actual detail of manufactured or fabricated items, indicate proper relation
  to adjoining work and amplify design details of mechanical and electrical
  equipment in proper relation to physical spaces in the structure.
- (3) Manufacturer's Instructions: Where any item of Work is required by the Contract Documents to be furnished, installed, or performed, at a minimum, in accordance with a specified product manufacturer's instructions, the Contractor shall procure and distribute copies of these to the County, the Architect, and all other concerned parties and shall furnish, install, or perform the work, at a minimum, in accordance with those instructions.

- B. Samples, Shop Drawings, Product Data, and other items as specified, in accordance with the following requirements:
  - (1) Contractor shall submit all Shop Drawings, Product Data, and Samples to the County, the Architect, and the Construction Manager.
  - (2) Contractor shall comply with all time frames herein and in the General Conditions and, in any case, shall submit required information in sufficient time to permit proper consideration and action before ordering any materials or items represented by such Shop Drawings, Product Data, and/or Samples.
  - (3) Contractor shall comply with all time frames herein and in the General Conditions and, in any case, shall allow sufficient time so that no delay occurs due to required lead time in ordering or delivery of any item to the Site. Contractor shall be responsible for any delay in progress of Work due to its failure to observe these requirements.
  - (4) Time for completion of Work shall not be extended on account of Contractor's failure to promptly submit Shop Drawings, Product Data, and/or Samples.
  - (5) Reference numbers on Shop Drawings shall have Architectural and/or Engineering Contract Drawings reference numbers for details, sections, and "cuts" shown on Shop Drawings. These reference numbers shall be in addition to any numbering system that Contractor chooses to use or has adopted as standard.
  - (6) When the magnitude or complexity of submittal material prevents a complete review within the stated time frame, Contractor shall make this submittal in increments to avoid extended delays.
  - (7) Contractor shall certify on submittals for review that submittals conform to Contract requirements. In event of any variance, Contractor shall specifically state in transmittal and on Shop Drawings, portions vary and require approval of a substitute. Also certify that Contractor-furnished equipment can be installed in allocated space.
  - (8) Unless specified otherwise, sampling, preparation of samples, and tests shall be in accordance with the latest standard of the American Society for Testing and Materials.
  - (9) Upon demand by Architect or County, Contractor shall submit samples of materials and/or articles for tests or examinations and consideration before Contractor incorporates same in Work. Contractor shall be solely

responsible for delays due to sample(s) not being submitted in time to allow for tests. Acceptance or rejection will be expressed in writing. Work shall be equal to approved samples in every respect. Samples that are of value after testing will remain the property of Contractor.

# C. Submittal Schedule:

- (1) Contractor shall prepare its proposed submittal schedule that is coordinated with its proposed construction schedule and submit both to the County within ten (10) days after the date of the Notice to Proceed. Contractor's proposed schedules shall become the Project Construction Schedule and the Project Submittal Schedule after each is approved by the County.
- (2) Contractor is responsible for all lost time should the initial submittal be rejected, marked "revised and resubmit", etc.
- (3) All Submittals shall be forwarded to the County by the date indicated on the approved Submittal Schedule, unless an earlier date is necessary to maintain the Construction Schedule, in which case those Submittals shall be forwarded to the County so as not to delay the Construction Schedule.

#### 1.03 SHOP DRAWINGS:

- A. Contractor shall submit one electronic copy of the submittal. The County will review and return the electronic copy to Contractor.
- B. Before commencing installation of any Work, the Contractor shall submit and receive approval of all drawings, descriptive data, and material list(s) as required to accomplish Work.
- C. Review of Shop Drawings is regarded as a service to assist Contractor and in all cases original Contract Documents shall take precedence as outlined under General Conditions.
- D. No claim for extra time or payment shall be based on work shown on Shop Drawings unless the claim is (1) noted on Contractor's transmittal letter accompanying Shop Drawings and (2) Contractor has complied with all applicable provisions of the General Conditions, including, without limitation, provisions regarding changes and payment, and all required written approvals.
- E. County shall not review Shop Drawings for quantities of materials or number of items supplied.
- F. County's and/or Architect's review of Shop Drawings will be general. County and/or Architect review does not relieve Contractor of responsibility for accuracy,

proper fitting, construction of Work, furnishing of materials, or Work required by Contract Documents and not indicated on Shop Drawings. Shop Drawings reviewed by County and/or Architect is not to be construed as approving departures from Contract Documents.

- G. Review of Shop Drawings and Schedules does not relieve Contractor from responsibility for any aspect of those Drawings or Schedules that is a violation of local, County, State, or Federal laws, rules, ordinances, or rules and regulations of commissions, boards, or other authorities or utilities having jurisdiction.
- H. Before submitting Shop Drawings for review, Contractor shall check Shop Drawings of its subcontractors for accuracy, and confirm that all Work contiguous with and having bearing on other work shown on Shop Drawings is accurately drawn and in conformance with Contract Documents.
- I. Submitted drawings and details must bear stamp of approval of Contractor:
  - (1) Stamp and signature shall clearly certify that Contractor has checked Shop Drawings for compliance with Drawings.
  - (2) If Contractor submits a Shop Drawing without an executed stamp of approval, or whenever it is evident (despite stamp) that Drawings have not been checked, the County and/or Architect will not consider them and will return them to the Contractor for revision and resubmission. In that event, it will be deemed that Contractor has not complied with this provision and Contractor shall bear risk of all delays to same extent as if it had not submitted any Shop Drawings or details.
- J. Submission of Shop Drawings (in either original submission or when resubmitted with correction) constitutes evidence that Contractor has checked all information thereon and that it accepts and is willing to perform Work as shown.
- K. Contractor shall pay for cost of any changes in construction due to improper checking and coordination. Contractor shall be responsible for all additional costs, including coordination. Contractor shall be responsible for costs incurred by itself, the County, the Architect, the Construction Manager, any other Subcontractor or contractor, etc., due to improperly checked and/or coordination of submittals.
- L. Shop Drawings must clearly delineate the following information:
  - (1) Project name and address.
  - (2) Architect's name and project number.
  - (3) Shop Drawing title, number, date, and scale.

- (4) Names of Contractor, Subcontractor(s) and fabricator.
- (5) Working and erection dimensions.
- (6) Arrangements and sectional views.
- (7) Necessary details, including complete information for making connections with other Work.
- (8) Kinds of materials and finishes.
- (9) Descriptive names of materials and equipment, classified item numbers, and locations at which materials or equipment are to be installed in the Work. Contractor shall use same reference identification(s) as shown on Contract Drawings.
- M. Contractor shall prepare composite drawings and installation layouts when required to solve tight field conditions.
  - (1) Shop Drawings shall consist of dimensioned plans and elevations and must give complete information, particularly as to size and location of sleeves, inserts, attachments, openings, conduits, ducts, boxes, structural interferences, etc.
  - (2) Contractor shall coordinate these composite Shop Drawings and installation layouts in the field between itself and its Subcontractor(s) for proper relationship to the Work, the work of other trades, and the field conditions. The Contractor shall check and approve the submittal(s) before submission for final review.

# 1.04 PRODUCT DATA OR NON REPRODUCIBLE SUBMITTALS:

- A. Contractor shall submit manufacturer's printed literature in original electronic form to the County. County shall return electronic form to the Contractor, who shall reproduce whatever additional copies it requires for distribution.
- B. Contractor shall submit electronic form of a complete list of all major items of mechanical, plumbing, and electrical equipment and materials in accordance with the approved Submittal Schedule, except as required earlier to comply with the approved Construction Schedule. Other items specified are to be submitted prior to commencing Work. Contractor shall submit items of like kind at one time in a neat and orderly manner. Partial lists will not be acceptable.
- C. Submittals shall include manufacturer's specifications, physical dimensions, and ratings of all equipment. Contractor shall furnish performance curves for all

pumps and fans. Where printed literature describes items in addition to that item being submitted, submitted item shall be clearly marked on sheet and superfluous information shall be crossed out. If highlighting is used, Contractor shall mark all copies.

D. Equipment submittals shall be complete and include space requirements, weight, electrical and mechanical requirements, performance data, and supplemental information that may be requested.

#### **1.05 SAMPLES:**

- A. Contractor shall submit for approval Samples as required and within the time frame in the Contract Documents. Materials such as concrete, mortar, etc., which require on-site testing will be obtained from Project Site.
- B. Contractor shall submit four (4) samples except where greater or lesser number is specifically required by Contract Documents including, without limitation, the Specifications.
  - (1) Samples must be of sufficient size and quality to clearly illustrate functional characteristics, with integrally related parts and attachment devices.
  - (2) Samples must show full range of texture, color, and pattern.
- C. Contractor shall make all Submittals, unless it has authorized Subcontractor(s) to submit and Contractor has notified the County in writing to this effect.
- D. Samples to be shipped prepaid or hand-delivered to the County.
- E. Contractor shall mark samples to show name of Project, name of Contractor submitting, Contract number and segment of Work where representative Sample will be used, all applicable Specifications Sections and documents, Contract Drawing Number and detail, and ASTM or FS reference, if applicable.
- F. Contractor shall not deliver any material to Site prior to receipt of County's and/or Architect's completed written review and approval. Contractor shall furnish materials equal in every respect to approved Samples and execute Work in conformance therewith.
- G. County's and/or Architect's review, acceptance, and/or approval of Sample(s) will not preclude rejections of any material upon discovery of defects in same prior to final acceptance of completed Work.
- H. After a material has been approved, no change in brand or make will be permitted.

- I. Contractor shall prepare its Submittal Schedule and submit Samples of materials requiring laboratory tests to specified laboratory for testing not less than ninety (90) days before such materials are required to be used in Work.
- J. Samples which are rejected must be resubmitted promptly after notification of rejection and be marked "Resubmitted Sample" in addition to other information required.
- K. Field Samples and Mock-Ups are to be removed by Contractor at County's direction:
  - (1) Size: As Specified.
  - (2) Furnish catalog numbers and similar data, as requested.

# 1.06 SAFETY PLAN:

- A. Submit a Safety Plan specific to this Contract to the County. Safety Plan to address requirements that are specific to construction performed at an operational hospital facility, including, but limited to infection control procedures and MOPs for working in occupied, staff, or patient care areas.
- B. No on-site work shall be started until Safety Plan has been reviewed and accepted by County. Acceptance of the Safety Plan shall not affect Contractor's responsibility for maintaining a safe working place and instituting safety programs in connection with project. Neither the County nor any of its representatives assume any responsibility for Contractor's safety related obligations. Contractor shall have sole responsibility for safety on and off the Site.

# 1.07 CONSTRUCTION SCHEDULE:

- A. See Section 013216 (Schedules and Reports) for schedule and report requirements.
- B. Submit one (1) electronic version of the schedule at each of the following times:
  - (1) Construction Schedule with each Pay Application.

# 1.08 REVIEW AND RESUBMISSION REQUIREMENTS:

A. The County will arrange for review of Sample(s), Shop Drawing(s), Product Data, and other submittal(s) by appropriate reviewer and return to Contractor as provided below within twenty one (21) days after receipt or within twenty one (21) days after receipt of all related information necessary for such review, whichever is later.

- B. One (1) electronic copy of product or materials data will be returned to Contractor with the review status.
- C. Samples to be incorporated into the Work will be returned to Contractor, together with a written notice designating the Sample with the appropriate review status and indicating errors discovered on review, if any. Other Samples will not be returned, but the same notice will be given with respect thereto, and that notice shall be considered a return of the Sample.
- D. Contractor shall revise and resubmit any Sample(s), Shop Drawing(s), Product Data, and other submittal(s) as required by the reviewer. Such resubmittals will be reviewed and returned in the same manner as original Sample(s), Shop Drawing(s), Product Data, and other submittal(s), within fourteen (14) days after receipt thereof or within fourteen (14) days after receipt of all related information necessary for such review.
- E. Contractor may proceed with any of the Work covered by Sample(s), Shop Drawing(s), Product Data, and other submittal(s) upon its return if designated as no exception taken, or revise as noted, provided the Contractor proceeds in accordance with the County's and/or the Architect's notes and comments.
- F. Contractor shall not begin any of the work covered by a Sample(s), Shop Drawing(s), Product Data, and other submittal(s), designated as revise and resubmit or rejected, until a revision or correction thereof has been reviewed and returned to Contractor.
- G. Sample(s), Shop Drawing(s), Product Data, and other submittal(s) designated as revise and resubmit or rejected and requiring resubmittal, shall be revised or corrected and resubmitted to the County no later than fourteen (14) days or a shorter period as required to comply with the approved Construction Schedule, after its return to Contractor.
- H. Neither the review nor the lack of review of any Sample(s), Shop Drawing(s), Product Data, and other submittal(s) shall waive any of the requirements of the Contract Documents, or relieve Contractor of any obligation thereunder.
- I. County's and/or Architect's review of Shop Drawings does not relieve the Contractor of responsibility for any errors that may exist. Contractor is responsible for the dimensions and design of adequate connections and details and for satisfactory construction of all the Work.

END OF DOCUMENT

# **DOCUMENT 01 41 00**

# **REGULATORY REQUIREMENTS**

#### PART 1 - GENERAL

# 1.01 RELATED DOCUMENTS AND PROVISIONS:

All Contract Documents should be reviewed for applicable provisions related to the provisions in this document, including without limitation:

- A. General Conditions, including, without limitation, Obtaining of Permits and Licenses and Work To Comply With All Applicable Regulations;
- B. Special Conditions;
- C. Quality Control.

#### 1.02 DESCRIPTION:

A. This section covers the general requirements for regulatory requirements pertaining to the Work and is supplementary to all other regulatory requirements mentioned or referenced elsewhere in the Contract Documents.

# 1.03 REQUIREMENTS OF REGULATORY AGENCIES:

- A. All statutes, ordinances, laws, rules, codes, regulations, standards, and the lawful orders of all public authorities having jurisdiction of the Work, are hereby incorporated into these Contract Documents as if repeated in full herein and are intended to be included in any reference to Code or Building Code, unless otherwise specified, including, without limitation, the references in the list below. Contractor shall make available at the Site copies of all the listed documents applicable to the Work as the County and/or Architect/Engineer may request, including, without limitation, applicable portions of the California Code of Regulations ("CCR").
- B. Items of deferred approval shall be clearly marked on the first sheet of the Architect's and/or Engineer's approved Drawings.
  - (1) Building Standards Administrative Code, Part 1, Title 24, CCR
  - (2) California Building Code (CBC), Part 2, Title 24, CCR; (Uniform Building code volumes 1-3 and California Amendments).
  - (3) California Electrical Code (CEC), Part 3, Title 24, CCR; (National Electrical Code and California Amendments).

- (4) California Mechanical Code (CMC), Part 4, Title 24, CCR; (Uniform Mechanical Code and California Amendments).
- (5) California Plumbing Code (CPC), Part 5, Title 24, CCR; (Uniform Plumbing Code and California Amendments).
- (6) California Fire Code (CFC), Part 9, Title 24, CCR; (Fire Plumbing Code and California Amendments).
- (7) California Referenced Standards Code, Part 12, Title 24, CCR
- (8) Title 19, CCR, Public Safety, State Fire Marshal Regulations.
- (9) Partial List of Applicable NFPA Standards:
  - (a) NFPA 13 Automatic Sprinkler System.
  - (b) NFPA 14 Standpipes Systems.
  - (c) NFPA 17A Wet Chemical System
  - (d) NFPA 24 Private Fire Mains.
  - (e) (California Amended) NFPA 72 National Fire Alarm Codes.
  - (f) NFPA 253 Critical Radiant Flux of Floor Covering System.
  - (g) FPA 2001 Clean Agent Fire Extinguishing Systems.
- C. Office of Statewide Health Planning and Development, (REGULATORY AGENCY), located at 1600 9th Street, Sacramento, CA, 95814, is the agency having jurisdiction over project design and construction.
  - (1) REGULATORY AGENCY will designate an inspector or inspectors for project who shall be allowed access to Project site at any time.
  - (2) REGULATORY AGENCY will require a Notice of Start of Construction be filed prior to start of work.
  - (3) REGULATORY AGENCY will require verified report forms to be filed during construction as required by the Testing, Inspection, and Observation Program established by the Architect/Engineer of Record and approved by REGULATORY AGENCY. Separate verified reports are required from Architect, each Consultant, Project Inspector, and Contractor.

# **Alameda County, General Services Agency**

**PROJECT # 2043** 

Highland Hospital – Koret Building Exhaust Fan and Waterproofing

- (4) REGULATORY AGENCY will require Change Order approval to be submitted and approved prior to work performed in the field.
- (5) REGULATORY AGENCY will require a Licensed Contractors' Declaration and a Worker's Compensation Declaration.
- D. Conflicts between regulatory requirements shall be addressed as follows:
  - (1) Between referenced regulatory requirements: Comply with the one establishing the more stringent requirement.
  - (2) Between referenced regulatory requirements and the Contract Documents: Comply with the one establishing the more stringent requirement.

END OF DOCUMENT

**DOCUMENT 01 42 13** 

Highland Hospital – Koret Building Exhaust Fan and Waterproofing

# **DOCUMENT 01 42 13**

# **ABBREVIATIONS AND ACRONYMS**

# PART 1 – GENERAL

# 1.01 RELATED DOCUMENTS AND PROVISIONS:

All Contract Documents should be reviewed for applicable provisions related to the provisions in this document, including without limitation:

- A. General Conditions including without limitation, Definitions
- B. Special Conditions

# 1.02 DOCUMENT INCLUDES:

ALAMEDA COUNTY GSA-CP

Page 1 of 3

- A. Abbreviations used throughout the Contract Documents.
- B. Reference to a technical society, organization, or body is by abbreviation, as follows:

1.	AA	Aluminum Association
2.	AAMA	Architectural Aluminum Manufacturers
		Association
3.	AASHTO	American Association of State Highway and
		Transportation Officials
4.	ABPA	Acoustical and Board Products Association
5.	ACI	American Concrete Institute
6.	AGA	American Gas Association
7.	AGC	Associated General Contractors
8.	AHC	Architectural Hardware Consultant
9.	AI	Asphalt Institute
10.	AIA	American Institute of Architects
11.	AIEE	American Institute of Electrical Engineers
12.	AISC	American Institute of Steel Construction
13.	AISI	American Iron and Steel Institute
14.	AMCA	Air Moving and Conditioning Association
15.	ANSI	American National Standards Institute
16.	APA	American Plywood Association
17.	ARI	Air Conditioning and Refrigeration Institute
18.	ASHRAE	American Society of Heating, Refrigeration and
		Air Conditioning Engineers
19.	ASME	American Society of Mechanical Engineers
20.	ASSE	American Society of Structural Engineers
21.	ASTM	American Society of Testing and Materials
OUNTY GS	SA-CP	ABBREVIATIONS AND ACRONYMS

**Alameda County, General Services Agency** Highland Hospital – Koret Building Exhaust Fan and Waterproofing

1	$\mathcal{E}$	
22.	AWPB	American Wood Preservers Bureau
23.	AWPI	American Wood preservers Institute
24.	AWS	American Welding Society
25.	AWSC	American Welding Society Code
26.	AWI	Architectural Woodwork Institute
27.	AWWA	American Water Works Association
28.	BIA	Brick Institute of America
29.	CCR	California Code of Regulations
30.	CLFMI	Chain Link Fence Manufacturers Institute
31.	CMG	California Masonry Guild
32.	CRA	California Redwood Association
33.	CRSI	Concrete Reinforcing Steel Institute
34.	CS	Commercial Standards
35.	CSI	Construction Specifications Institute
36.	CTI	Cooling Tower Institute
37.	FGMA	Flat Glass Manufacturer's Association
38.	FIA	Factory Insurance Association
39.	FM	Factory Mutual
40.	FS	Federal Specification
41.	FTI	Facing Title Institute
42.	GA	Gypsum Association
43.	ICBO	International Conference of Building Officials
44.	IEEE	Institute of Electrical and Electronic Engineers
45.	IES	Illumination Engineering Society
46.	LIA	Lead Industries Association
47.	MIA	Marble Institute of America
48.	MLMA	Metal Lath Manufacturers Association
49.	MS	Military Specifications
50.	NAAMM	National Association of Architectural Metal
		Manufacturers
51.	NBHA	National Builders Hardware Association
52.	NBFU	National Board of Fire Underwriters
53.	NBS	National Bureau of Standards
54.	NCMA	National Concrete Masonry Association
55.	NEC	National Electrical Code
56.	NEMA	National Electrical Manufacturers Association
57.	NFPA	National Fire Protection Association/National
		Forest Products Association
58.	NMWIA	National Mineral Wool Insulation Association
59.	NTMA	National Terrazzo and Mosaic Association
60.	NWMA	National Woodwork Manufacturer's Association
61.	ORS	Office of Regulatory Services (California)
62.	OSHA	Occupational Safety and Health Act
63.	PCI	Precast Concrete Institute
64.	PCA	Portland Cement Association
65.	PDCA	Painting and Decorating Contractors of America
OUNTY (		ABBREVIATIONS AND ACRONYN

ALAMEDA COUNTY GSA-CP Page 2 of 3

ABBREVIATIONS AND ACRONYMS **DOCUMENT 01 42 13** 

# Alameda County, General Services Agency

# **PROJECT # 2043**

Highland Hospital – Koret Building Exhaust Fan and Waterproofing

66.	PDI	Plumbing Drainage Institute
67.	PEI	Porcelain Enamel Institute
68.	PG&E	Pacific Gas & Electric Company
69.	PS	Product Standards
70.	SDI	Steel Door Institute; Steel Deck Institute
71.	SJI	Steel Joist Institute
72.	SSPC	Steel Structures Painting Council
73.	TCA	Tile Council of America
74.	TPI	Truss Plate Institute
75.	UBC	Uniform Building Code
76.	UL	Underwriters Laboratories Code
77.	UMC	Uniform Mechanical Code
78.	USDA	United States Department of Agriculture
79.	VI	Vermiculite Institute
80.	WCLA	West Coast Lumberman's Association
81	WCLB	West Coast Lumber Bureau
82.	WEUSER	Western Electric Utilities Service Engineering
		Requirements
83.	WIC	Woodwork Institute of California
84.	WPOA	Western Plumbing Officials Association

END OF DOCUMENT

# **DOCUMENT 01 42 16**

# **DEFINITIONS AND REFERENCE STANDARDS**

#### PART 1 - GENERAL

# 1.01 RELATED DOCUMENTS AND PROVISION

All Contract Documents should be reviewed for applicable provisions related to the provisions in this document, including without limitation:

- A. General Conditions including without limitation, Definitions;
- B. Special Conditions;

# 1.02 QUALITY ASSURANCE:

- A. For products or workmanship specified by association, trade, or Federal Standards, Contractor shall comply with requirements of the standard, except when more rigid requirements are specified in the Contract Documents, or are required by applicable codes.
- B. Contractor shall conform to current reference standard publication date in effect on the date of bid opening.
- C. Contractor shall obtain copies of standards unless specifically required not to by the Contract Documents.
- D. Contractor shall maintain a copy of all standards at jobsite during submittals, planning, and progress of the specific Work, until final completion, unless specifically required not to by the Contract Documents.
- E. Should specified reference standards conflict with Contract Documents, Contractor shall request clarification from the County and/or the Architect/Engineer before proceeding.
- F. The contractual relationship of the parties to the Contract shall not be altered from the contractual relationship as indicated in the Contract Documents by mention or inference otherwise in any referenced document.
- G. Governing Codes shall be as shown in the Contract Documents including, without limitation, the Specifications.

# 1.03 SCHEDULE OF REFERENCES:

The following information is intended only for the general assistance of the Contractor, and the County does not represent that all of the information is current. It is the Contractor's responsibility to verify the correct information for each of the entities listed.

AA	Aluminum Association 900 19 <sup>th</sup> Street NW, Suite 300 Washington, DC 20006 www.aluminum.org	202/862-5100
AABC	Associated Air Balance Council 1518 K Street, NW, Suite 503 Washington, DC 20005 www.aabchq.com	202/737-0202
AAMA	American Architectural Manufacturers Association 1827 Walden Office Sq., Suite 104 Schaumburg, IL 60173-4268 www.aamanet.org	847/303-5664
AASHTO	American Association of State Highway and Transportation Officials 444 North Capitol Street, Suite 249 Washington, DC 20001 www.aashto.org	202/624-5800
AATCC	American Association of Textile Chemists and Colorists P.O. Box 12215 One Davis Drive Research Triangle Park, NC 27709-2215 www.aatcc.org	919/549-8141
ACI	American Concrete Institute P.O. Box 9094 Farmington Hills, MI 48333-9094 www.aci-int.org	248/848-3700
ACPA	American Concrete Pipe Association 222 West Las Colinas Blvd., Suite 641 Irving, TX 75039-5423 www.concrete-pipe.org	972/506-7216

ADC Air Diffusion Council 312/201-0101

ALAMEDA COUNTY GSA-CP Page 2 of 16

Alameda	County.	<b>General Services A</b>	gencv
---------	---------	---------------------------	-------

**PROJECT # 2043** 

Highland Hospital – Koret Building Exhaust Fan and Waterproofing

11 South LaSalle St., Suite 1400

Chicago, IL 60603

AFPA American Forest and Paper Association 202/463-2700

1111 19th St., NW, Suite 800 Washington, DC 20036

AGA American Gas Association 703/841-8400

1515 Wilson Blvd. Arlington VA 22209 www.aga.com

AHA American Hardboard Association 847/934-8800

1210 W. Northwest Hwy Palatine, IL 60067-1897

AI Asphalt Institute 606/288-4960

Research Park Drive P.O. Box 14052

Lexington, KY 40512-4052 www.asphaltinstitute.org

AIA The American Institute of Architects 202/626-7300

1735 New York Avenue, NW Washington, DC 20006-5292

www.aia.org

AISC American Institute of Steel Construction 800/644-2400

One East Wacker Drive, Suite 3100

Chicago, IL 60601-2001

AITC American Institute of Timber Construction 303/792-9559

7012 S. Revere Pkwy., Suite 140

Englewood, CO 80112 www.aitc-glulam.org

ALCA Associated Landscape Contractors of America 703/620-6363

12200 Sunrise Valley Drive, Suite 150

Reston, VA 20191 www.alca.org

ALI Associated Laboratories, Inc. 214/565-0593

P.O. Box 152837 1323 Wall St. Dallas, TX 75315

<b>Alameda County, G</b> Highland Hospital – I	PROJECT # 2043	
ALSC	American Lumber Standards Committee P.O. Box 210 Germantown, MD 20875	301/972-1700
AMCA	Air Movement and Control Association International, Inc. 30 W. University Drive Arlington Heights, IL 60004-1893 www.amca.org	847/394-0150
ANLA	American Nursery and Landscape Association 1250 Eye Street, NW, Suite 500 Washington, DC 20005	202/789-2900
ANSI	American National Standards Institute 11 West 42nd Street, 13th Floor New York, NY 10036-8002 www.ansi.org	212/642-4900
APA	APA-The Engineered Wood Association P.O. Box 11700 Tacoma, WA 98411-0700 www.apawood.org	206/565-6600
APA	Architectural Precast Association P.O. Box 08669 Fort Myers, FL 33908-0669	941/454-6989
ARI	Air Conditioning and Refrigeration Institute 4301 Fairfax Drive, Suite 425 Arlington, VA 22203 www.ari/org	703/524-8800
ARMA	Asphalt Roofing Manufacturers Association Center Park 4041 Powder Mill Road, Suite 404 Calverton, MD 20705	301/231-9050
ASA	Acoustical Society of America 500 Sunnyside Blvd. Woodbury, NY 11797	516/576-2360

Alameda County, General Services Agency Highland Hospital – Koret Building Exhaust Fan and Waterproofing			
ASCE	American Society of Civil Engineers- World Headquarters 1801 Alexander Bell Drive Reston, VA 20190-4400 www.asce.org	800/548-2723 703/295-6000	
ASHRAE	American Society of Heating, Refrigerating and Air Conditioning Engineers 1791 Tullie Circle, NE Atlanta, GA 30329-2305 www.ashrae.org	800/527-4723 404/636-8400	
ASLA	American Society of Landscape Architects 4401 Connecticut Ave., NW, 5th Floor Washington, DC 20008-2369 www.asla.org	202/686-2752	
ASME	American Society of Mechanical Engineers 345 East 47 <sup>th</sup> Street New York, NY 10017-2392 www.asme.org	800/434-2763	
ASPE	American Society of Plumbing Engineers 3617 Thousand Oaks Blvd., Suite 210 Westlake, CA 91362-3649	805/495-7120	
ASQC	American Society for Quality Control 611 E. Wisconsin Avenue Milwaukee, WI 53201-3005 www.asqc.org	800/248-1946 414/272-8575	
ASSE	American Society of Sanitary Engineering 28901 Clemens Road Westlake, OH 44145 www.asse-plumbing.org	216/835-3040	
ASTM	American Society for Testing and Materials 100 Barr Harbor Drive West Conshohocken, PA 19428-2959	610/832-9500	

www.astm.org

Alameda County, G Highland Hospital –	PROJECT # 2043	
AWCI	Association of the Wall and Ceiling IndustriesInternational 307 E. Annandale Road, Suite 200 Falls Church, VA 22042-2433 www.awci.org	703/534-8300
AWPA	American Wood-Preservers' Association 3246 Fall Creek Highway, Suite 1900 Granbury, TX 76049-7979	817/326-6300
AWS	American Welding Society 550 NW LeJeune Road Miami, FL 33126 www.amweld.org	800/443-9373 305/443-9353
AWWA	American Water Works Association 6666 West Quincy Avenue Denver, CO 80235 www.awwa.org	800/926-7337 303/794-7711
ВНМА	Builders' Hardware Manufacturers Association 355 Lexington Avenue, 17th Floor New York, NY 10017-6603	212/661-4261
СВМ	Certified Ballast Manufacturers Association 1422 Euclid Avenue, Suite 402 Cleveland, OH 44115-2094	216/241-0711
CGA	Compressed Gas Association 1725 Jefferson Davis Hwy, Suite 1004 Arlington, VA 22202-4102 www.cganet.com	703/412-0900
CISCA	Ceilings & Interior Systems Construction Association 1500 Lincoln Hwy, Suite 202 St. Charles, IL 60174 www.cisca.org	630/584-1919
CISPI	Cast Iron Soil Pipe Institute 5959 Shallowford Road, Suite 419 Chattanooga, TN 37421	423/892-0137

• •	General Services Agency Koret Building Exhaust Fan and Waterproofing	PROJECT # 2043
CPSC	Consumer Product Safety Commission East West Towers 4330 East-West Hwy. Bethesda, MD 20814	800/638-2772
СРРА	Corrugated Polyethylene Pipe Association 432 N. Superior Street Toledo, OH 43604	800/510-2772 419/241-2221
CRA	California Redwood Association 405 Enfrente Drive, Suite 200 Novato, CA 94949	415/382-0662
CRI	Carpet and Rug Institute 310 S. Holiday Avenue Dalton, GA 30722-2048  www.carpet-rug.com	800/882-8846 706/278-3176
CRSI	Concrete Reinforcing Steel Institute 933 N. Plum Grove Road Schaumburg, IL 60173-4758 www.crsi.org	847/517-1200
CTI	Ceramic Tile Institute of America 12061 W. Jefferson Blvd. Culver City, CA 90230-6219	310/574-7800
DHI	Door and Hardware Institute 14170 Newbrook Drive Chantilly, VA 20151-2223 www.dhi.org	703/222-2010
DIPRA	Ductile Iron Pipe Research Association 245 Riverchase Pkwy East, Suite O Birmingham, AL 35244	205/988-9870
DOC	Department of Commerce 14 <sup>th</sup> Street and Constitution Avenue, NW Washington, DC 20230	202/482-2000
DOT	Department of Transportation 400 Seventh Street, SW Washington, DC 20590	202/366-4000
EJMA ALAMEDA COUNTY	Expansion Joint Manufacturers Association GSA-CP DEFINITIONS A	914/332-0040 AND REFERENCE STANDARDS

ALAMEDA COUNTY GSA-CP Page 7 of 16

Alameda	County	<b>General</b>	Services	<b>A</b> gency
Alameda	County.	General	Services	Agency

**PROJECT # 2043** 

Highland Hospital – Koret Building Exhaust Fan and Waterproofing

25 N. Broadway

Tarrytown, NY 10591-3201

EPA Environmental Protection Agency

202/260-2090

401 M Street, SW

Washington, DC 20460

FCICA Floor Covering Installation Contractors

706/226-5488

Association P.O. Box 948

Dalton, GA 30722-0948

FM Factory Mutual

781/255-4300

1151 Boston-Providence Turnpike

P.O. Box 9102

Norwood, MA 02062-9102 www.factorymutual.com

FS Federal Specifications Unit

202/619-8925

(Available from GSA)

470 East L'Enfant Plaza, SW, Suite 8100

Washington, DC 20407

GA Gypsum Association

202/289-5440

810 First Street NE, Suite 510 Washington, DC 20002

www.usg.com

GANA Glass Association of North America

913/266-7013

3310 SW Harrison Street Topeka, KS 66611-2279

www.glasswebsite.com/gana

HMA Hardwood Manufacturers Association

412/828-0770

400 Penn Center Blvd., Suite 530 Pittsburgh, PA 15235-5605

www.hardwood.org

HPVA Hardwood Plywood and Veneer Association

703/435-2900

1825 Michael Farraday Drive

P.O. Box 2789

Reston, VA 22195-0789

www.hpva.org

IEEE Institute of Electrical and Electronic Engineers

800/678-4333

345 E. 47<sup>th</sup> Street

212/705-7900

ALAMEDA COUNTY GSA-CP Page 8 of 16

New York, NY 10017-2394

www.ieee.org

IESNA Illuminating Engineering Society of North America 212/248-5000

120 Wall Street, 17th Floor New York, NY 10005-4001

www.iesna.org

ITS Intertek Testing Services 800/345-3851

P.O. Box 2040607/753-6711

3933 US Route 11

Cortland, NY 13045-7902 www.itsglobal.com

LMA Laminating Materials Association 201/664-2700

116 Lawrence Street

Hillsdale, NJ 07642-2730

www.lma.org

MCAA Mechanical Contractors Association of America 301/869-5800

1385 Piccard Drive

Rockville, MD 20850-4329

ML/SFA Metal Lath/Steel Framing Association 312/456-5590

(A Division of the NAAMM)

8 South Michigan Avenue, Suite 1000

Chicago, IL 60603

MSS Manufacturers Standardization Society for the 703/281-6613

Valve and Fittings Industry

127 Park Street, NE Vienna, VA 22180-4602

NAA National Arborist Association 800/733-2622

P.O. Box 1094603/673-3311 Amherst, NH 03031-1094

www.natlarb.com

NAAMM National Association of Architectural 312/782-5590

Metal Manufacturers

8 South Michigan Avenue, Suite 1000

Chicago, IL 60603 www.gss.net/naamm

NAIMA North American Insulation Manufacturers 703/684-0084

Association

ALAMEDA COUNTY GSA-CP Page 9 of 16

# **Alameda County, General Services Agency**

**PROJECT # 2043** 

Highland Hospital – Koret Building Exhaust Fan and Waterproofing

44 Canal Center Plaza, Suite 310 Alexandria, VA 22314

www.naima.org

NAPA National Asphalt Pavement Association 301/731-4748

NAPA Building 5100 Forbes Blvd.

Lanham, MD 20706-4413

NCSPA National Corrugated Steel Pipe Association 202/452-1700

1255 23rd Street, NW, Suite 850

Washington, DC 20037

www.ncspa.org

NEBB National Environmental Balancing Bureau 301/977-3698

8575 Grovemont Circle

Gaithersburg, MD 20877-4121

NECA National Electrical Contractors Association 301/657-3110

3 Bethesda Metro Center, Suite 1100

Bethesda, MD 20814-5372

NEI National Elevator Industry 201/944-3211

185 Bridge Plaza North, Suite 310

Fort Lee, NJ 07024

NEMA National Electrical Manufacturers' Association 703/841-3200

1300 N. 17th Street, Suite 1847

Rosslyn, VA 22209 www.nema.org

NFPA National Fire Protection Association 800/344-3555

One Batterymarch Park

P.O. Box 9101

Quincy, MA 02269-9101

www.nfpa.org

NHLA National Hardwood Lumber Association 901/377-1818

P.O. Box 34518

Memphis, TN 38184-0518 www.natlhardwood.org

NIA National Insulation Association 703/683-6422

99 Canal Center Plaza, Suite 222

Alexandria, VA 22314

www.insulation.org

ALAMEDA COUNTY GSA-CP Page 10 of 16 DEFINITIONS AND REFERENCE STANDARDS
DOCUMENT 01 42 16

617/770-3000

NPA	National Particleboard Association 18928 Premiere Court Gaithersburg, MD 20879-1569 www.pbmdf.com	301/670-0604
NPCA	National Paint and Coatings Association 1500 Rhode Island Avenue, NW Washington, DC 20005-5597 www.paint.org	202/462-6272
NRCA	National Roofing Contractors Association O'Hare International Center 10255 W. Higgins Road, Suite 600 Rosemont, IL 60018-5607 www.roofonline.org	800/323-9545
NRMCA	National Ready Mixed Concrete Association 900 Spring Street Silver Spring, MD 20910 www.nrmca.org	301/587-1400
NSF	NSF International P.O. Box 130140 Ann Arbor, MI 48113-0140 www.nsf.org	313/769-8010
NUSIG	National Uniform Seismic Installation Guidelines 12 Lahoma Court Alamo, CA 94526	510/946-0135
NWWDA	National Wood Window and Door Association 1400 E. Touhy Avenue, G-54847/299-5200 Des Plaines, IL 60018 www.nwwda.org	800/223-2301
SHA	Occupational Safety and Health Administration (U.S. Department of Labor) 200 Constitution Ave., NW Washington, DC 20210	202/219-8148
PCA	Portland Cement Association 5420 Old Orchard Road Skokie, IL 60077-1083 www.portcement.org	847/966-6200

Alameda County, G Highland Hospital –	PROJECT # 2043	
PDCA	Painting and Decorating Contractors of America 3913 Old Lee Hwy, Suite 33-B Fairfax, VA 22030 www.pdca.com	800/332-7322 703/359-0826
PDI	Plumbing and Drainage Institute 45 Bristol Drive, Suite 101 South Easton, MA 02375	800/589-8956 508/230-3516
RFCI	Resilient Floor Covering Institute 966 Hungerford Drive, Suite 12-B Rockville, MD 20805-1714	301/340-8580
RIS	Redwood Inspection Service c/o California Redwood Association 405 Enfrente Drive, Suite 200 Novato, CA 94949-7206	415/382-0662
SDI	Steel Deck Institute P.O. Box 25 Fox River Grove, IL 60012 www.sdi.org	847/462-1930
SDI	Steel Door Institute 30200 Detroit Road Cleveland, OH 44145-1967	216/889-0010
SMA	Stucco Manufacturers Association 14006 Ventura Blvd. Sherman Oaks, CA 91403	213/789-8733
SMACNA	Sheet Metal and Airconditioning Contractors National Association, Inc. P.O. Box 221230 Chantilly, VA 20151-1209 www.smacna.org	703/803-2980

	General Services Agency Koret Building Exhaust Fan and Waterproofing	PROJECT # 2043
SPI	Society of the Plastics Industry, Inc. Spray Polyurethane Division 202/974-5200 1801 K Street, NW, Suite 600K Washington, DC 20006 www.socplas.org	800/951-2001
SSPC	Steel Structures Painting Council 40 24th Street, 6th Floor Pittsburgh, PA 15222-4643	412/281-2331
TCA	Tile Council of America 100 Clemson Research Blvd. Anderson, SC 29625	864/646-8453
TPI	Turfgrass Producers International 1855-A Hicks Road Rolling Meadows, IL 60008	800/405-8873 847/705-9898
UL	Underwriters Laboratories, Inc. 333 Pfingston Road 847/272-8800 Northbrook, IL 60062 <a href="https://www.ul.com">www.ul.com</a>	800/704-4050
UNI	Uni-Bell PVC Pipe Association 2655 Villa Creek Drive, Suite 155 Dallas, TX 75234 www.members.aol.com/unibell1	972/243-3902
USDA	U.S. Department of Agriculture 14th St. and Independence Ave., SW Washington, DC 20250	202/720-8732
WA	Wallcoverings Association 401 N. Michigan Avenue Chicago, IL 60611-4267	312/644-6610
WCLIB	West Coast Lumber Inspection Bureau P.O. Box 23145 Portland, OR 97281-3145	503/639-0651
WCMA	Window Covering Manufacturers Association 355 Lexington Ave., 17th Floor New York, NY 10017-6603	212/661-4261
WIC	Woodwork Institute of California P.O. Box 980247	916/372-9943
AT AMEDA COUNTY	CCA CD DEFINITIONS A	NID DEFEDENCE STANDADD

West Sacramento, CA 95798-0247

WLPDIA Western Lath/Plaster/Drywall Industries 619/466-9070

Association 8635 Navajo Road

San Diego, CA 92119

WMMPA Wood Moulding & Millwork Producers 800/550-7889

Association 916/661-9591

507 First Street

Woodland, CA 95695 www.wmmpa.com

WRI Wire Reinforcement Institute 703/779-2339

203 Loudoun Street, SW Leesburg, VA 20175-2718

WWPA Western Wood Products Association 503/224-3930

Yeon Building

522 S.W. 5th Avenue Portland, OR 97204-2122

#### 1.04 **DEFINITIONS**:

Wherever any of the words or phrases defined below regardless of capitalization, or a pronoun used in place thereof, is used in any part of the Contract Documents, it shall have the meaning here set forth:

- (1) ADDENDA: Written or graphic instruments issued prior to the opening of Bids, which clarify, correct or change the bidding requirements or the Contract Documents. Addenda shall not include the minutes of the Pre-bid Conference and Site Visit.
- (2) APPROVED EQUAL: Approved in writing by County as being of equivalent quality, utility and appearance.
- (3) BUSINESS DAY: Any day other than Saturday, Sunday, and the following days that have been designated as holidays by County. If a holiday falls on a Saturday, the preceding Friday will be the holiday. If a holiday falls on a Sunday, the following Monday will be the holiday.
  - a. New Year's Day, January 1;
  - b. Martin Luther King Jr.'s Birthday, third Monday in January;
  - c. Lincoln's Birthday, February 12;
  - d. Presidents' Day, third Monday in February;
  - e. Memorial Day, last Monday in May;
  - f. Independence Day, July 4;

# **Alameda County, General Services Agency**

Highland Hospital – Koret Building Exhaust Fan and Waterproofing

- g. Labor Day, first Monday in September;
- h. Veterans' Day, November 11;
- i. Thanksgiving Day, as designated by the President;
- j. The Day following Thanksgiving Day;
- k. Christmas Day, December 25; and
- 1. Each day appointed by the Governor of California and formally recognized by the County as a day of mourning, thanksgiving, or special observance.
- (4) COMPLETION: The County's acceptance of the Work as satisfactorily completed in accordance with the Contract Documents. Requirements for Completion include, but are not limited to:
  - a. Generally all Work, except Contractor maintenance after Final Acceptance, having been completed in accordance with the Contract Documents.
  - b. All systems having been tested and accepted as having met requirements of the Contract Documents, including Final Test and Balance (TAB) report complete and accepted by MEOR.
  - c. All Commissioning, including by not limited to, final sequence of operations, functional and performance testing, and Building Management System programming, complete.
  - d. All punch list work, as directed by the County, having been completed by the Contractor.
  - e. All as built drawings and operations and maintenance manuals and Machine Inventory Sheets having been submitted by the Contractor, reviewed by and accepted by the County.
  - f. All required instructions and training sessions having been given by the Contractor.
  - g. Receipt of approved REGULATORY AGENCY Construction Advisory Construction Final ("CF") certification, or its REGULATORY AGENCY equivalent.
- (5) CONCEALED: Work not exposed to view in the finished Work, including within or behind various construction elements.
- (6) DEFECTIVE: An adjective which, when modifying the word "Work", refers to Work that is unsatisfactory or unsuited for the use intended, faulty, or deficient, that it does not conform to the Contract Documents, or does not meet the requirements of any inspection, reference standard, test or approval referred to in the Contract Documents (including but not limited to approval of samples and "or equal" items), or has been damaged prior to final payment (unless responsibility for the protection

- thereof has been assumed by County). The County is the judge of whether Work is defective.
- (7) ENGINEER: A person designated by the County authorizing such person to act as the Engineer on its behalf.
- (8) EQUAL: Equal in the opinion of the County. Burden of proof of equality is responsibility of the Contractor.
- (9) EXPOSED: Work exposed to view in the finished Work, including behind louvers, grilles, registers and various other construction elements.
- (10) FACILITY: Location where work is performed and organization operating from the location.
- (11) FIELD DIRECTIVE: A written work change directive to the Contractor from the County, ordering alterations or modifications, which do not result in change in Contract Sum or Contract Times, and do not substantially change the Contract Documents.
- (12) INSPECTOR: The person engaged by the County to inspect the workmanship, materials, or manner of construction of buildings or portions of buildings, to determine if such construction complies with the Contract Documents.
- (13) LATENT: Not apparent by reasonable inspection, including but not limited to, the inspections and research required as a condition to bidding under the General Conditions.
- (14) METHOD OF PROCEDURE (MOP): Written document outlining steps to be taken when performing work outside the Contractor's construction fence, outside the normal work hours, or work that impacts facility operations, e.g. utility shut down. MOP will be developed by the Contractor for review and acceptance by the County and Facility.
- (15) OFF SITE: Outside geographical location of the Project.

END OF DOCUMENT

# **DOCUMENT 01 43 00**

# **QUALITY ASSURANCE - MATERIALS AND EQUIPMENT**

# PART 1 - GENERAL

# 1.01 RELATED DOCUMENTS AND PROVISIONS

All Contract Documents should be reviewed for applicable provisions related to the provisions in this document, including without limitation:

- A. General Conditions, including, without limitation, Purchase of Materials and Equipment;
- B. Special Conditions;
- C. Imported Materials Certification.

# 1.02 MATERIAL AND EQUIPMENT

- A. Only items approved by the County and/or Architect/Engineer shall be used.
- B. Contractor shall submit lists of products and other product information in accordance with the Contract Documents, including, without limitation, the provisions regarding the submittals.

# 1.03 MATERIAL AND EQUIPMENT COLORS

- A. The County and/or Architect/Engineer will provide a schedule of colors.
- B. No individual color selections will be made until after approval of all pertinent materials and equipment and after receipt of appropriate samples in accordance with the Contract Documents, including, without limitation, the provisions regarding the submittals.
- C. Contractor shall request priority in writing for any item requiring advance ordering to maintain the approved Construction Schedule.

# 1.04 DELIVERY, STORAGE, AND HANDLING

A. Contractor shall deliver manufactured materials in original packages, containers, or bundles (with seals unbroken), bearing name or identification mark of manufacturer.

- B. Contractor shall deliver fabrications in as large assemblies as practicable; where specified as shop-primed or shop-finished, package or crate as required to preserve such priming or finish intact and free from abrasion.
- C. Contractor shall store materials in such a manner as necessary to properly protect them from damage. Materials or equipment damaged by handling, weather, dirt, or from any other cause will not be accepted.
- D. Materials are not be acceptable that have been warehoused for long periods of time, stored or transported in improper environment, improperly packaged, inadequately labeled, poorly protected, excessively shipped, deviated from normal distribution pattern, or reassembled.
- E. Contractor shall store material so as to cause no obstructions of sidewalks, roadways, and underground services. Contractor shall protect material and equipment furnished under Contract.
- F. Contractor may store materials on Site with prior written approval by the County, all material shall remain under Contractor's control and Contractor shall remain liable for any damage to the materials. Should the Project Site not have storage area available, the Contractor shall provide for off-site storage at no cost to County.
- G. When any room in Project is used as a shop or storeroom, the Contractor shall be responsible for any repairs, patching, or cleaning necessary due to that use. Location of storage space shall be subject to prior written approval by County.

# **PART 2 - PRODUCTS**

# 2.01 MANUFACTURERS

- A. Manufacturers listed in various sections of Contract Documents are names of those manufacturers that are believed to be capable of supplying one or more of items specified therein.
- B. The listing of a manufacturer does not imply that every product of that manufacturer is acceptable as meeting the requirements of the Contract Documents.

# 2.02 FACILITIES AND EQUIPMENT

A. Contractor shall provide, install, maintain, and operate a complete and adequate facility for handling, the execution, disposal, and distribution of material and equipment as required for proper and timely performance of Work connected with Contract.

# 2.03 MATERIAL REFERENCE STANDARDS

A. Where material is specified solely by reference to "standard specifications" and if requested by County, Contractor shall submit for review data on actual material proposed to be incorporated into Work of Contract listing name and address of vendor, manufacturer, or producer, and trade or brand names of those materials, and data substantiating compliance with standard specifications.

# **PART 3 - EXECUTION**

#### 3.01 WORKMANSHIP

- A. Where not more specifically described in any other Contract Documents, workmanship shall conform to methods and operations of best standards and accepted practices of trade or trades involved and shall include items of fabrication, construction, or installation regularly furnished or required for completion (including finish and for successful operation, as intended).
- B. Work shall be executed by tradespersons skilled in their respective lines of Work. When completed, parts shall have been durably and substantially built and present a neat appearance.

# 3.02 COORDINATION

- A. Contractor shall coordinate installation of Work so as to not interfere with installation of others. Adjustment or rework because of Contractor's failure to coordinate will be at no additional cost to County.
- B. Contractor shall examine in-place work for readiness, completeness, fitness to be concealed or to receive other work, and in compliance with Contract Documents. Concealing or covering Work constitutes acceptance of additional cost which will result should in-place Work be found unsuitable for receiving other Work or otherwise deviating from the requirements of the Contract Documents.

# 3.03 COMPLETENESS

A. Contractor shall provide all portions of the Work, unless clearly stated otherwise, installed complete and operational with all elements, accessories, anchorages, utility connections, etc., in manner to assure well-balanced performance, in accordance with manufacturer's recommendations and by Contract Documents. For example, electric water coolers require water, electricity, and drain services; roof drains require drain system; sinks fit within countertop, etc. Terms such as "installed complete," "operable condition," "for use intended," "connected to all utilities," "terminate with proper cap," "adequately anchored," "patch and refinish," "to match similar," should be assumed to apply in all cases, except

where completeness of functional or operable condition is specifically stated as not required.

# 3.04 APPROVED INSTALLER OR APPLICATOR

A. Installation by a manufacturer's approved installer or applicator is an understood part of Specifications and only approved installer or applicator is to provide onsite Work where specified manufacturer has on-going program of approving (i.e. certifying, bonding, re-warranting) installers or applicators. Newly established relationships between a manufacturer and an installer or applicator that does not have other approved applicator work in progress or completed is not approved for this Project.

# 3.05 MANUFACTURER'S RECOMMENDATIONS

A. All installations shall be in accordance with manufacturer's published recommendations and specific written directions of manufacturer's representative. Should Contract Documents differ from recommendations of manufacturer or directions of his representative, Contractor shall analyze differences, make recommendations to the County and the Architect/Engineer in writing, and shall not proceed until interpretation or clarification has been issued by the County and/or the Architect/Engineer.

END OF DOCUMENT

# **DOCUMENT 01 45 00**

# **QUALITY CONTROL**

#### PART 1 - GENERAL

# 1.01 RELATED DOCUMENTS AND PROVISIONS:

All Contract Documents should be reviewed for applicable provisions related to the provisions in this document, including without limitation:

- A. General Conditions, including, without limitation, Inspections and Tests, Uncovering of Work and Non-conforming of Work and Correction of Work;
- B. Special Conditions.

# 1.02 RELATED CODES:

A. The Work is governed by requirements of Title 24, California Code of Regulations ("CCR"), and the Contractor shall keep a copy of these available at the job Site for ready reference during construction.

#### 1.03 DEFINITIONS:

- A. <u>Quality Control Services</u>: Tests, inspections, procedures, and related actions during and after execution of the Work to evaluate that actual products incorporated into the Work and completed construction comply with requirements. Services do not include contract enforcement activities performed by County's Representative.
- B. NRTL: A nationally recognized testing laboratory according to 29 CFR 1910.7.
- C. <u>NVLAP</u>: A testing agency accredited according to the National Institute of Standards and Technology's (NIST's) National Voluntary Laboratory Accreditation Program.
- D. <u>Preconstruction Testing</u>: Tests and inspections that are performed specifically for the Project before products and materials are incorporated into the Work to verify performance or compliance with specified criteria.
- E. <u>Product Testing</u>: Tests and inspections that are performed by an NRTL, an NVLAP, or a testing agency qualified to conduct product testing and acceptable to County's Representative, to establish product performance and compliance with industry standards.

- F. <u>Source Quality Control Testing</u>: Tests and inspections that are performed at the source (i.e., a plant, mill, factory, or shop).
- G. <u>Field Quality Control Testing</u>: Tests and inspections that are performed on-site for installation of the Work and for completed Work.
- H. <u>Testing Agency</u>: An entity engaged to perform specific tests, inspections, or both that is certified as meeting the requirements applicable to the Work. Testing laboratory shall mean the same as testing agency.
- I. <u>Testing, Inspection and Observation (TIO) Program</u>: A program prepared for Regulatory Agency approval prior to issuance of the building permit, pursuant to Section 7-141, California Building Standards Code, Part 1, Title 24, CCR. The program identifies the materials and tests to be performed on a project and the firm(s) and/or individual(s) responsible for performing those tests including, at a minimum, those required by applicable sections of the California Building Standards Code.
- J. <u>Installer/Applicator/Erector</u>: Design Builder or another entity engaged by Design Builder as an employee or Subcontractor of any tier to perform a particular construction operation, including installation, erection, application, and similar operations.
- K. <u>Experienced</u>: As used herein, an individual or entity that has successfully completed a minimum of five previous projects similar in size and scope to this Project; being familiar with special requirements indicated; and having complied with requirements of authorities having jurisdiction to work in California.

# 1.04 CONFLICTING REQUIREMENTS

- A. <u>General</u>: If compliance with two or more standards is specified and the standards establish different or conflicting requirements for minimum quantities or quality levels, comply with the most stringent requirement. Refer uncertainties and requirements that are different, but apparently equal, to County's Representative for a decision before proceeding.
- B. <u>Minimum Quantity or Quality Levels</u>: The quantity or quality level shown or specified shall be the minimum provided or performed. The actual installation may comply exactly with the minimum quantity or quality specified, or it may exceed the minimum within reasonable limits. To comply with these requirements, indicated numeric values are minimum or maximum, as appropriate, for the context of requirements. Refer uncertainties to County's Representative for a decision before proceeding.

The County, Architect, and Engineer or their appointed representatives will review the Work and the Contractor shall provide facilities and access to the Work at all times as required to facilitate this review. Administration by the Architect and any consulting Engineer will be in accordance with applicable regulations.

# 1.04 TESTING AGENCIES:

- A. Testing agencies and tests shall be in conformance with the General Documents.
- B. Testing and inspection in connection with earthwork shall be under the direction of the County's consulting soils engineer, if any, referred to hereinafter as the "Soils Engineer."
- C. Testing and inspection of construction materials and workmanship shall be performed by a qualified laboratory, referred to hereinafter as the "Testing Laboratory." The Testing Laboratory shall be under direction of an engineer registered in the State of California, shall conform to requirements of ASTM E329, and shall be employed by or in contract with the County.

#### 1.05 TESTS AND INSPECTIONS:

- A. Materials to be furnished under the Contract Documents are subject to testing and inspection for compliance with the Drawings and Specifications. Testing and inspecting services are required to verify compliance with requirements specified or indicated. These services do not relieve Contractor of responsibility for compliance with the Contract Document requirements.
  - (1) Specific quality assurance and control requirements for individual construction activities are specified in the Sections that specify those activities. Requirements in those Sections may also cover production of standard products.
  - (2) Specified tests, inspections, and related actions do not limit Contractor's other quality assurance and control procedures that facilitate compliance with the Contract Document requirements.
  - (3) Requirements for Contractor to provide quality assurance and control services required by County, or Regulatory Agency are not limited by provisions of this Section.

- B. The County shall employ and assign an Inspector of Record (IOR) to the Work in accordance with the requirements of the California Code of Regulations, Title 24, Part 1. The Inspector's duties are specifically defined in Title 24.
  - (1) The Work in all stages of progress shall be subject to the continuous observation of the Inspector. The Inspector shall have free access to all parts of the Work at any time. The Contractor shall keep the IOR fully informed regarding the progress and manner of the Work and the character of the materials. Inspection of the Work shall not relieve the Contractor of any obligation contained in the Contract Documents.
- C. The Contractor shall be responsible for notifying the Construction Manager of all required tests and inspections. Contractor shall notify the Construction Manager seventy-two (472) hours in advance of performing any Work requiring testing or inspection.
- D. The Contractor shall provide access to Work to be tested and furnish incidental labor, equipment, and facilities to facilitate all inspections and tests.
- E. The County will pay for first inspections and tests required by the "CCR", and other inspections or tests that the County and/or the Architect/Engineer may direct to have made, including the following principal items:
  - (1) Tests and observations for earthwork and paving.
  - (2) Tests for concrete mix designs, including tests of trial batches.
  - (3) Tests and inspections for structural steel work.
  - (4) Field tests for framing lumber moisture content.
  - (5) Additional tests directed by the County that establish that materials and installation comply with the Contract Documents.
  - (6) Test and observation of welding and expansion anchors.
- F. The County may at its discretion, pay and back charge the Contractor for:
  - (1) Retests or reinspections, if required, and tests or inspections required due to Contractor error or lack of required identifications of material.
  - (2) Uncovering of work in accordance with Contract Documents.
  - (3) Testing done on weekends, holidays, and overtime will be chargeable to the Contractor for the overtime portion.

- (4) Testing done off Site.
- G. Testing and inspection reports and certifications:
  - (1) If initially received by Contractor, Contractor shall provide to each of the following a copy of the agency or laboratory report of each test or inspection or certification.
    - a. The County;
    - b. The Construction Manager, if any;
    - c. The Architect;
    - d. The Consulting Engineer, if any;
    - e. Other Engineers on the Project, as appropriate; and
    - f. The Inspector.
- H. OSHPD required TIO Program:
  - (1) Contractor will be responsible for testing, inspections, and observations required in the OSHPD TIO form for each project. TIOs are included immediately following this document.

# **PART 2 - PRODUCTS**

# 2.01 TYPE OF TEST AND INSPECTIONS:

- A. Slump Test ASTM C 143
- B. Concrete Tests

  Testing agency shall test concrete used in the work per the following paragraphs:
  - (1) Compressive Strength:
    - a. Minimum number of tests required: One (1) set of three (3) cylinders for each 100 cubic yards (Sec. 2604(h) 01) of concrete or major fraction thereof, placed in one (1) day. See Title 24, Section 2605(g).

- b. Two cylinders of each set shall be tested at twenty-eight (28) days. One (1) cylinder shall be held in reserve and tested only when directed by the Architect, Engineer, or County.
- c. Concrete shall test the minimum ultimate compressive strength in 28 days, as specified on the structural drawings.
- d. In the event that the twenty-eight (28) day test falls below the minimum specified strength, the effective concrete in place shall be tested by taking cores in accordance with UBC Standard No. 26-13 and tested as required for cylinders.
- e. In the event that the test on core specimens falls below the minimum specified strength, the concrete will be deemed defective and shall be removed and replaced upon such direction of the Architect/Engineer.
- C. Reinforcing, Steel
- D. Structural Steel Per Title 24 and as noted:
  - (1) Material: Steel per Table in Title 24, Section 2712.
  - (2) Qualification of Welders (UBC Std. 27-6).
  - (3) Shop fabrication (Section 2712(d). Structural steel only).
  - (4) Shop and field welding (Section 2712(e)).

END OF DOCUMENT



# OFFICE OF STATEWIDE HEALTH PLANNING AND DEVELOPMENT

FACILITIES DEVELOPMENT DIVISION – www.oshpd.ca.gov/fdd 400 R Street, Suite 200 ~ Sacramento, California 95811 700 N. Alameda Street, Suite 2-500 ~ Los Angeles, California 90012

Phone (916) 440-8300 Phone (213) 897-0166 FAX (916) 324-9188 FAX (213) 897-0168

# **Testing, Inspection and Observation Program**

2016 California Building Standards Code - OSHPD 1

This program is prepared and submitted for an OSHPD 1 project. OSHPD 1 projects include all construction and remodel projects for general acute-care hospitals and acute psychiatric hospitals. OSHPD 1 projects also include construction and remodel of skilled nursing facilities and/or intermediate care facilities except those of single-story, Type V. wood or light steel-frame construction.

NA		PROJECT INFORMATION					
SELECTION CONTRACTOR	Facility Name:	TRUE STATE	Project #:	Sub #:			
Highland Hospital			S192339-01-00	4 4			
1411 East 31st Street				1.46			
Oakland	County:	Alameda	that your care	4 2 4 2			
(Scope of Project):	Koret Building En	nergency Department Exh	aust Fan				
	1411 East 31st Street Oakland	Facility Name: Highland Hospital  1411 East 31st Street  Oakland  County:	Facility Name: Highland Hospital  1411 East 31st Street  Oakland  County: Alameda	Facility Name: Project #: Highland Hospital S192339-01-00  1411 East 31st Street  Oakland County: Alameda			



OFFICE OF STATEWIDE HEALTH PLANNING AND DEVELOPMENT FACILITIES DEVELOPMENT DIVISION – www.oshpd.ca.gov/fdd 400 R Street, Suite 200 ~ Sacramento, California 95811 Phone (916) 4700 N. Alameda Street, Suite 2-500 ~ Los Angeles, California 90012 Phone (213) 8

Phone (916) 440-8300 Phone (213) 897-0166

FAX (916) 324-9188 FAX (213) 897-0168

# **Testing, Inspection and Observation Program** 2016 California Building Standards Code – OSHPD 1

S	EC	TION B		Approved agencies, individuals, and all changes to the TIO proed by the DPOR and approved by OSHPD prior to proceeding v		
Facility #: Facility Name:		Facility Name:		Project #:		Sub#
1	11288 Highland Hospital		S192339-01-00			0
DURING CONSTRUCTION DOCUMENT SUBMITTAL		DURING CONSTRUCTION				
lndex #	REQUIRED (Select)	TESTS	PERFORMED OFF-SITE	RESPONSIBLE APPROVED AGENCY AND/OR INDIVIDUAL	COMPLIANCE VERIFICATION BY IOR	OSHPD/FDD
		AL TESTS	FeW.			1000
Conc	rete				and the second	med
B-C15	х	Post-installed anchors CBC 1910A.5 Installation verification test		CONTRACTOR		DSE:
Vons	tructi	ural Components, Supports and Attachments				
B-N2	х	Special Seismic Certification CBC 1705A.13.3 Shake Table test				DSE:
CTRI	ICAL	TESTS				
B-E1	Х	Ground fault protection CEC 230.95(C) & 517.17(D) Performance test		CONTRACTOR		CO:
B-E3	Х	Wet Location 2-15 NFPA 99 6.3.2.2.8.5 (B)		CONTRACTOR		CO:
CHA	NIC	AL TESTS				
B-ME9	Х	Ventilation system CMC 407.3.1 & Table 4-A Areas tested and balanced		INDEPENDENT CONTRACTOR		CO:



Phone (916) 440-8300 Phone (213) 897-0166 FAX (916) 324-9188 FAX (213) 897-0168

SE	EC	TION C		Approved agencies, individuals, and all changes to the TIO proged by the DPOR and approved by OSHPD prior to proceeding w						
Facili	ity #:	Facility Name:		Project #:	i a valida	Sub #:				
11	1288	Highland Hospital	The Top 1	[8851]	0					
DI	URI	NG CONSTRUCTION DOCUMENT SUBMITTAL	AL DURING CONSTRUCTION							
# xəpul	REQUIRED (Select)	SPECIAL INSPECTIONS	PERFORMED OFF-SITE	RESPONSIBLE APPROVED AGENCY AND/OR INDIVIDUAL	COMPLIANCE VERIFICATION BY IOR	OSHPD/FDD USE				
JCT oncr	-	AL SPECIAL INSPECTIONS		the state of the s		_				
J	ete	la martina de la companya della companya della companya de la companya della comp	T .	T		DSE:				
ဂ်	Χ	Concrete CBC 1705A.3; ACI 318 26.5.2 & 26.13 Placement of concrete				DOL.				
C-C4	Х	Concrete CBC 1705A.3; ACI 318 26.13 Reinforcing steel & prestressing steel				DSE:				
C-C5	Х	Concrete CBC 1705A.3 CIP & Post-installed anchors				DSE:				
onst	ructi	ural components, supports and attachments								
C-N5	Х	Plumbing, mechanical and electrical components CBC 1705A.12.6 Anchorage, bracing, and vibration isolators		CONTRACTOR		CO:				
C-N6	X	Special Seismic Certification CBC 1705A.12.4 Special Seismic Certification label, anchorage and mounting				CO:				
PRO	OTE	CTION SPECIAL INSPECTIONS		-						
C-FP3	Х	Penetration firestops CBC 1705A.17.1 Penetration firestop systems that are tested and listed				FLSO:				



Phone (916) 440-8300 Phone (213) 897-0166

FAX (916) 324-9188 FAX (213) 897-0168

SEC1	TION D		C	ONS	TRI	UCT	ION	I VE	RIF	ICA	TIC	N
Facility #:	Facilit	Name: Amplifigati 4 Project #:							Sub #:			
11288	Highlan	nd Hospital S192339-01-00							1385 FT			
in .	VERIFIED CONSTRUCTION	N INSPECTION	ON AND	OBSER	VATION	REPOR	TING	TWE CO				
Reference	PROJECT MILESTONE OR INTERVAL	(See	VERIFIED COMPLIANCE REPORT REQUIRED AS INDICATED (Form OSH-FD-123) (See "PERSONAL KNOWLEDGE" as defined in California Administrative Code, Section 7-151)						Code,	FOR OSHPE USE ONLY		
		GEOR	AOR	SEOR	MEOR	EEOR	CONT	IOR	SP INSP	TEST LAB		OSHPD FDD
	PROJECT COMPLETION			х	х		Х	Х	7		-	



Phone (916) 440-8300 Phone (213) 897-0166 FAX (916) 324-9188 FAX (213) 897-0168

SECTION F	PLAN R	EVIEW APPROVAL				
Facility #:	Facility Name:	Project #:	Sub #:			
11288	Highland Hospital	S192339-01-00	Jer i			
NOTE: For test	ing, Inspection and Observation Program Instructions, visit	our website: http://oshpd.ca.gov/FDD/Plan_Review/T	IO.html#TIO			
Submitted by: I have reviewed the a as "required" on this f	pproved construction documents for this project and form.	d all tests and special inspections required by 0	Code are marked			
Timothy J. Belke	carar fello agab mile ameni le (*) as bossino	Twistly & Shelle	12-19-2019			
Architect/Engineer of Reco	ord (Print Name)	Architect/Engineer of Record (Signature)	Date			
Maryann T. Phipps	s	Maryann J. Phipps 12-19-2				
Structural Engineer of Rec	ord (Print Name)	Structural Engineer of Record (Signature) Date				
	FOR OSHPD US	E ONLY				
OSHPD Plan Approval:  Name	REVIEWED IN ACCORDANCE WITH THE REQUIREMENTS OF T24, CCR  APPROVED  Dale	A AC D	Abbits none with a plate and a			
Comments:	Connie Christensen, Sr. Architect Office of Statewide Health Planning & Development FACILITIES DEVELOPMENT DIVISION	CAPRO TO THE SERVICE DESCRIPTION OF THE SERVICE OF	mages, and the mages done manner holling call and manner			



Phone (916) 440-8300 Phone (213) 897-0166

FAX (916) 324-9188 FAX (213) 897-0168

SECTION G	BUILDING	PERMIT APPROVAL	
Facility #:	Facility Name:	Project #:	Sub #:
11288	Highland Hospital	S192339-01-00	
NOTE: For testing, Inspection	and Observation Program Instructions, visit o	ur website: http://oshpd.ca.gov/FDD/Plan_Review/	TIO.html#TIO
Attached  To be provided following determine to proceeding with the work of the prior to proceeding with the work of the prior to proceeding with the work of the project has not applicable. In addition to the preprinted tests a indicated:  Other Tests  Other Special Inspections  See Attachment  Verification that approved test and 1703A.1.1:  Verification of independence and in accordance with the CAC Simple projects for general acute care in the project for general acute care in the project for general g	rmination of the responsible firm(s) or individually that requires tests or special inspections. To required tests or special inspections. To required tests or special inspections. To required tests or special inspections.  Ports shall be prepared and submitted to the Control of test and inspection agencians acceptance of test and inspection agencians approval or approval of testing agencies the for approval.  If and submitted for an OSHPD 1 project, appricals and acute psychiatric hospitals.	n, this program includes additional tests and spec inpetent and independent as required by the G ies by Registered Design Professional (RDP) in	ion of all tests cial inspections as CBC 2016 Section responsible charge
Timothy J. Belke	M20577	Trinothy 9 helle	12-19-2019
Architect/Engineer of Record (Print Name)	Professional License #	Architect/Engineer of Record (Signature)	Date
	FOR OSHPD USE	ONLY	
OSHPD TI&O Program Approval:	*****		
Name	Date	A AC D	
Comments:			



Phone (916) 440-8300 Phone (213) 897-0166

FAX (916) 324-9188 FAX (213) 897-0168

SECT	TION H	NOTE: Note all changes shall be approved 7-153, or "concurred with" as non-materia	SUMMARY OF CHANGES TO THE TIO PROGRAM  NOTE: Note all changes shall be approved as amended construction documents per 7-153, or "concurred with" as non-material alterations per CAC 7-153(b), by OSHPD to proceeding with the related work.						
Facility #:	Facility #: Facility Name: Project #:								
11288	Hig	hland Hospital							
		ATION PROGRAM NON-MATERIALLY ALTER CHANG D / ENGINEER OF RECORD IN RESPONSIBLE CHARG							
	umn F or G indicates the indicated DPOR has revie o any approved agencies/testing laboratories and sp	wed the changes made and the acceptance statement i pecial inspectors added.	in Section I of this Form is	OSHPD FDD CONCURRENCE					
REVISION #	SYNOPSIS OF CHANGE	Architect/Engineer of Structural Engineer Record Signature Record Signature							
1	Added C-C5.	Tringthy ghelle Maryann J. Ah	ipps 12/19/2019						



**FACILITIES DEVELOPMENT DIVISION** – www.oshpd.ca.gov/fdd 400 R Street, Suite 200 ~ Sacramento, California 95811 700 N. Alameda Street, Suite 2-500 ~ Los Angeles, California 90012

Phone (916) 440-8300 Phone (213) 897-0166 FAX (916) 324-9188 FAX (213) 897-0168

# **Testing, Inspection and Observation Program**

2016 California Building Standards Code - OSHPD 1

This program is prepared and submitted for an OSHPD 1 project. OSHPD 1 projects include all construction and remodel projects for general acute-care hospitals and acute psychiatric hospitals. OSHPD 1 projects also include construction and remodel of skilled nursing facilities and/or intermediate care facilities except those of single-story, Type V, wood or light steel-frame construction.

SECTION	ON A	PROJEC	PROJECT INFORMATION							
Facility #:		Project #:	Sub #:							
11288	Н	S192916-01-00	1							
Street Address:	1411 E 31st Street		•							
City:	Oakland	County: Alameda								
Record Nam	ne (Scope of Project):	Building K Exterior Maintenance								
Abbreviations: CAC: California	Administrative Code CBC: California Building	Code RDP: Registered Design Professiona	I .	Version: R02.5						



**FACILITIES DEVELOPMENT DIVISION** – www.oshpd.ca.gov/fdd 400 R Street, Suite 200 ~ Sacramento, California 95811 700 N. Alameda Street, Suite 2-500 ~ Los Angeles, California 90012

Phone (916) 440-8300 Phone (213) 897-0166 FAX (916) 324-9188 FAX (213) 897-0168

# **Testing, Inspection and Observation Program**

2016 California Building Standards Code - OSHPD 1

SECT	TION D		C	ONS	TRI	JCT	ION	VE	RIF	ICAT	IC	N
Facility #:	Facility	Name:	Name: Project #:							Sub #:		
11288	Highland	d Hospital S192916-01-00										
	VERIFIED CONSTRUCTION	INSPECTION	ON AND	OBSER	VATION	REPOR	TING					
Reference	VERIFIED COMPLIANCE REPORT REQUIRED AS INDICATED (Form OSH-FD-123) (Form OSH-FD-123) (See "PERSONAL KNOWLEDGE" as defined in California Administrative Section 7-151)							ode,	FOR OSHPO USE ONLY			
		GEOR	AOR	SEOR	MEOR	EEOR	CONT	IOR	SP INSP	TEST LAB		OSHPD FDE
	PROJECT COMPLETION		X,				х	Х				



FACILITIES DEVELOPMENT DIVISION – www.oshpd.ca.gov/fdd 400 R Street, Suite 200 ~ Sacramento, California 95811 700 N. Alameda Street, Suite 2-500 ~ Los Angeles, California 90012

Phone (916) 440-8300 Phone (213) 897-0166 FAX (916) 324-9188 FAX (213) 897-0168

# **Testing, Inspection and Observation Program**

2016 California Building Standards Code - OSHPD 1

SECTION	SECTION F PLAN REVIEW APPROVAL					
Facility #:	Facility Name:	Project #:	Sub #:			
11288	Highland Hospital	S192916-01-00				
NOTE: For t	esting, Inspection and Observation Program Instruction	ns, visit our website: http://oshpd.ca.gov/FDD/Plan_Review/TIO.ht	ml#TIO			
Submitted by: I have reviewed the as "required" on the		ect and all tests and special inspections required by Code	e are marked			
Christopher P. I	Decareau	CARD	1/13/2020			
Architect/Engineer of R	ecord (Print Name)	Architect/Engineer of Record (Signature)	Date			
Structural Engineer of I	Record (Print Name)	Structural Engineer of Record (Signature)	Date			
	FOR OSH	PD USE ONLY				
OSHPD Plan Approva Name	REVIEWED IN ACCORDANCE WITH THE REQUIREMENTS OF T24, CCR  APPROVED  Date of the province of th	te A AC D				
Comments:	Connie Christensen, Sr. Architect  Office of Statewide Health Planning & Development FACILITIES DEVELOPMENT DIVISION					
	5/4/2020, 4:50:33 PM S192916-01-00					



FACILITIES DEVELOPMENT DIVISION – www.oshpd.ca.gov/fdd 400 R Street, Suite 200 ~ Sacramento, California 95811 700 N. Alameda Street, Suite 2-500 ~ Los Angeles, California 90012

Phone (916) 440-8300 Phone (213) 897-0166 FAX (916) 324-9188 FAX (213) 897-0168

# **Testing, Inspection and Observation Program**

2016 California Building Standards Code - OSHPD 1

SECTION G	BUILDING	PERMIT	APPROVA	L
Facility #:	Facility Name:		Project #:	Sub #:
11288	Highland Hospital		S192916-01-00	
NOTE: For testing, Inspection a	nd Observation Program Instructions, visit ou	r website: http://os	hpd.ca.gov/FDD/Plan_Re	eview/TIO.html#TIO
Samples of Test and Inspection Attached To be provided following deterprior to proceeding with the work  X Not applicable. Project has not repand inspections. If not designate In addition to the preprinted tests a indicated: Other Tests Other Special Inspections See Attachment  Verification that approved test at 1703A.1.1: Verification of independence a In accordance with the CAC S X Testing agency qualification fo Inspection agency qualification This program has been prepared projects for general acute care he	Reports are: (NOT required for tests per mination of the responsible firm(s) or individually that requires tests or special inspections. The required tests or special inspections are required tests or special inspections. The required tests or special inspections are required tests or special inspections. The reports shall be submitted to the O and special inspections identified on this form and inspection agencies are objective, control acceptance of test and inspection agencies the reports approval or approval of testing agencies the	poshpo 1 project OSHPD 1 project OSHPD 1 project OSHPD 1 project	in days of the coalendar days.  Judes additional tests and pendent as required by Design Professional (Ruram.	agh OPAA Program)  diapproved by the Office,  empletion of all tests  and special inspections as  by the CBC 2016 Section  DP) in responsible charge
Christopher P. Decareau	25403 (CA Architect)	Chi.	LPD.	1/13/2020
Architect/Engineer of Record (Print Name)	Professional License #	Architect/Enginee	er of Record (Signature)	Date
	FOR OSHPD USE (	ONLY		
OSHPD TI&O Program Approval:				
Name	Date	A	AC	D
Comments:				

#### **DOCUMENT 01 50 00**

### **TEMPORARY FACILITIES AND CONTROLS**

#### PART 1 – GENERAL

#### 1.01 RELATED DOCUMENTS AND PROVISIONS:

All Contract Documents should be reviewed for applicable provisions related to the provisions in this document, including without limitation:

- A. General Conditions
- B. Special Conditions

#### 1.02 TEMPORARY UTILITIES:

- A. Electric Power and Lighting
  - (1) The County will furnish and pay for power during the course of the work to the extent power is available in the building(s) or on the Site. The Contractor shall be responsible for providing temporary facilities required to deliver that power service from its existing location in the building(s) or on the Site to point of intended use.
  - (2) Contractor shall verify characteristics of power available in building(s) or on the Site. Contractor shall take all actions required to make modifications where power of higher voltage or different phases of current are required. Contractor shall be fully responsible for providing that service and shall pay all costs required therefor.
  - (3) The Contractor shall furnish, wire for, install, and maintain temporary electrical lights wherever it is necessary to provide illumination for the proper performance and/or observation of the Work: a minimum of 20 foot-candles for rough work and 50 foot-candles for finish work.

#### B. Heat and Ventilation

(1) Contractor shall provide temporary heat to maintain environmental conditions to facilitate progress of the Work, to meet specified minimum conditions for the installation and curing of materials, and to protect materials and finishes from damage due to improper temperature and humidity conditions. Portable heaters shall be standard units complete with controls.

- (2) Contractor shall provide forced ventilation and dehumidification, as required, of enclosed areas for proper installation and curing of materials, to disperse humidity, and to prevent hazardous accumulations of dust, fumes, vapors, and gases.
- (3) Contractor shall pay the costs of installation, maintenance, operation, and removal of temporary heat and ventilation, including costs for fuel consumed, required for the performance of the Work.

#### C. Water

- (1) The County will furnish and pay for water during the course of the work to the extent water is then available in the building(s) or on the Site. The Contractor shall be responsible for providing temporary facilities required to deliver such utility service from its existing location in the building(s) or on the Site to point of intended use.
- (2) Contractor shall use backflow preventers on water lines at point of connection to County's water supply. Backflow preventers shall comply with requirements of Uniform Plumbing Code.
- (3) Contractor shall make potable water available for human consumption.

### D. Sanitary Facilities

- (1) Contractor shall provide sanitary temporary facilities in no fewer numbers than required by law and such additional facilities as may be directed by the County for the use of all workers. The facilities shall be maintained in a sanitary condition at all times and shall be left at the Site until removal is directed by the County or Contractor completes all other work at the Site.
- (2) Use of toilet facilities in the Work under construction shall not be permitted except by consent of the County.

# E. Telephone Service

- (1) Contractor shall arrange with local telephone service company for telephone service for the performance of the Work. Contractor shall, at a minimum, provide in its field office one line for telephone and one line for fax machine.
- (2) Contractor shall pay the costs for telephone and fax lines installation, maintenance, service, and removal.

#### F. Fire Protection:

- (1) Contractor shall provide and maintain fire extinguishers and other equipment for fire protection. Such equipment shall be designated for use for fire protection only and shall comply with all requirements of the California Fire, State Fire Marshall and/or its designee.
- (2) Where on-site welding and burning of steel is unavoidable, Contractor shall provide protection for adjacent surfaces. Contractor shall comply with Facility procedures and processes for "hot work".

#### G. Trash Removal:

(1) See Document 01 35 13.26

#### 1.03 CONSTRUCTION AIDS:

- A. Plant and Equipment:
  - (1) Contractor shall furnish, operate, and maintain a complete plant for fabricating, handling, conveying, installing, and erecting materials and equipment; and for conveyances for transporting workmen. Include elevators, hoists, debris chutes, and other equipment, tools, and appliances necessary for performance of the Work.
  - (2) Contractor shall maintain plant and equipment in safe and efficient operating condition. Damages due to defective plant and equipment, and uses made thereof, shall be repaired by Contractor at no expense to the County.
- B. None of the County's tools and equipment shall be used by Contractor for the performance of the Work

#### 1.04 BARRIERS AND ENCLOSURES:

- A. Contractor shall obtain the County's written permission for locations and types of temporary barriers and enclosures, including fire-rated materials proposed for use, prior to their installation.
- B. Contractor shall provide and maintain temporary enclosures to prevent public entry and to protect persons using other buildings and portions of the Site and/or Premises, the public, and workers. Contractor shall also protect the Work and existing facilities from the elements, and adjacent construction and improvements, persons, and trees and plants from damage and injury from demolition and construction operations. Barriers shall be chain link fencing with fabric except areas with a safety concern or those that are highly visible to the public, e.g. front entry of facility. In visible areas painted plywood barriers are required.

C. Contractor shall provide site access to existing facilities for persons using other buildings and portions of the Site, the public, and for deliveries and other services and activities.

#### D. Public and Private Property Protection:

- (1) Contractor shall protect, shore, brace, support, and maintain all underground pipes, conduits, drains, and other underground construction uncovered or otherwise affected by its construction operations. All pavement, surfacing, driveways, curbs, walks, buildings, utility poles, guy wires, fences, and other surface structures affected by construction operations, together with all sod and shrubs in yards, parkways, and medians, shall be restored to their original condition, whether within or outside the easement. All replacements shall be made with new materials.
- (2) Contractor shall be responsible for all damage to streets, roads, highways, shoulders, ditches, embankments, culverts, bridges, and other public or private property, regardless of location or character, which may be caused by transporting equipment, materials, or workers to or from the Work, Site or any part thereof, whether by Contractor or Subcontractors. Contractor shall make satisfactory and acceptable arrangements with the County, or the agency or authority having jurisdiction over the damaged property, concerning its repair or replacement or payment of costs incurred in connection with the damage.
- (3) All fire hydrants and water control valves shall be kept free from obstruction and available for use at all times.

#### E. Tree and Plant Protection:

- (1) Contractor shall preserve and protect existing trees and plants on the Premises that are not designated or required to be removed, and those adjacent to the Premises.
- (2) Contractor shall provide barriers to a minimum height of 4'-0" around drip line of each tree and plant, around each group of trees and plants, as applicable, in the proximity of demolition and construction operations.
- (3) Contractor shall not park trucks, store materials, perform Work or cross over landscaped areas. Contractor shall not dispose of paint thinners, water from cleaning, plastering or concrete operations, or other deleterious materials in landscaped areas, storm drain systems, or sewers. Plant materials damaged as a result of the performance of the Work shall, at the option of the County and at Contractor's expense, either be replaced with new plant materials equal in size to those damaged or by payment of an

- amount representing the value of the damaged materials as determined by the County.
- (4) Contractor shall remove soil that has been contaminated during the performance of the Work by oil, solvents, and other materials which could be harmful to trees and plants, and replace with good soil, at Contractor's expense
- (5) Excavation Around Trees:
  - (a) Excavation within drip lines of trees shall be done only where absolutely necessary and with written permission from the County.
  - (b) Where trenching for utilities is required within drip lines, tunneling under and around roots shall be by hand digging and shall be approved by the County. Main lateral roots and taproots shall not be cut. All roots 2 inches in diameter and larger shall be tunneled under and heavily wrapped with wet burlap so as to prevent scarring or excessive drying. Smaller roots that interfere with installation of new work may be cut with prior approval by the County. Roots must first be cut with a Vermeer, or equivalent, root cutter prior to any trenching.
  - (c) Where excavation for new construction is required within drip line of trees, hand excavation shall be employed to minimize damage to root system. Roots shall be relocated in backfill areas wherever possible. If encountered immediately adjacent to location of new construction, roots shall be cut approximately 6 inches back from new construction.
  - (d) Approved excavations shall be carefully backfilled with the excavated materials approved for backfilling. Backfill shall conform to adjacent grades without dips, sunken areas, humps, or other surface irregularities. Do not use mechanical equipment to compact backfill. Tamp carefully using hand tools, refilling and tamping until Final Acceptance as necessary to offset settlement.
  - (e) Exposed roots shall not be allowed to dry out before permanent backfill is placed. Temporary earth cover shall be provided, or roots shall be wrapped with four layers of wet, untreated burlap and temporarily supported and protected from damage until permanently relocated and covered with backfill.
  - (f) Accidentally broken roots should be sawed cleanly 3 inches behind ragged end.

#### 1.05 SECURITY AND PARKING:

- A. The Contractor shall be responsible for project security for materials, tools, equipment, supplies, and completed and partially completed Work.
- B. Contractor is responsible for any charges associated with parking of Contractor and Contractor employee's vehicles either at County facilities or off-site. Unless otherwise stated, all parking at County facilities requires payment. 10 parking spaces near the Highland Campus Wing B1 will be made available to the contractor and associated personnel for the duration of the project.

#### 1.06 TEMPORARY CONTROLS:

#### A. Noise Control

- (1) Contractor acknowledge, that adjacent facilities are in operation during all of the Work, and it shall take all reasonable precautions to minimize noise as required by applicable laws and the Contract Documents.
- (2) Notice of proposed noisy operations, including without limitation, operation of pneumatic demolition tools, concrete saws, and other equipment, shall be submitted to the County a minimum of forty-eight (48) hours in advance of their performance.

#### B. Erosion Control

- (1) Contractor shall prevent soil erosion on the Site and adjacent property resulting from its construction activities consistent with an approved Storm Water Pollution Prevention Plan ("SWPPP"). Effective measures shall be initiated prior to the commencement of clearing, grading, excavation, or other operations that will disturb the natural protection.
- (2) Work shall be scheduled to expose areas subject to erosion for the shortest possible time, and natural vegetation shall be preserved to the greatest extent practicable. Temporary storage and construction buildings shall be located, and construction traffic routed, to minimize erosion. Temporary fast-growing vegetation or other suitable ground cover shall be provided as necessary to control runoff.

### C. Traffic Control

(1) All traffic associated with the construction, including without limitation delivery and mail trucks, shall enter Contractor's access gate and road. Contractor shall provide signs directing construction and delivery traffic to this gate.

- (2) Contractor shall take all necessary steps to minimize inconvenience to the general public throughout all work under this Contract. No driveways or private roads shall be blocked without notifying the property owner, and access must be restored during all non-working hours. Safe access must be maintained for pedestrian traffic throughout any public work area at all times.
- (3) At least one lane of traffic in each direction on all roads used on the Project must be kept open at all times unless prior approval is provided by the County and any affected agency. No roads shall be blocked or made inaccessible, due to Contractor's work, without prior written approval of the County and the affected agencies in the form of an encroachment permit. Contractor shall not block or obstruct fire lanes at any time.
- (4) Traffic control shall be in accordance with the California Department of Transportation Traffic Manual. Contractor shall submit its traffic control plans to the appropriate agency for approval prior to work or public streets. Traffic control shall include signs, warning lights, reflectors, barriers, and other necessary safety devices and measures, including sufficient flaggers to direct vehicular traffic through the construction areas. No material or equipment shall be stored/parked where it will interfere with the free and safe passage of public traffic, and at the end of each day's work, and at other times when construction operations are suspended for any reason, Contractor shall remove all equipment and other obstructions from the public right-of-way.

### D. Noise and Vibration

- (1) Equipment and impact tools shall have intake and exhaust mufflers.
- (2) Contractor shall cooperate with County to minimize and/or seize the use of noisy and vibratory equipment if that equipment becomes objectionable by its longevity.

#### E. Dust and Dirt

- (1) Contractor shall conduct demolition and construction operations to minimize the generation of dust and dirt, and prevent dust and dirt from interfering with the progress of the Work and from accumulating in the Work and adjacent areas including, without limitation, occupied facilities.
- (2) Contractor shall periodically water exterior demolition and construction areas to minimize the generation of dust and dirt.
- (3) Contractor shall ensure that all hauling equipment and trucks carrying loads of soil and debris shall have their loads sprayed with water or

covered with tarpaulins, and as otherwise required by local and state ordinance.

(4) Contractor shall prevent dust and dirt from accumulating on walks, roadways, parking areas, and planting, and from washing into sewer and storm drain lines.

#### F. Water

Contractor shall not permit surface and subsurface water, and other liquids, to accumulate in or about the vicinity of the Premises. Should accumulation develop, Contractor shall control the water or other liquid, and suitably dispose of it by means of temporary pumps, piping, drainage lines, troughs, ditches, dams, or other methods.

#### G. Pollution

- (1) No burning of refuse, debris, or other materials shall be permitted on or in the vicinity of the Premises.
- (2) Contractor shall comply with applicable regulatory requirements and antipollution ordinances during the conduct of the Work including, without limitation, demolition, construction, and disposal operations.

## H. Lighting:

If portable lights are used after dark, all light must be located so as not to direct light into neighboring property.

#### 1.07 **JOB SIGN(S)**:

#### A. General:

- (1) If required, contractor shall provide and maintain a Project identification sign with the design, text, and colors designated by the County; locate sign as approved by the County.
- (2) Signs other than the specified Project sign and or signs required by law, for safety, or for egress, shall not be permitted, unless otherwise approved in advance by the County.

#### B. Materials:

(1) Structure and Framing: Structurally sound, new or used wood or metal; wood shall be nominal 3/4-inch exterior grade plywood.

- (2) Sign Surface: Minimum 3/4-inch exterior grade plywood.
- (3) Rough Hardware: Galvanized.
- (4) Paint: Exterior quality, of type and colors selected by the County...

#### C. Fabrication:

- (1) Contractor shall fabricate to provide smooth, even surface for painting.
- (2) Size: 4'-0" x 8'-0", unless otherwise indicated.
- (3) Contractor shall paint exposed surfaces of supports, framing, and surface material with exterior grade paint: one coat of primer and one coat of finish paint.
- (4) Text and Graphics: As indicated.

#### 1.08 CONSTRUCTION FACILITIES:

#### A. General:

- (1) The County shall provide the Contractor with a Field Office Space at Wing B1 on the Highland Hospital campus. Furniture, equipment, and maintenance shall be supplied by the Contractor.
- (2) Any additional Field Office Space required by the Contractor to execute the work shall be supplied by the Contractor, including all furniture, equipment, utilities, maintenance, janitorial services required to utilize the additional space.
- (3) Contractor shall properly remove the Office Space and contents from the Facility upon completion of the Contract, or as directed by the County in writing. Patch and repair affected areas; replace damaged items with new items. Carefully and properly inventory, clean, pack, store, and protect County property; submit County property to the County at a date, time and location as directed by the County.

#### B. Regulatory Requirements:

(1) In addition to any other regulatory compliance required by the Contract Documents, if Contractor provides Office Space in part or in whole by means of trailers, any such space shall display California Commercial Coach Insignia indicating that the entire complies with the California law including, but not necessarily limited to, the Construction and Fire Safety requirements of the California Code of Regulations.

#### 1.09 PUBLICITY RELEASES:

A. Contractor shall not release any information, story, photograph, plan, or drawing relating information about the Project to anyone, including press and other public communications medium, including, without limitation, on website(s).

END OF DOCUMENT

#### **DOCUMENT 01 62 00**

### PRODUCT OPTIONS AND SUBSTITUTIONS

#### PART 1 - GENERAL

#### 1.01 RELATED DOCUMENTS AND PROVISIONS

All Contract Documents should be reviewed for applicable provisions related to the provisions in this document, including without limitation:

- A. Instructions to Bidders;
- B. General Conditions, including, without limitation, Substitutions For Specified Items;
- C. Special Conditions.

# 1.02 SUBSTITUTIONS OF MATERIALS AND EQUIPMENT:

- A. The County will consider requests for substitution if received within thirty (30) Days after the Notice to Proceed. Requests received after that time may be considered or rejected at discretion of County.
- B. No substitutions that would require additional regulatory agency approval<sup>1</sup> shall be allowed except as may be deemed necessary by the County's Representative because:
  - (1) Previously specified or approved manufactured products are no longer manufactured.
  - (2) The substitution is required due to a County-initiated Change Order.
  - (3) The substitution is in the best interests of the County.
- C. The County will consider Contractor's request for substitution when the following conditions are satisfied. If the following conditions are not satisfied, County will return requests without action, except to record noncompliance with these requirements:
  - (1) Requested substitution offers County a substantial advantage in cost, time, energy conservation, or other benefits.

\_\_\_

- (2) Requested substitution is consistent with the Contract Documents and will produce required results.
- (3) Substitution request is fully documented and properly submitted.
- (4) If requested substitution involves more than one contractor, requested substitution has been coordinated with other portions of the Work, is uniform and consistent, is compatible with other products, and is acceptable to all contractors involved.
- (5) One or more of the following conditions must be satisfied:
  - a. The specified product or method of construction cannot be provided within the Contract Time. The request for substitution will not be considered if the product or method cannot be provided due to the Contractor's failure to pursue the Work promptly or coordinate activities properly.
  - b. The specified product or method of construction cannot receive necessary approval by a governing authority, and the requested substitution can be approved.
  - c. The specified product or method of construction cannot be provided in a manner that is compatible with other materials, and where the Contractor certifies that the substitution will overcome the incompatibility.
  - d. The specified product or method of construction cannot be coordinated with other materials, and where the Contractor certifies that the proposed substitution can be coordinated.
  - e. The specified product or method of construction cannot provide a warranty required by the Contract Documents and where the Contractor certifies that the proposed substitution can provide the required warranty.
  - f. The specified product or material has been discontinued or is no longer available.
- (6) The County's acceptance of any substitution shall not relieve the Contractor of the responsibility to comply with the requirements of the Drawings and Specifications.
- (7) The Contractor shall be responsible for all costs of any changes resulting from substitutions that affect other parts of the Work or the work of Separate Contractors.

- D. Catalog numbers and specific brands or trade names followed by the designation "or equal" are used in conjunction with material and equipment required by the Specifications to establish the standards of quality, utility, and appearance required. Substitutions which are equal in quality, utility, and appearance to those specified may be reviewed subject to the provisions of the General Conditions.
- E. Wherever more than one manufacturer's product is specified, the first-named product is the basis for the design used in the work and the use of alternative-named manufacturers' products or substitutes may require modifications in that design. If such alternatives are proposed by Contractor and are approved by the County and/or the Architect and/or Engineer, Contractor shall assume all costs required to make necessary revisions and modifications of the design resulting from the substitutions requested by the Contractor.
- F. When materials and equipment are specified by first manufacturer's name and product number, second manufacturer's name and "or approved equal," supporting data for the second product, if proposed by Contractor, shall be submitted in accordance with the requirements for substitutions.
- G. If the County and/or Architect/Engineer, in reviewing proposed substitute materials and equipment, require revisions or corrections to be made to previously accepted Shop Drawings and supplemental supporting data to be resubmitted, Contractor shall promptly do so. If any proposed substitution is judged by the County and/or Architect/Engineer to be unacceptable, the specified material or equipment shall be provided.
- H. Samples may be required. Tests required by the County and/or Architect/Engineer for the determination of quality and utility shall be made at the expense of Contractor, with acceptance of the test procedure first given by the County.
- I. In reviewing the supporting data submitted for substitutions, the County and/or Architect/Engineer will use for purposes of comparison all the characteristics of the specified material or equipment as they appear in the manufacturer's published data even though all the characteristics may not have been particularly mentioned in the Contract Documents. If more than two (2) submissions of supporting data are required, the cost of reviewing the additional supporting data shall be borne by Contractor, and the County will deduct the costs from the Contract Price.

END OF DOCUMENT

#### **DOCUMENT 01 65 00**

# **DELIVERY, STORAGE AND HANDLING**

#### **PART 1 - GENERAL**

#### 1.01 RELATED DOCUMENTS AND PROVISIONS

All Contract Documents should be reviewed for applicable provisions related to the provisions in this document, including without limitation:

- A. General Conditions, including, without limitation, Site Access, Conditions and Requirements;
- B. Special Conditions.

#### 1.02 PRODUCTS

- A. Products are as defined in the General Conditions.
- B. Contractor shall not use and/or reuse materials and/or equipment removed from existing Premises, except as specifically permitted by the Contract Documents.
- C. Contractor shall provide interchangeable components of the same manufacturer, for similar components.

#### 1.03 TRANSPORTATION AND HANDLING

- A. Contractor shall transport and handle Products in accordance with manufacturer's instructions.
- B. Contractor shall promptly inspect shipments to confirm that Products comply with requirements, quantities are correct, and products are undamaged.
- C. Contractor shall provide equipment and personnel to handle Products by methods to prevent soiling, disfigurement, or damage.

#### 1.04 STORAGE AND PROTECTION

- A. Contractor shall store and protect Products in accordance with manufacturer's instructions, with seals and labels intact and legible. Contractor shall store sensitive products in weather-tight, climate controlled enclosures.
- B. For exterior storage of fabricated Products, Contractor shall place on sloped supports, above ground.

- C. Contractor shall provide off-site storage and protection when Site does not permit on-site storage or protection.
- D. Contractor shall cover products subject to deterioration with impervious sheet covering and provide ventilation to avoid condensation.
- E. Contractor shall store loose granular materials on solid flat surfaces in a well-drained area and prevent mixing with foreign matter.
- F. Contractor shall provide equipment and personnel to store Products by methods to prevent soiling, disfigurement, or damage.
- G. Contractor shall arrange storage of Products to permit access for inspection and periodically inspect to assure Products are undamaged and are maintained under specified conditions.

END OF DOCUMENT

#### **DOCUMENT 01 71 23**

#### FIELD ENGINEERING

#### **PART 1 - GENERAL**

#### 1.01 RELATED DOCUMENTS AND PROVISIONS:

All Contract Documents should be reviewed for applicable provisions related to the provisions in this document, including without limitation:

- A. General Conditions, including, without limitation, Site Investigation, and Soils Investigation Report;
- B. Special Conditions;
- C. Site-Visit Certification.

# 1.02 REQUIREMENTS INCLUDED:

- A. Contractor shall provide and pay for field engineering services by a Californiaregistered engineer, required for the project, including, without limitations:
  - (1) Survey work required in execution of the Project.
  - (2) Civil or other professional engineering services specified, or required to execute Contractor's construction methods.

#### 1.03 QUALIFICATIONS OF SURVEYOR OR ENGINEERS:

A. Contractor shall only use a qualified licensed engineer or registered land surveyor, to whom County makes no objection.

#### 1.04 SURVEY REFERENCE POINTS:

- A. Existing basic horizontal and vertical control points for the Project are those designated on the Drawings.
- B. Contractor shall locate and protect control points prior to starting Site Work and preserve all permanent reference points during construction. In addition Contractor shall:
  - (1) Make no changes or relocation without prior written notice to County and Architect.

- (2) Report to County and Architect when any reference point is lost or destroyed, or requires relocation because of necessary changes in grades or locations.
- (3) Require surveyor to replace Project control points based on original survey control that may be lost or destroyed.

#### 1.05 RECORDS:

A. Contractor shall maintain a complete, accurate log of all control and survey work as it progresses.

#### 1.06 SUBMITTALS:

- A. Contractor shall submit name and address of Surveyor and Professional Engineer to County and Architect prior to its/their work on the Project.
- B. On request of County and Architect, Contractor shall submit documentation to verify accuracy of field engineering work, at no additional cost to the County.
- C. Contractor shall submit a certificate signed by registered engineer or surveyor certifying that elevations and locations of improvements are in conformance or nonconformance with Contract Documents.

#### **PART 2 – PRODUCTS**

Not Used.

#### **PART 3 - EXECUTION**

- 3.01 Contractor is responsible for meeting all applicable codes, OSHA, safety and shoring requirements.
- 3.02 Contractor is responsible for any re-surveying required by correction of nonconforming work.
- 3.03 Contractor shall request assistance from County two (2) Business Days prior to date assistance required.

#### END OF DOCUMENT

#### **DOCUMENT 01 73 29**

# **CUTTING AND PATCHING**

#### 1. PART 1 – GENERAL

#### 1.01 RELATED DOCUMENTS AND PROVISIONS:

All Contract Documents should be reviewed for applicable provisions related to the provisions in this document, including without limitation:

- A. General Conditions;
- B. Special Conditions;
- C. Hazardous Materials Procedures and Requirements;
- D. Hazardous Materials Certification;
- E. Imported Materials Certification.

#### 1.02 CUTTING AND PATCHING:

- A. Contractor shall be responsible for all cutting, fitting, and patching, including associated excavation and backfill, required to complete the Work or to:
  - (1) Make several parts fit together properly.
  - (2) Uncover portions of Work to provide for installation of ill-timed Work.
  - (3) Remove and replace defective Work.
  - (4) Remove and replace Work not conforming to requirements of Contract Documents.
  - (5) Remove Samples of installed Work as specified for testing.
  - (6) Provide routine penetrations of non-structural surfaces for installation of piping and electrical conduit.
  - (7) Attach new materials to existing remodeling areas, including painting (or other finishes) to match existing conditions.

- B. In addition to Contract requirements, upon written instructions from the County, Contractor shall uncover Work to provide for observations of covered Work in accordance with the Contract Documents; remove samples of installed materials for testing as directed by County; and remove Work to provide for alteration of existing Work.
- C. Contractor shall not cut or alter Work, or any part of it, in such a way that endangers or compromises the integrity of the Work, the Project, or work of others.

#### 1.03 SUBMITTALS:

- A. Prior to any cutting or alterations that may affect the safety of Project and Facility, or work of others, and well in advance of executing such cutting or alterations, Contractor shall submit written notice to County pursuant to the applicable notice provisions of the Contract Documents, requesting consent to proceed with the cutting or alteration, including the following:
  - (1) The Work of the County or other trades.
  - (2) Structural value or integrity of any element of Project.
  - (3) Integrity or effectiveness of weather-exposed or weather-resistant elements or systems.
  - (4) Efficiency, operational life, maintenance or safety of operational elements.
  - (5) Visual qualities of sight-exposed elements.
- B. Contractor's Request shall also include:
  - (1) Identification of Project.
  - (2) Description of affected Work.
  - (3) Necessity for cutting, alteration, or excavations.
  - (4) Effects of Work on County, other trades, or structural or weatherproof integrity of Project.
  - (5) Description of proposed Work:
    - (a) Scope of cutting, patching, alteration, or excavation.

- (b) Trades that will execute Work.
- (c) Products proposed to be used.
- (d) Extent of refinishing to be done.
- (6) Alternates to cutting and patching.
- (7) Cost proposal, when applicable.
- (8) The scheduled date the Contractor intends to perform the Work and the duration of time to complete the Work.
- (9) Written permission of other trades whose Work will be affected.

## 1.04 QUALITY ASSURANCE:

- A. Contractor shall ensure that cutting, fitting, and patching shall achieve security, strength, weather protection, appearance for aesthetic match, efficiency, operational life, maintenance, safety of operational elements, and the continuity of existing fire ratings.
- B. Contractor shall ensure that cutting, fitting, and patching shall successfully duplicate undisturbed adjacent profiles, materials, textures, finishes, colors, and that materials shall match existing construction. Where there is dispute as to whether duplication is successful or has been achieved to a reasonable degree, the County's decision shall be final.

#### 1.05 PAYMENT FOR COSTS:

- A. Cost caused by ill-timed or defective Work or Work not conforming to Contract Documents, including costs for additional services of the County, its consultants, including but not limited to the Construction Manager, the Architect, the Project Inspector(s), Engineers, and Agents, will be paid by Contractor and/or deducted from the Contract by the County.
- B. County shall only pay for cost of Work if it is part of the original Contract Price or if a change has been made to the contract in compliance with the provisions of the General Conditions. Cost of Work performed upon instructions from the County, other than defective or nonconforming Work, will be paid by County on approval of written Change Order. Contractor shall provide written cost proposals prior to proceeding with cutting and patching.

#### **PART 2 - PRODUCTS**

#### 2.01 MATERIALS:

- A. Contractor shall provide for replacement and restoration of Work removed. Contractor shall comply with the Contract Documents and with the Industry Standard(s), for the type of Work, and the Specification requirements for each specific product involved. If not specified, Contractor shall first recommend a product of a manufacturer or appropriate trade association for approval by the County.
- B. Materials to be cut and patched include those damaged by the performance of the Work.

#### **PART 3 – EXECUTION**

#### 3.01 INSPECTION:

- A. Contractor shall inspect existing conditions of the Site and the Work, including elements subject to movement or damage during cutting and patching, excavating and backfilling. After uncovering Work, Contractor shall inspect conditions affecting installation of new products.
- B. Contractor shall report unsatisfactory or questionable conditions in writing to County as indicated in the General Conditions and shall proceed with Work as indicated in the General Conditions by County.

#### 3.02 PREPARATION:

- A. Contractor shall provide shoring, bracing and supports as required to maintain structural integrity for all portions of the Project, including all requirements of the Project.
- B. Contractor shall provide devices and methods to protect other portions of Project from damage.
- C. Contractor shall, provide all necessary protection from weather and extremes of temperature and humidity for the Project, including without limitation, any work that may be exposed by cutting and patching Work. Contractor shall keep excavations free from water.

#### 3.03 ERECTION, INSTALLATION AND APPLICATION:

A. With respect to performance, Contractor shall:

- (1) Execute fitting and adjustment of products to provide finished installation to comply with and match specified tolerances and finishes.
- (2) Execute cutting and demolition by methods that will prevent damage to other Work, and provide proper surfaces to receive installation of repairs and new Work.
- (3) Execute cutting, demolition excavating, and backfilling by methods that will prevent damage to other Work and damage to settlement.
- B. Contractor shall employ original installer or fabricator to perform cutting and patching for:
  - (1) Weather-exposed surfaces and moisture-resistant elements such as roofing, sheet metal, sealants, waterproofing, and other trades.
  - (2) Sight-exposed finished surfaces.
- C. Contractor shall execute fitting and adjustment of products to provide a finished installation to comply with specified products, functions, tolerances, and finishes as shown or specified in the Contract Documents including, without limitation, the Drawings and Specifications.
- D. Contractor shall fit Work airtight to pipes, sleeves, ducts, conduit, and other penetrations through surfaces. Contractor shall conform to all Code requirements for penetrations or the Drawings and Specifications, whichever calls for a higher quality or more thorough requirement. Contractor shall maintain integrity of both rated and non-rated fire walls, ceilings, floors, etc.
- E. Contractor shall restore Work which has been cut or removed. Contractor shall install new products to provide completed Work in accordance with requirements of the Contract Documents and as required to match surrounding areas and surfaces.
- F. Contractor shall refinish all continuous surfaces to nearest intersection as necessary to match the existing finish to any new finish.

END OF DOCUMENT

# SECTION 01 74 19 CONSTRUCTION WASTE MANAGEMENT

#### PART 1 - GENERAL

#### 1.01 SUMMARY

- A. This section specifies the requirements for the diversion of demolition (non-hazardous) and construction debris from landfill and submittal of the Waste Management Plan.
- B. Performance Requirement: Divert a minimum of 65% of construction and demolition (non-hazardous) debris from landfill. Diversion of 75% of the construction & demolition debris is the project goal.

#### 1.03 DEFINITIONS

- A. "Conversion Rate" means the rate set forth in the standardized Conversion Rate Table approved by the County of Alameda for use in estimating the weight of materials identified in the Waste Management Plan and Waste Management Table.
- B. "Divert" means to use material for any purpose other than disposal in a landfill or transfer facility.
- C. "Good faith" shall be as defined by law.
- Provides processing of material and diversion from landfill.
- E. "Hauler" means the entity who transports construction and demolition debris to either a landfill or a recycling service.
- F. "Recycling Wizard" means an online database maintained by StopWaste.Org of approved recycling vendors that can be accessed at <a href="https://www.StopWaste.Org">www.StopWaste.Org</a> or by calling 1-877-STOPWASTE.
- G. "Waste Management Plan or Plan" means a waste management plan required under this contract that is used to track and report the means of disposal of all construction debris generated on this project.
- H. "WasteTracking.com powered by Green Halo Systems" means the web based system required for use in developing a waste management plan, uploading recycling data throughout the construction process and submitting the final report demonstrating the project compliance online. More information about WasteTracking.com powered by Green Halo Systems is available by visiting them online at <a href="https://www.wastetracking.com">www.wastetracking.com</a> or calling (888) 525-1301.

#### 1.04 QUALITY ASSURANCE

- A. The Contractor shall obtain all special permits and licenses and meet all special requirements for performance and completion of the work of this section.
- B. Regulatory requirements

- 1. Approval of the Waste Management Plan and Waste Management Table using the WasteTracking.com powered by Green Halo Systems by the Owner's Representative is required before beginning construction or demolition.
- C. Recycling service company qualifications Submit certification for recycling services listed in the approved Waste Management Plan that accepted waste will be diverted from landfill. Certification shall be demonstrated in one of the following ways:
  - 1. Recycling service is listed in the Green Halo System as an approved recycler for Alameda County projects; or
  - Recycling service is listed within Alameda County Waste Management Authority's Recycling Wizard, a Directory of where to recycle construction and demolition debris materials, or equivalent database/listing at StopWaste.Org. A database of construction and demolition recyclers can be found at <a href="https://www.StopWaste.Org">www.StopWaste.Org</a> or can be obtained by calling 1-877-STOPWASTE; or.
  - 3. If not listed as described in items 1 and 2 above, submit certification in writing from any recycling services that verifies accepted waste will be diverted from landfill.

#### 1.05 SUBMITTALS

- A. Submit specified Waste Management Plan to indicate how waste will be diverted from landfills. Plan to include procedures and schedule for debris disposal. Submittal shall be made using WasteTracking.com powered by Green Halo Systems using the following URL: <u>ACGSA.WasteTracking.com</u>. Submittal is required within 7 calendar days after receipt of Notice to Proceed; and
- B. Submit written documentation from recycling services that are not listed in the WasteTracking.com powered by Green Halo Systems or the Stopwaste.Org Recycling Wizard identifying where the construction and demolition material is taken, what method or process is being used to recycle the material, and identifying applicable state and local permits held by the recycling service provider and recycling facility; and
- C. Submit completed Waste Management Plan to report on the means of disposal of waste generated from project at the following project milestones:
  - 1. Fifty (50) percent progress payment; and
  - 2. One hundred (100) percent construction complete. County approval of final Waste Management Plan is required before full release of retention.

#### 1.06 WASTE MANAGEMENT PLAN

- A. Plan Development: Using the website <u>ACGSA.WasteTracking.com</u> develop a plan for diverting the specified percentage of construction debris from landfill. The plan shall include the following:
  - 1. Submit within 7-calendar days after receipt of Notice to Proceed.

- 2. Propose means and methods for collecting and separating each type of debris deemed reusable or recyclable.
- Identify the off-site recycling service and hauler of each designated debris item, who
  has agreed to accept and divert that item from landfill, in the proposed quantities
  anticipated. Schedule each item and list off-site recycling service and hauler company
  name, telephone number, address, and person contacted.
- 4. Include a "good faith" estimate of each type of construction waste that would be generated if no diversion methods were implemented. Submit with calculations based upon weight or volume of each. The following items are subject to the "good faith" estimate and diversion requirement:
  - a. Asphalt & Concrete
  - b. Brick/Masonry/Tiles
  - c. Building Materials (doors, windows, fixtures, etc.)
  - d. Cardboard and other paper products
  - e. Carpet/Carpet Padding/Foam
  - f. Ceiling Tiles (acoustic)
  - g. Drywall
  - h. Electrical Components (light fixtures, cables, etc)
  - i. Film Plastic & Styrofoam Blocks
  - j. Landscape Debris (Plant & tree trimmings)
  - k. Mechanical Debris (ducts, controls, plumbing fixtures, etc)
  - I. Scrap Metal
  - m. Unpainted Wood and Pallets
  - n. Other (painted wood & drywall, roofing, etc)
  - Mixed C&D (defined as a mixture of three or more materials from construction or demolition sites that will be taken to a "qualified" facility for recycling.)
  - p. Trash/garbage
- Construction waste quantities entered in volume will be converted automatically using the defined Conversion Rate approved by Alameda County that is integrated into WasteTracking.com powered by Green Halo Systems.

#### B. Plan Implementation

- 1. Contractor shall do all of the following:
  - Retain a copy of, and upload into WasteTracking.com powered by Green Halo Systems, all weight tickets, copies of receipts and invoices and any other documentation related to the reuse, recycling, and disposal of generated waste/debris from demolition and construction activities; and
  - Maintain a log of each load of each category item diverted from landfill. Log in separately debris sent to a Class III landfill and materials sent to recycling facilities.
    - Include in log: type of load, load weight, name of hauling service, name of recycling service or landfill, and date accepted by recycling service or by landfill.
    - 2) Owner reserves the right to audit the log at any time. Contractor shall retain and provide to the Owner all weight tickets, copies of receipts

and invoices and any other documentation related to the disposal or recycling of generated waste/debris from demolition and construction activities.

c. Units of measure: Use same units as stated in the approved plan "good faith" estimate of construction waste that would be generated if no remedial methods were implemented.

#### 2. Material handling

#### a. Separation facilities

- 1) Designate a specific on site area or areas to facilitate separation of materials for potential reuse, salvage, recycling, and return.
- Keep waste bins and pile areas neat and clean. Clearly mark bins for each category of waste. Do not co-mingle non-recyclable waste with materials designated for reuse or recycling.
- b. Environmental controls during handling, storage, or transport: Do not permit designated materials to become contaminated or to contaminate site or surrounding areas.

#### 3. Training and coordination

- a. Provide access and training as needed to subcontractors to the online construction and demolition debris waste management tool Green Halo Systems for the purposes of input of waste management information.
- b. Furnish copies of the Waste Management Plan to all on-site supervisors, each subcontractor, and the Owner.
- c. Instruction: Provide on-site instruction of appropriate separation, handling, and recycling, salvage, reuse, and return methods to be used by all entities at the appropriate stages of the Project.
- d. Meetings: Include construction waste management on the agenda of meetings. At a minimum, discuss waste management goals and issues at the following meetings:
  - 1) Pre-bid meetings.
  - 2) Pre-construction meeting.
  - 3) Regularly scheduled job-site meetings.

#### **PART 2 - PRODUCTS**

# 2.01 MATERIALS, EQUIPMENT AND FACILITIES

Furnish all materials, tools, equipment, devices, appurtenances, facilities, and services required for performing waste management of debris covered under this Section.

#### **PART 3 - EXECUTION**

#### 3.01 EXAMINATION AND PREPARATION

- A. Set up and maintain in good standing a project account with WasteTracking.com powered by Green Halo Systems using the website <a href="ACGSA.WasteTracking.com">ACGSA.WasteTracking.com</a> to be used exclusively for this project to develop a waste management plan, upload all reuse, recycling and waste disposal data throughout the construction process and submit the final online report demonstrating project compliance.
- B. Perform as required in the approved Waste Management Plan.

#### 3.02 DISPOSAL OF DEBRIS

- A. Dispose of waste, trash and debris in a safe, acceptable manner, in accordance with applicable laws and ordinances and as prescribed by authorities having jurisdiction. Burying of trash and debris on the site is prohibited.
- B. Remove demolished materials from site as work progresses. Remove debris from the site so that its presence will not delay the progress of the work.
- C. Debris shall be the property of the Contractor and shall be removed and disposed of in a legal manner off the County's property in accordance with the approved Waste Management Plan described herein. Location of recycling facility or dump and length of haul shall be the Contractor's responsibility.

#### **END OF SECTION**

### **DOCUMENT 01 76 00**

# PROTECTING INSTALLED CONSTRUCTION

### PART 1 – GENERAL

### 1.01 RELATED DOCUMENTS AND PROVISIONS:

All Contract Documents should be reviewed for applicable provisions related to the provisions in this document, including without limitation:

- A. General Conditions;
- B. Special Conditions.

### **PART 2 - PRODUCTS**

### 2.01 PRODUCTS FOR PATCHING AND EXTENDING WORK:

- A. New Materials: As specified in the Contract Documents including, without limitation, in the Specifications, Contractor shall match existing products, conditions, and work for patching and extending work.
- B. Type and Quality of Existing Products: Contractor shall determine by inspection, by testing products where necessary, by referring to existing conditions and to the Work as a standard.

### **PART 3 - EXECUTION**

# 3.01 EXAMINATION:

- A. Contractor shall verify that demolition is complete and that areas are ready for installation of new Work.
- B. By beginning restoration Work, Contract or acknowledges and accepts the existing conditions.

### 3.02 PREPARATION:

- A. Contractor shall cut, move, or remove items as necessary for access to alterations and renovation Work. Contractor shall replace and restore these at completion.
- B. Contractor shall remove unsuitable material not as salvage unless otherwise indicated in the Contract Documents. Unsuitable material may include, without

limitation, rotted wood, corroded metals, and deteriorated masonry and concrete. Contractor shall replace materials as specified for finished Work.

- C. Contractor shall remove debris and abandoned items from all areas of the Site and from concealed spaces.
- D. Contractor shall prepare surface and remove surface finishes to provide for proper installation of new Work and finishes.
- E. Contractor shall close openings in exterior surfaces to protect existing work from weather and extremes of temperature and humidity. Contractor shall insulate ductwork and piping to prevent condensation in exposed areas. Contractor shall insulate building cavities for thermal and/or acoustical protection, as detailed.

#### 3.03 INSTALLATION:

- A. Contractor shall coordinate Work of all alternations and renovations to expedite completion and to accommodate County occupancy.
- B. Designated Areas and Finishes: Contractor shall complete all installations in all respects, including operational, mechanical work and electrical work.
- C. Contractor shall remove, cut, and patch Work in a manner to minimize damage and to provide a means of restoring Products and finishes to original or specified condition.
- D. Contractor shall refinish visible existing surfaces to remain in renovated rooms and spaces, to specified condition for each material, with a neat transition to adjacent finishes.
- E. Contractor shall install products as specified in the Contract Documents, including without limitation, the Specifications.

## 3.04 TRANSITIONS:

- A. Where new Work abuts or aligns with existing, Contractor shall perform a smooth and even transition. Patched Work must match existing adjacent work in texture and appearance.
- B. When finished surfaces are cut so that a smooth transition with new Work is not possible, Contractor shall terminate existing surface along a straight line at a natural line of division and make a recommendation for resolution to the County and the Architect and/or Engineer for review and approval.

### 3.05 ADJUSTMENTS:

- A. Where removal of partitions or walls results in adjacent spaces becoming one, Contractor shall rework floors, walls, and ceilings to a smooth plane without breaks, steps, or bulkheads.
- B. Where a change of plane of 1/4 inch or more occurs, Contractor shall submit a recommendation for providing a smooth transition to the County and the Architect and/or Engineer for review and approval.
- C. Contractor shall trim existing doors as necessary to clear new floor finish and refinish trim as required.
- D. Contractor shall fit Work at penetrations of surfaces.

### 3.06 REPAIR OF DAMAGED SURFACES:

- A. Contractor shall patch or replace portions of existing surfaces which are damaged, lifted, discolored, or showing other imperfections.
- B. Contractor shall repair substrate prior to patching finish.

### 3.07 CULTIVATED AREAS AND OTHER SURFACE IMPROVEMENTS:

- A. Cultivated or planted areas and other surface improvements which are damaged by actions of the Contractor shall be restored by Contractor to their original condition or better, where indicated.
- B. Contractor shall protect and replace, if damaged, all existing guard posts, barricades, and fences.
- C. Contractor shall give special attention to avoid damaging or killing trees, bushes and/or shrubs on the Premises and/or identified the Contract Documents, including without limitation, the Drawings.

### **3.08 FINISHES:**

- A. Contractor shall finish surfaces as specified in the Contract Documents, including without limitations, the provisions of all Divisions of the Specifications.
- B. Contractor shall finish patches to produce uniform finish and texture over entire area. When finish cannot be matched, Contractor shall refinish entire surface to nearest intersections.

### 3.09 CLEANING:

# **Alameda County, General Services Agency**

**PROJECT # 2043** 

Highland Hospital – Koret Building Exhaust Fan and Waterproofing

A. Contractor shall continually clean the Site and the Premises as indicated in the Contract Documents, including without limitation, the provisions in the General Conditions and the Specifications regarding cleaning.

END OF DOCUMENT

### **SECTION 01 77 00**

# **CONTRACT CLOSEOUT**

### **PART 1 - GENERAL**

### 1.01 RELATED DOCUMENTS AND PROVISIONS

All Contract Documents should be reviewed for applicable provisions related to the provisions in this document, including without limitation:

- A. General Conditions, including, without limitation, Completion of Work;
- B. Special Conditions;
- C. Temporary Facilities and Controls.

### 1.02 CLOSEOUT PROCEDURES

Contractor shall comply with all closeout provisions as indicated in the General Conditions.

### 1.03 COMPLETION

A. Completion is defined in Section 014216 (Definitions and Reference Standards). Contractor must meet all requirements of Completion as defined in Section 014216 in order to achieve Completion.

# 1.04 REMOVAL OF TEMPORARY CONSTRUCTION FACILITIES

- A. Remove temporary materials, equipment, services, and construction prior to Final inspection. Temporary construction facilities to be removed upon Completion.
- B. Clean and repair damage caused by installation or use of temporary facilities.
- C. Restore permanent facilities used during construction to specified condition.

### 1.05 FINAL CLEANING

- A. Contractor shall execute final cleaning prior to final inspection.
- B. Contractor shall clean interior and exterior glass and surfaces exposed to view; remove temporary labels, tape, stains, and foreign substances, polish transparent and glossy surfaces, wax and polish new vinyl floor surfaces, vacuum carpeted and soft surfaces.
- C. Contractor shall clean equipment and fixtures to a sanitary condition.

- D. Contractor shall replace filters of operating equipment.
- E. Contractor shall clean debris from roofs, gutters, down spouts, and drainage systems.
- F. Contractor shall clean Site, sweep paved areas, and rake clean landscaped surfaces.
- G. Contractor shall remove waste and surplus materials, rubbish, and construction facilities from the Site.

### 1.06 ADJUSTING

- A. Contractor shall adjust operating products and equipment to ensure smooth and unhindered operation.
- B. Record Documents and Shop Drawings: Contractor shall legibly mark each item to record actual construction, including:
  - (1) Measured depths of foundations in relation to finish floor datum.
  - (2) Measured horizontal and vertical locations of underground utilities and appurtenances, referenced to permanent surface improvements.
  - (3) Measured locations of internal utilities and appurtenances concealed in construction, referenced to visible and accessible features of the Work.
  - (4) Field changes of dimension and detail.
  - (5) Details not on original Contract Drawings.
  - (6) Changes made by modification(s).
  - (7) References to related Shop Drawings and modifications.
- C. County will provide one set of reproducible drawings to Contractor.
- D. Contractor shall submit all required documents to County and/or Architect prior to or with its final Application for Payment.

### **1.07** INSTRUCTION OF COUNTY PERSONNEL:

A. Before final inspection, at agreed upon times, Contractor shall instruct County's designated personnel in operation, adjustment, and maintenance of products, equipment, and systems.

- B. For equipment requiring seasonal operation, Contractor shall perform instructions for other seasons within six months.
- C. Contractor shall use operation and maintenance manuals as basis for instruction. Contractor shall review contents of manual with personnel in detail to explain all aspects of operation and maintenance.
- D. Contractor shall prepare and insert additional data in Operation and Maintenance Manual when need for such data becomes apparent during instruction.
- E. Contractor shall use operation and maintenance manuals as basis for instruction. Contractor shall review contents of manual with personnel in detail to explain all aspects of operation and maintenance.

### **1.08** SPARE PARTS AND MAINTENANCE MATERIALS:

- A. Contractor shall provide products, spare parts, maintenance, and extra materials in quantities specified in the Specifications and in Manufacturer's recommendations.
- B. Contractor shall provide County all required Operation and Maintenance Data.

#### **1.09** TURN-IN

A. The Contract Documents will not be closed out and final payment will not be made until all personnel Identification Media, vehicle permits and keys issued to Contractor during prosecution of Work are turned in to County.

### 1.10 LABOR COMPLIANCE, ECOP, SLEB

### A. LABOR COMPLIANCE

- (1) The Contract will not be closed out and final payment will not be made until all Labor Compliance documents are received, weekly payrolls have been certified and cleared of violations, and penalties are paid, if applicable. The following Labor Compliance documents must be received and approved by Contractor and County from Contractor's subcontractors:
  - a. Apprentice Certification (Federal and State)
  - b. Apprentice Wage Rate and Ratio
  - c. DAS 140
  - d. DAS 142

# **Alameda County, General Services Agency**

**PROJECT # 2043** 

Highland Hospital – Koret Building Exhaust Fan and Waterproofing

- e. Fringe Benefit Statement
- f. Fringe Benefits Proof of Payment
- g. 1-9 Employment Eligibility Verification
- h. CAC-2
- i. Checklist of Labor Law Requirements
- (2) ECOP, SLEB
- (3) The Contract will not be closed out and final payment will not be made until all ECOP, SLEB payments are made and confirmed in the Elation system, as well as all change orders approved and paid to the satisfaction of Contractor and County from Contractor's subcontractors. Additionally, Contractor and its subcontractors are required to meet and/or exceed the County's and/or their committed SLEB/ECOP goal or provide documentation to support the lack thereof to the satisfaction of Contractor and County.

PART 2 – PRODUCTS Not used.

**PART 3 – EXECUTION** Not used.

**END OF SECTION** 

### **DOCUMENT 01 78 23**

# **OPERATION AND MAINTENANCE INSTRUCTIONS**

### PART 1 – GENERAL

### 1.01 RELATED DOCUMENTS AND PROVISIONS:

All Contract Documents should be reviewed for applicable provisions related to the provisions in this document, including without limitation:

- A. General Conditions, including, without limitation, Completion of the Work;
- B. Special Conditions.

# 1.02 QUALITY ASSURANCE:

A. Contractor shall prepare instructions and data by personnel experienced in maintenance and operation of described products.

#### **1.03 FORMAT**:

- A. Contractor shall prepare data in the form of an instructional manual entitled "OPERATIONS AND MAINTENANCE MANUAL & INSTRUCTIONS" ("Manual").
- B. Binders: Contractor shall use commercial quality, 8-1/2 by 11 inch, three-side rings, with durable plastic covers; two inch maximum ring size. When multiple binders are used, Contractor shall correlate data into related consistent groupings.
- C. Cover: Contractor shall identify each binder with typed or printed title "OPERATION AND MAINTENANCE MANUAL & INSTRUCTIONS"; and shall list title of Project and identify subject matter of contents.
- D. Contractor shall arrange content by systems process flow under section numbers and sequence of Table of Contents of the Contract Documents.
- E. Contractor shall provide tabbed fly leaf for each separate product and system, with typed description of product and major component parts of equipment.
- F. Text: The content shall include Manufacturer's printed data, or typewritten data on 24 pound paper.
- G. Drawings: Contractor shall provide with reinforced punched binder tab and shall bind in with text; folding larger drawings to size of text pages.

### 1.04 CONTENTS, EACH VOLUME:

- A. Table of Contents: Contractor shall provide title of Project; names, addresses, and telephone numbers of the Architect, any engineers, subconsultants, Subcontractor(s), and Contractor with name of responsible parties; and schedule of products and systems, indexed to content of the volume.
- B. For Each Product or System: Contractor shall list names, addresses, and telephone numbers of Subcontractor(s) and suppliers, including local source of supplies and replacement parts.
- C. Product Data: Contractor shall mark each sheet to clearly identify specific products and component parts, and data applicable to installation. Delete inapplicable information.
- D. Drawings: Contractor shall supplement product data to illustrate relations of component parts of equipment and systems, to show control and flow diagrams. Contractor shall not use Project Record Documents as maintenance drawings.
- E. Text: The Contractor shall include any and all information as required to supplement product data. Contractor shall provide logical sequence of instructions for each procedure, incorporating manufacturer's instructions.
- F. Warranties and Bonds: Contractor shall bind in one copy of each.

### 1.05 MANUAL FOR MATERIALS AND FINISHES:

- A. Building Products, Applied Materials, and Finishes: Contractor shall include product data, with catalog number, size, composition, and color and texture designations. Contractor shall provide information for re-ordering custom manufactured products.
- B. Instructions for Care and Maintenance: Contractor shall include Manufacturer's recommendations for cleaning agents and methods, precautions against detrimental agents and methods, and recommended schedule for cleaning and maintenance.
- C. Moisture Protection and Weather Exposed Products: Contractor shall include product data listing applicable reference standards, chemical composition, and details of installation. Contractor shall provide recommendations for inspections, maintenance, and repair.
- D. Additional Requirements: Contractor shall include all additional requirements as specified in the Specifications.

E. Contractor shall provide a listing in Table of Contents for design data, with tabbed fly sheet and space for insertion of data.

### 1.06 MANUAL FOR EQUIPMENT AND SYSTEMS:

- A. Each Item of Equipment and Each System: Contractor shall include description of unit or system, and component parts and identify function, normal operating characteristics, and limiting conditions. Contractor shall include performance curves, with engineering data and tests, and complete nomenclature, and commercial number of replaceable parts.
- B. Panelboard Circuit Directories: Contractor shall provide electrical service characteristics, controls, and communications.
- C. Contractor shall include color coded wiring diagrams as installed.
- D. Operating Procedures: Contractor shall include start-up, break-in, and routine normal operating instructions and sequences. Contractor shall include regulation, control, stopping, shut-down, and emergency instructions. Contractor shall include summer, winter, and any special operating instructions.
- E. Maintenance Requirements: Contractor shall include routine procedures and guide for trouble-shooting; disassembly, repair, and reassembly instructions; and alignment, adjusting, balancing, and checking instructions.
- F. Contractor shall provide servicing and lubrication schedule, and list of lubricants required.
- G. Contractor shall include manufacturer's printed operation and maintenance instructions.
- H. Contractor shall include sequence of operation by controls manufacturer.
- I. Contractor shall provide original manufacturer's parts list, illustrations, assembly drawings, and diagrams required for maintenance.
- J. Contractor shall provide control diagrams by controls manufacturer as installed.
- K. Contractor shall provide Contractor's coordination drawings, with color coded piping diagrams as installed.
- L. Contractor shall provide charts of valve tag numbers, with location and function of each valve, keyed to flow and control diagrams.
- M. Contractor shall provide list of original manufacturer's spare parts, current prices, and recommended quantities to be maintained in storage.

- N. Additional Requirements: Contractor shall include all additional requirements as specified in Specification(s).
- O. Contractor shall provide a listing in Table of Contents for design data, with tabbed fly sheet and space for insertion of data.

### 1.08 SUBMITTAL:

- A. Contractor shall submit to the County for review electronic copy of preliminary draft or proposed formats and outlines of the contents of the Manual within thirty (30) days of Contractor's start of Work.
- B. For equipment, or component parts of equipment put into service during construction and to be operated by County, Contractor shall submit draft content for that portion of the Manual within ten (10) days after acceptance of that equipment or component.
- C. Contractor shall submit electronic copy of a complete Manual in final form prior to final Application for Payment. Copy will be returned with County and/or Architect/Engineer comments. Contractor must revise the content of the Manual as required by County prior to County's approval of Contractor's final Application for Payment.
- D. Contractor must submit two (2) hard copies of revised Manual in final form within ten (10) days after final inspection.

END OF DOCUMENT

#### **DOCUMENT 01 78 36**

### **WARRANTIES**

### PART 1 - GENERAL

### 1.01 RELATED DOCUMENTS AND PROVISIONS

All Contract Documents should be reviewed for applicable provisions related to the provisions in this document, including without limitation:

- A. General Conditions, including, without limitation, Warranty/Guarantee Information;
- B. Special Conditions.

#### 1.02 FORMAT

- A. Binders: Contractor shall use commercial quality, 8-1/2 by 11 inch, three-side rings, with durable plastic covers; two inch maximum ring size.
- B. Cover: Contractor shall identify each binder with typed or printed title "WARRANTIES" and shall list title of Project.
- C. Table of Contents: Contractor shall provide title of Project; name, address, and telephone number of Contractor and equipment supplier, and name of responsible principal. Contractor shall identify each item with the number and title of the specific Specification, document, provision, or section in which the name of the product or work item is specified.
- D. Contractor shall separate each warranty with index tab sheets keyed to the Table of Contents listing, providing full information and using separate typed sheets as necessary. Contractor shall list each applicable and/or responsible subcontractor(s), supplier(s), and/or manufacturer(s), with name, address, and telephone number of each responsible principal(s).

#### 1.03 PREPARATION:

- A. Contractor shall obtain warranties, executed in duplicate by each applicable and/or responsible subcontractor(s), supplier(s), and manufacturer(s), within ten (10) days after completion of the applicable item or work. Except for items put into use with County's permission, Contractor shall leave date of beginning of time of warranty until the date of completion is determined.
- B. Contractor shall verify that documents are in proper form, contain full information, and are notarized, when required.

- C. Contractor shall co-execute submittals when required.
- D. Contractor shall retain warranties until time specified for submittal.

### **1.04** TIME OF SUBMITTALS:

- A. For equipment or component parts of equipment put into service during construction with County's permission, Contractor shall submit a draft warranty for that equipment or component within ten (10) days after acceptance of that equipment or component.
- B. Contractor shall submit for County approval all warranties and related documents within ten (10) days after date of completion. Contractor must revise the warranties as required by the County prior to County's approval of Contractor's final Application for Payment.
- C. For items of work delayed beyond date of completion, provide updated submittal within ten days after acceptance, listing the date of acceptance as start of warranty period.

END OF DOCUMENT

### **DOCUMENT 01 78 39**

# PROJECT RECORD DOCUMENTS

#### PART 1 - GENERAL

### 1.01 RELATED DOCUMENTS AND PROVISIONS:

All Contract Documents should be reviewed for applicable provisions related to the provisions in this document, including without limitation:

- A. General Conditions, including, without limitation, Documents on Work;
- B. Special Conditions.

### **PART 2 - RECORD DRAWINGS**

### **2.01 GENERAL:**

- A. The County will provide Contractor with one electronic set (Autocad) of the original Contract Drawings.
- B. Contractor shall maintain at each Project Site one set of marked-up prints and each month, or as otherwise agreed, shall transfer all changes and information to those marked-up prints. Contractor shall submit to the Project County one electronic set of the Project Record Drawings ("As-Builts") showing all changes incorporated into the Work since the preceding monthly submittal. The Record Drawings shall be available at the Project Site. The Contractor shall submit the electronic set at the conclusion of the Project following review of the prints.
- C. Label and date each Record Drawing "RECORD DOCUMENT" in legibly printed letters.
- D. All deviations in construction, including but not limited to pipe and conduit locations and deviations caused by without limitation Change Orders, Construction Claim Directives, RFI's, and Addenda, shall be accurately and legibly recorded by Contractor.
- E. Locations and changes shall be done by Contractor in a neat and legible manner and, where applicable, indicated by drawing a "cloud" around the changed or additional information.

## 2.02 RECORD DRAWING INFORMATION:

A. Contractor shall record the following information:

- (1) Locations of Work buried under or outside each building, including, without limitation, all utilities, plumbing and electrical lines, and conduits.
- (2) Actual numbering of each electrical circuit.
- (3) Locations of significant Work concealed inside each building whose general locations are changed from those shown on the Contract Drawings.
- (4) Locations of all items, not necessarily concealed, which vary from the Contract Documents.
- (5) Installed location of all cathodic protection anodes.
- (6) Deviations from the sizes, locations, and other features of installations shown in the Contract Documents.
- (7) Locations of underground work, points of connection with existing utilities, changes in direction, valves, manholes, catch basins, capped stubouts, invert elevations, etc.
- (8) Sufficient information to locate Work concealed in each building with reasonable ease and accuracy. In some instances, this may be by dimension, in others, it may be in relation to the spaces in the building near which it was installed.
- B. Contractor shall provide additional drawings as necessary for clarification.
- C. Contractor shall provide reproducible record drawings, made from final Shop Drawings marked "No Exceptions Taken" or "Approved as Noted."

## **PART 3 - RECORD SPECIFICATIONS**

## 3.01 GENERAL:

- A. Contractor shall mark each section legibly to record manufacturer, trade name, catalog number, and supplier of each Product and item of equipment actually installed.
- B. Contractor shall submit an electronic file of the record specifications. Format shall be Microsoft Word.

### PART 4 - MAINTENANCE OF RECORD DOCUMENTS

# 4.01 GENERAL

- A. Contractor shall store Record Documents apart from documents used for construction:
  - (1) Provide files and racks for storage of Record Documents.
  - (2) Maintain Record Documents in a clean, dry, legible condition and in good order
- B. Do not use Record Documents for construction purposes.

END OF DOCUMENT

### DOCUMENT 01 91 13

# GENERAL COMMISSIONING REQUIREMENTS

### PART 1 - GENERAL

### 1.01 RELATED DOCUMENTS

- A. Drawings and general provisions of the Contract, including General and Supplementary Conditions and other Division 01 Specification Sections, apply to this Section.
- B. OPR and BoD documentation prepared by Owner and Architect contains requirements that apply to this Section.

#### 1.02 SUMMARY

A. This Section includes general requirements that apply to implementation of commissioning without regard to systems, subsystems, and equipment being commissioned.

### 1.2 DEFINITIONS

- A. BoD: Basis of Design.
- B. CxA: Commissioning Authority.
- C. OPR: Owner's Project Requirements.
- D. Systems, Subsystems, and Equipment: Where these terms are used together or separately, they shall mean "as-built" systems, subsystems, and equipment.
- E. TAB: Testing, Adjusting, and Balancing.

### 1.3 COMMISSIONING TEAM

- A. Members Appointed by Contractor(s): Individuals, each having authority to act on behalf of the entity he or she represents, explicitly organized to implement the commissioning process through coordinated actions. The commissioning team shall consist of, but not be limited to, representatives of the Contractor, including Project superintendent and subcontractors, installers, suppliers, and specialists deemed appropriate by the CxA.
- B. Members Appointed by Owner:

- 1. CxA: The designated person, company, or entity that plans, schedules, and coordinates the commissioning team to implement the commissioning process. Owner will engage the CxA under a separate contract.
- 2. Representatives of the facility user and operation and maintenance personnel.
- 3. Architect and engineering design professionals.

#### 1.4 OWNER'S RESPONSIBILITIES

- A. Provide the OPR documentation to the CxA and Contractor for use in developing the commissioning plan; systems manual; operation and maintenance training plan; and testing plans and checklists.
- B. Assign operation and maintenance personnel and schedule them to participate in commissioning team activities including, but not limited to, the following:
  - 1. Coordination meetings.
  - 2. Training in operation and maintenance of systems, subsystems, and equipment.
  - 3. Testing meetings.
  - 4. Demonstration of operation of systems, subsystems, and equipment.
- C. Provide utility services required for the commissioning process.
- D. Provide the BoD documents, prepared by Architect and approved by Owner, to the CxA and Contractor for use in developing the commissioning plan, systems manual, and operation and maintenance training plan.

#### 1.5 CONTRACTOR'S RESPONSIBILITIES

- A. Provide utility services required for the commissioning process.
- B. Contractor shall assign representatives with expertise and authority to act on behalf of the Contractor and schedule them to participate in and perform commissioning team activities including, but not limited to, the following:
  - 1. Participate in construction-phase coordination meetings.
  - 2. Participate in maintenance orientation and inspection.
  - 3. Participate in operation and maintenance training sessions.
  - 4. Participate in final review at acceptance meeting.
  - 5. Certify that Work is complete and systems are operational according to the Contract Documents, including calibration of instrumentation and controls.
  - 6. Evaluate performance deficiencies identified in test reports and, in collaboration with entity responsible for system and equipment installation, recommend corrective action.
  - 7. Review and approve final commissioning documentation.

- C. Subcontractors shall assign representatives with expertise and authority to act on behalf of subcontractors and schedule them to participate in and perform commissioning team activities including, but not limited to, the following:
  - 1. Participate in construction-phase coordination meetings.
  - 2. Participate in maintenance orientation and inspection.
  - 3. Participate in procedures meeting for testing.
  - 4. Participate in final review at acceptance meeting.
  - 5. Provide schedule for operation and maintenance data submittals, equipment startup, and testing to CxA for incorporation into the commissioning plan. Update schedule on a weekly basis throughout the construction period.
  - 6. Provide information to the CxA for developing construction-phase commissioning plan.
  - 7. Participate in training sessions for Owner's operation and maintenance personnel.
  - 8. Provide updated Project Record Documents to the CxA on a daily basis.
  - 9. Gather and submit operation and maintenance data for systems, subsystems, and equipment to the CxA, as specified in Division 1 Section "Operation and Maintenance Data."
  - 10. Provide technicians who are familiar with the construction and operation of installed systems and who shall develop specific test procedures and participate in testing of installed systems, subsystems, and equipment.

### 1.6 CxA'S RESPONSIBILITIES

- A. Organize and lead the commissioning team.
- B. Prepare a construction-phase commissioning plan. Collaborate with Contractor and with subcontractors to develop test and inspection procedures. Include design changes and scheduled commissioning activities coordinated with overall Project schedule. Identify commissioning team member responsibilities, by name, firm, and trade specialty, for performance of each commissioning task.
- C. Review and comment on submittals from Contractor for compliance with the OPR, BoD, Contract Documents, and construction-phase commissioning plan. Review and comment on performance expectations of systems and equipment and interfaces between systems relating to the OPR and BoD.
- D. Convene commissioning team meetings for the purpose of coordination, communication, and conflict resolution; discuss progress of the commissioning processes. Responsibilities include arranging for facilities, preparing agenda and attendance lists, and notifying participants. The CxA shall prepare and distribute minutes to commissioning team members and attendees within five (5) workdays of the commissioning meeting.
- E. At the beginning of the construction phase, conduct an initial construction-phase coordination meeting for the purpose of reviewing the commissioning activities and

- establishing tentative schedules for operation and maintenance submittals; operation and maintenance training sessions; TAB Work; and Project completion.
- F. Observe and inspect construction and report progress and deficiencies. In addition to compliance with the OPR, BoD, and Contract Documents, inspect systems and equipment installation for adequate accessibility for maintenance and component replacement or repair.
- G. Prepare Project-specific test and inspection procedures and checklists.
- H. Schedule, direct, witness, and document tests, inspections, and systems startup.
- I. Compile test data, inspection reports, and certificates and include them in the systems manual and commissioning report.
- J. Certify date of acceptance and startup for each item of equipment for start of warranty periods.
- K. Review Project Record Documents for accuracy. Request revisions from Contractor to achieve accuracy. Project Record Documents requirements are specified in Division 1 Section "Project Record Documents."
- Review and comment on operation and maintenance documentation and systems manual outline for compliance with the OPR, BoD, and Contract Documents.
   Operation and maintenance documentation requirements are specified in Division 1 Section "Operation and Maintenance Data."
- M. Prepare operation and maintenance training program and provide qualified instructors to conduct operation and maintenance training. Operation and maintenance training is specified in Division 1 Section "Demonstration and Training."
- N. Videotape and edit training sessions.
- O. Prepare commissioning reports.
- P. Assemble the final commissioning documentation, including the commissioning report and Project Record Documents.

### 1.7 COMMISSIONING DOCUMENTATION

- A. Index of Commissioning Documents: CxA shall prepare an index to include storage location of each document.
- B. OPR: A written document, prepared by Owner, which details the functional requirements of Project and expectations of how it will be used and operated. This document includes Project and design goals, measurable performance criteria, budgets, schedules, success criteria, and supporting information.

- C. BoD Document: A document, prepared by Architect, that records concepts, calculations, decisions, and product selections used to meet the OPR and to satisfy applicable regulatory requirements, standards, and guidelines. The document includes both narrative descriptions and lists of individual items that support the design process.
- D. Commissioning Plan: A document, prepared by CxA, that outlines the schedule, allocation of resources, and documentation requirements of the commissioning process, and shall include, but is not limited to the following:
  - 1. Plan for delivery and review of submittals, systems manuals, and other documents and reports. Identification of the relationship of these documents to other functions and a detailed description of submittals that are required to support the commissioning processes. Submittal dates shall include the latest date approved submittals must be received without adversely affecting commissioning plan.
  - 2. Description of the organization, layout, and content of commissioning documentation (including systems manual) and a detailed description of documents to be provided along with identification of responsible parties.
  - 3. Identification of systems and equipment to be commissioned.
  - 4. Description of schedules for testing procedures along with identification of parties involved in performing and verifying tests.
  - 5. Identification of items that must be completed before the next operation can proceed.
  - 6. Description of responsibilities of commissioning team members.
  - 7. Description of observations to be made.
  - 8. Description of requirements for operation and maintenance training, including required training materials.
  - 9. Description of expected performance for systems, subsystems, equipment, and controls.
  - 10. Schedule for commissioning activities with specific dates coordinated with overall construction schedule.
  - 11. Identification of installed systems, subsystems, and equipment, including design changes that occurred during the construction phase.
  - 12. Process and schedule for documenting changes on a continuous basis to appear in Project Record Documents.
  - 13. Process and schedule for completing prestart and startup checklists for systems, subsystems, and equipment to be verified and tested.
  - 14. Step-by-step procedures for testing systems, subsystems, and equipment with descriptions for methods of verifying relevant data, recording the results obtained, and listing parties involved in performing and verifying tests.
- E. Test Checklists: CxA shall develop test checklists for each system, subsystem, or equipment including interfaces and interlocks, and include a separate entry, with space for comments, for each item to be tested. Prepare separate checklists for each mode of operation and provide space to indicate whether the mode under test responded as required. Provide space for testing personnel to sign off on each checklist. Each checklist, regardless of system, subsystem, or equipment being tested, shall include, but not be limited to, the following:

- 1. Name and identification code of tested item.
- 2. Test number.
- 3. Time and date of test.
- 4. Indication of whether the record is for a first test or retest following correction of a problem or issue.
- 5. Dated signatures of the person performing test and of the witness, if applicable.
- 6. Individuals present for test.
- 7. Deficiencies.
- 8. Issue number, if any, generated as the result of test.
- F. Certificate of Readiness: Certificate of Readiness shall be signed by the Contractor, Subcontractor(s), Installer(s), and CxA certifying that systems, subsystems, equipment, and associated controls are ready for testing. Completed test checklists signed by the responsible parties shall accompany this certificate.
- G. Test and Inspection Reports: CxA shall record test data, observations, and measurements on test checklists. Photographs, forms, and other means appropriate for the application shall be included with data. CxA shall compile test and inspection reports and test and inspection certificates and include them in systems manual and commissioning report.
- H. Corrective Action Documents: CxA shall document corrective action taken for systems and equipment that fail tests. Include required modifications to systems and equipment and revisions to test procedures, if any. Retest systems and equipment requiring corrective action and document retest results.
- I. Issues Log: CxA shall prepare and maintain an issues log that describes design, installation, and performance issues that are at variance with the OPR, BoD, and Contract Documents. Identify and track issues as they are encountered, documenting the status of unresolved and resolved issues.
  - 1. Creating an Issues Log Entry:
    - a. Identify the issue with unique numeric or alphanumeric identifier by which the issue may be tracked.
    - b. Assign a descriptive title of the issue.
    - c. Identify date and time of the issue.
    - d. Identify test number of test being performed at the time of the observation, if applicable, for cross-reference.
    - e. Identify system, subsystem, and equipment to which the issue applies.
    - f. Identify location of system, subsystem, and equipment.
    - g. Include information that may be helpful in diagnosing or evaluating the issue.
    - h. Note recommended corrective action.
    - i. Identify commissioning team member responsible for corrective action.
    - j. Identify expected date of correction.
    - k. Identify person documenting the issue.

### **Alameda County, General Services Agency**

Highland Hospital – Koret Building Exhaust Fan and Waterproofing

- 2. Documenting Issue Resolution:
  - a. Log date correction is completed or the issue is resolved.
  - b. Describe corrective action or resolution taken. Include description of diagnostic steps taken to determine root cause of the issue, if any.
  - c. Identify changes to the OPR, BoD, or Contract Documents that may require action.
  - d. State that correction was completed and system, subsystem, and equipment is ready for retest, if applicable.
  - e. Identify person(s) who corrected or resolved the issue.
  - f. Identify person(s) documenting the issue resolution.
- 3. Issues Log Report: On a periodic basis, but not less than for each commissioning team meeting, CxA shall prepare a written narrative for review of outstanding issues and a status update of the issues log. As a minimum, CxA shall include the following information in the issues log and expand it in the narrative:
  - a. Issue number and title.
  - b. Date of the identification of the issue.
  - c. Name of the commissioning team member assigned responsibility for resolution.
  - d. Expected date of correction.
- J. Commissioning Report: CxA shall document results of the commissioning process including unresolved issues and performance of systems, subsystems, and equipment. The commissioning report shall indicate whether systems, subsystems, and equipment have been completed and are performing according to the OPR, BoD, and Contract Documents. The commissioning report shall include, but is not limited to, the following:
  - 1. Lists and explanations of substitutions; compromises; variances in the OPR, BoD, and Contract Documents; record of conditions; and, if appropriate, recommendations for resolution. This report shall be used to evaluate systems, subsystems, and equipment and shall serve as a future reference document during Owner occupancy and operation. It shall describe components and performance that exceed requirements of the OPR, BoD, and Contract Documents and those that do not meet requirements of the OPR, BoD, and Contract Documents. It may also include a recommendation for accepting or rejecting systems, subsystems, and equipment.
  - 2. OPR and BoD documentation.
  - 3. Commissioning plan.
  - 4. Testing plans and reports.
  - 5. Corrective modification documentation.
  - 6. Issues log.
  - 7. Completed test checklists.
  - 8. Listing of off-season test(s) not performed and a schedule for their completion.

- K. Systems Manual: CxA shall gather required information and compile systems manual. Systems manual shall include, but is not limited to, the following:
  - 1. OPR and BoD, including system narratives, schematics, and changes made throughout the Project.
  - 2. Project Record Documents as specified in Division 1 Section "Project Record Documents."
  - 3. Final commissioning plan.
  - 4. Commissioning report.
  - 5. Operation and maintenance data as specified in Division 1 Section "Operation and Maintenance Data."

### 1.8 SUBMITTALS

- A. Commissioning Plan Final Submittal: CxA shall submit an electronic format of the final commissioning plan.
- B. Test Checklists and Report Forms: CxA shall submit sample checklists and forms to the Contractor quality-control manager and subcontractors for review and comment.
- C. Certificates of Readiness: CxA shall submit Certificates of Readiness.
- D. Test and Inspection Reports: CxA shall submit test and inspection reports.
- E. Corrective Action Documents: CxA shall submit corrective action documents.
- F. Final Commissioning Report Submittal: CxA shall submit an electronic format of the final commissioning report.

### 1.9 QUALITY ASSURANCE

- A. Instructor Qualifications: Factory-authorized service representatives, experienced in training, operation, and maintenance procedures for installed systems, subsystems, and equipment.
- B. Test Equipment Calibration: Comply with test equipment manufacturer's calibration procedures and intervals. Recalibrate test instruments immediately whenever instruments have been repaired following damage or dropping. Affix calibration tags to test instruments. Instruments shall have been calibrated within six months prior to use.

#### 1.10 COORDINATION

A. Coordinating Meetings: CxA shall conduct coordination meetings of the commissioning team to review progress on the commissioning plan, to discuss scheduling conflicts, and to discuss upcoming commissioning process activities.

- B. Pretesting Meetings: CxA shall conduct pretest meetings of the commissioning team to review startup reports, pretest inspection results, testing procedures, testing personnel and instrumentation requirements, and manufacturers' authorized service representative services for each system, subsystem, equipment, and component to be tested.
- C. Testing Coordination: CxA shall coordinate sequence of testing activities to accommodate required quality-assurance and -control services with a minimum of delay and to avoid necessity of removing and replacing construction to accommodate testing and inspecting.
  - 1. Schedule times for tests, inspections, obtaining samples, and similar activities.
- D. Manufacturers' Field Services: CxA shall coordinate services of manufacturers' field services.

PART 2 - PRODUCTS (Not Used)

#### PART 3 - EXECUTION

# 3.1 OPERATION AND MAINTENANCE TRAINING REQUIREMENTS

- A. Training Preparation Conference: Before operation and maintenance training, CxA shall convene a training preparation conference to include Owner's operation and maintenance personnel, Contractor, and subcontractors. In addition to requirements specified in Division 1 Section "Demonstration and Training," perform the following:
  - 1. Review the OPR and BoD.
  - 2. Review installed systems, subsystems, and equipment.
  - 3. Review instructor qualifications.
  - 4. Review instructional methods and procedures.
  - 5. Review training module outlines and contents.
  - 6. Review course materials (including operation and maintenance manuals).
  - 7. Inspect and discuss locations and other facilities required for instruction.
  - 8. Review and finalize training schedule and verify availability of educational materials, instructors, audiovisual equipment, and facilities needed to avoid delays.
  - 9. For instruction that must occur outside, review weather and forecasted weather conditions and procedures to follow if conditions are unfavorable.
- B. Training Modules: Develop an instruction program that includes individual training modules for each system, subsystem, and equipment as specified in Division 1 Section "Demonstration and Training."

#### END OF SECTION

### **SECTION 079200**

#### JOINT SEALANTS

#### PART 1 - GENERAL

#### 1.1 GENERAL REQUIREMENTS

A. Drawings and general provisions of the Contract, including General and Supplementary Conditions and other Division 1 Specification Sections, apply to this Section.

### 1.2 SUMMARY

- A. Remove existing sealant and prepare substrates for sealant installation.
- B. Install sealant and backing for the following applications:
  - 1. Plaster floor line joints.
  - 2. Joints between plaster assembly and window framing.
  - 3. Joints between plaster assembly and louvers.
  - 4. Window glazing to metal frame. Infill metal panel wall cladding joints
  - 5. Plaster assembly terminations.
  - 6. Parapet caps.
  - 7. Penetrations.
  - 8. Miscellaneous existing exterior sealant joints
- C. Install extruded silicone tape for the following applications:
  - 1. Window frame joints

#### 1.3 RELATED SECTIONS

- A. Work related to the Section includes, but is not limited to, the following:
  - 1. Section 099653-Elastomeric Coating

#### 1.4 REFERENCES

- A. American Society for Testing and Materials, ASTM.
- B. ANSI/ASTM D1056 Flexible Cellular Materials Sponge or Expanded Rubber.
- C. ANSI/ASTM D1565 Flexible Cellular Materials Vinyl Polymers and Copolymers.
- D. FS TT-S-00230 Sealing Compound: Elastomeric Type, Single Component.

E. SWI (Sealing and Waterproofer's Institute) — Sealant and Caulking Guide Specification.

### 1.5 SUBMITTALS

- A. Product data and samples, for each type and color of joint sealant.
- B. Product test reports, from an independent qualified testing agency, for compliance with:
  - 1. Testing in accordance with ASTM C719 and ASTM C920 using building substrates and production run materials for this project and application procedures in this specification.
- C. Submit manufacturer's installation instructions.
- D. Job site mock-up:
  - 1. Incorporate the work specified in this Section.
  - 2. Rebuild the mock-up as many times as required to meet the Engineer's approval.
  - 3. Schedule the mock-up installation to leave adequate time for sealant cure, testing and reconstruction, if needed, without delay of the project.
  - 4. Coordinate mock-up construction with other trades.
  - 5. Approved mock-up will become part of the finished work and serve as a standard of acceptability for the remaining work

### 1.6 QUALITY ASSURANCE

- A. Installer Qualifications: An experienced installer, approved by the sealant manufacturer, who has specialized in installing joint sealants similar in material, design and extent to those indicated for this Project and whose work has resulted in joint-sealant installations with minimum 5 years record of successful in-service performance.
- B. Testing and Inspection
  - 1. Sealant Adhesion Field Tests
    - a. Prior to start of sealant application, construct a mock-up using specified surface preparation and sealant installation on aluminum frame and cement plaster accessories.
    - b. Notify Engineer at least 48 hours prior to start of installation and testing so he can be present.
  - 2. Procedure for Sealant Adhesion Field Test
    - a. Construct three 10 in.-long x 1 in.-wide x 1/4 in.-thick strips of sealant over each substrate. Apply bond breaker tape to the substrate under the last 2 in. of the sealant at each end of the strips to provide a tab for peel testing.
    - b. For all existing materials, cure the samples 21 days at min. temperature of 50° F.

- c. After curing grasp strip sample end tabs and pull 90 degrees to the surface.
- d. Acceptable application, cohesive failure (tearing within itself) with no adhesive (debonding) failure.
- e. If sample debonds, the sealant manufacturer shall make written recommendations regarding changes in surface preparations or primers. Submit recommendations to the Engineer for approval.
- f. Repeat sealant adhesion trials as required to produce an acceptable application.

#### 3. Sealant Manufacturer

- a. Observe mock-up construction and testing.
- b. Observe initial day of sealant installation.
- c. Visit site and observe work at least once every two weeks.
- d. Issue a written report recording surface preparation procedures, sealant application procedures, and stating the work conforms to the manufacturer's recommendations and these specifications. Note departures from these requirements and recommendations for necessary corrective actions.
- 4. Owner reserves the right to perform water testing of any area that has been repaired.
- 5. Test joint backings and release tapes for compatibility.

### 1.7 DELIVERY, STORAGE AND HANDLING

- A. All materials are to be of recent manufacture and delivered to the job site in original unopened containers with the manufacturer's name, number, batch identification and date of production.
- B. Do not use materials whose shelf life has expired.
- C. Remove from the job site any materials or incomplete waterproofing work exposed to moisture or rejected by the Engineer.

### 1.8 ENVIRONMENTAL REQUIREMENTS

- A. Maintain temperature and humidity recommended by the sealant manufacturer during and after installation.
- B. Store all materials between 40°F and 80°F; if exposed to lower temperatures, restore to proper temperatures before use.
- C. Do not proceed with installation of joint sealants under the following conditions:
  - 1. When ambient and substratre temperature conditions are outside limits permitted by manufacturer or are below 40°F or over 95°F.

- 2. When joint substrates are damp, or humidity conditions are outside limits permitted by sealant manufacturer.
- 3. If inclement weather is forecast within 36 hours.

#### 1.9 SEQUENCING AND SCHEDULING

- A. Coordinate the work of this Section with all Sections referencing this Section.
- B. Attend preconstruction conference to discuss the work and coordinate with other trades.

#### 1.10 WARRANTY

- A. Manufacturer's warranty: Written warranty, signed by manufacturer, agreeing to furnish joint sealants and accessories to repair or replace those that do not comply with performance and other requirements specified in this Section within specified warranty period. Include coverage of installed sealants and accessories that fail to achieve airtight and watertight seal, exhibit loss of adhesion or cohesion, or do not cure.
  - 1. Warranty period: 20 years from date of Substantial Completion.
- B. Contractor's warranty: Written warranty, signed by Installer, agreeing to repair or replace sealants that do not comply with performance and other requirements specified in this Section within specified warranty period.
  - 1. Warranty period: 5 years from date of Substantial Completion.

#### PART 2 - PRODUCTS

#### 2.1 GENERAL

A. Compatibility: Provide joint sealants, backings, and other related materials that are compatible with one another, with sealant and with joint substrates under conditions of service and application, as demonstrated by sealant manufacturer based on testing and field experience.

### 2.2 SEALANTS

#### A. Products

- 1. Cement Plaster Accessory Joints:
  - a. 795 Silicone Building Sealant by Dow Corning.
- 2. Window, Metal Panel, and Wet Seal Glazing Joints:

- a. 795 Silicone Building Sealant by Dow Corning.
- B. Colors, to be approved by Owner.
- C. Acceptable alternate: Equivalent GE Momentive Silicone Sealant.

#### 2.3 PREFORMED SILICONE-SEALANT SYSTEM

#### A. Products

- 1. Extruded Tape at Mullion Frame Joints:
  - a. 123 Silicone Seal by Dow Corning.
- B. Colors, to be approved by Owner.
- C. Acceptable Alternate: Equivalent GE Momentive Preformed Silicone-Sealant System.
  - 1. Coordinate with products submitted under Specification Section 099653
    - a. Elastomeric Coating and Silicone Sealants must be from same Manufacturer

#### 2.4 ACCESSORIES

- A. Primer: Non-staining type, recommended by sealant manufacturer for each different substrate.
- B. Joint Cleaner: Non-corrosive and non-staining type, recommended by sealant manufacturer for each different substrate.
- C. Bond Breaker Tape: Pressure-sensitive polyethylene, adhesive-backed, 0.006 in. thick tape, in widths required, recommended by sealant manufacturer to suit application and compatible with joint sealants.
- D. Backer Rod: Closed-cell, non-gassing, polyethylene foam, 30 to 50 percent larger than joint width.
- E. Masking Tape: Non-staining, nonabsorbent material compatible with joint sealants and surfaces adjacent to joints.

#### PART 3 - EXECUTION

#### 3.1 EXAMINATION

A. Examine joints indicated to receive joint sealants for compliance with requirements for joint configuration, installation tolerances and other conditions affecting joint-sealant performance.

- B. Verify site conditions and field dimensions by field measurement. Notify the Engineer immediately of any inconsistency between existing conditions and the specifications. The Engineer will determine what modifications or additional repairs are necessary.
- C. Examine all surfaces to receive sealant for roughness, contaminants, or other conditions that will impair the sealant application. Inform the Engineer in writing of such conditions. Do not commence work until all defects are remedied.
- D. Beginning installation means installer accepts existing substrate.

### 3.2 PREPARATION

- A. All surfaces must be clean and dry before preparation begins.
- B. Remove all foreign material from joint substrates, including dust, oil, and grease.
- C. Prior to applying primer and installing backer rod, clean all substrates.
- D. Clean porous joint substrate surfaces by brushing, grinding, mechanical abrading or a combination of these methods to produce a clean, sound substrate. Remove loose particles remaining from cleaning operations by vacuuming or blowing out joints with oil-free compressed air. Use non-metallic brushes or grinding blade.
- E. Clean nonporous surfaces with chemical, non-staining cleaners.
  - 1. Use two clean, white, soft, lint-free cotton cloths and clean, fresh joint cleaner as required to clean the different substrates.
    - a. Wet one cloth with cleaner and wipe the surface vigorously. Use second cloth to clean the surface before solvent evaporates. Continue two-cloth procedure until surface does not discolor cloth. Repeat once more.
    - b. Do not dip cloth into solvent to avoid contamination of solvent. Pump solvent on to first cloth.
    - c. Do not solvent clean at temperatures below 40° F.
    - d. Dispose of cloths and chemical containers daily.

### 3.3 INSTALLATION

### A. Primer

- 1. Mask all surfaces, do not allow primer on exposed surfaces.
- 2. Pour primer in clean container for use. Do not pour more than a 10-minute supply into container.
- 3. Replace cap on primer immediately, remove any thickened or degraded primer from site.
- 4. Apply primer with a clean brush to clean and dry substrates. Prime all surfaces to receive sealants. Use only one coat of primer.
- 5. Allow primer to dry as required by manufacturer's specifications.

6. Do not allow primer to become wet or dirty before sealant application.

#### B. Backer Rod

- 1. Install clean, dry backer rod into joint openings against dry substrates. Use specialized tool to accurately set depth of rod. Account for concave tooling of joints.
- 2. Place the rod so the sealant depth measured at the center of the joint after tooling is one-half of the sealant joint width.
  - a. Minimum depth: 1/4 in.
  - b. Maximum depth: 1/2 in.
- 3. Change rod sizes, as required by variation in joint widths.
- 4. Butt ends together, do not twist rods together.
- 5. Do not touch or otherwise contaminate substrate surfaces while inserting backer rod.
- 6. Do not rupture skin of backer rod; remove any rod containing punctures.
- 7. Install only as much backer rod as can be sealed in the same day.

# C. Bond Breaker Tape

- 1. Install bond breaker tape at locations as detailed against dry substrates.
- 2. Install only as much bond breaker tape as can be sealed in the same day.

#### D. Sealant

- 1. Apply sealant only to clean, dry, primed surfaces at ambient temperatures above 40° F.
- 2. Seal joints within 10 hours of primer application.
- 3. Inspect sealant cartridge and verify the production date is within 6 months of the date of application. Remove from site all sealant more than 6 months old.
- 4. Verify correct backer rod or bond breaker tape positioning before applying sealant.
- 5. Follow manufacturer's recommendations for proper joint geometry.
- 6. Fill all joints solidly and continuously with sealant, neatly applied with standard caulking gun in a continuous motion using a slight pressure. "Push" the sealant bead ahead of the nozzle. Do not drag the nozzle.
- 7. Tool joint surface within 5 minutes of sealant application and before skin develops on sealant. Use a concave tool to insure intimate contact with the substrate and eliminate air bubbles. Do not use fingers to tool sealants. Avoid 3-sided adhesion at all sealant joints.
- 8. Do not use chemicals, soaps, water, waxes or any liquids on tooling equipment or for tooling. Provide a smooth, uniform finished surface. Provide flush joint configuration for fillet seals and concave joint configuration for all other seals, unless otherwise indicated, per ASTM C 1193.
- 9. Sequence work to prevent contamination of fresh sealant by dust or other debris.

### E. Preformed Silicone-Sealant Systems

- 1. Apply masking tape to each side of joint, outside of area to be covered by sealant system.
- Apply a bead of silicone sealant to each side of joint to produce a bead of size complying with preformed silicone-sealant system manufacturer's printed schedule and covering a bonded area of not less than 3/8 inch. Hold edge of sealant bead inside of masking tape by 1/4 inch. Apply sealant bed where indicated.
- Within 10 minutes of sealant application, press silicone extrusion or molded pieces into sealant to wet extrusion and substrate. Use a roller to apply consistent pressure and uniform contact between sealant and both extrusion and substrate. Install fillet bead of sealant along edges of extrusions and molded pieces.
- 4. Complete installation of horizontal joints before installing vertical joints. Lap vertical joints over horizontal joints. At end of joints, cut silicone extrusion with a razor knife, or sharp scissors.

#### F. Wet Seal

- 1. Cut gasket flush with surface of window frame using approved cutting jig. Do not allow cutting blade to contact glass.
- 2. Prepare metal and surfaces per Section 3.2.
- 3. Apply sealant and tool surface within 5 minutes of sealant application and before skin develops on sealant. Use a concave tool to ensure intimate contact with the substrate and eliminate air bubbles. Provide 1/4 in. (minimum) bond to glass and metal surfaces.
- 4. Do not use any liquid for tooling. Provide a smooth, uniform finished surface and straight lines on the glass surface.

#### 3.4 CLEANING

- A. Clean off excess sealants or sealant smears adjacent to joints as the work progresses by methods approved in writing by manufacturers.
- B. Upon completion of a drop, clean excess materials from sealant removal and repairs from the façade.
  - 1. Use pressure washer
    - a. Do not damage surfaces.

### 3.5 PROTECTION OF FINISHED WORK

A. Protect joints sealants during and after curing period from contact with contaminating substances and from damage resulting from construction operations or other causes, so that sealants are without deterioration or damage at time of Substantial Completion.

If damage occurs, cut out and remove damaged or deteriorated joint sealants immediately and repair so installations are indistinguishable from the original work.

**END OF SECTION** 

#### **SECTION 099653**

### **ELASTOMERIC COATING**

#### PART 1 - GENERAL

#### 1.1 RELATED DOCUMENTS

A. Drawings and general provisions of the Contract, including General and Supplementary Conditions and Division 01 Specification Sections, apply to this Section.

### 1.2 SUMMARY

- A. This Section includes the following work:
  - 1. Preparing existing plaster substrate and installing elastomeric coating.

#### 1.3 RELATED SECTIONS

A. Section 079200 - Joint Sealants

#### 1.4 REFERENCES

- A. ASTM D16 Definitions of Terms Relating to Paint, Varnish, Lacquer, and Related Products.
- B. PDCA (Painting and Decorating Contractors of America) Painting Architectural Specifications Manual.
- C. SSPC (Steel Structures Painting Council).

### 1.5 SUBMITTALS

- A. Comply with provisions of Division 1 Section "Submittals".
- B. Product Data: For each elastomeric coating system specified. Include crack fillers, and primers.
  - 1. Material List: An inclusive list of required coating materials. Indicate each material and cross-reference the specific coating, finish system, and application.
  - 2. Manufacturer's Information: Technical information including label analysis and instructions for handling, storing, and applying each coating material.
  - 3. Certification by elastomeric coating manufacturer that products supplied comply with local VOC regulations.

- C. Samples for Initial Selection: Manufacturer's standard color chart for each type of topcoat product indicated.
- D. Adhesion Test Reports, from coating manufacturer, indicating the following:
  - Material forming coating has been tested for compatibility and adhesion with coating substrates.
- E. Manufacturer's interpretation of test results and written recommendations for primers and substrate preparation needed for adhesion. Field Test Report Log: For each coating application.
- F. Application Schedule: Provide a mark-up of the elevations depicting starting and stopping points at inconspicuous locations to avoid appearance of discontinuity in coating application.

#### 1.6 QUALIFICATIONS

A. Installer Qualifications: Engage an experienced installer who has completed work similar in material, design, and extent to that indicated for this Project. Applicator should have personnel that have been trained by a manufacturer's representative to install manufacturer's products.

### 1.7 REGULATORY REQUIREMENTS

- A. California Air Resources Board (CARB).
- B. Bay Area Air Quality Management District (BAAQMD).

# 1.8 DELIVERY, STORAGE, AND HANDLING

- A. Deliver products to site in sealed and labeled containers; inspect to verify acceptability.
- B. Include on container label: Manufacturer's name, type of paint, brand name, lot number, brand code, coverage, surface preparation, drying time, cleanup requirements, color designation, and instructions for mixing and reducing.
- C. Store paint materials at minimum ambient temperature of 45°F and a maximum of 90°F, in ventilated area, and as required by manufacturer's instructions.

# 1.9 ENVIRONMENTAL REQUIREMENTS

- A. Do not apply materials if substrate is wet or damp.
- B. Do not apply materials when surface and ambient temperatures are lower than 40°F or higher than 100°F or outside of range permitted by coating manufacturer.

- C. Do not apply exterior coatings during rain or snow, or when relative humidity is outside the humidity ranges required by the coating manufacturer.
- D. Do not apply materials until sealants have been allowed to cure a minimum of 5 days and are fully skinned over.

### 1.10 EXTRA MATERIALS

- A. Provide two gallons of each coating color to the Owner for attic stock.
- B. Label each container with color, type, and texture, in addition to the manufacturer's label.

### 1.11 MOCK-UP

- A. Construct a mock-up of coating installation at each substrate type and color before proceeding with work. The location of the mock-up will be selected by the Engineer.
- B. Allow to dry a minimum of 7 days.
- C. Perform tape adhesion test in the presence of the manufacturer's representative, owner's representative, and Engineer of mock-up coating to confirm visual and performance acceptability.

#### 1.12 QUALITY CONTROL

- A. Manufacturer's representative will perform tape adhesion tests on the coating after application to ensure proper bond of coating to substrate.
- B. Manufacturer's representative will inspect and measure dry film thickness at various locations.
  - 1. If inspection shows that proper adhesion or specified dry film thickness has not been achieved, Perform corrective work to the satisfaction of the Architect/Engineer at no extra cost to the Owner.
- C. Obtain block fillers, primers and undercoat materials for each coating system from the same manufacturer as the finish coats.

### 1.13 WARRANTY

A. Provide a ten year labor and material manufacturer warranty. All coating shall have full adhesion and there shall be no evidence of blisters running, peeling, scaling, chalking, streaks, fading, or stains at the end of ten years.

### PART 2 - PRODUCTS

### 2.1 GENERAL

- A. Products included in this Section which form basis of design, or approved equals.
  - 1. Basis of Design: AllGuard by Dow Corning
    - a. And associated primers, fillers, and accessories.
    - b. Use compatible sealant under Specification Section 079200 "Joint Sealants."
  - 2. Approved alternate: SilShield SEC2400 by GE Silicones
    - a. And associated primers, fillers, and accessories.
    - b. Use compatible sealant under Specification Section 079200 "Joint Sealants"

#### 2.2 MATERIALS

- A. Coating Primer: Recommended by Manufacturer.
- B. Elastomeric Coating: AllGuard by Dow Corning
  - 1. Thickness: 2 coats, 10 wet mils per coat (10 dry mils total)
  - 2. Performance Requirements:
    - a. Tensile Strength at 7 days: >145 psi minimum; ASTM D-412 modified.
    - b. Elongation at Break: 600% minimum at 0°F; ASTM D-412 modified.
    - c. Resistance to wind driven rain: No passage of water.
    - d. Weathering: No chalking or cracking at 10,000 hours: ASTM G-23.
- C. Crack filler: 1/16 in. to 1/4 in.: Silicone Sealant- Dowsil 790.
- D. Colors: As selected by Owner.

### PART 3 - EXECUTION

#### 3.1 EXAMINATION

- A. Verify that substrate conditions are ready to receive work as instructed by the product manufacturer, where applicable.
  - 1. Verify that all plaster repairs have been completed.
  - 2. Verify that all sealant work has been completed and sealant has fully cured.
- B. Beginning of installation means installer accepts existing substrate conditions.

### 3.2 PREPARATION

A. Remove items already in place that are not to be painted and are scheduled to be removed. Mask items not scheduled to be removed before surface preparation and

coating, including but not limited to windows, signs, traffic signals and security cameras.

- B. Plaster: Wait a minimum of 3 days after completing repairs before painting.
- C. Plaster Substrates: Remove release agents, curing compounds, efflorescence, and chalk. Do not paint surfaces if moisture content or alkalinity of surfaces to be painted exceeds that permitted in manufacturer's written instructions.
- D. All surfaces must be clean, dry, sound, and frost-free with all residues and other contaminants removed. Pressure wash plaster surfaces to remove all residues. Take all appropriate measures to contain water so as to avoid water overspray onto street.

#### 3.3 APPLICATION - GENERAL

- A. Apply all products in accordance with manufacturer's instructions.
- B. Fill all surface irregularities, rough areas and cracks per manufacturer's guidelines.
  - 1. Cracks up to 1/16 Inch: no additional preparation required.
  - 2. Cracks up to 1/4 Inch: Rout open cracks to 1/4 inch wide and 1/8 inch deep. Clean cracks and surrounding area removing dust, dirt, and other impurities. Apply crack filler primer recommended by manufacturer and fill cracks with crack filler applied with a putty knife or trowel. If excessive shrinkage occurs, reapply crack filler.
- C. Do not apply coating to surfaces that are not dry. Allow sufficient drying time after walls have been pressure washed. Moisture content must be below 5%.
- D. Apply primer as recommended by manufacturer following mock-up installation.
- E. Wall Coating Application:
  - 1. Apply single coat of coating primer to substrate.
  - 2. Allow applied coating primer to dry a minimum of 24 hours prior to application of elastomeric coating.
  - 3. Apply two coats in uniform finish using roller, airless spray, or nylon brush. Apply carefully to provide a smooth, even coating without runs or sags.
    - a. Apply first coat of elastomeric coating. Allow applied coating to dry a minimum of 4 hours prior to application of second coat.
    - b. Apply second coat of elastomeric coating.
    - c. Total nominal dry film thickness of elastomeric coating: 10 mils.
    - d. At higher temperatures, work carefully to maintain a wet edge.
  - 4. Finished elastomeric coating shall have no pinholes, holidays, or any other visible defects and shall fully coat the wall surface.
  - 5. Finish surfaces shall be uniform in finish and color and be free from flash spots and brush marks. In all cases, the film produced shall be satisfactory in all respects to the Owner
  - 6. Correct defective work at no additional cost to the Owner.

F. Complete application of each coat in one day. Start and stop coating application at inconspicuous locations, as depicted on submitted schedule and approved by Owner, each day to avoid discontinuities at noticeable surfaces.

#### 3.4 FIELD QUALITY CONTROL

- A. Perform a minimum of three adhesion tests for each coating type in the presence of the manufacturer's representative or as required to ensure compliance with warranty.
- B. Periodically inspect wet mil thickness requirements during application by using a wet mil gage.

#### 3.5 CLEANING AND PROTECTION

- A. Clean work under provisions of Division 1 Specification Sections.
- B. Collect waste material which may constitute a fire hazard, place in closed metal containers, and remove daily from site.
- C. Clean spattered surfaces. Remove spattered paints by methods recommended by manufacturer. Do not scratch or damage adjacent finished surfaces.
- D. Correct damage to adjacent surfaces by cleaning, repairing, replacing, and refinishing, as approved by the Owner, and leave in an undamaged condition.
- E. Provide "Wet Paint" signs to protect newly coated finishes. After completing coating operations, remove temporary protective wrappings.
- F. Touch up and restore damaged or defaced painted surfaces.

**END OF SECTION**